

County of Los Angeles Chief Executive Office

PUBLIC SAFETY CLUSTER AGENDA REVIEW MEETING

FESIA A. DAVENPORT Chief Executive Officer

DATE: Wednesday, June 16, 2021 TIME: 9:30 a.m.

DUE TO CLOSURE OF ALL COUNTY BUILDING, TO PARTICIPATE IN THE MEETING CALL TELECONFERENCE NUMBER: (323) 776-6996 ID: 169948309#

Click here to join the meeting

<u>AGENDA</u>

Members of the Public may address the Public Safety Cluster on any agenda item by submitting a written request prior to the meeting. Two (2) minutes are allowed per person in total for each item.

- 1. CALL TO ORDER
- 2. GENERAL PUBLIC COMMENT (15 Minutes)
- **3. INFORMATIONAL ITEM(S):** [Any Information Item is subject to discussion and/or presentation at the request of two or more Board offices with advance notification]:
 - A. Board Letter: APPROVAL OF CONTRACTS FOR TEMPORARY PERSONNEL SERVICES Speaker(s): Theresa Barrera (Fire)
 - B. Board Letter: APPROVAL OF SOLE SOURCE CONTRACT WITH ALACO LADDER COMPANY FOR WOODEN LADDER REPAIR AND MAINTENANCE SERVICES Speaker(s): Theresa Barrera (Fire)
 - C. Board Letter:

APPROVAL OF TRANSIT LAW ENFORCEMENT SERVICES AGREEMENT BY AND BETWEEN COUNTY OF LOS ANGELES AND FOOTHILL TRANSIT AUTHORITY Speaker(s): Bryan Aguilera and Kevin Piper (Sheriff)

D. Board Letter: ADOPT RESOLUTION TO AUTHORIZE PARTICIPATION IN THE CALIFORNIA BOATING SAFETY AND ENFORCEMENT FINANCIAL AID PROGRAM FOR FISCAL YEAR 2021-22 Speaker(s): Tom Ewald (Fire) and Jack Ewell (Sheriff) Wednesday, June 16, 2021

4. **PRESENTATION/DISCUSSION ITEM(S)**:

- A. Board Briefing: CIVILIAN OVERSIGHT COMMISSION and OFFICE OF INSPECTOR GENERAL BRIEFING Speaker(s): Brian Williams (COC) and Max Huntsman (OIG)
- B. Board Briefing: PUBLIC SAFETY DEPARTMENTAL BUDGET BRIEFING Speaker(s): Rene Phillips (CEO)
- C. Board Briefing: DIVISION OF JUVENILE JUSTICE (DJJ) TRANSITION COMMITTEE BRIEFING Speaker(s): Brandon Nichols (Probation)
- D. Board Briefing: OFFICE OF DIVERSION AND RE-ENTRY (ODR) MONTHLY BRIEFING Speaker(s): Peter Espinoza (ODR)

5. PUBLIC COMMENTS

6. ADJOURNMENT

7. UPCOMING ITEMS:

A. Board Letter:

APPROVE THE ACCEPTANCE OF GRANT FUNDS FROM THE FEDERAL EMERGENCY MANAGEMENT AGENCY DEPARTMENT OF HOMELAND SECURITY – FISCAL YEAR 2019 FIRE PREVENTION AND SAFETY GRANT Speaker(s): Theresa Barrera and Nick Duvally (Fire)

B. Board Briefing: ALTERNATIVES TO INCARCERATION (ATI) BRIEFING Speaker(s): Songhai Armstead (ODR)

IF YOU WOULD LIKE TO EMAIL A COMMENT ON AN ITEM ON THE PUBLIC SAFETY CLUSTER AGENDA, PLEASE USE THE FOLLOWING EMAIL AND INCLUDE THE AGENDA NUMBER YOU ARE COMMENTING ON:

PUBLIC_SAFETY_COMMENTS@CEO.LACOUNTY.GOV



COUNTY OF LOS ANGELES FIRE DEPARTMENT

1320 NORTH EASTERN AVENUE LOS ANGELES, CALIFORNIA 90063-3294 (323) 881-2426 www.fire.lacounty.gov

"Proud Protectors of Life, Property, and the Environment"

BOARD OF SUPERVISORS

HILDA L. SOLIS FIRST DISTRICT

HOLLY J. MITCHELL SECOND DISTRICT

> SHEILA KUEHL THIRD DISTRICT

JANICE HAHN FOURTH DISTRICT

KATHRYN BARGER FIFTH DISTRICT

DARYL L. OSBY FIRE CHIEF FORESTER & FIRE WARDEN

June 29, 2021

The Honorable Board of Supervisors County of Los Angeles 383 Kenneth Hahn Hall of Administration 500 West Temple Street Los Angeles, California 90012

Dear Supervisors:

APPROVAL OF CONTRACTS FOR TEMPORARY PERSONNEL SERVICES (ALL DISTRICTS) (3 VOTES)

SUBJECT

The Consolidated Fire Protection District of Los Angeles County (District) is requesting Board of Supervisors (Board) approval to establish contracts with 22nd Century Technologies, Inc. (TSCTI); CathyJon Enterprises, Inc. (dba) HB Staffing; Trust Temporary Services, Inc. (dba) Helpmates Staffing Services (Helpmates); and Howroyd-Wright Employment Services, Inc. (dba) AppleOne Employment Services (AppleOne) to provide as-needed and intermittent Temporary Personnel Services for the District's emergency and business operations.

IT IS RECOMMENDED THAT THE BOARD, ACTING AS THE GOVERNING BODY OF THE CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY:

- 1. Approve and instruct the Fire Chief, or his designee, to sign the four attached contracts (Attachment A) between the District and TSCTI, HB Staffing, Helpmates, and AppleOne to provide Temporary Personnel Services.
- 2. Authorize the combined maximum contract sum of \$4,140,000, including the initial three-year term, two one-year, and twelve month-to-month extension options. The maximum contract sum is comprised of annual expenditures not to exceed a combined total for all four contracts of \$690,000 per contract year.

AGOURA HILLS ARTESIA AZUSA BALDWIN PARK BELL BELL GARDENS BELLFLOWER BRADBURY CALABASAS

CARSON CERRITOS CLAREMONT COMMERCE COVINA CUDAHY DIAMOND BAR DUARTE

FI MONTE GARDENA GLENDORA HAWAIIAN GARDENS HAWTHORNE HERMOSA BEACH HIDDEN HILLS HUNTINGTON PARK INDUSTRY

SERVING THE UNINCORPORATED AREAS OF LOS ANGELES COUNTY AND THE CITIES OF: INGLEWOOD IRWINDALE I A HABRA LA MIRADA LA PUENTE LAKEWOOD LANCASTER

LA CANADA-FLINTRIDGE

LAWNDALE LOMITA LYNWOOD MALIBU MAYWOOD NORWALK PALMDALE PALOS VERDES ESTATES PARAMOUNT

PICO RIVERA POMONA RANCHO PALOS VERDES ROLLING HILLS ROLLING HILLS ESTATES ROSEMEAD SAN DIMAS SANTA CLARITA

SIGNAL HILL SOUTH EL MONTE SOUTH GATE TEMPLE CITY VERNON WALNUT WEST HOLLYWOOD WESTLAKE VILLAGE WHITTIER

The Honorable Board of Supervisors June 29, 2021 Page 2

- 3. Delegate authority to the Fire Chief, or his designee, to execute amendments, suspensions, or terminations if deemed necessary, including the extension options as described in recommendation number two (2) above, in accordance with the approved contract terms and conditions, and with prior review and approval as to form by County Counsel.
- 4. Find these contracts are exempt from the provisions of the California Environmental Quality Act (CEQA).

PURPOSE/JUSTIFICATION OF RECOMMENDED ACTION

The recommended actions will enable the District to obtain temporary personnel services on an as-needed and intermittent basis when utilization of permanent District employees is not feasible. It is necessary for the District to hire temporary assistance to maintain business continuity during peak loads, emergency situations, and temporary absences. Temporary personnel services will be utilized to fill vacancies caused by approved leave of absence, attrition, and when a certification list is not available for any items.

Implementation of Strategic Plan Goals

Approval of the recommended actions is consistent with the County's Strategic Plan Goal No. III.3 – Pursue Operational Effectiveness, Fiscal Responsibility, and Accountability by maximizing the use of County assets, and ensuring resources are expended in a responsible, efficient, and strategic manner. Contracting administrative support is fiscally cost effective and supports the District's ability to ensure an organization of capable staff to provide superior service to the public.

FISCAL IMPACT/FINANCING

Sufficient funding is available in the District's Fiscal Year 2021-22 Adopted Budget. The District will continue to allocate the necessary funds to obtain the required services. There is no impact to net County cost. These contracts do not include an allowance for Cost of Living Adjustments (COLAs).

FACTS AND PROVISIONS/LEGAL REQUIREMENTS

The District is authorized to enter into contracts for temporary personnel services in accordance with California Government Code Section 31000.4, to obtain temporary assistance during peak load, temporary absences, or emergencies other than labor disputes. Use of these services is limited to a period not to exceed ninety (90) days for any single peak load, temporary absence, or emergency situation.

The attached contracts specify the District has no obligation to pay expenditures beyond the contract payment mechanisms. Furthermore, the firms will not be asked to perform services which exceed the approved scope of work or contract term. TSCTI, HB Staffing, Helpmates, and AppleOne are in compliance with all Board and Chief Executive Office (CEO)

The Honorable Board of Supervisors June 29, 2021 Page 3

requirements, including contractor Employee Jury Service, Safely Surrendered Baby Law, and the Defaulted Property Tax Reduction Program, and agree to maintain compliance with all requirements throughout the term of the contracts.

The CEO's Risk Management Section (CEO RMS) reviewed the contracts prior to the release of the solicitation and concurred with the provisions relating to insurance and indemnification. The contracts have been approved as to form by County Counsel and have been signed by TSCTI, HB Staffing, Helpmates, and AppleOne. On final analysis and consideration of the award, these firms were selected without regard to race, color, creed, or national origin.

ENVIRONMENTAL DOCUMENTATION

The services provided through these proposed contracts will not have a significant effect on the environment; and therefore, the contracts are exempt from CEQA, pursuant to Section 15061 (b) (3) of the CEQA Guidelines.

CONTRACTING PROCESS

On November 22, 2020, the District released a Request for Proposals (RFP) to solicit qualified proposers to provide temporary personnel services to the District. In addition to posting the announcement on the County's WebVen and the District's contracting webpage, advertisements were placed in the Los Angeles Times newspaper.

On December 22, 2020, the District received thirteen (13) proposals. The proposals were forwarded to an evaluation committee for review and scoring. The evaluation committee was comprised of subject matter experts from the District's Human Resources and Materials Management Divisions. The committee's evaluation was based on criteria set forth in the RFP, which included cost, references, experience/background, financial capability, approach to provide services, and quality control.

The evaluation committee evaluated and scored all thirteen (13) proposals. Upon completion of the evaluations and based on an informed averaging process, the top four (4) scored proposals were selected for contract award recommendation. The nine (9) remaining proposers were notified they were not selected and offered a debriefing on their proposal evaluations. Six (6) proposers participated in the debriefing process and upon completion of the debriefings, were provided a Notice of Intent to Request a Proposed Contractor Selection Review (PCSR) to submit at their discretion. No PCSR notices were received and proposers expressed satisfaction with the debriefing process and results.

AppleOne, one of the four selected proposers submitted an exception to the contract in its proposal, however, with review and concurrence from County Counsel, the District completed contract negotiations and AppleOne agreed to the terms and conditions as described in the attached contract.

The Honorable Board of Supervisors June 29, 2021 Page 4

The District reviewed the Contractor Alert Reporting Database and Better Business Bureau to assess all proposers' past performances, negative experiences, and complaints with other agencies and entities. The District found there are currently no negative findings or complaints that would prevent any of the proposers from contracting with the District.

IMPACT ON CURRENT SERVICES

Approval of the recommended actions will support the District's ability to provide life safety services and fire protection by continuing to provide administrative support staff, essential to the District's emergency operations and business services.

Award of these contracts will not result in the displacement of any County employees as these services are provided on an as-needed and intermittent basis. The contract will not result in a reduction of service and there is no change in risk exposure to the County.

CONCLUSION

Upon approval by your Honorable Board, please instruct the Executive Officer of the Board to return the adopted stamped Board Letter to the following:

Consolidated Fire Protection District of Los Angeles County Executive Office - Business Operations Attention: Zuleyda Santana, Administrative Services Manager II 1320 North Eastern Avenue Los Angeles, CA 90063 Zuleyda.Santana@fire.lacounty.gov

The District's contact can be reached at (323) 881-6173.

Respectfully submitted,

DARYL L. OSBY, FIRE CHIEF

DLO:ah

Enclosures

c: Chief Executive Officer Executive Officer, Board of Supervisors County Counsel

ATTACHMENT A

CONTRACT NO.: _____

CONTRACT



BY AND BETWEEN

CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY

AND

22nd CENTURY TECHNOLOGIES, INC.

FOR

TEMPORARY PERSONNEL SERVICES

CONTRACT PROVISIONS TABLE OF CONTENTS

PARA	-		PAGE		
		EXHIBITS			
		CABLE DOCUMENTS			
		ITIONS			
	WORK				
-		OF CONTRACT	-		
		RACT SUM			
-	5.1	Maximum Contract Sum	-		
		Written Approval for Reimbursement			
	5.3	Notification of 75% of Total Contract Sum			
	5.4	No Payment for Services Provided Following Expiration-Termination of Contract			
	5.5	Invoices and Payments	7		
	5.6	Intentionally Omitted	9		
	5.7 5.8	Default Method of Payment: Direct Deposit or Electronic Funds Transf Travel, Transportation, and Parking			
6	ADMINISTRATION OF CONTRACT- DISTRICT				
	6.1	District Administration	10		
	6.2	District's Project Director			
	6.3	District's Project Manager	10		
	6.4	District's Contract Administrator	11		
7		NISTRATION OF CONTRACT-CONTRACTOR	11		
	7.1	Contractor Administration	11		
	7.2	Contractor's Project Manager	11		
	7.3	Approval of Contractor's Staff	12		
	7.4	Contractor's Staff Identification	12		
	7.5	Background and Security Investigations	12		
	7.6	Confidentiality	13		
8	STANDARD TERMS AND CONDITIONS14				
	8.1	Amendments	14		
	8.2	Assignment and Delegation/Mergers or Acquisitions	15		
	8.3	Authorization Warranty	16		
	8.4	Budget Reductions	16		
	8.5	Complaints			
	8.6	Compliance with Applicable Law	17		

18 20
20
21
21
21
24
25
25
25
26
26
27
27
28
28
29
29
34
35
37
37
38
38
39
39
39
39
40
40
40
41
42
43

8.41	Termination for Breach of Warranty to Maintain Compliance with County's Child Support Compliance Program	. 44
8.42	Termination for Convenience	. 44
8.43	Termination for Default	. 45
8.44	Termination for Improper Consideration	. 46
8.45	Termination for Insolvency	. 47
8.46	Termination for Non-Adherence of County Lobbyist Ordinance	. 48
8.47	Termination for Non-Appropriation of Funds	. 48
8.48	Validity	. 48
8.49	Waiver	. 48
8.50	Warranty Against Contingent Fees	. 48
8.51	Warranty of Compliance with County's Defaulted Property Tax	
	Reduction Program	. 49
8.52	Termination for Breach of Warranty to Maintain Compliance with	
	County's Defaulted Property Tax Reduction Program	. 49
8.53	Time off for Voting	. 49
8.54	Compliance with County's Zero Tolerance Policy on Human Trafficking	. 50
8.55	Intentionally Omitted	. 50
8.56	Compliance with Fair Chance Employment Practices	
8.57	Compliance with the County Policy of Equity	. 50
8.58	Prohibition from Participation in Future Solicitation(s)	. 51
UNIQ	UE TERMS AND CONDITIONS	
9.1	Intentionally Omitted	. 51
9.2	Health Insurance Portability and Accountability Act of 1996 ("HIPAA")	. 51
9.3	Intentionally Omitted	. 52
9.4	Intentionally Omitted	. 52
9.5	Intentionally Omitted	. 52
9.6	Data Destruction	. 52
9.7	Intentionally Omitted	. 53
9.8	Intentionally Omitted	
9.9	Intentionally Omitted	. 53
9.10	Mandatory Requirement to Register on County's WebVen	. 53
9.11	Limitation on Corporate Acts	. 53
9.12	Modifications	. 54
9.13	Remedies of Non-Compliance	. 54
9.14	Suspension	. 54
9.15	Transition of Contract Services	. 56

9

NATURES

STANDARD EXHIBITS

- A Statement of Work
- B Pricing Sheet
- C Independent Price Determination
- D Contractor's EEO Certification
- E District's Administration
- F Contractor's Administration
- G Contractor's Confidentiality Acknowledgement
- H Jury Service Ordinance
- I Safely Surrendered Baby Law

CONTRACT BETWEEN CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY AND 22nd CENTURY TECHNOLOGIES, INC. FOR TEMPORARY PERSONNEL SERVICES

This Contract ("Contract") made and entered into this 1st day of July, 2021 by and between the Consolidated Fire Protection District of Los Angeles County, hereinafter referred to as "District" and 22nd Century Technologies, Inc., hereinafter referred to as "Contractor." Contractor is located at 818 W. Seventh Street, Suite 930, Los Angeles, CA 90017.

RECITALS

WHEREAS, the District may contract with private businesses for temporary personnel services when certain requirements are met; and

WHEREAS, the Contractor is a private firm specializing in providing temporary personnel services; and

WHEREAS, the District is authorized under California Government Code Section 31000 to contract for special services, and under 31000.4 to contract with temporary help firms to assist the District during any peak load, temporary absence, or emergency other than a labor dispute; and

WHEREAS, the District is in need of temporary personnel services on an asneeded and intermittent basis and is authorized under Health and Safety Code Section 13861 to contract for services; and

NOW THEREFORE, in consideration of the mutual covenants contained herein, and for good and valuable consideration, the parties agree to the following:

1 APPLICABLE DOCUMENTS

Exhibits A, B, C, D, E, F, G, H, and I are attached to and form a part of this Contract. In the event of any conflict or inconsistency in the definition or interpretation of any word, responsibility, schedule, or the contents or description of any task, deliverable, goods, service, or other work, or otherwise between the base Contract and the Exhibits, or between Exhibits, such conflict or inconsistency shall be resolved by giving precedence first to the terms and conditions of the Contract and then to the Exhibits according to the following priority.

Standard Exhibits:

- 1.1 Exhibit A Statement of Work
- 1.2 Exhibit B Pricing Sheet
- 1.3 Exhibit C Independent Price Determination
- 1.4 Exhibit D Contractor's EEO Certification
- 1.5 Exhibit E District's Administration
- 1.6 Exhibit F Contractor's Administration
- 1.7 Exhibit G Acknowledgement and Confidentiality Agreement
- 1.8 Exhibit H Jury Service Ordinance
- 1.9 Exhibit I Safely Surrendered Baby Law

2 **DEFINITIONS**

The headings herein contained are for convenience and reference only and are not intended to define the scope of any provision thereof. The following words as used herein shall be construed to have the following meaning, unless otherwise apparent from the context in which they are used.

- 2.1 **Board of Supervisors (Board):** The Board of Supervisors of the County of Los Angeles; the governing body of the District and the County of Los Angeles.
- 2.2 **Contract:** This agreement executed between District and Contractor. Included are all supplemental agreements amending or extending the service to be performed. The Contract sets forth the terms and conditions for the issuance and performance of all tasks, deliverables, services and other work
- 2.3 **Contractor:** The person or persons, sole proprietor, partnership, joint venture, corporation or other legal entity who has entered into an agreement with the District to perform or execute the work covered by this Contract.
- 2.4 <u>**Contractor Project Manager:**</u> The person designated by the Contractor to administer the Contract operations under this Contract.
- 2.5 **<u>County</u>**: The County of Los Angeles, a political subdivision of the State of California.
- 2.6 **<u>Dav(s)</u>**: Calendar day(s) unless otherwise specified.
- 2.7 <u>District</u>: The Consolidated Fire Protection District of Los Angeles County.
- 2.8 **District Project Director:** Person designated by District with authority for District on contractual or administrative matters relating to this Contract that cannot be resolved by the District's Project Manager.
- 2.9 <u>**District Project Manager</u>:** Person designated by District's Project Director to manage the operations under this Contract.</u>
- 2.10 **Fiscal Year**: The twelve (12) month period beginning July 1st and ending the following June 30th.
- 2.11 <u>Statement of Work</u>: The directions, provisions, and requirements provided herein and special provisions pertaining to the method, frequency, manner and place of performing the contract services.

- 2.12 **<u>Subcontract</u>**: An agreement by the contractor to employ a subcontractor to provide services to fulfill this contract.
- 2.13 **Subcontractor:** Any individual, person or persons, sole proprietor, firm, partnership, joint venture, corporation, or other legal entity furnishing supplies, services of any nature, equipment, and/or materials to contractor in furtherance of contractor's performance of this contract, at any tier, under oral or written agreement.

3 WORK

- 3.1 Pursuant to the provisions of this Contract, the Contractor shall fully perform, complete and deliver on time, all tasks, deliverables, services and other work as set forth in herein.
- 3.2 If the Contractor provides any tasks, deliverables, goods, services, or other work, other than as specified in this contract, the same shall be deemed to be a gratuitous effort on the part of the contractor, and the contractor shall have no claim whatsoever against the District.

4 TERM OF CONTRACT

4.1 The term of this Contract shall be three (3) years commencing after approval by County's Board of Supervisors (Board), and execution by the Fire Chief or his designee, unless sooner terminated or extended, in whole or in part, as provided in this Contract.

The District shall have the sole option to extend this Contract term for up to two (2) additional one-year periods and twelve (12) month to month extension options, for a maximum total Contract term of six (6) years, pursuant to the same terms and conditions. Each such extension option may be exercised at the sole discretion of the Fire Chief or his designee as authorized by the Board.

- 4.2 The District maintains databases that track/monitor contractor performance history. Information entered into such databases may be used for a variety of purposes, including determining whether the District will exercise a contract term extension option.
- 4.3 The Contractor shall notify the District when this Contract is within six (6) months of the expiration of the term as provided for hereinabove. Upon occurrence of this event, the Contractor shall send written notification to the District at the address herein provided in Exhibit E District's Administration.

5 CONTRACT SUM

5.1 Maximum Contract Sum

5.1.1 The amount the District shall expend from its own funds during the Contract's entire term for as-needed Temporary Personnel Services shall not exceed \$690,000 per Contract year. The Maximum Contract Sum shall be the maximum monetary amount available that is payable by the District to the Contractor for supplying all the Services, Deliverables, Work, etc.

The annual Maximum Contract Sum under the terms of this Contract shall be \$690,000, shared amongst all contracts for temporary personnel services.

This is not a commitment or offer on the part of the District to expend the Maximum Contract Sum. The Contractor shall perform and complete all Services required of the Contractor under this Contract as set forth in Exhibit B (Pricing Sheet), but in any event, not in excess of the Maximum Contract Sum.

5.1.2 The Contractor acknowledges and agrees the Maximum Contract Sum is an all-inclusive, not-to-exceed price that cannot be adjusted for any costs or expenses whatsoever of Contractor. This Contract includes the full amount of compensation and reimbursement the District will be asked to provide to the Contractor in order for the Contractor to fully perform all of its obligations under this Contract, with such amount of compensation and reimbursement subject to any executed Amendments if applicable. The Contractor understands the District is entering into this Contract in reliance upon the premise that the Contractor shall fully perform all of its obligations under this Contract without seeking any additional compensation or reimbursement beyond that already provided for in this Contract, subject to any Amendments, if applicable. It is the Contractor's risk and responsibility to achieve and timely deliver the Services in accordance with the requirements of the Contract.

5.2 Written Approval for Reimbursement

The Contractor shall not be entitled to payment or reimbursement for any tasks or services performed, nor for any incidental or administrative expenses whatsoever incurred in or incidental to performance hereunder, except as specified herein. Assumption or takeover of any of the Contractor's duties, responsibilities, or obligations, or performance of same by any person or entity other than the Contractor, whether through assignment, subcontract, delegation, merger, buyout, or any other mechanism, with or without consideration for any reason whatsoever, shall not occur except with the District's express prior written approval.

5.3 Notification of 75% of Total Contract Sum

The Contractor shall maintain a system of record keeping that will allow the Contractor to determine when it has incurred seventy-five percent (75%) of the total contract sum under this Contract. Upon occurrence of this event, the Contractor shall send written notification to the District at the address herein provided in Exhibit E - District's Administration.

5.4 No Payment for Services Provided Following Expiration-Termination of Contract

The Contractor shall have no claim against District for payment of any money or reimbursement, of any kind whatsoever, for any service provided by the Contractor after the expiration or other termination of this Contract. Should the Contractor receive any such payment it shall immediately notify the District and shall immediately repay all such funds to the District. Payment by the District for services rendered after expiration-termination of this Contract shall not constitute a waiver of the District's right to recover such payment from the Contractor. This provision shall survive the expiration or other termination of this Contract.

5.5 Invoices and Payments

5.5.1 The Contractor shall invoice the District only for providing the tasks, deliverables, goods, services, and other work specified in Exhibit A - Statement of Work and elsewhere hereunder. The Contractor shall prepare invoices, which shall include the charges owed to the Contractor by the District under the terms of this Contract. The Contractor's payments shall be as provided in Exhibit B - Pricing Sheet and the Contractor shall be paid only for the tasks, deliverables, goods, services, and other work approved in writing by the District. If the District does not approve work in writing no payment shall be due to the Contractor for that work.

- 5.5.2 The Contractor's invoices shall be priced in accordance with Exhibit B Pricing Sheet.
- 5.5.3 The Contractor's invoices shall contain the information set forth in Exhibit A - Statement of Work describing the tasks, deliverables, goods, services, work hours, and facility and/or other work for which payment is claimed.
- 5.5.4 The Contractor shall submit the monthly invoices to the District by the 15th calendar day of the month following the month of service.
- 5.5.5 Payment to Contractor shall be made on an arrears basis, upon acceptance of completed work by the District, provided the Contractor is not in default under any provisions of this Contract. The Contractor shall email one (1) copy of the invoice to the following:
 - 1. <u>Bianca.Moya@fire.lacounty.gov</u> <u>Vinrose.Napoles@fire.lacounty.gov</u> for review and approval of all invoices; and
 - 2. <u>Fire-InvoiceSubmission@fire.lacounty.gov</u> for payment of all invoices.

The Contractor's invoices shall include the following:

- Contract Number
- Date(s) of Service
- Brief description of services
- A breakdown of labor hours and hourly rate i.e.: 3 hours @ \$20/hour = \$60.00.
 NO OVERTIME MAY BE ACCUMULATED OR PAID.
- Employee Name and Employee Number of District Employee who ordered or authorized the service.
- Signature of authorized District employee. Contractor's failure to obtain the signature of District employee authorizing the work may result in a delay of payment.

5.5.6 **District Approval of Invoices**

All invoices submitted by the Contractor for payment must have the written approval of the District's Project Manager prior to any payment thereof. In no event shall the District be liable or responsible for any payment prior to such written approval. Approval for payment will not be unreasonably withheld.

5.5.7 Local Small Business Enterprises – Prompt Payment Program

Certified Local Small Business Enterprises (LSBEs) will receive prompt payment for services they provide to the District. Prompt payment is defined as fifteen (15) calendar days after receipt of an undisputed invoice.

5.6 Intentionally Omitted

5.7 Default Method of Payment: Direct Deposit or Electronic Funds Transfer

- 5.7.1 The County, at its sole discretion, has determined the most efficient and secure default form of payment for goods and/or services provided under an agreement/ contract with the County shall be Electronic Funds Transfer (EFT) or direct deposit, unless an alternative method of payment is deemed appropriate by the Auditor-Controller (A-C).
- 5.7.2 The Contractor shall submit a direct deposit authorization request via the website https://directdeposit.lacounty.gov with banking and vendor information, and any other information that the A-C determines is reasonably necessary to process the payment and comply with all accounting, record keeping, and tax reporting requirements.
- 5.7.3 Any provision of law, grant, or funding agreement requiring a specific form or method of payment other than EFT or direct deposit shall supersede this requirement with respect to those payments.
- 5.7.4 At any time during the duration of the agreement/contract, a Contractor may submit a written request for an exemption to this requirement. Such request must be based on specific legal, business or operational needs and explain why the

payment method designated by the A-C is not feasible and an alternative is necessary. The A-C, in consultation with the contracting department(s), shall decide whether to approve exemption requests.

5.8 Travel, Transportation, and Parking

- 5.8.1 All travel-related costs are the responsibility of the Contractor. The District will not be responsible for paying or reimbursing Contractor or its employees for any travel related costs.
- 5.8.2 The District is not responsible for paying or reimbursing parking or transportation costs for temporary personnel.

6 ADMINISTRATION OF CONTRACT – DISTRICT

6.1 District Administration

A listing of all District Administration referenced in the following subparagraphs are designated in Exhibit E - District's Administration. The District will notify the Contractor in writing of any change in the names or addresses shown.

6.2 District's Project Director

The responsibilities of the District's Project Director include:

- Coordinating with Contractor and ensuring Contractor's performance of the Contract; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby; and
- Upon request of the Contractor, providing direction to the Contractor, as appropriate in areas relating to District policy, information requirements, and procedural requirements; however, in no event, shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby.

6.3 District's Project Manager

The role of the District's Project Manager is to oversee the day-today administration of this Contract; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby. The District's Project Manager's responsibilities include:

- Meeting with the Contractor's Project Manager on a regular basis; and
- Inspecting any and all tasks, deliverables, goods, services, or other work provided by or on behalf of the Contractor; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby.

The District's Project Manager is not authorized to make any changes in any of the terms and conditions of this Contract and is not authorized to further obligate District in any respect whatsoever.

6.4 District's Contract Administrator

The responsibilities of the District's Contract Administrator include:

- Ensuring that the objectives of this Contract are met; and
- Making changes in the terms and conditions of this Contract in accordance with Sub-paragraph 8.1, Amendments; and
- Providing direction to Contractor in the areas relating to District policy, information requirements, and procedural requirements.

7 ADMINISTRATION OF CONTRACT - CONTRACTOR

7.1 Contractor Administration

A listing of all of Contractor's Administration referenced in the following paragraphs is designated in Exhibit F - Contractor's Administration. The Contractor will notify the District in writing of any change in the names or addresses shown.

7.2 Contractor's Project Manager

- 7.2.1 The Contractor's Project Manager is designated in Exhibit F - Contractor's Administration. The Contractor shall notify the District in writing of any change in the name or address of the Contractor's Project Manager.
- 7.2.2 The Contractor's Project Manager shall be responsible for the Contractor's day-to-day activities as related to this

Contract and shall meet and coordinate with the District's Project Manager on a regular basis.

7.3 Approval of Contractor's Staff

District has the absolute right to approve or disapprove all of the Contractor's staff performing work hereunder and any proposed changes in the Contractor's staff, including, but not limited to, the Contractor's Project Manager.

7.4 Contractor's Staff Identification

All of the Contractor's employees assigned to District facilities are required to have a District Identification (ID) badge on their person and visible at all times. The Contractor may bear the expense of any badge not returned to the District when the temporary personnel's assignment has ended.

- 7.4.1 Contractor is responsible to ensure employees have obtained a District ID badge before they are assigned to work in a District facility. Contractor personnel may be asked by a District representative to leave a District facility if they do not have the proper District ID badge on their person and Contractor personnel must immediately comply with such request.
- 7.4.2 Contractor shall notify the District within one business day when staff is terminated from working under this Contract. Contractor shall retrieve and return an employee's District ID badge to the District on the next business day after the employee has terminated employment with the Contractor.
- 7.4.3 If District requests the removal of Contractor's staff, Contractor shall retrieve and return an employee's District ID badge to the District on the next business day after the employee has been removed from working on the District's Contract.

7.5 Background and Security Investigations

7.5.1 Each of Contractor's staff performing services under this Contract, as determined by District in District's sole discretion, shall undergo and pass a background investigation to the satisfaction of District as a condition of beginning and continuing to perform services under this Contract. Such background investigation must be obtained through fingerprints submitted to the California Department of Justice to include State, local, and federal-level review, which may include, but shall not be limited to, criminal conviction information. The fees associated with the background investigation shall be at the expense of the Contractor, regardless of whether the member of Contractor's staff passes or fails the background investigation. The District, may at its sole discretion and expense, provide live scan services for Contractor's staff assigned to District facilities. The District reserves the right to request reimbursement for the live scan services.

If a member of Contractor's staff does not pass the background investigation, the District may request that the member of Contractor's staff be removed immediately from performing services under the Contract. Contractor shall comply with District's request at any time during the term of the Contract. District will not provide to Contractor or to Contractor's staff any information obtained through the District's background investigation.

- 7.5.2 District shall immediately deny or terminate facility access to any member of Contractor's staff that does not pass such investigation to the satisfaction of the District or whose background or conduct is incompatible with District facility access.
- 7.5.3 Disqualification of any member of Contractor's staff pursuant to this Paragraph 7.5 shall not relieve Contractor of its obligation to complete all work in accordance with the terms and conditions of this Contract.

7.6 Confidentiality

- 7.6.1 Contractor shall maintain the confidentiality of all records and information in accordance with all applicable Federal, State and local laws, rules, regulations, ordinances, directives, guidelines, policies and procedures relating to confidentiality, including, without limitation, County policies concerning information technology security and the protection of confidential records and information.
- 7.6.2 Contractor shall indemnify, defend, and hold harmless County, its officers, employees, and agents, from and against any and all claims, demands, damages, liabilities, losses, costs and expenses, including, without limitation,

defense costs and legal, accounting and other expert, consulting, or professional fees, arising from, connected with, or related to any failure by Contractor, its officers, employees, agents, or subcontractors, to comply with this Paragraph 7.6, as determined by County in its sole judgment. Any legal defense pursuant to contractor's indemnification obligations under this Paragraph 7.6 shall be conducted by contractor and performed by counsel selected by Contractor and approved by District. Notwithstanding the preceding sentence, District shall have the right to participate in any such defense at its sole cost and expense, except that in the event Contractor fails to provide District with a full and adequate defense, as determined by District in its sole judgment, District shall be entitled to retain its own counsel, including, without limitation, County Counsel, and to reimbursement from Contractor for all such costs and expenses incurred by County in doing so. Contractor shall not have the right to enter into any settlement, agree to any injunction, or make any admission, in each case, on behalf of District without District's prior written approval.

- 7.6.3 Contractor shall inform all of its officers, employees, agents and subcontractors providing services hereunder of the confidentiality provisions of this Contract.
- 7.6.4 Contractor shall sign and adhere to the provisions of Exhibit G - "Contractor Acknowledgement and Confidentiality Agreement."

8 STANDARD TERMS AND CONDITIONS

8.1 Amendments

- 8.1.1 For any change which affects the scope of work, term, contract sum, payments, or any term or condition included under this Contract, an amendment to the Contract shall be prepared and executed by the contractor and by the Fire Chief or his designee.
- 8.1.2 The County's Board of Supervisors or Chief Executive Officer or designee may require the addition and/or change of certain terms and conditions in the Contract during the term of this Contract. The District reserves the right to add and/or change such provisions as required by the County's Board of Supervisors or Chief Executive Officer. To implement such changes, an Amendment to the Contract shall be prepared

and executed by the contractor and by the Fire Chief or his designee.

8.1.3 The Fire Chief or his designee may at his sole discretion, authorize extensions of time as defined in Paragraph 4 - Term of Contract. The contractor agrees that such extensions of time shall not change any other term or condition of this Contract during the period of such extensions. To implement an extension of time, an Amendment to the Contract shall be prepared and executed by the contractor and by the Fire Chief or his designee.

8.2 Assignment and Delegation/Mergers or Acquisitions

- 8.2.1 The contractor shall notify the District of any pending acquisitions/mergers of its company unless otherwise legally prohibited from doing so. If the contractor is restricted from legally notifying the District of pending acquisitions/mergers, District then it should notifv the of the actual acquisitions/mergers as soon as the law allows and provide to the District the legal framework that restricted it from notifying the District prior to the actual acquisitions/mergers.
- 8.2. 2 The contractor shall not assign its rights or delegate its duties under this Contract, or both, whether in whole or in part, without the prior written consent of District, in its discretion, and any attempted assignment or delegation without such consent shall be null and void. For purposes of this paragraph, District consent shall require a written Amendment to the Contract, which is formally approved and executed by the parties. Any payments by the District to any approved designee or assignee on any claim under this Contract shall be deductible, at District's sole discretion, against the claims, which the contractor may have against the District.
- 8.2.3 Shareholders, partners, members, or other equity holders of contractor may transfer, sell, exchange, assign, or divest themselves of any interest they may have therein. However, in the event any such sale, transfer, exchange, assignment, or divestment is effected in such a way as to give majority control of contractor to any person(s), corporation, partnership, or legal entity other than the majority controlling interest therein at the time of execution of the Contract, such disposition is an assignment requiring the prior written consent of District in accordance with applicable provisions of this Contract.

8.2.4 Any assumption, assignment, delegation, or takeover of any of the contractor's duties, responsibilities, obligations, or performance of same by any person or entity other than the contractor, whether through assignment, subcontract, delegation, merger, buyout, or any other mechanism, with or without consideration for any reason whatsoever without District's express prior written approval, shall be a material breach of the Contract which may result in the termination of this Contract. In the event of such termination. District shall be entitled to pursue the same remedies against contractor as it could pursue in the event of default by contractor.

8.3 Authorization Warranty

The contractor represents and warrants that the person executing this Contract for the contractor is an authorized agent who has actual authority to bind the contractor to each and every term, condition, and obligation of this Contract and that all requirements of the contractor have been fulfilled to provide such actual authority.

8.4 Budget Reductions

In the event the County's Board of Supervisors adopts, in any fiscal year, a County Budget which provides for reductions in the salaries and benefits paid to the majority of County employees and imposes similar reductions with respect to County contracts, the District reserves the right to reduce its payment obligation under this Contract correspondingly for that fiscal year and any subsequent fiscal year during the term of this Contract (including any extensions), and the services to be provided by the contractor under this Contract shall also be reduced correspondingly. The District's notice to the contractor regarding said reduction in payment obligation shall be provided within thirty (30) calendar days of the Board's approval of such actions. Except as set forth in the preceding sentence, the contractor shall continue to provide all of the services set forth in this Contract.

8.5 Complaints

- 8.5.1 The contractor shall develop, maintain and operate procedures for receiving, investigating and responding to complaints.
- 8.5.2 Complaint Procedures

- 8.5.2.1 Within forty-five (45) business days after the Contract effective date, the contractor shall provide the District with the contractor's policy for receiving, investigating and responding to user complaints.
- 8.5.2.2 The District will review the contractor's policy and provide the contractor with approval of said plan or with requested changes.
- 8.5.2.3 If the District requests changes in the contractor's policy, the contractor shall make such changes and resubmit the plan within five (5) business days for District approval.
- 8.5.2.4 If, at any time, the contractor wishes to change the contractor's policy, the contractor shall submit proposed changes to the District for approval before implementation.
- 8.5.2.5 The contractor shall preliminarily investigate all complaints and notify the District's Project Manager of the status of the investigation within five (5) business days of receiving the complaint.
- 8.5.2.6 When complaints cannot be resolved informally, a system of follow-through shall be instituted which adheres to formal plans for specific actions and strict time deadlines.
- 8.5.2.7 Copies of all written responses shall be sent to the District's Project Manager within three (3) business days of mailing to the complainant.

8.6 Compliance with Applicable Law

- 8.6.1 In the performance of this Contract, contractor shall comply with all applicable Federal, State and local laws, rules, regulations, ordinances, directives, guidelines, policies and procedures, and all provisions required thereby to be included in this Contract are hereby incorporated herein by reference.
- 8.6.2 Contractor shall indemnify, defend, and hold harmless County, its officers, employees, and agents, from and against any and all claims, demands, damages, liabilities, losses, costs, and expenses, including, without limitation,

defense costs and legal, accounting and other expert, consulting or professional fees, arising from, connected with, or related to any failure by contractor, its officers, employees, agents, or subcontractors, to comply with any such laws, rules, regulations, ordinances, directives, guidelines, policies, or procedures, as determined by County in its sole judgment. Any legal defense pursuant to contractor's indemnification obligations under Paragraph 8.6 (Compliance with Applicable Law) shall be conducted by contractor and performed by counsel selected by contractor and approved by County. Notwithstanding the preceding sentence, County shall have the right to participate in any such defense at its sole cost and expense, except that in the event contractor fails to provide County with a full and adequate defense, as determined by County in its sole judgment, County shall be entitled to retain its own counsel, including, without limitation, County Counsel, and to reimbursement from contractor for all such costs and expenses incurred by County in doing so. Contractor shall not have the right to enter into any settlement, agree to any injunction or other equitable relief, or make any admission, in each case, on behalf of County without County's prior written approval.

8.7 Compliance with Civil Rights Laws

The contractor hereby assures that it will comply with Subchapter VI of the Civil Rights Act of 1964, 42 USC Sections 2000 (e) (1) through 2000 (e) (17), to the end that no person shall, on the grounds of race, creed, color, sex, religion, ancestry, age, condition of physical handicap, marital status, political affiliation, or national origin, be excluded from participation in, be denied the benefits of, or be otherwise subjected to discrimination under this Contract or under any project, program, or activity supported by this Contract. The contractor shall comply with Exhibit D - Contractor's EEO Certification.

8.8 Compliance with the County's Jury Service Program

8.8.1 Jury Service Program:

This Contract is subject to the provisions of the County's ordinance entitled Contractor Employee Jury Service ("Jury Service Program") as codified in Sections 2.203.010 through 2.203.090 of the Los Angeles County Code, a copy of which

is attached as Exhibit H and incorporated by reference into and made a part of this Contract.

8.8.2 Written Employee Jury Service Policy.

- 1. Unless the contractor has demonstrated to the District's satisfaction either that the contractor is not a "contractor" as defined under the Jury Service Program (Section 2.203.020 of the County Code) or that the contractor qualifies for an exception to the Jury Service Program (Section 2.203.070 of the County Code), the contractor shall have and adhere to a written policy that provides that its Employees shall receive from the contractor, on an annual basis, no less than five days of regular pay for actual jury service. The policy may provide that Employees deposit any fees received for such jury service with the contractor or that the contractor deduct from the Employee's regular pay the fees received for jury service.
- 2. For purposes of this paragraph, "contractor" means a person, partnership, corporation or other entity which has a contract with the County or a subcontract with a County contractor and has received or will receive an aggregate sum of fifty thousand dollars (\$50,000) or more in any twelve (12) month period under one or more County contracts or subcontracts. "Employee" means any California resident who is a full-time employee of the contractor. "Full-time" means forty (40) hours or more worked per week, or a lesser number of hours if: 1) the lesser number is a recognized industry standard as determined by the District, or 2) contractor has a long-standing practice that defines the lesser number of hours as full-time. Full-time employees providing short-term, temporary services of ninety (90) days or less within a twelve (12) month period are not considered full-time for purposes of the Jury Service Program. If the contractor uses any subcontractor to perform services for the District under the Contract, the subcontractor shall also be subject to the provisions of this paragraph. The provisions of this paragraph shall be inserted into any such subcontract agreement and a copy of the Jury Service Program shall be attached to the agreement.

- 3. If the contractor is not required to comply with the Jury Service Program when the Contract commences, the contractor shall have a continuing obligation to review the applicability of its "exception status" from the Jury Service Program, and the contractor shall immediately notify the District if the contractor at any time either comes within the Jury Service Program's definition of "contractor" or if the contractor no longer gualifies for an exception to the Jury Service Program. In either event, the contractor shall immediately implement a written policy consistent with the Jury Service Program. The District may also require, at any time during the Contract and at its sole discretion, that the contractor demonstrate, to the District's satisfaction that the contractor either continues to remain outside of the Jury Service Program's definition of "contractor" and/or that the contractor continues to qualify for an exception to the Program.
- 4. Contractor's violation of this paragraph of the Contract may constitute a material breach of the Contract. In the event of such material breach, District may, in its sole discretion, terminate the Contract and/or bar the contractor from the award of future County contracts for a period of time consistent with the seriousness of the breach.

8.9 Conflict of Interest

- 8.9.1 No County employee whose position with the County enables such employee to influence the award of this Contract or any competing Contract, and no spouse or economic dependent of such employee, shall be employed in any capacity by the contractor or have any other direct or indirect financial interest in this Contract. No officer or employee of the contractor who may financially benefit from the performance of work hereunder shall in any way participate in the District's approval, or ongoing evaluation, of such work, or in any way attempt to unlawfully influence the District's approval or ongoing evaluation of such work.
- 8.9.2 The contractor shall comply with all conflict of interest laws, ordinances, and regulations now in effect or hereafter to be enacted during the term of this Contract. The contractor warrants that it is not now aware of any facts that create a conflict of interest. If the contractor hereafter becomes

aware of any facts that might reasonably be expected to create a conflict of interest, it shall immediately make full written disclosure of such facts to the District. Full written disclosure shall include, but is not limited to, identification of all persons implicated and a complete description of all relevant circumstances. Failure to comply with the provisions of this paragraph shall be a material breach of this Contract.

8.10 Consideration of Hiring County Employees Targeted for Layoff or Re-Employment List

Should the contractor require additional or replacement personnel after the effective date of this Contract to perform the services set forth herein, the contractor shall give first consideration for such employment openings to qualified, permanent County employees who are targeted for layoff or qualified, former County employees who are on a re-employment list during the life of this Contract.

8.11 Consideration of Hiring GAIN-GROW Participants

- 8.11.1 Should the contractor require additional or replacement personnel after the effective date of this Contract, the contractor shall give consideration for any such employment openings to participants in the County's Department of Public Social Services Greater Avenues for Independence (GAIN) Program or General Relief Opportunity for Work (GROW) Program who meet the contractor's minimum qualifications for the open position. For this purpose, consideration shall mean that the contractor will interview qualified candidates. The County will refer GAIN-GROW participants by job category to the contractor. Contractors shall report all job openings with job requirements to: GAINGROW@DPSS.LACOUNTY.GOV and BSERVICES@WDACS.LACOUNTY.GOV and DPSS will refer qualified GAIN/GROW job candidates.
- 8.11.2 In the event that both laid-off County employees and GAIN/GROW participants are available for hiring, County employees shall be given first priority.

8.12 Contractor Responsibility and Debarment

8.12.1 **Responsible Contractor**

A responsible contractor is a contractor who has demonstrated the attribute of trustworthiness, as well as quality, fitness, capacity and experience to satisfactorily perform the contract. It is the County's policy to conduct business only with responsible contractors.

8.12.2 Chapter 2.202 of the County Code

The contractor is hereby notified that, in accordance with Chapter 2.202 of the County Code, if the County acquires information concerning the performance of the contractor on this or other contracts which indicates that the contractor is not responsible, the County may, in addition to other remedies provided in the Contract, debar the contractor from bidding or proposing on, or being awarded, and/or performing work on County contracts for a specified period of time, which generally will not exceed five (5) years but may exceed five (5) years or be permanent if warranted by the circumstances, and terminate any or all existing contracts the contractor may have with the County.

8.12.3 Non-responsible contractor

The County may debar a contractor if the Board of Supervisors finds, in its discretion, that the contractor has done any of the following: 1) violated a term of a contract with the County or a nonprofit corporation created by the County, 2) committed an act or omission which negatively reflects on the contractor's quality, fitness or capacity to perform a contract with the County, any other public entity, or a nonprofit corporation created by the County, or engaged in a pattern or practice which negatively reflects on same, 3) committed an act or offense which indicates a lack of business integrity or business honesty, or 4) made or submitted a false claim against the County or any other public entity.

8.12.4 Contractor Hearing Board

8.12.4.1 If there is evidence that the contractor may be subject to debarment, the County will notify the contractor in writing of the evidence which is the basis for the proposed debarment and will advise the contractor of the scheduled date for a debarment hearing before the Contractor Hearing Board.

- 8.12.4.2 The Contractor Hearing Board will conduct a hearing where evidence on the proposed debarment is presented. The contractor and/or the contractor's representative shall be given an opportunity to submit evidence at that hearing. After the hearing, the Contractor Hearing Board shall prepare a tentative proposed decision, which shall contain а recommendation regarding whether the contractor should be debarred, and, if so, the appropriate length of time of the debarment. The contractor and the District shall be provided an opportunity to object to the tentative proposed decision prior to its presentation to the Board of Supervisors.
- 8.12.4.3 After consideration of any objections, or if no objections are submitted, a record of the hearing, the proposed decision, and any other recommendation of the Contractor Hearing Board shall be presented to the Board of Supervisors. The Board of Supervisors shall have the right to modify, deny, or adopt the proposed decision and recommendation of the Contractor Hearing Board.
- 8.12.4.4 If a contractor has been debarred for a period longer than five (5) years, that contractor may after the debarment has been in effect for at least five (5) years, submit a written request for review of the debarment determination to reduce the period of debarment or terminate the debarment. The County may, in its discretion, reduce the period of debarment or terminate the debarment if it finds that the contractor has adequately demonstrated one or more of the 1) elimination of the grounds for followina: which the debarment was imposed; 2) a bona fide change in ownership or management; 3) material evidence discovered after debarment was imposed; or 4) any other reason that is in the best interests of the County.
- 8.12.4.5 The Contractor Hearing Board will consider a request for review of a debarment determination only where 1) the contractor has been debarred for a period longer than five (5) years; 2) the debarment has been in effect for at least five (5)

years; and 3) the request is in writing, states one or more of the grounds for reduction of the debarment period or termination of the debarment. includes and supporting documentation. Upon receiving an appropriate request, the Contractor Hearing Board will provide notice of the hearing on the request. At the hearing, the Contractor Hearing Board shall conduct a hearing where evidence on the proposed reduction of debarment period or termination of debarment is presented. This hearing shall be conducted and the request for review decided by the Contractor Hearing Board pursuant to the same procedures as for a debarment hearing.

8.12.4.6 The Contractor Hearing Board's proposed decision shall contain a recommendation on the request to reduce the period of debarment or terminate the debarment. The Contractor Hearing Board shall present its proposed decision and recommendation to the Board of Supervisors. The Board of Supervisors shall have the right to modify, deny, or adopt the proposed decision and recommendation of the Contractor Hearing Board.

8.12.5 Subcontractors of Contractor

These terms shall also apply to subcontractors of County contractors.

8.13 Contractor's Acknowledgement of County's Commitment to Safely Surrendered Baby Law

The contractor acknowledges that the County places a high priority on the implementation of the Safely Surrendered Baby Law. The contractor understands that it is the County's policy to encourage all County contractors to voluntarily post the County's "Safely Surrendered Baby Law" poster, in Exhibit I, in a prominent position at the contractor's place of business. The contractor will also encourage its subcontractors, if any, to post this poster in a prominent position in the subcontractor's place of business. Information and posters for printing are available at www.babysafela.org.

8.14 Contractor's Warranty of Adherence to County's Child Support Compliance Program

- 8.14.1 The contractor acknowledges that the County has established a goal of ensuring that all individuals who benefit financially from the County through contracts are in compliance with their court-ordered child, family and spousal support obligations in order to mitigate the economic burden otherwise imposed upon the County and its taxpayers.
- 8.14.2 As required by the County's Child Support Compliance Program (County Code Chapter 2.200) and without limiting the contractor's duty under this Contract to comply with all applicable provisions of law, the contractor warrants that it is now in compliance and shall during the term of this Contract maintain in compliance with employment and wage reporting requirements as required by the Federal Social Security Act (42 USC Section 653a) and California Unemployment Insurance Code Section 1088.5, and shall implement all lawfully served Wage and Earnings Withholding Orders or Child Support Services Department Notices of Wage and Earnings Assignment for Child, Family or Spousal Support, pursuant to Code of Civil Procedure Section 706.031 and Family Code Section 5246(b).

8.15 District's Quality Assurance Plan

The District or its agent(s) will monitor the contractor's performance under this Contract on not less than an annual basis. Such monitoring will include assessing the contractor's compliance with all Contract terms and conditions and performance standards. Contractor deficiencies which the District determines are significant or continuing and that may place performance of the Contract in jeopardy if not corrected will be reported to the Board of Supervisors and listed in the appropriate contractor performance database. The report to the Board will include improvement/corrective action measures taken by the District and the contractor. If improvement does not occur consistent with the corrective action measures, the District may terminate this Contract or impose other penalties as specified in this Contract.

8.16 Damage to District Property, Facilities, Buildings or Grounds

8.16.1 The contractor shall repair, or cause to be repaired, at its own cost, any damage to the District's real or personal

property used or occupied by the temporary personnel beyond wear from normal usage, as determined by District, any and all damage to District facilities, buildings, or grounds caused by the contractor or employees or agents of the contractor. Such repairs shall be made immediately after the contractor has become aware of such damage, but in no event later than thirty (30) days after the occurrence.

8.16.2 If the contractor fails to make timely repairs, District may make any necessary repairs. All costs incurred by District, as determined by District, for such repairs shall be repaid by the contractor by cash payment upon demand.

8.17 Employment Eligibility Verification

- 8.17.1 The contractor warrants that it fully complies with all Federal and State statutes and regulations regarding the employment of aliens and others and that all its employees performing work under this Contract meet the citizenship or alien status requirements set forth in Federal and State statutes and regulations. The contractor shall obtain, from all employees performing work hereunder, all verification and other documentation of employment eligibility status required by Federal and State statutes and regulations including, but not limited to, the Immigration Reform and Control Act of 1986, (P.L. 99-603), or as they currently exist and as they may be hereafter amended. The contractor shall retain all such documentation for all covered employees for the period prescribed by law.
- 8.17.2 The contractor shall indemnify, defend, and hold harmless, the County, its agents, officers, and employees from employer sanctions and any other liability which may be assessed against the contractor or the County or both in connection with any alleged violation of any Federal or State statutes or regulations pertaining to the eligibility for employment of any persons performing work under this Contract.

8.18 Counterparts and Electronic Signatures and Representations

This Contract may be executed in two or more counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same Contract. The facsimile, email or electronic signature of the Parties shall be deemed to constitute original signatures, and facsimile or electronic copies hereof shall be deemed to constitute duplicate originals.

The County and the Contractor hereby agree to regard electronic representations of original signatures of authorized officers of each party, when appearing in appropriate places on the Amendments prepared pursuant to Paragraph 8.1 (Amendments) and received via communications facilities (facsimile, email or electronic signature), as legally sufficient evidence that such legally binding signatures have been affixed to Amendments to this Contract.

8.19 Fair Labor Standards

The contractor shall comply with all applicable provisions of the Federal Fair Labor Standards Act and shall indemnify, defend, and hold harmless the County and its agents, officers, and employees from any and all liability, including, but not limited to, wages, overtime pay, liquidated damages, penalties, court costs, and attorneys' fees arising under any wage and hour law, including, but not limited to, the Federal Fair Labor Standards Act, for work performed by the contractor's employees for which the County may be found jointly or solely liable.

8.20 Force Majeure

- 8.20.1 Neither party shall be liable for such party's failure to perform its obligations under and in accordance with this Contract, if such failure arises out of fires, floods, epidemics, quarantine restrictions. other natural occurrences, strikes, lockouts (other than a lockout by such party or any of such party's subcontractors), freight embargoes, or other similar events to those described above, but in every such case the failure to perform must be totally beyond the control and without any fault or negligence of such party (such events are referred to in this paragraph as "force majeure events").
- 8.20.2 Notwithstanding the foregoing, a default by a subcontractor of contractor shall not constitute a force majeure event, unless such default arises out of causes beyond the control of both contractor and such subcontractor, and without any fault or negligence of either of them. In such case, contractor shall not be liable for failure to perform, unless the goods or services to be furnished by the subcontractor were obtainable from other sources in sufficient time to permit contractor to meet the required performance

schedule. As used in this subparagraph, the term "subcontractor" and "subcontractors" mean subcontractors at any tier.

8.20.3 In the event contractor's failure to perform arises out of a force majeure event, contractor agrees to use commercially reasonable best efforts to obtain goods or services from other sources, if applicable, and to otherwise mitigate the damages and reduce the delay caused by such force majeure event.

8.21 Governing Law, Jurisdiction, and Venue

This Contract shall be governed by, and construed in accordance with, the laws of the State of California. The contractor agrees and consents to the exclusive jurisdiction of the courts of the State of California for all purposes regarding this Contract and further agrees and consents that venue of any action brought hereunder shall be exclusively in the County of Los Angeles.

8.22 Independent Contractor Status

- 8.22.1 This Contract is by and between the District and the contractor and is not intended, and shall not be construed, to create the relationship of agent, servant, employee, partnership, joint venture, or association, as between the District and the contractor. The employees and agents of one party shall not be, or be construed to be, the employees or agents of the other party for any purpose whatsoever.
- 8.22.2 The contractor shall be solely liable and responsible for providing to, or on behalf of, all persons performing work pursuant to this Contract all compensation and benefits. The District shall have no liability or responsibility for the payment of any salaries, wages, unemployment benefits, disability benefits, Federal, State, or local taxes, or other compensation, benefits, or taxes for any personnel provided by or on behalf of the contractor.
- 8.22.3 The contractor understands and agrees that all persons performing work pursuant to this Contract are, for purposes of Workers' Compensation liability, solely employees of the contractor and not employees of the District. The contractor shall be solely liable and responsible for furnishing any and all Workers' Compensation benefits to any person as a result of any injuries arising from or connected with any

work performed by or on behalf of the contractor pursuant to this Contract.

8.22.4 The contractor shall adhere to the provisions stated in Paragraph 7.6 (Confidentiality).

8.23 Indemnification

The contractor shall indemnify, defend and hold harmless the County, its Special Districts, elected and appointed officers, employees, agents and volunteers (County Indemnitees) from and against any and all liability, including but not limited to demands, claims, actions, fees, costs and expenses (including attorney and expert witness fees), arising from and/or relating to this Contract, except for such loss or damage arising from the sole negligence or willful misconduct of the County indemnitees.

8.24 General Provisions for all Insurance Coverage

Without limiting Contractor's indemnification of District, and in the performance of this Contract and until all of its obligations pursuant to this Contract have been met, Contractor shall provide and maintain at its own expense insurance coverage satisfying the requirements specified in Paragraphs 8.24 and 8.25 of this Contract. These minimum insurance coverage terms, types and limits (the "Required Insurance") also are in addition to and separate from any other contract. The District in no way warrants that the Required Insurance is sufficient to protect the Contractor for liabilities which may arise from or relate to this Contract.

8.24.2 Evidence of Coverage and Notice to District

- 8.24.2.1 Certificate(s) of insurance coverage (Certificate) satisfactory to District, and a copy of an Additional Insured endorsement confirming County and its Agents (defined below) has been given Insured status under the Contractor's General Liability policy, shall be delivered to District at the address shown below and provided prior to commencing services under this Contract.
- 8.24.2.2 Renewal Certificates shall be provided to District not less than ten (10) days prior to

contractor's policy expiration dates. The District reserves the right to obtain complete, certified copies of any required contractor and/or subcontractor insurance policies at any time.

- 8.24.2.3 Certificates shall identify all Required Insurance coverage types and limits specified herein. reference this Contract by name or number, and be signed by an authorized representative of the insurer(s). The Insured party named on the Certificate shall match the name of the contractor identified as the contracting party in this Contract. Certificates shall provide the full name of each insurer providing coverage, its NAIC (National Association of Insurance identification number, Commissioners) its financial rating, the amounts of any policy deductibles or self-insured retentions exceeding fifty thousand dollars (\$50,000), and list any District required endorsement forms.
- 8.24.2.4 Neither the District's failure to obtain, nor the District's receipt of, or failure to object to a noncomplying insurance certificate or endorsement, or any other insurance documentation or information provided by the contractor, its insurance broker(s) and/or insurer(s), shall be construed as a waiver of any of the Required Insurance provisions.
- 8.24.2.5 Certificates and copies of any required endorsements shall be sent to:

<u>Consolidated Fire Protection District of</u> <u>Los Angeles County</u> <u>Materials Management Division/Contracts Section</u> <u>5801 S. Eastern Avenue, Suite 100</u> <u>Commerce, California 90040-4001</u>

8.24.2.6 Contractor also shall promptly report to District any injury or property damage accident or incident, including any injury to a contractor employee occurring on District property, and any loss, disappearance, destruction, misuse, or theft of District property, monies or securities entrusted to contractor. Contractor also shall promptly notify District of any third-party claim or suit filed against contractor or any of its subcontractors which arises from or relates to this Contract, and could result in the filing of a claim or lawsuit against contractor and/or County.

8.24.3 Additional Insured Status and Scope of Coverage

The County of Los Angeles, it's Special Districts, Elected Officials, Officers, Agents, employees and volunteers (collectively County and its Agents) shall be provided additional insured status under contractor's General Liability policy with respect to liability arising out of contractor's ongoing and completed operations performed on behalf of the District. County and its Agents additional insured status shall apply with respect to liability and defense of suits arising out of the contractor's acts or omissions, whether such liability is attributable to the contractor or to the District. The full policy limits and scope of protection also shall apply to the County and its Agents as an additional insured, even if they exceed the County's minimum Required Insurance specifications herein. Use of an automatic additional insured endorsement form is acceptable providing it satisfies the Required Insurance provisions herein.

8.24.4 Cancellation of or Changes in Insurance

Contractor shall provide District with, or contractor's insurance policies shall contain a provision that District shall receive, written notice of cancellation or any change in Required Insurance, including insurer, limits of coverage, term of coverage or policy period. The written notice shall be provided to District at least ten (10) days in advance of cancellation for non-payment of premium and thirty (30) days in advance for any other cancellation or policy change. Failure to provide written notice of cancellation or any change in Required Insurance may constitute a material breach of the Contract, in the sole discretion of the District, upon which the District may suspend or terminate this Contract.

8.24.5 Failure to Maintain Insurance

Contractor's failure to maintain or to provide acceptable evidence that it maintains the Required Insurance shall constitute a material breach of the Contract, upon which District immediately may withhold payments due to contractor, and/or suspend or terminate this Contract. District, at its sole discretion, may obtain damages from contractor resulting from said breach. Alternatively, the District may purchase the Required Insurance, and without further notice to contractor, deduct the premium cost from sums due to contractor or pursue contractor reimbursement.

8.24.6 Insurer Financial Ratings

Coverage shall be placed with insurers acceptable to the District with A.M. Best ratings of not less than A:VII unless otherwise approved by District.

8.24.7 Contractor's Insurance Shall Be Primary

Contractor's insurance policies, with respect to any claims related to this Contract, shall be primary with respect to all other sources of coverage available to contractor. Any District maintained insurance or self-insurance coverage shall be in excess of and not contribute to any contractor coverage.

8.24.8 Waivers of Subrogation

To the fullest extent permitted by law, the contractor hereby waives its rights and its insurer(s)' rights of recovery against County under all the Required Insurance for any loss arising from or relating to this Contract. The contractor shall require its insurers to execute any waiver of subrogation endorsements which may be necessary to effect such waiver.

8.24.9 Subcontractor Insurance Coverage Requirements

Contractor shall include all subcontractors as insureds under contractor's own policies, or shall provide District with each subcontractor's separate evidence of insurance coverage. Contractor shall be responsible for verifying each subcontractor complies with the Required Insurance provisions herein, and shall require that each subcontractor name the County and contractor as additional insureds on the subcontractor's General Liability policy. Contractor shall obtain County's prior review and approval of any subcontractor request for modification of the Required Insurance.

8.24.10 Deductibles and Self-Insured Retentions (SIRs)

Contractor's policies shall not obligate the County to pay any portion of any contractor deductible or SIR. The District retains the right to require contractor to reduce or eliminate policy deductibles and SIRs as respects the District, or to provide a bond guaranteeing contractor's payment of all deductibles and SIRs, including all related claims investigation, administration and defense expenses. Such bond shall be executed by a corporate surety licensed to transact business in the State of California.

8.24.11 Claims Made Coverage

If any part of the Required Insurance is written on a claims made basis, any policy retroactive date shall precede the effective date of this Contract. Contractor understands and agrees it shall maintain such coverage for a period of not less than three (3) years following Contract expiration, termination or cancellation.

8.24.12 Application of Excess Liability Coverage

Contractors may use a combination of primary, and excess insurance policies which provide coverage as broad as ("follow form" over) the underlying primary policies, to satisfy the Required Insurance provisions.

8.24.13 Separation of Insureds

All liability policies shall provide cross-liability coverage as would be afforded by the standard ISO (Insurance Services Office, Inc.) separation of insureds provision with no insured versus insured exclusions or limitations.

8.24.14 Alternative Risk Financing Programs

The District reserves the right to review, and then approve, Contractor use of self-insurance, risk retention groups, risk purchasing groups, pooling arrangements and captive insurance to satisfy the Required Insurance provisions. The County and its Agents shall be designated as an Additional Covered Party under any approved program.

8.24.15 District Review and Approval of Insurance Requirements

The District reserves the right to review and adjust the Required Insurance provisions, conditioned upon District's determination of changes in risk exposures.

8.25 Insurance Coverage

8.25.1 Commercial General Liability insurance (providing scope of coverage equivalent to ISO policy form CG 00 01), naming County and its Agents as an additional insured, with limits of not less than:

General Aggregate: \$2 million

Products/Completed Operations Aggregate: \$1 million

Personal and Advertising Injury: \$1 million

Each Occurrence: \$1 million

- **8.25.2** Automobile Liability insurance (providing scope of coverage equivalent to ISO policy form CA 00 01) with limits of not less than \$1 million for bodily injury and property damage, in combined or equivalent split limits, for each single accident. Insurance shall cover liability arising out of contractor's use of autos pursuant to this Contract, including owned, leased, hired, and/or non-owned autos, as each may be applicable.
- 8.25.3 Workers Compensation and Employers' Liability insurance or gualified self- insurance satisfying statutory requirements, which includes Employers' Liability coverage with limits of not less than \$1 million per accident. If Contractor will provide leased employees, or, is an employee leasing or temporary staffing firm or a professional employer organization (PEO), coverage also shall include an Alternate Employer Endorsement (providing scope of coverage equivalent to ISO policy form WC 00 03 01 A) naming the County as the Alternate Employer. The written notice shall be provided to District at least ten (10) days in advance of cancellation

for non-payment of premium and thirty (30) days in advance for any other cancellation or policy change. If applicable to Contractor's operations, coverage also shall be arranged to satisfy the requirements of any federal workers or workmen's compensation law or any federal occupational disease law.

8.25.4 Unique Insurance Coverage

8.25.4.1 **Professional Liability-Errors and Omissions**

Insurance covering Contractor's liability arising from or related to this Contract, with limits of not less than \$1 million per claim and \$2 million aggregate. Further, Contractor understands and agrees it shall maintain such coverage for a period of not less than three (3) years following this Agreement's expiration, termination or cancellation.

8.25.4.2 Privacy/Network Security (Cyber) Liability

coverage providing Insurance protection against liability for (1) privacy breaches [liability arising from the loss or disclosure of confidential information no matter how it occurs]; (2) system breach; (3) denial or loss of service; (4) introduction, implantation. spread of malicious or software code; (5) unauthorized access to or use of computer systems with limits of not less than \$2 million. No exclusion/restriction for unencrypted portable devices/media may be on the policy.

8.26 Liquidated Damages

8.26.1 If, in the judgment of the Fire Chief or his designee, the contractor is deemed to be non-compliant with the terms and obligations assumed hereby, the Fire Chief or his designee, at his/her option, in addition to, or in lieu of, other remedies provided herein, may withhold the entire monthly payment or deduct pro rata from the contractor's invoice for work not performed. A description of the work not performed and the amount to be withheld or deducted from payments to the contractor from the District, will be

forwarded to the contractor by the Fire Chief or his designee, in a written notice describing the reasons for said action.

- 8.26.2 If the Fire Chief or his designee, determines that there are deficiencies in the performance of this Contract that the Fire Chief or his designee, deems are correctable by the contractor over a certain time span, the Fire Chief or his designee, will provide a written notice to the contractor to correct the deficiency within specified time frames. Should the contractor fail to correct deficiencies within said time frame, the Fire Chief, or his designee, may: (a) Deduct from the contractor's payment, pro rata, those applicable portions of the Monthly Contract Sum; and/or (b) Deduct liquidated damages. The parties agree that it will be impracticable or extremely difficult to fix the extent of actual damages resulting from the failure of the contractor to correct a deficiency within the specified time frame. The parties hereby agree that under the current circumstances a reasonable estimate of such damages is one hundred dollars (\$100) per day per infraction, or as specified in the Exhibit 2 - Performance Requirements Summary (PRS) Chart, Statement of Work Exhibits, hereunder, and that the contractor shall be liable to the District for liquidated damages in said amount. Said amount shall be deducted from the District's payment to the contractor; and/or (c) Upon giving five (5) days notice to the contractor for failure to correct the deficiencies, the District may correct any and all deficiencies and the total costs incurred by the District for completion of the work by an alternate source, whether it be District forces or separate private contractor, will be deducted and forfeited from the payment to the contractor from the District, as determined by the District.
- 8.26.3 The action noted in Paragraph 8.26.2 shall not be construed as a penalty, but as adjustment of payment to the contractor to recover the District cost due to the failure of the contractor to complete or comply with the provisions of this Contract.
- 8.26.4 This Paragraph shall not, in any manner, restrict or limit the District's right to damages for any breach of this Contract provided by law or as specified in the PRS or Paragraph 8.26.2, and shall not, in any manner, restrict or limit the District's right to terminate this Contract as agreed to herein.

8.27 Most Favored Public Entity

If the contractor's prices decline, or should the contractor at any time during the term of this Contract provide the same goods or services under similar quantity and delivery conditions to the State of California or any county, municipality, or district of the State at prices below those set forth in this Contract, then such lower prices shall be immediately extended to the District.

8.28 Nondiscrimination and Affirmative Action

- 8.28.1 The contractor certifies and agrees that all persons employed by it, its affiliates, subsidiaries, or holding companies are and shall be treated equally without regard to or because of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, in compliance with all applicable Federal and State anti-discrimination laws and regulations.
- 8.28.2 The contractor shall certify to, and comply with, the provisions of Exhibit D Contractor's EEO Certification.
- 8.28.3 The contractor shall take affirmative action to ensure that applicants are employed, and that employees are treated during employment, without regard to race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, in compliance with all applicable Federal and State anti-discrimination laws and regulations. Such action shall include, but is not limited to: employment, upgrading, demotion, transfer, recruitment or recruitment advertising, layoff or termination, rates of pay or other forms of compensation, and selection for training, including apprenticeship.
- 8.28.4 The contractor certifies and agrees that it will deal with its subcontractors, bidders, or vendors without regard to or because of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation.
- 8.28.5 The contractor certifies and agrees that it, its affiliates, subsidiaries, or holding companies shall comply with all applicable Federal and State laws and regulations to the end that no person shall, on the grounds of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, be

excluded from participation in, be denied the benefits of, or be otherwise subjected to discrimination under this Contract or under any project, program, or activity supported by this Contract.

- 8.28.6 The contractor shall allow District representatives access to the contractor's employment records during regular business hours to verify compliance with the provisions of this Paragraph 8.28 Nondiscrimination and Affirmative Action when so requested by the District.
- 8.28.7 If the District finds that any provisions of this Paragraph 8.28 (Nondiscrimination and Affirmative Action) have been violated, such violation shall constitute a material breach of this Contract upon which the District may terminate or suspend this Contract. While the District reserves the right to determine independently that the anti-discrimination provisions of this Contract have been violated, in addition, a determination by the California Fair Employment and Housing Commission or the Federal Equal Employment Opportunity Commission that the contractor has violated Federal or State anti-discrimination laws or regulations shall constitute a finding by the District that the contractor has violated the anti-discrimination provisions of this Contract.
- 8.28.8 The parties agree that in the event the contractor violates any of the anti-discrimination provisions of this Contract, the District shall, at its sole option, be entitled to the sum of five hundred dollars (\$500) for each such violation pursuant to California Civil Code Section 1671 as liquidated damages in lieu of terminating or suspending this Contract.

8.29 Non Exclusivity

Nothing herein is intended nor shall be construed as creating any exclusive arrangement with the contractor. This Contract shall not restrict District from acquiring similar, equal or like goods and/or services from other entities or sources.

8.30 Notice of Delays

Except as otherwise provided under this Contract, when either party has knowledge that any actual or potential situation is delaying or threatens to delay the timely performance of this Contract, that party shall, within one (1) business day, give notice thereof, including all relevant information with respect thereto, to the other party.

8.31 Notice of Disputes

The contractor shall bring to the attention of the District's Project Manager and/or District's Project Director any dispute between the District and the contractor regarding the performance of services as stated in this Contract. If the District's Project Manager or District's Project Director is not able to resolve the dispute, the Fire Chief or his designee shall resolve it.

8.32 Notice to Employees Regarding the Federal Earned Income Credit

The contractor shall notify its employees, and shall require each subcontractor to notify its employees, that they may be eligible for the Federal Earned Income Credit under the federal income tax laws. Such notice shall be provided in accordance with the requirements set forth in Internal Revenue Service Notice No. 1015.

8.33 Notice to Employees Regarding the Safely Surrendered Baby Law

The contractor shall notify and provide to its employees, and shall require each subcontractor to notify and provide to its employees, information regarding the Safely Surrendered Baby Law, its implementation in Los Angeles County, and where and how to safely surrender a baby. The information is set forth in Exhibit I, Safely Surrendered Baby Law of this Contract. Additional information is available at <u>www.babysafela.org</u>.

8.34 Notices

All notices or demands required or permitted to be given or made under this Contract shall be in writing and shall be hand delivered with signed receipt or mailed by first-class registered or certified mail, postage prepaid, addressed to the parties as identified in Exhibits E -District's Administration and F - Contractor's Administration. Addresses may be changed by either party giving ten (10) days prior written notice thereof to the other party. The Fire Chief or his designee shall have the authority to issue all notices or demands required or permitted by the District under this Contract.

8.35 Prohibition Against Inducement or Persuasion

Notwithstanding the above, the contractor and the District agree that, during the term of this Contract and for a period of one year thereafter, neither party shall in any way intentionally induce or persuade any employee of one party to become an employee or agent of the other party. No bar exists against any hiring action initiated through a public announcement.

8.36 Public Records Act

- 8.36.1 Any documents submitted by the contractor; all information obtained in connection with the District's right to audit and inspect the contractor's documents, books, and accounting records pursuant to Paragraph 8.38 (Record Retention and Inspection-Audit Settlement) of this Contract; as well as those documents which were required to be submitted in response to the Request for Proposals (RFP) used in the solicitation process for this Contract, become the exclusive property of the District. All such documents become a matter of public record and shall be regarded as public records. Exceptions will be those elements in the California Government Code Section 6250 et seq. (Public Records Act) and which are marked "trade secret," "confidential," or "proprietary." The District shall not in any way be liable or responsible for the disclosure of any such records including. without limitation, those so marked, if disclosure is required by law, or by an order issued by a court of competent jurisdiction.
- 8.36.2 In the event the District is required to defend an action on a Public Records Act request for any of the aforementioned documents, information, books, records, and/or contents of a proposal marked "trade secret," "confidential," or "proprietary," the contractor agrees to defend and indemnify the District from all costs and expenses, including reasonable attorney's fees, in action or liability arising under the Public Records Act.

8.37 Publicity

8.37.1 The contractor shall not disclose any details in connection with this Contract to any person or entity except as may be otherwise provided hereunder or required by law. However, in recognizing the contractor's need to identify its services and related clients to sustain itself, the District shall not inhibit the contractor from publishing its role under this Contract within the following conditions:

- 8.37.1.1 The contractor shall develop all publicity material in a professional manner; and
- 8.37.1.2 During the term of this Contract, the contractor shall not, and shall not authorize another to, publish or disseminate any commercial advertisements, press releases, feature articles, or other materials using the name of the District without the prior written consent of the District's Project Director. The District shall not unreasonably withhold written consent.
- 8.37.2 The contractor may, without the prior written consent of District, indicate in its proposals and sales materials that it has been awarded this Contract with the County of Los Angeles, provided that the requirements of this Paragraph 8.37 (Publicity) shall apply.

8.38 Record Retention and Inspection-Audit Settlement

8.38.1 The Contractor shall maintain accurate and complete financial records of its activities and operations relating to this Contract in accordance with generally accepted accounting principles, and which meet the requirements for contract accounting described in Auditor-Controller Contract Accounting and Administration Handbook. The Contractor shall also maintain accurate and complete employment and other records relating to its performance of this Contract. The contractor agrees that the District, or its authorized representatives, shall have access to and the right to examine, audit, excerpt, copy, or transcribe any pertinent transaction, activity, or record relating to this Contract. All such material, including, but not limited to, all financial records, bank statements, cancelled checks or other proof of payment, timecards, sign-in/sign-out sheets and other time and employment records, and proprietary data and information, shall be kept and maintained by the contractor and shall be made available to the District during the term of this Contract and for a period of five (5) years thereafter unless the District's written permission is given to dispose of any such material prior to such time. All such material shall be maintained by the contractor at a location in Los Angeles County, provided that if any such material is located outside Los Angeles County, then, at the District's option, the contractor shall pay the District for travel, per diem, and other costs incurred by the District to examine, audit, excerpt, copy, or transcribe such material at such other location.

- 8.38.2 In the event that an audit of the contractor is conducted specifically regarding this Contract by any Federal or State auditor, or by any auditor or accountant employed by the contractor or otherwise, then the contractor shall file a copy of such audit report with the County's Auditor-Controller within thirty (30) days of the contractor's receipt thereof, unless otherwise provided by applicable Federal or State law or under this Contract. Subject to applicable law, the District shall make a reasonable effort to maintain the confidentiality of such audit report(s). Failure on the part of the contractor to comply with any of the provisions of this subparagraph 8.38 shall constitute a material breach of this Contract.
- 8.38.3 If, at any time during the term of this Contract or within five (5) years after the expiration or termination of this Contract, representatives of the District conduct an audit of the contractor regarding the work performed under this Contract, and if such audit finds that the District's dollar liability for any such work is less than payments made by the District to the contractor, then the difference shall be either: a) repaid by the contractor to the District by cash payment upon demand or b) at the sole option of the County's Auditor-Controller, deducted from any amounts due to the contractor from the District, whether under this Contract or otherwise. If such audit finds that the District's dollar liability for such work is more than the payments made by the District to the contractor, then the difference shall be paid to the contractor by the District by cash payment, provided that in no event shall the District's maximum obligation for this Contract exceed the funds appropriated by the District for the purpose of this Contract.

8.39 Recycled Bond Paper

Consistent with the Board of Supervisors' policy to reduce the amount of solid waste deposited at the County landfills, the contractor agrees to use recycled-content paper to the maximum extent possible on this Contract.

8.40 Subcontracting

- 8.40.1 The requirements of this Contract may not be subcontracted by the contractor **without the advance written approval of the District**. Any attempt by the contractor to subcontract without the prior consent of the District may be deemed a material breach of this Contract.
- 8.40.2 If the contractor desires to subcontract, the contractor shall provide the following information promptly at the District's request:
 - 8.40.2.1 A description of the work to be performed by the subcontractor;
 - 8.40.2.2 A draft copy of the proposed subcontract; and
 - 8.40.2.3 Other pertinent information and/or certifications requested by the District.
- 8.40.3 The contractor shall indemnify, defend, and hold the County harmless with respect to the activities of each and every subcontractor in the same manner and to the same degree as if such subcontractor(s) were the contractor employees.
- 8.40.4 The contractor shall remain fully responsible for all performances required of it under this Contract, including those that the contractor has determined to subcontract, notwithstanding the District's approval of the contractor's proposed subcontract.
- 8.40.5 The District's consent to subcontract shall not waive the District's right to prior and continuing approval of any and all personnel, including subcontractor employees, providing services under this Contract. The contractor is responsible to notify its subcontractors of this District right.
- 8.40.6 The District's Project Director is authorized to act for and on behalf of the District with respect to approval of any subcontract and subcontractor employees. After approval of the subcontract by the District, contractor shall forward a fully executed subcontract to the District for their files.
- 8.40.7 The contractor shall be solely liable and responsible for all payments or other compensation to all subcontractors and their officers, employees, agents, and successors in interest

arising through services performed hereunder, notwithstanding the District's consent to subcontract.

8.40.8 The contractor shall obtain certificates of insurance, which establish that the subcontractor maintains all the programs of insurance required by the District from each approved subcontractor. Before any subcontractor employee may perform any work hereunder, contractor shall ensure delivery of all such documents to:

Consolidated Fire Protection District of Los Angeles County Materials Management Division / Contracts Section 5801 S. Eastern Avenue, Suite 100 Commerce, California 90040-4001

8.41 Termination for Breach of Warranty to Maintain Compliance with County's Child Support Compliance Program

Failure of the contractor to maintain compliance with the requirements set forth in Paragraph 8.14 (Contractor's Warranty of Adherence to County's Child Support Compliance Program) shall constitute default under this Contract. Without limiting the rights and remedies available to the District under any other provision of this Contract, failure of the contractor to cure such default within ninety (90) calendar days of written notice shall be grounds upon which the District may terminate this Contract pursuant to Paragraph 8.43 (Termination for Default) and pursue debarment of the contractor, pursuant to County Code Chapter 2.202.

8.42 Termination for Convenience

- 8.42.1 This Contract may be terminated, in whole or in part, from time to time, when such action is deemed by the District, in its sole discretion, to be in its best interest. Termination of work hereunder shall be effected by notice of termination to the contractor specifying the extent to which performance of work is terminated and the date upon which such termination becomes effective. The date upon which such termination becomes effective shall be no less than ten (10) days after the notice is sent.
- 8.42.2 After receipt of a notice of termination and except as otherwise directed by the District, the contractor shall:
 - 8.42.2.1 Stop work under this Contract on the date and to the extent specified in such notice, and

- 8.42.2.2 Complete performance of such part of the work as shall not have been terminated by such notice.
- 8.42.3 All material including books, records, documents, or other evidence bearing on the costs and expenses of the contractor under this Contract shall be maintained by the contractor in accordance with Paragraph 8.38 (Record Retention and Inspection-Audit Settlement).

8.43 Termination for Default

- 8.43.1 The District may, by written notice to the contractor, terminate the whole or any part of this Contract, if, in the judgment of District's Project Director:
 - 8.43.1.1 Contractor has materially breached this Contract; or
 - 8.43.1.2 Contractor fails to timely provide and/or satisfactorily perform any task, deliverable, service, or other work required either under this Contract; or
 - 8.43.1.3 Contractor fails to demonstrate a high probability of timely fulfillment of performance requirements under this Contract, or of any obligations of this Contract and in either case, fails to demonstrate convincing progress toward a cure within five (5) working days (or such longer period as the District may authorize in writing) after receipt of written notice from the District specifying such failure.
- 8.43.2 In the event the District terminates this Contract in whole or in part as provided in Paragraph 8.43.1, the District may procure, upon such terms and in such manner as the District may deem appropriate, goods and services similar to those so terminated. The contractor shall be liable to the District for any and all excess costs incurred by the District, as determined by the District, for such similar goods and services. The contractor shall continue the performance of this Contract to the extent not terminated under the provisions of this paragraph.
- 8.43.3 Except with respect to defaults of any subcontractor, the contractor shall not be liable for any such excess costs of

the type identified in Paragraph 8.43.2 if its failure to perform this Contract arises out of causes beyond the control and without the fault or negligence of the contractor. Such causes may include, but are not limited to: acts of God or of the public enemy, acts of the District in either its sovereign or contractual capacity, acts of Federal or State governments in their sovereign capacities, fires, floods, restrictions, epidemics, quarantine strikes. freiaht embargoes, and unusually severe weather; but in every case, the failure to perform must be beyond the control and without the fault or negligence of the contractor. If the failure to perform is caused by the default of a subcontractor, and if such default arises out of causes beyond the control of both the contractor and subcontractor, and without the fault or negligence of either of them, the contractor shall not be liable for any such excess costs for failure to perform, unless the goods or services to be furnished by the subcontractor were obtainable from other sources in sufficient time to permit the contractor to meet the required performance schedule. As used in this paragraph, "subcontractor(s)" the term means subcontractor(s) at any tier.

- 8.43.4 If, after the District has given notice of termination under the provisions of Paragraph 8.43 (Termination for Default) it is determined by the District that the contractor was not in default under the provisions of Paragraph 8.43 (Termination for Default) or that the default was excusable under the provisions of subparagraph 8.43.3, the rights and obligations of the parties shall be the same as if the notice of termination had been issued pursuant to Paragraph 8.42 (Termination for Convenience).
- 8.43.5 The rights and remedies of the District provided in this Paragraph 8.43 (Termination for Default) shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.44 Termination for Improper Consideration

8.44.1 The District may, by written notice to the contractor, immediately terminate the right of the contractor to proceed under this Contract if it is found that consideration, in any form, was offered or given by the contractor, either directly or through an intermediary, to any District officer, employee, or agent with the intent of securing this Contract or securing favorable treatment with respect to the award, amendment, or extension of this Contract or the making of any determinations with respect to the contractor's performance pursuant to this Contract. In the event of such termination, the District shall be entitled to pursue the same remedies against the contractor as it could pursue in the event of default by the contractor.

- 8.44.2 The contractor shall immediately report any attempt by a District officer or employee to solicit such improper consideration. The report shall be made either to the District manager charged with the supervision of the employee or to the County Auditor-Controller's Employee Fraud Hotline at (800) 544-6861.
- 8.44.3 Among other items, such improper consideration may take the form of cash, discounts, services, the provision of travel or entertainment, or tangible gifts.

8.45 Termination for Insolvency

- 8.45.1 The District may terminate this Contract forthwith in the event of the occurrence of any of the following:
 - 8.45.1.1 Insolvency of the contractor. The contractor shall be deemed to be insolvent if it has ceased to pay its debts for at least sixty (60) days in the ordinary course of business or cannot pay its debts as they become due, whether or not a petition has been filed under the Federal Bankruptcy Code and whether or not the contractor is insolvent within the meaning of the Federal Bankruptcy Code;
 - 8.45.1.2 The filing of a voluntary or involuntary petition regarding the contractor under the Federal Bankruptcy Code;
 - 8.45.1.3 The appointment of a Receiver or Trustee for the contractor; or
 - 8.45.1.4 The execution by the contractor of a general assignment for the benefit of creditors.
- 8.45.2 The rights and remedies of the District provided in this Paragraph 8.45 (Termination for Insolvency) shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.46 Termination for Non-Adherence of County Lobbyist Ordinance

The contractor, and each County Lobbyist or County Lobbying firm as defined in County Code Section 2.160.010 retained by the contractor, shall fully comply with the County's Lobbyist Ordinance, County Code Chapter 2.160. Failure on the part of the contractor or any County Lobbyist or County Lobbying firm retained by the contractor to fully comply with the County's Lobbyist Ordinance shall constitute a material breach of this Contract, upon which the District may in its sole discretion, immediately terminate or suspend this Contract.

8.47 Termination for Non-Appropriation of Funds

Notwithstanding any other provision of this Contract, the District shall not be obligated for the contractor's performance hereunder or by any provision of this Contract during any of the District's future fiscal years unless and until the County's Board of Supervisors appropriates funds for this Contract in the District's Budget for each such future fiscal year. In the event that funds are not appropriated for this Contract, then this Contract shall terminate as of June 30 of the last fiscal year for which funds were appropriated. The District shall notify the contractor in writing of any such non-allocation of funds at the earliest possible date.

8.48 Validity

If any provision of this Contract or the application thereof to any person or circumstance is held invalid, the remainder of this Contract and the application of such provision to other persons or circumstances shall not be affected thereby.

8.49 Waiver

No waiver by the District of any breach of any provision of this Contract shall constitute a waiver of any other breach or of such provision. Failure of the District to enforce at any time, or from time to time, any provision of this Contract shall not be construed as a waiver thereof. The rights and remedies set forth in this paragraph 8.49 shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.50 Warranty Against Contingent Fees

8.50.1 The contractor warrants that no person or selling agency has been employed or retained to solicit or secure this

Contract upon any Contract or understanding for a commission, percentage, brokerage, or contingent fee, excepting bona fide employees or bona fide established commercial or selling agencies maintained by the contractor for the purpose of securing business.

8.50.2 For breach of this warranty, the District shall have the right to terminate this Contract and, at its sole discretion, deduct from the Contract price or consideration, or otherwise recover, the full amount of such commission, percentage, brokerage, or contingent fee.

8.51 Warranty of Compliance with County's Defaulted Property Tax Reduction Program

Contractor acknowledges that County has established a goal of ensuring that all individuals and businesses that benefit financially from County through contract are current in paying their property tax obligations (secured and unsecured roll) in order to mitigate the economic burden otherwise imposed upon County and its taxpayers.

Unless contractor qualifies for an exemption or exclusion, contractor warrants and certifies that to the best of its knowledge it is now in compliance, and during the term of this contract will maintain compliance, with Los Angeles County Code Chapter 2.206.

8.52 Termination for Breach of Warranty to Maintain Compliance with County's Defaulted Property Tax Reduction Program

Failure of contractor to maintain compliance with the requirements set forth in Paragraph 8.51 "Warranty of Compliance with County's Defaulted Property Tax Reduction Program" shall constitute default under this contract. Without limiting the rights and remedies available to County under any other provision of this contract, failure of contractor to cure such default within ten (10) days of notice shall be grounds upon which District may terminate this contract and/or pursue debarment of contractor, pursuant to County Code Chapter 2.206.

8.53 Time Off for Voting

The contractor shall notify its employees, and shall require each subcontractor to notify and provide to its employees, information regarding the time off for voting law (Elections Code Section 14000). Not less than ten (10) days before every statewide election, every contractor and subcontractors shall keep posted conspicuously at the place of work, if practicable, or elsewhere where it can be seen as employees come or go to their place of work, a notice setting forth the provisions of Section 14000.

8.54 Compliance with County's Zero Tolerance Policy on Human Trafficking

Contractor acknowledges that the County has established a Zero Tolerance Policy on Human Trafficking prohibiting contractors from engaging in human trafficking.

If a Contractor or member of Contractor's staff is convicted of a human trafficking offense, the District shall require that the Contractor or member of Contractor's staff be removed immediately from performing services under the Contract. District will not be under any obligation to disclose confidential information regarding the offenses other than those required by law.

Disqualification of any member of Contractor's staff pursuant to this paragraph shall not relieve Contractor of its obligation to complete all work in accordance with the terms and conditions of this Contract.

8.55 Intentionally Omitted

8.56 Compliance with Fair Chance Employment Practices

Contractor shall comply with fair chance employment hiring practices set forth in California Government Code Section 12952, Employment Discrimination: Conviction History. Contractor's violation of this paragraph of the Contract may constitute a material breach of the Contract. In the event of such material breach, District may, in its sole discretion, terminate the Contract.

8.57 Compliance with the County Policy of Equity

The contractor acknowledges that the County takes its commitment to preserving the dignity and professionalism of the workplace very seriously, as set forth in the County Policy of Equity (CPOE) (<u>https://ceop.lacounty.gov/</u>). The contractor further acknowledges that the County strives to provide a workplace free from discrimination, harassment, retaliation and inappropriate conduct based on a protected characteristic, and which may violate the CPOE. The contractor, its employees and subcontractors acknowledge and certify receipt and understanding of the CPOE. Failure of the contractor, its employees or its subcontractors to uphold the County's expectations of a workplace free from harassment and discrimination, including inappropriate conduct based on a protected characteristic, may subject the contractor to termination of contractual agreements as well as civil liability.

8.58 **Prohibition from Participation in Future Solicitation(s)**

A Proposer, or a Contractor or its subsidiary or Subcontractor ("Proposer/Contractor"), is prohibited from submitting a bid or proposal in a County solicitation if the Proposer/Contractor has provided advice or consultation for the solicitation. A Proposer/Contractor is also prohibited from submitting a bid or proposal in a County solicitation if the Proposer/Contractor has developed or prepared any of the solicitation materials on behalf of the County. A violation of this provision shall result in the disqualification of the Contractor/Proposer from participation in the County solicitation or the termination or cancellation of any resultant County contract. This provision shall survive the expiration, or other termination of this Agreement.

9 UNIQUE TERMS AND CONDITIONS

9.1 Intentionally Omitted

9.2 Health Insurance Portability and Accountability Act of 1996 (HIPAA)

- 9.2.1 Contractor expressly acknowledges and agrees that the provision of services under this Agreement does not require or permit access by Contractor or any of its officers, employees, or agents, to any patient medical records/patient information. Accordingly, Contractor shall instruct its officers, employees, and agents that they are not to pursue, or gain access to, patient medical records/patient information for any reason whatsoever.
- 9.2.2 Notwithstanding the forgoing, the parties acknowledge that in the course of the provision of services hereunder, Contractor or its officers, employees, and agents, may have inadvertent access to patient medical records/patient information. Contractor understands and agrees that neither it nor its officers, employees, or agents, are to take advantage of such access for any purpose whatsoever.
- 9.2.3 Additionally, in the event of such inadvertent access, Contractor and its officers, employees, and agents, shall

maintain the confidentiality of any information obtained and shall notify Director that such access has been gained immediately, or upon the first reasonable opportunity to do so. In the event of any access, whether inadvertent or intentional, Contractor shall indemnify, defend, and hold harmless County, its officers, employees, and agents, from and against any and all liability, including but not limited to, actions, claims, costs, demands, expenses, and fees (including attorney and expert witness fees) arising from or connected with Contractor's or its officers', employees', or agents', access to patient medical records/patient information. Contractor agrees to provide appropriate training to its employees regarding their obligations as described hereinabove.

9.3 Intentionally Omitted

9.4 Intentionally Omitted

9.5 Intentionally Omitted

9.6 Data Destruction

Contractor(s) and Vendor(s) that have maintained, processed, or stored the County of Los Angeles' ("County") data and/or information, implied or expressed, have the sole responsibility to certify that the data and information have been appropriately destroyed consistent with the National Institute of Standards and Technology (NIST) Special Publication SP 800-88 titled *Guidelines for Media Sanitization*. Available at:

http://csrc.nist.gov/publications/PubsDrafts.html#SP-800-88 Rev.%201

The data and/or information may be stored on purchased, leased, or rented electronic storage equipment (e.g., printers, hard drives) and electronic devices (e.g., servers, workstations) that are geographically located within the County, or external to the County's boundaries. The District must receive within ten (10) business days, a signed document from Contractor(s) and Vendor(s) that certifies and validates the data and information were placed in one or more of the following stored states: unusable, unreadable, and indecipherable.

Vendor shall certify that any District data stored on purchased, leased, or rented electronic storage equipment and electronic devices, including, but not limited to printers, hard drives, servers, and/or workstations are destroyed consistent with the current National Institute of Standard and Technology (NIST) Special Publication SP-800-88, *Guidelines for Media Sanitization*. Vendor shall provide District with written certification, within ten (10) business days of removal of any electronic storage equipment and devices that validates that any and all District data was destroyed and is unusable, unreadable, and/or undecipherable.

9.7 Intentionally Omitted

9.8 Intentionally Omitted

9.9 Intentionally Omitted

9.10 Mandatory Requirement to Register on County's WebVen

As a pre-condition to the award of this Contract, Contractor represents and warrants that it has registered in Los Angeles County's vendor registration system (hereafter "WebVen"). The WebVen contains Contractor's business profile and identifies the goods/services being provided by Contractor. Contractor shall ensure that it updates its vendor profile whenever changes occur to Contractor's operations by accessing the WebVen site located on-line at: <u>http://camisvr.co.la.ca.us/webven</u> County shall use the data obtained from Contractor's WebVen profile to ensure that Contractor's legal name, as reflected in its WebVen profile, shall be used in Contract documents).

9.11 Limitation on Corporate Acts

- 9.11.1 Contractor shall not amend its articles of incorporation or bylaws, move to dissolve or transfer any assets obtained using Contract Funds, or take any other steps which may materially affect the performance of this Contract without first notifying the District in writing no less than thirty (30) days prior to said action. Contractor shall notify the District's Contract Administrator immediately in writing of any change in Contractor's corporate name.
- 9.11.2 If, in the District's sole discretion, the steps taken by Contractor are determined to materially affect Contractor's performance of this Contract, the District may, at its sole discretion, take any (or all) of the following actions:

- 9.11.2.1 Require Contractor to remedy the areas that affect Contractor's ability to perform its obligations under this Contract.
- 9.11.2.2 Suspend Contractor from performing (and receiving payment for) Contract tasks until a remedy has been reached.
- 9.11.2.3 Terminate this Contract pursuant to Paragraph 8.43 (Termination for Default).

9.12 Modifications

This Contract fully expresses the agreement of the parties. Any modification to this Contract must be by means of a separate written document approved by the District. No oral conversation between any officer, employee or agent of the parties shall modify or otherwise amend this Contract in any way.

9.13 Remedies of Non-Compliance

Contractor agrees to comply with the requirements set forth in the entirety of this Contract as well as the requirements contained in any applicable directives, notices, guidelines and instructions used by the District. Contractor's failure to comply with such requirements shall subject Contractor to remedies which are available under this Contract and as provided by law. These remedies include but are not limited to the following: suspension of payment(s); suspension of Service(s); assessment and collection of liquidated damages; de-obligation of Contract Funds (for purposes of this Contract, de-obligation is the partial or full removal of Contract Funds from Contractor); debarment; and/or termination of Contract. The District shall have the sole discretion to determine which remedy(ies) will be applied as a result of Contractor's noncompliance.

9.14 Suspension

9.14.1 Contractor may be placed on suspension if District determines that Contractor is not in compliance with any Service, Work, task, deliverable or requirement outlined in this Contract and/or Contractor has demonstrated a consistent and significant lack of achievement of the Contract goals (including, but not limited to, meeting the requirements for work performance, the Pricing Sheet, staffing, administration, etc.). The District shall notify the

Contractor in writing in the event that Contractor is placed on suspension.

- 9.14.2 Suspension as used herein shall mean a specified period of time (as determined by the District) during which the District shall withhold payment from Contractor. During the suspension, Contractor has a continuing obligation to remedy the areas of non-compliance which have been identified by the District or its duly authorized representative(s). The District shall monitor Contractor's adherence to such remedy(ies) during the suspension period. When applicable, the District may also provide the Contractor with a written determination stating whether or not the Contractor may continue to provide non-suspended Services, if any, during the suspension period.
- 9.14.3 District's written notice of suspension shall set forth the conditions of Contractor's non-compliance as well as the period in which Contractor must correct noted deficiencies. In response to the notice of suspension, Contractor shall submit a written Corrective Action Plan to the District's Contract Manager within ten (10) days of the date indicated on the notice from the District. Contractor's Corrective Action Plan shall address all of the deficiencies noted by the District.
- 9.14.4 The District shall review Contractor's Corrective Action Plan, and will determine whether it meets the requirements for District's approval. The District reserves the right to suspend/deduct payments for or to terminate all or any part of this Contract (and/or any Contractor's other contracts with the District) when Contractor submits a Corrective Action Plan that is not acceptable to the District.
- 9.14.5 Contractor shall implement the Corrective Action Plan upon receiving District's final written approval of the Corrective Action Plan. Contractor's failure to comply with an approved Corrective Action Plan will be cause for material breach of Contract upon which the District may pursue the remedies for default of Contract.

9.15 Transition of Contract Services

9.15.1 Completion of Contract

9.15.1.1 Within sixty (60) calendar days prior to the expiration of this Contract (or shorter time period if notified in writing by District), Contractor shall allow the District or a newly selected contractor a transition period for orientation purposes and the orderly transition of Contractor's current Services without additional cost to the District. Contractor shall continue to provide Services timely and accurately so that the Services are current at the expiration of this Contract.

9.15.2 Transition Plan

- 9.15.2.2 If this Contract (or any part thereof) is pursuant to any terminated of the termination provisions outlined herein or if it expires pursuant to Paragraph 4.0 (Term of Contract), Contractor shall provide a Transition Plan to the District. Contractor shall submit said Transition Plan to the District's Contract Administrator within the timeframe designated by the District in the notice of termination or Contractor shall submit it at least sixty (60) days prior to the expiration of this Contract as noted in Paragraph 4.0 (Term of Contract).
- 9.15.2.3 The District shall review Contractor's Transition Plan and will determine whether it meets the requirements for District's approval. The District reserves the right to suspend/deduct payments under this Contract and/or under any of Contractor's other contracts with the District when Contractor submits a Transition Plan that is not acceptable to the District. Contractor shall adhere to the Transition Plan which, at a minimum, shall include all of the elements outlined below.

9.15.3 Elements of the Transition Plan

- 9.15.3.1 Contractor's method to communicate with other organizations that can assist in locating alternative Services.
- 9.15.3.2 Contractor's plan to assure for the provision of adequate staff to provide continued Services through the remaining term of this Contract.
- 9.15.3.3 Any additional information which may be necessary to ensure Contract Services are being provided.

9.15.4 Implementation of the Transition Plan

Contractor shall implement the Transition Plan that is approved by the District. Contractor's failure to provide and/or implement the Transition Plan as prescribed herein shall mean that the District will provide Contractor a Transition Plan and Contractor will implement the Transition Plan provided by the District. The District will monitor Contractor's progress in carrying out all elements of the Transition Plan. IN WITNESS WHEREOF, the Board of Supervisors of the County of Los Angeles has caused this Contract to be executed by the Fire Chief of the Consolidated Fire Protection District of Los Angeles County (or designee) and approved by County Counsel, and Contractor has caused this Contract to be executed in its behalf by its duly authorized officer, this 1st day of July, 2021.

CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY

Ву _____

Fire Chief

By <u>22nd Century Technologies, Inc.</u> Contractor

Signed: _____

Printed: Eva Gaddis-McKnight

Title: Administrator

APPROVED AS TO FORM:

RODRIGO CASTRO-SILVA County Counsel

By_

Senior Deputy County Counsel

ATTACHMENT A

CONTRACT NO.: _____

CONTRACT



BY AND BETWEEN

CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY

AND

CATHYJON ENTERPRISES, INC. DBA HB STAFFING

FOR

TEMPORARY PERSONNEL SERVICES

CONTRACT PROVISIONS TABLE OF CONTENTS

	GRAPI		PAGE
		EXHIBITS	
		ICABLE DOCUMENTS	
ו 2		ITIONS	-
3		<pre></pre>	
4	_	OF CONTRACT	-
5		RACT SUM	
•	5.1	Maximum Contract Sum	-
	••••	Written Approval for Reimbursement	
	5.3	Notification of 75% of Total Contract Sum	
	5.4	No Payment for Services Provided Following Expiration-Termination of Contract	
	5.5	Invoices and Payments	7
	5.6	Intentionally Omitted	9
	5.7 5.8	Default Method of Payment: Direct Deposit or Electronic Funds Transf Travel, Transportation, and Parking	
6	ADMI 6.1	NISTRATION OF CONTRACT- DISTRICT District Administration	
	6.2	District's Project Director	10
	6.3	District's Project Manager	10
	6.4	District's Contract Administrator	11
7		NISTRATION OF CONTRACT-CONTRACTOR	11
	7.1	Contractor Administration	11
	7.2	Contractor's Project Manager	11
	7.3	Approval of Contractor's Staff	12
	7.4	Contractor's Staff Identification	12
	7.5	Background and Security Investigations	12
	7.6	Confidentiality	13
8	STAN	DARD TERMS AND CONDITIONS	14
	8.1	Amendments	14
	8.2	Assignment and Delegation/Mergers or Acquisitions	15
	8.3	Authorization Warranty	16
	8.4	Budget Reductions	16
	8.5	Complaints	16
	8.6	Compliance with Applicable Law	17

8.7	Compliance with Civil Rights Laws	. 18
8.8	Compliance with the County's Jury Service Program	. 18
8.9	Conflict of Interest	. 20
8.10	Consideration of Hiring County Employees Targeted for Layoff or Re- Employment List	. 21
8.11	Consideration of Hiring GAIN-GROW Participants	. 21
8.12	Contractor Responsibility and Debarment	. 22
8.13	Contractor's Acknowledgement of County's Commitment to Safely Surrendered Baby Law	. 24
8.14	Contractor's Warranty of Adherence to County's Child Support	
	Compliance Program	. 25
8.15	District's Quality Assurance Plan	. 25
8.16	Damage to District Facilities, Buildings or Grounds	. 26
8.17	Employment Eligibility Verification	. 26
8.18	Counterparts and Electronic Signatures and Representations	. 27
8.19	Fair Labor Standards	. 27
8.20	Force Majeure	. 27
8.21	Governing Law, Jurisdiction, and Venue	. 28
8.22	Independent Contractor Status	. 28
8.23	Indemnification	. 29
8.24	General Provisions for all Insurance Coverage	. 29
8.25	Insurance Coverage	. 34
8.26	Liquidated Damages	. 36
8.27	Most Favored Public Entity	. 37
8.28	Nondiscrimination and Affirmative Action	. 37
8.29	Non Exclusivity	. 39
8.30	Notice of Delays	. 39
8.31	Notice of Disputes	. 39
8.32	Notice to Employees Regarding the Federal Earned Income Credit	. 39
8.33	Notice to Employees Regarding the Safely Surrendered Baby Law	. 39
8.34	Notices	. 40
8.35	Prohibition Against Inducement or Persuasion	. 40
8.36	Public Records Act	. 40
8.37	Publicity	. 41
8.38	Record Retention and Inspection-Audit Settlement	. 41
8.39	Recycled Bond Paper	. 43
8.40	Subcontracting	. 43

8.41	Termination for Breach of Warranty to Maintain Compliance with County's Child Support Compliance Program	. 44
8.42	Termination for Convenience	. 44
8.43	Termination for Default	. 45
8.44	Termination for Improper Consideration	. 47
8.45	Termination for Insolvency	. 47
8.46	Termination for Non-Adherence of County Lobbyist Ordinance	. 48
8.47	Termination for Non-Appropriation of Funds	. 48
8.48	Validity	. 48
8.49	Waiver	. 49
8.50	Warranty Against Contingent Fees	. 49
8.51	Warranty of Compliance with County's Defaulted Property Tax	
	Reduction Program	. 49
8.52	Termination for Breach of Warranty to Maintain Compliance with	
	County's Defaulted Property Tax Reduction Program	. 50
8.53	Time off for Voting	. 50
8.54	Compliance with County's Zero Tolerance Policy on Human Trafficking	. 50
8.55	Intentionally Omitted	. 50
8.56	Compliance with Fair Chance Employment Practices	
8.57	Compliance with the County Policy of Equity	. 51
8.58	Prohibition from Participation in Future Solicitation(s)	. 51
UNIQ	UE TERMS AND CONDITIONS	
9.1	Intentionally Omitted	. 52
9.2	Health Insurance Portability and Accountability Act of 1996 ("HIPAA")	. 52
9.3	Intentionally Omitted	. 52
9.4	Intentionally Omitted	. 52
9.5	Intentionally Omitted	. 52
9.6	Data Destruction	. 53
9.7	Local Small Business Enterprise (LSBE) Preference Program	. 53
9.8	Intentionally Omitted	. 54
9.9	Intentionally Omitted	. 54
9.10	Mandatory Requirement to Register on County's WebVen	. 54
9.11	Limitation on Corporate Acts	. 55
9.12	Modifications	. 55
9.13	Remedies of Non-Compliance	. 56
9.14	Suspension	. 56
9.15	Transition of Contract Services	. 57

9

STANDARD EXHIBITS

- A Statement of Work
- B Pricing Sheet
- C Independent Price Determination
- D Contractor's EEO Certification
- E District's Administration
- F Contractor's Administration
- G Contractor's Confidentiality Acknowledgement
- H Jury Service Ordinance
- I Safely Surrendered Baby Law

CONTRACT BETWEEN CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY AND CATHYJON ENTERPRISES, INC. DBA HB STAFFING FOR

TEMPORARY PERSONNEL SERVICES

This Contract ("Contract") made and entered into this 1st day of July, 2021 by and between the Consolidated Fire Protection District of Los Angeles County, hereinafter referred to as "District" and CathyJon Enterprises, Inc. DBA HB Staffing, hereinafter referred to as "Contractor." Contractor is located at 2120 Main Street, Suite 250, Huntington Beach, CA 92648.

RECITALS

WHEREAS, the District may contract with private businesses for temporary personnel services when certain requirements are met; and

WHEREAS, the Contractor is a private firm specializing in providing temporary personnel services; and

WHEREAS, the District is authorized under California Government Code Section 31000 to contract for special services, and under 31000.4 to contract with temporary help firms to assist the District during any peak load, temporary absence, or emergency other than a labor dispute; and

WHEREAS, the District is in need of temporary personnel services on an asneeded and intermittent basis and is authorized under Health and Safety Code Section 13861 to contract for services; and

NOW THEREFORE, in consideration of the mutual covenants contained herein, and for good and valuable consideration, the parties agree to the following:

1 APPLICABLE DOCUMENTS

Exhibits A, B, C, D, E, F, G, H, and I are attached to and form a part of this Contract. In the event of any conflict or inconsistency in the definition or interpretation of any word, responsibility, schedule, or the contents or description of any task, deliverable, goods, service, or other work, or otherwise between the base Contract and the Exhibits, or between Exhibits, such conflict or inconsistency shall be resolved by giving precedence first to the terms and conditions of the Contract and then to the Exhibits according to the following priority.

Standard Exhibits:

- 1.1 Exhibit A Statement of Work
- 1.2 Exhibit B Pricing Sheet
- 1.3 Exhibit C Independent Price Determination
- 1.4 Exhibit D Contractor's EEO Certification
- 1.5 Exhibit E District's Administration
- 1.6 Exhibit F Contractor's Administration
- 1.7 Exhibit G Acknowledgement and Confidentiality Agreement
- 1.8 Exhibit H Jury Service Ordinance
- 1.9 Exhibit I Safely Surrendered Baby Law

2 **DEFINITIONS**

The headings herein contained are for convenience and reference only and are not intended to define the scope of any provision thereof. The following words as used herein shall be construed to have the following meaning, unless otherwise apparent from the context in which they are used.

- 2.1 **Board of Supervisors (Board):** The Board of Supervisors of the County of Los Angeles; the governing body of the District and the County of Los Angeles.
- 2.2 **Contract:** This agreement executed between District and Contractor. Included are all supplemental agreements amending or extending the service to be performed. The Contract sets forth the terms and conditions for the issuance and performance of all tasks, deliverables, services and other work
- 2.3 **Contractor:** The person or persons, sole proprietor, partnership, joint venture, corporation or other legal entity who has entered into an agreement with the District to perform or execute the work covered by this Contract.
- 2.4 <u>**Contractor Project Manager:**</u> The person designated by the Contractor to administer the Contract operations under this Contract.
- 2.5 **<u>County</u>**: The County of Los Angeles, a political subdivision of the State of California.
- 2.6 **<u>Dav(s)</u>**: Calendar day(s) unless otherwise specified.
- 2.7 <u>District</u>: The Consolidated Fire Protection District of Los Angeles County.
- 2.8 **District Project Director:** Person designated by District with authority for District on contractual or administrative matters relating to this Contract that cannot be resolved by the District's Project Manager.
- 2.9 <u>**District Project Manager</u>:** Person designated by District's Project Director to manage the operations under this Contract.</u>
- 2.10 **Fiscal Year**: The twelve (12) month period beginning July 1st and ending the following June 30th.
- 2.11 <u>Statement of Work</u>: The directions, provisions, and requirements provided herein and special provisions pertaining to the method, frequency, manner and place of performing the contract services.

- 2.12 **<u>Subcontract</u>**: An agreement by the contractor to employ a subcontractor to provide services to fulfill this contract.
- 2.13 **Subcontractor:** Any individual, person or persons, sole proprietor, firm, partnership, joint venture, corporation, or other legal entity furnishing supplies, services of any nature, equipment, and/or materials to contractor in furtherance of contractor's performance of this contract, at any tier, under oral or written agreement.

3 WORK

- 3.1 Pursuant to the provisions of this Contract, the Contractor shall fully perform, complete and deliver on time, all tasks, deliverables, services and other work as set forth in herein.
- 3.2 If the Contractor provides any tasks, deliverables, goods, services, or other work, other than as specified in this contract, the same shall be deemed to be a gratuitous effort on the part of the contractor, and the contractor shall have no claim whatsoever against the District.

4 TERM OF CONTRACT

4.1 The term of this Contract shall be three (3) years commencing after approval by County's Board of Supervisors (Board), and execution by the Fire Chief or his designee, unless sooner terminated or extended, in whole or in part, as provided in this Contract.

The District shall have the sole option to extend this Contract term for up to two (2) additional one-year periods and twelve (12) month to month extension options, for a maximum total Contract term of six (6) years, pursuant to the same terms and conditions. Each such extension option may be exercised at the sole discretion of the Fire Chief or his designee as authorized by the Board.

- 4.2 The District maintains databases that track/monitor contractor performance history. Information entered into such databases may be used for a variety of purposes, including determining whether the District will exercise a contract term extension option.
- 4.3 The Contractor shall notify the District when this Contract is within six (6) months of the expiration of the term as provided for hereinabove. Upon occurrence of this event, the Contractor shall send written notification to the District at the address herein provided in Exhibit E District's Administration.

5 CONTRACT SUM

5.1 Maximum Contract Sum

5.1.1 The amount the District shall expend from its own funds during the Contract's entire term for as-needed Temporary Personnel Services shall not exceed \$690,000 per Contract year. The Maximum Contract Sum shall be the maximum monetary amount available that is payable by the District to the Contractor for supplying all the Services, Deliverables, Work, etc.

The annual Maximum Contract Sum under the terms of this Contract shall be \$690,000, shared amongst all contracts for temporary personnel services.

This is not a commitment or offer on the part of the District to expend the Maximum Contract Sum. The Contractor shall perform and complete all Services required of the Contractor under this Contract as set forth in Exhibit B (Pricing Sheet), but in any event, not in excess of the Maximum Contract Sum.

5.1.2 The Contractor acknowledges and agrees the Maximum Contract Sum is an all-inclusive, not-to-exceed price that cannot be adjusted for any costs or expenses whatsoever of Contractor. This Contract includes the full amount of compensation and reimbursement the District will be asked to provide to the Contractor in order for the Contractor to fully perform all of its obligations under this Contract, with such amount of compensation and reimbursement subject to any executed Amendments if applicable. The Contractor understands the District is entering into this Contract in reliance upon the premise that the Contractor shall fully perform all of its obligations under this Contract without seeking any additional compensation or reimbursement beyond that already provided for in this Contract, subject to any Amendments, if applicable. It is the Contractor's risk and responsibility to achieve and timely deliver the Services in accordance with the requirements of the Contract.

5.2 Written Approval for Reimbursement

The Contractor shall not be entitled to payment or reimbursement for any tasks or services performed, nor for any incidental or administrative expenses whatsoever incurred in or incidental to performance hereunder, except as specified herein. Assumption or takeover of any of the Contractor's duties, responsibilities, or obligations, or performance of same by any person or entity other than the Contractor, whether through assignment, subcontract, delegation, merger, buyout, or any other mechanism, with or without consideration for any reason whatsoever, shall not occur except with the District's express prior written approval.

5.3 Notification of 75% of Total Contract Sum

The Contractor shall maintain a system of record keeping that will allow the Contractor to determine when it has incurred seventy-five percent (75%) of the total contract sum under this Contract. Upon occurrence of this event, the Contractor shall send written notification to the District at the address herein provided in Exhibit E - District's Administration.

5.4 No Payment for Services Provided Following Expiration-Termination of Contract

The Contractor shall have no claim against District for payment of any money or reimbursement, of any kind whatsoever, for any service provided by the Contractor after the expiration or other termination of this Contract. Should the Contractor receive any such payment it shall immediately notify the District and shall immediately repay all such funds to the District. Payment by the District for services rendered after expiration-termination of this Contract shall not constitute a waiver of the District's right to recover such payment from the Contractor. This provision shall survive the expiration or other termination of this Contract.

5.5 Invoices and Payments

5.5.1 The Contractor shall invoice the District only for providing the tasks, deliverables, goods, services, and other work specified in Exhibit A - Statement of Work and elsewhere hereunder. The Contractor shall prepare invoices, which shall include the charges owed to the Contractor by the District under the terms of this Contract. The Contractor's payments shall be as provided in Exhibit B - Pricing Sheet and the Contractor shall be paid only for the tasks, deliverables, goods, services, and other work approved in writing by the District. If the District does not approve work in writing no payment shall be due to the Contractor for that work.

- 5.5.2 The Contractor's invoices shall be priced in accordance with Exhibit B Pricing Sheet.
- 5.5.3 The Contractor's invoices shall contain the information set forth in Exhibit A - Statement of Work describing the tasks, deliverables, goods, services, work hours, and facility and/or other work for which payment is claimed.
- 5.5.4 The Contractor shall submit the monthly invoices to the District by the 15th calendar day of the month following the month of service.
- 5.5.5 Payment to Contractor shall be made on an arrears basis, upon acceptance of completed work by the District, provided the Contractor is not in default under any provisions of this Contract. The Contractor shall email one (1) copy of the invoice to the following:
 - 1. <u>Bianca.Moya@fire.lacounty.gov</u> <u>Vinrose.Napoles@fire.lacounty.gov</u> for review and approval of all invoices; and
 - 2. <u>Fire-InvoiceSubmission@fire.lacounty.gov</u> for payment of all invoices.

The Contractor's invoices shall include the following:

- Contract Number
- Date(s) of Service
- Brief description of services
- A breakdown of labor hours and hourly rate i.e.: 3 hours @ \$20/hour = \$60.00.
 NO OVERTIME MAY BE ACCUMULATED OR PAID.
- Employee Name and Employee Number of District Employee who ordered or authorized the service.
- Signature of authorized District employee. Contractor's failure to obtain the signature of District employee authorizing the work may result in a delay of payment.

5.5.6 **District Approval of Invoices**

All invoices submitted by the Contractor for payment must have the written approval of the District's Project Manager prior to any payment thereof. In no event shall the District be liable or responsible for any payment prior to such written approval. Approval for payment will not be unreasonably withheld.

5.5.7 Local Small Business Enterprises – Prompt Payment Program

Certified Local Small Business Enterprises (LSBEs) will receive prompt payment for services they provide to the District. Prompt payment is defined as fifteen (15) calendar days after receipt of an undisputed invoice.

5.6 Intentionally Omitted

5.7 Default Method of Payment: Direct Deposit or Electronic Funds Transfer

- 5.7.1 The County, at its sole discretion, has determined the most efficient and secure default form of payment for goods and/or services provided under an agreement/ contract with the County shall be Electronic Funds Transfer (EFT) or direct deposit, unless an alternative method of payment is deemed appropriate by the Auditor-Controller (A-C).
- 5.7.2 The Contractor shall submit a direct deposit authorization request via the website https://directdeposit.lacounty.gov with banking and vendor information, and any other information that the A-C determines is reasonably necessary to process the payment and comply with all accounting, record keeping, and tax reporting requirements.
- 5.7.3 Any provision of law, grant, or funding agreement requiring a specific form or method of payment other than EFT or direct deposit shall supersede this requirement with respect to those payments.
- 5.7.4 At any time during the duration of the agreement/contract, a Contractor may submit a written request for an exemption to this requirement. Such request must be based on specific legal, business or operational needs and explain why the

payment method designated by the A-C is not feasible and an alternative is necessary. The A-C, in consultation with the contracting department(s), shall decide whether to approve exemption requests.

5.8 Travel, Transportation, and Parking

- 5.8.1 All travel-related costs are the responsibility of the Contractor. The District will not be responsible for paying or reimbursing Contractor or its employees for any travel related costs.
- 5.8.2 The District is not responsible for paying or reimbursing parking or transportation costs for temporary personnel.

6 ADMINISTRATION OF CONTRACT – DISTRICT

6.1 District Administration

A listing of all District Administration referenced in the following subparagraphs are designated in Exhibit E - District's Administration. The District will notify the Contractor in writing of any change in the names or addresses shown.

6.2 District's Project Director

The responsibilities of the District's Project Director include:

- Coordinating with Contractor and ensuring Contractor's performance of the Contract; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby; and
- Upon request of the Contractor, providing direction to the Contractor, as appropriate in areas relating to District policy, information requirements, and procedural requirements; however, in no event, shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby.

6.3 District's Project Manager

The role of the District's Project Manager is to oversee the day-today administration of this Contract; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby. The District's Project Manager's responsibilities include:

- Meeting with the Contractor's Project Manager on a regular basis; and
- Inspecting any and all tasks, deliverables, goods, services, or other work provided by or on behalf of the Contractor; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby.

The District's Project Manager is not authorized to make any changes in any of the terms and conditions of this Contract and is not authorized to further obligate District in any respect whatsoever.

6.4 District's Contract Administrator

The responsibilities of the District's Contract Administrator include:

- Ensuring that the objectives of this Contract are met; and
- Making changes in the terms and conditions of this Contract in accordance with Sub-paragraph 8.1, Amendments; and
- Providing direction to Contractor in the areas relating to District policy, information requirements, and procedural requirements.

7 ADMINISTRATION OF CONTRACT - CONTRACTOR

7.1 Contractor Administration

A listing of all of Contractor's Administration referenced in the following paragraphs is designated in Exhibit F - Contractor's Administration. The Contractor will notify the District in writing of any change in the names or addresses shown.

7.2 Contractor's Project Manager

- 7.2.1 The Contractor's Project Manager is designated in Exhibit F - Contractor's Administration. The Contractor shall notify the District in writing of any change in the name or address of the Contractor's Project Manager.
- 7.2.2 The Contractor's Project Manager shall be responsible for the Contractor's day-to-day activities as related to this

Contract and shall meet and coordinate with the District's Project Manager on a regular basis.

7.3 Approval of Contractor's Staff

District has the absolute right to approve or disapprove all of the Contractor's staff performing work hereunder and any proposed changes in the Contractor's staff, including, but not limited to, the Contractor's Project Manager.

7.4 Contractor's Staff Identification

All of the Contractor's employees assigned to District facilities are required to have a District Identification (ID) badge on their person and visible at all times. The Contractor may bear the expense of any badge not returned to the District when the temporary personnel's assignment has ended.

- 7.4.1 Contractor is responsible to ensure employees have obtained a District ID badge before they are assigned to work in a District facility. Contractor personnel may be asked by a District representative to leave a District facility if they do not have the proper District ID badge on their person and Contractor personnel must immediately comply with such request.
- 7.4.2 Contractor shall notify the District within one business day when staff is terminated from working under this Contract. Contractor shall retrieve and return an employee's District ID badge to the District on the next business day after the employee has terminated employment with the Contractor.
- 7.4.3 If District requests the removal of Contractor's staff, Contractor shall retrieve and return an employee's District ID badge to the District on the next business day after the employee has been removed from working on the District's Contract.

7.5 Background and Security Investigations

7.5.1 Each of Contractor's staff performing services under this Contract, as determined by District in District's sole discretion, shall undergo and pass a background investigation to the satisfaction of District as a condition of beginning and continuing to perform services under this Contract. Such background investigation must be obtained through fingerprints submitted to the California Department of Justice to include State, local, and federal-level review, which may include, but shall not be limited to, criminal conviction information. The fees associated with the background investigation shall be at the expense of the Contractor, regardless of whether the member of Contractor's staff passes or fails the background investigation. The District, may at its sole discretion and expense, provide live scan services for Contractor's staff assigned to District facilities. The District reserves the right to request reimbursement for the live scan services.

If a member of Contractor's staff does not pass the background investigation, the District may request that the member of Contractor's staff be removed immediately from performing services under the Contract. Contractor shall comply with District's request at any time during the term of the Contract. District will not provide to Contractor or to Contractor's staff any information obtained through the District's background investigation.

- 7.5.2 District shall immediately deny or terminate facility access to any member of Contractor's staff that does not pass such investigation to the satisfaction of the District or whose background or conduct is incompatible with District facility access.
- 7.5.3 Disqualification of any member of Contractor's staff pursuant to this Paragraph 7.5 shall not relieve Contractor of its obligation to complete all work in accordance with the terms and conditions of this Contract.

7.6 Confidentiality

- 7.6.1 Contractor shall maintain the confidentiality of all records and information in accordance with all applicable Federal, State and local laws, rules, regulations, ordinances, directives, guidelines, policies and procedures relating to confidentiality, including, without limitation, County policies concerning information technology security and the protection of confidential records and information.
- 7.6.2 Contractor shall indemnify, defend, and hold harmless County, its officers, employees, and agents, from and against any and all claims, demands, damages, liabilities, losses, costs and expenses, including, without limitation,

defense costs and legal, accounting and other expert, consulting, or professional fees, arising from, connected with, or related to any failure by Contractor, its officers, employees, agents, or subcontractors, to comply with this Paragraph 7.6, as determined by County in its sole judgment. Any legal defense pursuant to contractor's indemnification obligations under this Paragraph 7.6 shall be conducted by contractor and performed by counsel selected by Contractor and approved by District. Notwithstanding the preceding sentence, District shall have the right to participate in any such defense at its sole cost and expense, except that in the event Contractor fails to provide District with a full and adequate defense, as determined by District in its sole judgment, District shall be entitled to retain its own counsel, including, without limitation, County Counsel, and to reimbursement from Contractor for all such costs and expenses incurred by County in doing so. Contractor shall not have the right to enter into any settlement, agree to any injunction, or make any admission, in each case, on behalf of District without District's prior written approval.

- 7.6.3 Contractor shall inform all of its officers, employees, agents and subcontractors providing services hereunder of the confidentiality provisions of this Contract.
- 7.6.4 Contractor shall sign and adhere to the provisions of Exhibit G - "Contractor Acknowledgement and Confidentiality Agreement."

8 STANDARD TERMS AND CONDITIONS

8.1 Amendments

- 8.1.1 For any change which affects the scope of work, term, contract sum, payments, or any term or condition included under this Contract, an amendment to the Contract shall be prepared and executed by the contractor and by the Fire Chief or his designee.
- 8.1.2 The County's Board of Supervisors or Chief Executive Officer or designee may require the addition and/or change of certain terms and conditions in the Contract during the term of this Contract. The District reserves the right to add and/or change such provisions as required by the County's Board of Supervisors or Chief Executive Officer. To implement such changes, an Amendment to the Contract shall be prepared

and executed by the contractor and by the Fire Chief or his designee.

8.1.3 The Fire Chief or his designee may at his sole discretion, authorize extensions of time as defined in Paragraph 4 - Term of Contract. The contractor agrees that such extensions of time shall not change any other term or condition of this Contract during the period of such extensions. To implement an extension of time, an Amendment to the Contract shall be prepared and executed by the contractor and by the Fire Chief or his designee.

8.2 Assignment and Delegation/Mergers or Acquisitions

- 8.2.1 The contractor shall notify the District of any pending acquisitions/mergers of its company unless otherwise legally prohibited from doing so. If the contractor is restricted from legally notifying the District of pending acquisitions/mergers, District then it should notifv the of the actual acquisitions/mergers as soon as the law allows and provide to the District the legal framework that restricted it from notifying the District prior to the actual acquisitions/mergers.
- 8.2. 2 The contractor shall not assign its rights or delegate its duties under this Contract, or both, whether in whole or in part, without the prior written consent of District, in its discretion, and any attempted assignment or delegation without such consent shall be null and void. For purposes of this paragraph, District consent shall require a written Amendment to the Contract, which is formally approved and executed by the parties. Any payments by the District to any approved designee or assignee on any claim under this Contract shall be deductible, at District's sole discretion, against the claims, which the contractor may have against the District.
- 8.2.3 Shareholders, partners, members, or other equity holders of contractor may transfer, sell, exchange, assign, or divest themselves of any interest they may have therein. However, in the event any such sale, transfer, exchange, assignment, or divestment is effected in such a way as to give majority control of contractor to any person(s), corporation, partnership, or legal entity other than the majority controlling interest therein at the time of execution of the Contract, such disposition is an assignment requiring the prior written consent of District in accordance with applicable provisions of this Contract.

8.2.4 Any assumption, assignment, delegation, or takeover of any of the contractor's duties, responsibilities, obligations, or performance of same by any person or entity other than the contractor, whether through assignment, subcontract, delegation, merger, buyout, or any other mechanism, with or without consideration for any reason whatsoever without District's express prior written approval, shall be a material breach of the Contract which may result in the termination of this Contract. In the event of such termination. District shall be entitled to pursue the same remedies against contractor as it could pursue in the event of default by contractor.

8.3 Authorization Warranty

The contractor represents and warrants that the person executing this Contract for the contractor is an authorized agent who has actual authority to bind the contractor to each and every term, condition, and obligation of this Contract and that all requirements of the contractor have been fulfilled to provide such actual authority.

8.4 Budget Reductions

In the event the County's Board of Supervisors adopts, in any fiscal year, a County Budget which provides for reductions in the salaries and benefits paid to the majority of County employees and imposes similar reductions with respect to County contracts, the District reserves the right to reduce its payment obligation under this Contract correspondingly for that fiscal year and any subsequent fiscal year during the term of this Contract (including any extensions), and the services to be provided by the contractor under this Contract shall also be reduced correspondingly. The District's notice to the contractor regarding said reduction in payment obligation shall be provided within thirty (30) calendar days of the Board's approval of such actions. Except as set forth in the preceding sentence, the contractor shall continue to provide all of the services set forth in this Contract.

8.5 Complaints

- 8.5.1 The contractor shall develop, maintain and operate procedures for receiving, investigating and responding to complaints.
- 8.5.2 Complaint Procedures

- 8.5.2.1 Within forty-five (45) business days after the Contract effective date, the contractor shall provide the District with the contractor's policy for receiving, investigating and responding to user complaints.
- 8.5.2.2 The District will review the contractor's policy and provide the contractor with approval of said plan or with requested changes.
- 8.5.2.3 If the District requests changes in the contractor's policy, the contractor shall make such changes and resubmit the plan within five (5) business days for District approval.
- 8.5.2.4 If, at any time, the contractor wishes to change the contractor's policy, the contractor shall submit proposed changes to the District for approval before implementation.
- 8.5.2.5 The contractor shall preliminarily investigate all complaints and notify the District's Project Manager of the status of the investigation within five (5) business days of receiving the complaint.
- 8.5.2.6 When complaints cannot be resolved informally, a system of follow-through shall be instituted which adheres to formal plans for specific actions and strict time deadlines.
- 8.5.2.7 Copies of all written responses shall be sent to the District's Project Manager within three (3) business days of mailing to the complainant.

8.6 Compliance with Applicable Law

- 8.6.1 In the performance of this Contract, contractor shall comply with all applicable Federal, State and local laws, rules, regulations, ordinances, directives, guidelines, policies and procedures, and all provisions required thereby to be included in this Contract are hereby incorporated herein by reference.
- 8.6.2 Contractor shall indemnify, defend, and hold harmless County, its officers, employees, and agents, from and against any and all claims, demands, damages, liabilities, losses, costs, and expenses, including, without limitation,

defense costs and legal, accounting and other expert, consulting or professional fees, arising from, connected with, or related to any failure by contractor, its officers, employees, agents, or subcontractors, to comply with any such laws, rules, regulations, ordinances, directives, guidelines, policies, or procedures, as determined by County in its sole judgment. Any legal defense pursuant to contractor's indemnification obligations under Paragraph 8.6 (Compliance with Applicable Law) shall be conducted by contractor and performed by counsel selected by contractor and approved by County. Notwithstanding the preceding sentence, County shall have the right to participate in any such defense at its sole cost and expense, except that in the event contractor fails to provide County with a full and adequate defense, as determined by County in its sole judgment, County shall be entitled to retain its own counsel, including, without limitation, County Counsel, and to reimbursement from contractor for all such costs and expenses incurred by County in doing so. Contractor shall not have the right to enter into any settlement, agree to any injunction or other equitable relief, or make any admission, in each case, on behalf of County without County's prior written approval.

8.7 Compliance with Civil Rights Laws

The contractor hereby assures that it will comply with Subchapter VI of the Civil Rights Act of 1964, 42 USC Sections 2000 (e) (1) through 2000 (e) (17), to the end that no person shall, on the grounds of race, creed, color, sex, religion, ancestry, age, condition of physical handicap, marital status, political affiliation, or national origin, be excluded from participation in, be denied the benefits of, or be otherwise subjected to discrimination under this Contract or under any project, program, or activity supported by this Contract. The contractor shall comply with Exhibit D - Contractor's EEO Certification.

8.8 Compliance with the County's Jury Service Program

8.8.1 Jury Service Program:

This Contract is subject to the provisions of the County's ordinance entitled Contractor Employee Jury Service ("Jury Service Program") as codified in Sections 2.203.010 through 2.203.090 of the Los Angeles County Code, a copy of which

is attached as Exhibit H and incorporated by reference into and made a part of this Contract.

8.8.2 Written Employee Jury Service Policy.

- 1. Unless the contractor has demonstrated to the District's satisfaction either that the contractor is not a "contractor" as defined under the Jury Service Program (Section 2.203.020 of the County Code) or that the contractor qualifies for an exception to the Jury Service Program (Section 2.203.070 of the County Code), the contractor shall have and adhere to a written policy that provides that its Employees shall receive from the contractor, on an annual basis, no less than five days of regular pay for actual jury service. The policy may provide that Employees deposit any fees received for such jury service with the contractor or that the contractor deduct from the Employee's regular pay the fees received for jury service.
- 2. For purposes of this paragraph, "contractor" means a person, partnership, corporation or other entity which has a contract with the County or a subcontract with a County contractor and has received or will receive an aggregate sum of fifty thousand dollars (\$50,000) or more in any twelve (12) month period under one or more County contracts or subcontracts. "Employee" means any California resident who is a full-time employee of the contractor. "Full-time" means forty (40) hours or more worked per week, or a lesser number of hours if: 1) the lesser number is a recognized industry standard as determined by the District, or 2) contractor has a long-standing practice that defines the lesser number of hours as full-time. Full-time employees providing short-term, temporary services of ninety (90) days or less within a twelve (12) month period are not considered full-time for purposes of the Jury Service Program. If the contractor uses any subcontractor to perform services for the District under the Contract, the subcontractor shall also be subject to the provisions of this paragraph. The provisions of this paragraph shall be inserted into any such subcontract agreement and a copy of the Jury Service Program shall be attached to the agreement.

- 3. If the contractor is not required to comply with the Jury Service Program when the Contract commences, the contractor shall have a continuing obligation to review the applicability of its "exception status" from the Jury Service Program, and the contractor shall immediately notify the District if the contractor at any time either comes within the Jury Service Program's definition of "contractor" or if the contractor no longer gualifies for an exception to the Jury Service Program. In either event, the contractor shall immediately implement a written policy consistent with the Jury Service Program. The District may also require, at any time during the Contract and at its sole discretion, that the contractor demonstrate, to the District's satisfaction that the contractor either continues to remain outside of the Jury Service Program's definition of "contractor" and/or that the contractor continues to qualify for an exception to the Program.
- 4. Contractor's violation of this paragraph of the Contract may constitute a material breach of the Contract. In the event of such material breach, District may, in its sole discretion, terminate the Contract and/or bar the contractor from the award of future County contracts for a period of time consistent with the seriousness of the breach.

8.9 Conflict of Interest

- 8.9.1 No County employee whose position with the County enables such employee to influence the award of this Contract or any competing Contract, and no spouse or economic dependent of such employee, shall be employed in any capacity by the contractor or have any other direct or indirect financial interest in this Contract. No officer or employee of the contractor who may financially benefit from the performance of work hereunder shall in any way participate in the District's approval, or ongoing evaluation, of such work, or in any way attempt to unlawfully influence the District's approval or ongoing evaluation of such work.
- 8.9.2 The contractor shall comply with all conflict of interest laws, ordinances, and regulations now in effect or hereafter to be enacted during the term of this Contract. The contractor warrants that it is not now aware of any facts that create a conflict of interest. If the contractor hereafter becomes

aware of any facts that might reasonably be expected to create a conflict of interest, it shall immediately make full written disclosure of such facts to the District. Full written disclosure shall include, but is not limited to, identification of all persons implicated and a complete description of all relevant circumstances. Failure to comply with the provisions of this paragraph shall be a material breach of this Contract.

8.10 Consideration of Hiring County Employees Targeted for Layoff or Re-Employment List

Should the contractor require additional or replacement personnel after the effective date of this Contract to perform the services set forth herein, the contractor shall give first consideration for such employment openings to qualified, permanent County employees who are targeted for layoff or qualified, former County employees who are on a re-employment list during the life of this Contract.

8.11 Consideration of Hiring GAIN-GROW Participants

- 8.11.1 Should the contractor require additional or replacement personnel after the effective date of this Contract, the contractor shall give consideration for any such employment openings to participants in the County's Department of Public Social Services Greater Avenues for Independence (GAIN) Program or General Relief Opportunity for Work (GROW) Program who meet the contractor's minimum qualifications for the open position. For this purpose, consideration shall mean that the contractor will interview qualified candidates. The County will refer GAIN-GROW participants by job category to the contractor. Contractors shall report all job openings with job requirements to: GAINGROW@DPSS.LACOUNTY.GOV and BSERVICES@WDACS.LACOUNTY.GOV and DPSS will refer qualified GAIN/GROW job candidates.
- 8.11.2 In the event that both laid-off County employees and GAIN/GROW participants are available for hiring, County employees shall be given first priority.

8.12 Contractor Responsibility and Debarment

8.12.1 Responsible Contractor

A responsible contractor is a contractor who has demonstrated the attribute of trustworthiness, as well as quality, fitness, capacity and experience to satisfactorily perform the contract. It is the County's policy to conduct business only with responsible contractors.

8.12.2 Chapter 2.202 of the County Code

The contractor is hereby notified that, in accordance with Chapter 2.202 of the County Code, if the County acquires information concerning the performance of the contractor on this or other contracts which indicates that the contractor is not responsible, the County may, in addition to other remedies provided in the Contract, debar the contractor from bidding or proposing on, or being awarded, and/or performing work on County contracts for a specified period of time, which generally will not exceed five (5) years but may exceed five (5) years or be permanent if warranted by the circumstances, and terminate any or all existing contracts the contractor may have with the County.

8.12.3 Non-responsible contractor

The County may debar a contractor if the Board of Supervisors finds, in its discretion, that the contractor has done any of the following: 1) violated a term of a contract with the County or a nonprofit corporation created by the County, 2) committed an act or omission which negatively reflects on the contractor's quality, fitness or capacity to perform a contract with the County, any other public entity, or a nonprofit corporation created by the County, or engaged in a pattern or practice which negatively reflects on same, 3) committed an act or offense which indicates a lack of business integrity or business honesty, or 4) made or submitted a false claim against the County or any other public entity.

8.12.4 Contractor Hearing Board

8.12.4.1 If there is evidence that the contractor may be subject to debarment, the County will notify the contractor in writing of the evidence which is the

basis for the proposed debarment and will advise the contractor of the scheduled date for a debarment hearing before the Contractor Hearing Board.

- 8.12.4.2 The Contractor Hearing Board will conduct a hearing where evidence on the proposed debarment is presented. The contractor and/or the contractor's representative shall be given an opportunity to submit evidence at that hearing. After the hearing, the Contractor Hearing Board shall prepare a tentative proposed decision, which shall contain а recommendation regarding whether the contractor should be debarred, and, if so, the appropriate length of time of the debarment. The contractor and the District shall be provided an opportunity to object to the tentative proposed decision prior to its presentation to the Board of Supervisors.
- 8.12.4.3 After consideration of any objections, or if no objections are submitted, a record of the hearing, the proposed decision, and any other recommendation of the Contractor Hearing Board shall be presented to the Board of Supervisors. The Board of Supervisors shall have the right to modify, deny, or adopt the proposed decision and recommendation of the Contractor Hearing Board.
- 8.12.4.4 If a contractor has been debarred for a period longer than five (5) years, that contractor may after the debarment has been in effect for at least five (5) years, submit a written request for review of the debarment determination to reduce the period of debarment or terminate the debarment. The County may, in its discretion, reduce the period of debarment or terminate the debarment if it finds that the contractor has adequately demonstrated one or more of the 1) elimination of the grounds for followina: which the debarment was imposed; 2) a bona fide change in ownership or management; 3) material evidence discovered after debarment was imposed; or 4) any other reason that is in the best interests of the County.

- 8.12.4.5 The Contractor Hearing Board will consider a request for review of a debarment determination only where 1) the contractor has been debarred for a period longer than five (5) years; 2) the debarment has been in effect for at least five (5) years; and 3) the request is in writing, states one or more of the grounds for reduction of the debarment period or termination of the debarment. includes and supporting documentation. Upon receiving an appropriate request, the Contractor Hearing Board will provide notice of the hearing on the request. At the hearing, the Contractor Hearing Board shall conduct a hearing where evidence on the proposed reduction of debarment period or termination of debarment is presented. This hearing shall be conducted and the request for review decided by the Contractor Hearing Board pursuant to the same procedures as for a debarment hearing.
- 8.12.4.6 The Contractor Hearing Board's proposed decision shall contain a recommendation on the request to reduce the period of debarment or terminate the debarment. The Contractor Hearing Board shall present its proposed decision and recommendation to the Board of Supervisors. The Board of Supervisors shall have the right to modify, deny, or adopt the proposed decision and recommendation of the Contractor Hearing Board.

8.12.5 Subcontractors of Contractor

These terms shall also apply to subcontractors of County contractors.

8.13 Contractor's Acknowledgement of County's Commitment to Safely Surrendered Baby Law

The contractor acknowledges that the County places a high priority on the implementation of the Safely Surrendered Baby Law. The contractor understands that it is the County's policy to encourage all County contractors to voluntarily post the County's "Safely Surrendered Baby Law" poster, in Exhibit I, in a prominent position at the contractor's place of business. The contractor will also encourage its subcontractors, if any, to post this poster in a prominent position in the subcontractor's place of business. Information and posters for printing are available at <u>www.babysafela.org.</u>

8.14 Contractor's Warranty of Adherence to County's Child Support Compliance Program

- 8.14.1 The contractor acknowledges that the County has established a goal of ensuring that all individuals who benefit financially from the County through contracts are in compliance with their court-ordered child, family and spousal support obligations in order to mitigate the economic burden otherwise imposed upon the County and its taxpayers.
- 8.14.2 As required by the County's Child Support Compliance Program (County Code Chapter 2.200) and without limiting the contractor's duty under this Contract to comply with all applicable provisions of law, the contractor warrants that it is now in compliance and shall during the term of this Contract maintain in compliance with employment and wage reporting requirements as required by the Federal Social Security Act (42 USC Section 653a) and California Unemployment Insurance Code Section 1088.5, and shall implement all lawfully served Wage and Earnings Withholding Orders or Child Support Services Department Notices of Wage and Earnings Assignment for Child, Family or Spousal Support, pursuant to Code of Civil Procedure Section 706.031 and Family Code Section 5246(b).

8.15 District's Quality Assurance Plan

The District or its agent(s) will monitor the contractor's performance under this Contract on not less than an annual basis. Such monitoring will include assessing the contractor's compliance with all Contract terms and conditions and performance standards. Contractor deficiencies which the District determines are significant or continuing and that may place performance of the Contract in jeopardy if not corrected will be reported to the Board of Supervisors and listed in the appropriate contractor performance database. The report to the Board will include improvement/corrective action measures taken by the District and the contractor. If improvement does not occur consistent with the corrective action measures, the District may terminate this Contract or impose other penalties as specified in this Contract.

8.16 Damage to District Property, Facilities, Buildings or Grounds

- 8.16.1 The contractor shall repair, or cause to be repaired, at its own cost, any damage to the District's real or personal property used or occupied by the temporary personnel beyond wear from normal usage, as determined by District, any and all damage to District facilities, buildings, or grounds caused by the contractor or employees or agents of the contractor. Such repairs shall be made immediately after the contractor has become aware of such damage, but in no event later than thirty (30) days after the occurrence.
- 8.16.2 If the contractor fails to make timely repairs, District may make any necessary repairs. All costs incurred by District, as determined by District, for such repairs shall be repaid by the contractor by cash payment upon demand.

8.17 Employment Eligibility Verification

- The contractor warrants that it fully complies with all 8.17.1 Federal and State statutes and regulations regarding the employment of aliens and others and that all its employees performing work under this Contract meet the citizenship or alien status requirements set forth in Federal and State statutes and regulations. The contractor shall obtain, from all employees performing work hereunder, all verification and other documentation of employment eligibility status required by Federal and State statutes and regulations including, but not limited to, the Immigration Reform and Control Act of 1986, (P.L. 99-603), or as they currently exist and as they may be hereafter amended. The contractor shall retain all such documentation for all covered employees for the period prescribed by law.
- 8.17.2 The contractor shall indemnify, defend, and hold harmless, the County, its agents, officers, and employees from employer sanctions and any other liability which may be assessed against the contractor or the County or both in connection with any alleged violation of any Federal or State statutes or regulations pertaining to the eligibility for employment of any persons performing work under this Contract.

8.18 Counterparts and Electronic Signatures and Representations

This Contract may be executed in two or more counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same Contract. The facsimile, email or electronic signature of the Parties shall be deemed to constitute original signatures, and facsimile or electronic copies hereof shall be deemed to constitute duplicate originals.

The County and the Contractor hereby agree to regard electronic representations of original signatures of authorized officers of each party, when appearing in appropriate places on the Amendments prepared pursuant to Paragraph 8.1 (Amendments) and received via communications facilities (facsimile, email or electronic signature), as legally sufficient evidence that such legally binding signatures have been affixed to Amendments to this Contract.

8.19 Fair Labor Standards

The contractor shall comply with all applicable provisions of the Federal Fair Labor Standards Act and shall indemnify, defend, and hold harmless the County and its agents, officers, and employees from any and all liability, including, but not limited to, wages, overtime pay, liquidated damages, penalties, court costs, and attorneys' fees arising under any wage and hour law, including, but not limited to, the Federal Fair Labor Standards Act, for work performed by the contractor's employees for which the County may be found jointly or solely liable.

8.20 Force Majeure

- Neither party shall be liable for such party's failure to 8.20.1 perform its obligations under and in accordance with this Contract, if such failure arises out of fires, floods, epidemics. quarantine restrictions. natural other occurrences, strikes, lockouts (other than a lockout by such party or any of such party's subcontractors), freight embargoes, or other similar events to those described above, but in every such case the failure to perform must be totally beyond the control and without any fault or negligence of such party (such events are referred to in this paragraph as "force majeure events").
- 8.20.2 Notwithstanding the foregoing, a default by a subcontractor of contractor shall not constitute a force majeure event, unless such default arises out of causes beyond the control

of both contractor and such subcontractor, and without any fault or negligence of either of them. In such case, contractor shall not be liable for failure to perform, unless the goods or services to be furnished by the subcontractor were obtainable from other sources in sufficient time to permit contractor to meet the required performance schedule. As used in this subparagraph, the term "subcontractor" and "subcontractors" mean subcontractors at any tier.

8.20.3 In the event contractor's failure to perform arises out of a force majeure event, contractor agrees to use commercially reasonable best efforts to obtain goods or services from other sources, if applicable, and to otherwise mitigate the damages and reduce the delay caused by such force majeure event.

8.21 Governing Law, Jurisdiction, and Venue

This Contract shall be governed by, and construed in accordance with, the laws of the State of California. The contractor agrees and consents to the exclusive jurisdiction of the courts of the State of California for all purposes regarding this Contract and further agrees and consents that venue of any action brought hereunder shall be exclusively in the County of Los Angeles.

8.22 Independent Contractor Status

- 8.22.1 This Contract is by and between the District and the contractor and is not intended, and shall not be construed, to create the relationship of agent, servant, employee, partnership, joint venture, or association, as between the District and the contractor. The employees and agents of one party shall not be, or be construed to be, the employees or agents of the other party for any purpose whatsoever.
- 8.22.2 The contractor shall be solely liable and responsible for providing to, or on behalf of, all persons performing work pursuant to this Contract all compensation and benefits. The District shall have no liability or responsibility for the payment of any salaries, wages, unemployment benefits, disability benefits, Federal, State, or local taxes, or other compensation, benefits, or taxes for any personnel provided by or on behalf of the contractor.
- 8.22.3 The contractor understands and agrees that all persons

performing work pursuant to this Contract are, for purposes of Workers' Compensation liability, solely employees of the contractor and not employees of the District. The contractor shall be solely liable and responsible for furnishing any and all Workers' Compensation benefits to any person as a result of any injuries arising from or connected with any work performed by or on behalf of the contractor pursuant to this Contract.

8.22.4 The contractor shall adhere to the provisions stated in Paragraph 7.6 (Confidentiality).

8.23 Indemnification

The contractor shall indemnify, defend and hold harmless the County, its Special Districts, elected and appointed officers, employees, agents and volunteers (County Indemnitees) from and against any and all liability, including but not limited to demands, claims, actions, fees, costs and expenses (including attorney and expert witness fees), arising from and/or relating to this Contract, except for such loss or damage arising from the sole negligence or willful misconduct of the County indemnitees.

8.24 General Provisions for all Insurance Coverage

Without limiting Contractor's indemnification of District, and in the performance of this Contract and until all of its obligations pursuant to this Contract have been met, Contractor shall provide and maintain at its own expense insurance coverage satisfying the requirements specified in Paragraphs 8.24 and 8.25 of this Contract. These minimum insurance coverage terms, types and limits (the "Required Insurance") also are in addition to and separate from any other contract. The District in no way warrants that the Required Insurance is sufficient to protect the Contractor for liabilities which may arise from or relate to this Contract.

8.24.2 Evidence of Coverage and Notice to District

8.24.2.1 Certificate(s) of insurance coverage (Certificate) satisfactory to District, and a copy of an Additional Insured endorsement confirming County and its Agents (defined below) has been given Insured status under the Contractor's General Liability policy, shall be delivered to District at the address shown below and provided prior to commencing services under this Contract.

- 8.24.2.2 Renewal Certificates shall be provided to District not less than ten (10) days prior to contractor's policy expiration dates. The District reserves the right to obtain complete, certified copies of any required contractor and/or subcontractor insurance policies at any time.
- 8.24.2.3 Certificates shall identify all Required Insurance coverage types and limits specified herein, reference this Contract by name or number, and be signed by an authorized representative of the insurer(s). The Insured party named on the Certificate shall match the name of the contractor identified as the contracting party in this Contract. Certificates shall provide the full name of each insurer providing coverage, its NAIC (National Association of Insurance identification number. Commissioners) its financial rating, the amounts of any policy deductibles or self-insured retentions exceeding fifty thousand dollars (\$50,000), and list any District required endorsement forms.
- 8.24.2.4 Neither the District's failure to obtain, nor the District's receipt of, or failure to object to a noncomplying insurance certificate or endorsement, or any other insurance documentation or information provided by the contractor, its insurance broker(s) and/or insurer(s), shall be construed as a waiver of any of the Required Insurance provisions.
- 8.24.2.5 Certificates and copies of any required endorsements shall be sent to:

<u>Consolidated Fire Protection District of</u> <u>Los Angeles County</u> <u>Materials Management Division/Contracts Section</u> <u>5801 S. Eastern Avenue, Suite 100</u> <u>Commerce, California 90040-4001</u> 8.24.2.6 Contractor also shall promptly report to District any injury or property damage accident or incident, including any injury to a contractor employee occurring on District property, and any loss, disappearance, destruction, misuse, or theft of District property, monies or securities entrusted to contractor. Contractor also shall promptly notify District of any third-party claim or suit filed against contractor or any of its subcontractors which arises from or relates to this Contract, and could result in the filing of a claim or lawsuit against contractor and/or County.

8.24.3 Additional Insured Status and Scope of Coverage

The County of Los Angeles, it's Special Districts, Elected Officials, Officers, Agents, employees and volunteers (collectively County and its Agents) shall be provided additional insured status under contractor's General Liability policy with respect to liability arising out of contractor's ongoing and completed operations performed on behalf of the District. County and its Agents additional insured status shall apply with respect to liability and defense of suits arising out of the contractor's acts or omissions, whether such liability is attributable to the contractor or to the District. The full policy limits and scope of protection also shall apply to the County and its Agents as an additional insured, even if they exceed the County's minimum Required Insurance specifications herein. Use of an automatic additional insured endorsement form is acceptable providing it satisfies the Required Insurance provisions herein.

8.24.4 Cancellation of or Changes in Insurance

Contractor shall provide District with, or contractor's insurance policies shall contain a provision that District shall receive, written notice of cancellation or any change in Required Insurance, including insurer, limits of coverage, term of coverage or policy period. The written notice shall be provided to District at least ten (10) days in advance of cancellation for non-payment of premium and thirty (30) days in advance for any other cancellation or policy change. Failure to provide written notice of cancellation or any change in Required Insurance may constitute a

material breach of the Contract, in the sole discretion of the District, upon which the District may suspend or terminate this Contract.

8.24.5 Failure to Maintain Insurance

Contractor's failure to maintain or to provide acceptable evidence that it maintains the Required Insurance shall constitute a material breach of the Contract, upon which District immediately may withhold payments due to contractor, and/or suspend or terminate this Contract. District, at its sole discretion, may obtain damages from contractor resulting from said breach. Alternatively, the District may purchase the Required Insurance, and without further notice to contractor, deduct the premium cost from sums due to contractor or pursue contractor reimbursement.

8.24.6 Insurer Financial Ratings

Coverage shall be placed with insurers acceptable to the District with A.M. Best ratings of not less than A:VII unless otherwise approved by District.

8.24.7 Contractor's Insurance Shall Be Primary

Contractor's insurance policies, with respect to any claims related to this Contract, shall be primary with respect to all other sources of coverage available to contractor. Any District maintained insurance or self-insurance coverage shall be in excess of and not contribute to any contractor coverage.

8.24.8 Waivers of Subrogation

To the fullest extent permitted by law, the contractor hereby waives its rights and its insurer(s)' rights of recovery against County under all the Required Insurance for any loss arising from or relating to this Contract. The contractor shall require its insurers to execute any waiver of subrogation endorsements which may be necessary to effect such waiver.

8.24.9 Subcontractor Insurance Coverage Requirements

Contractor shall include all subcontractors as insureds under contractor's own policies, or shall provide District with each subcontractor's separate evidence of insurance coverage. Contractor shall be responsible for verifying each subcontractor complies with the Required Insurance provisions herein, and shall require that each subcontractor name the County and contractor as additional insureds on the subcontractor's General Liability policy. Contractor shall obtain County's prior review and approval of any subcontractor request for modification of the Required Insurance.

8.24.10 Deductibles and Self-Insured Retentions (SIRs)

Contractor's policies shall not obligate the County to pay any portion of any contractor deductible or SIR. The District retains the right to require contractor to reduce or eliminate policy deductibles and SIRs as respects the District, or to provide a bond guaranteeing contractor's payment of all deductibles and SIRs, including all related claims investigation, administration and defense expenses. Such bond shall be executed by a corporate surety licensed to transact business in the State of California.

8.24.11 Claims Made Coverage

If any part of the Required Insurance is written on a claims made basis, any policy retroactive date shall precede the effective date of this Contract. Contractor understands and agrees it shall maintain such coverage for a period of not less than three (3) years following Contract expiration, termination or cancellation.

8.24.12 Application of Excess Liability Coverage

Contractors may use a combination of primary, and excess insurance policies which provide coverage as broad as ("follow form" over) the underlying primary policies, to satisfy the Required Insurance provisions.

8.24.13 Separation of Insureds

All liability policies shall provide cross-liability coverage as would be afforded by the standard ISO (Insurance Services

Office, Inc.) separation of insureds provision with no insured versus insured exclusions or limitations.

8.24.14 Alternative Risk Financing Programs

The District reserves the right to review, and then approve, Contractor use of self-insurance, risk retention groups, risk purchasing groups, pooling arrangements and captive insurance to satisfy the Required Insurance provisions. The County and its Agents shall be designated as an Additional Covered Party under any approved program.

8.24.15 District Review and Approval of Insurance Requirements

The District reserves the right to review and adjust the Required Insurance provisions, conditioned upon District's determination of changes in risk exposures.

8.25 Insurance Coverage

8.25.1 Commercial General Liability insurance (providing scope of coverage equivalent to ISO policy form CG 00 01), naming County and its Agents as an additional insured, with limits of not less than:

General Aggregate:	\$2 million
Products/Completed Operations Aggregate:	\$1 million
Personal and Advertising Injury:	\$1 million

Each Occurrence: \$1 million

- **8.25.2** Automobile Liability insurance (providing scope of coverage equivalent to ISO policy form CA 00 01) with limits of not less than \$1 million for bodily injury and property damage, in combined or equivalent split limits, for each single accident. Insurance shall cover liability arising out of contractor's use of autos pursuant to this Contract, including owned, leased, hired, and/or non-owned autos, as each may be applicable.
- 8.25.3 Workers Compensation and Employers' Liability insurance or qualified self- insurance satisfying statutory requirements, which includes Employers' Liability coverage with limits of not less than \$1 million per

accident. If Contractor will provide leased employees, or, is an employee leasing or temporary staffing firm or a professional employer organization (PEO), coverage also shall include an Alternate Employer Endorsement (providing scope of coverage equivalent to ISO policy form WC 00 03 01 A) naming the County as the Alternate Employer. The written notice shall be provided to District at least ten (10) days in advance of cancellation for non-payment of premium and thirty (30) days in advance for any other cancellation or policy change. If applicable to Contractor's operations, coverage also shall be arranged to satisfy the requirements of any federal workers or workmen's compensation law or any federal occupational disease law.

8.25.4 Unique Insurance Coverage

8.25.4.1 **Professional Liability-Errors and Omissions**

Insurance covering Contractor's liability arising from or related to this Contract, with limits of not less than \$1 million per claim and \$2 million aggregate. Further, Contractor understands and agrees it shall maintain such coverage for a period of not less than three (3) years following this Agreement's expiration, termination or cancellation.

8.25.4.2 Privacy/Network Security (Cyber) Liability

Insurance coverage providing protection against liability for (1) privacy breaches **[liability**] arising from the loss or disclosure of confidential information no matter how it occurs]; (2) system breach; (3) denial or loss of service; (4) introduction, implantation. spread of malicious or software code; (5) unauthorized access to or use of computer systems with limits of not less than \$2 million. No exclusion/restriction for unencrypted portable devices/media may be on the policy.

8.26 Liquidated Damages

- 8.26.1 If, in the judgment of the Fire Chief or his designee, the contractor is deemed to be non-compliant with the terms and obligations assumed hereby, the Fire Chief or his designee, at his/her option, in addition to, or in lieu of, other remedies provided herein, may withhold the entire monthly payment or deduct pro rata from the contractor's invoice for work not performed. A description of the work not performed and the amount to be withheld or deducted from payments to the contractor from the District, will be forwarded to the contractor by the Fire Chief or his designee, in a written notice describing the reasons for said action.
- 8.26.2 If the Fire Chief or his designee, determines that there are deficiencies in the performance of this Contract that the Fire Chief or his designee, deems are correctable by the contractor over a certain time span, the Fire Chief or his designee, will provide a written notice to the contractor to correct the deficiency within specified time frames. Should the contractor fail to correct deficiencies within said time frame, the Fire Chief, or his designee, may: (a) Deduct from the contractor's payment, pro rata, those applicable portions of the Monthly Contract Sum; and/or (b) Deduct liquidated damages. The parties agree that it will be impracticable or extremely difficult to fix the extent of actual damages resulting from the failure of the contractor to correct a deficiency within the specified time frame. The parties hereby agree that under the current circumstances a reasonable estimate of such damages is one hundred dollars (\$100) per day per infraction, or as specified in the Exhibit 2 - Performance Requirements Summary (PRS) Chart, Statement of Work Exhibits, hereunder, and that the contractor shall be liable to the District for liquidated damages in said amount. Said amount shall be deducted from the District's payment to the contractor; and/or (c) Upon giving five (5) days notice to the contractor for failure to correct the deficiencies, the District may correct any and all deficiencies and the total costs incurred by the District for completion of the work by an alternate source, whether it be District forces or separate private contractor, will be deducted and forfeited from the payment to the contractor from the District, as determined by the District.

- 8.26.3 The action noted in Paragraph 8.26.2 shall not be construed as a penalty, but as adjustment of payment to the contractor to recover the District cost due to the failure of the contractor to complete or comply with the provisions of this Contract.
- 8.26.4 This Paragraph shall not, in any manner, restrict or limit the District's right to damages for any breach of this Contract provided by law or as specified in the PRS or Paragraph 8.26.2, and shall not, in any manner, restrict or limit the District's right to terminate this Contract as agreed to herein.

8.27 Most Favored Public Entity

If the contractor's prices decline, or should the contractor at any time during the term of this Contract provide the same goods or services under similar quantity and delivery conditions to the State of California or any county, municipality, or district of the State at prices below those set forth in this Contract, then such lower prices shall be immediately extended to the District.

8.28 Nondiscrimination and Affirmative Action

- 8.28.1 The contractor certifies and agrees that all persons employed by it, its affiliates, subsidiaries, or holding companies are and shall be treated equally without regard to or because of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, in compliance with all applicable Federal and State anti-discrimination laws and regulations.
- 8.28.2 The contractor shall certify to, and comply with, the provisions of Exhibit D Contractor's EEO Certification.
- 8.28.3 The contractor shall take affirmative action to ensure that applicants are employed, and that employees are treated during employment, without regard to race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, in compliance with all applicable Federal and State anti-discrimination laws and regulations. Such action shall include, but is not limited to: employment, upgrading, demotion, transfer, recruitment or recruitment advertising, layoff or termination, rates of pay or other forms of compensation, and selection for training, including apprenticeship.

- 8.28.4 The contractor certifies and agrees that it will deal with its subcontractors, bidders, or vendors without regard to or because of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation.
- 8.28.5 The contractor certifies and agrees that it, its affiliates, subsidiaries, or holding companies shall comply with all applicable Federal and State laws and regulations to the end that no person shall, on the grounds of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, be excluded from participation in, be denied the benefits of, or be otherwise subjected to discrimination under this Contract or under any project, program, or activity supported by this Contract.
- 8.28.6 The contractor shall allow District representatives access to the contractor's employment records during regular business hours to verify compliance with the provisions of this Paragraph 8.28 Nondiscrimination and Affirmative Action when so requested by the District.
- 8.28.7 If the District finds that any provisions of this Paragraph 8.28 (Nondiscrimination and Affirmative Action) have been violated, such violation shall constitute a material breach of this Contract upon which the District may terminate or suspend this Contract. While the District reserves the right to determine independently that the anti-discrimination provisions of this Contract have been violated, in addition, a determination by the California Fair Employment and Housing Commission or the Federal Equal Employment Opportunity Commission that the contractor has violated Federal or State anti-discrimination laws or regulations shall constitute a finding by the District that the contractor has violated the anti-discrimination provisions of this Contract.
- 8.28.8 The parties agree that in the event the contractor violates any of the anti-discrimination provisions of this Contract, the District shall, at its sole option, be entitled to the sum of five hundred dollars (\$500) for each such violation pursuant to California Civil Code Section 1671 as liquidated damages in lieu of terminating or suspending this Contract.

8.29 Non Exclusivity

Nothing herein is intended nor shall be construed as creating any exclusive arrangement with the contractor. This Contract shall not restrict District from acquiring similar, equal or like goods and/or services from other entities or sources.

8.30 Notice of Delays

Except as otherwise provided under this Contract, when either party has knowledge that any actual or potential situation is delaying or threatens to delay the timely performance of this Contract, that party shall, within one (1) business day, give notice thereof, including all relevant information with respect thereto, to the other party.

8.31 Notice of Disputes

The contractor shall bring to the attention of the District's Project Manager and/or District's Project Director any dispute between the District and the contractor regarding the performance of services as stated in this Contract. If the District's Project Manager or District's Project Director is not able to resolve the dispute, the Fire Chief or his designee shall resolve it.

8.32 Notice to Employees Regarding the Federal Earned Income Credit

The contractor shall notify its employees, and shall require each subcontractor to notify its employees, that they may be eligible for the Federal Earned Income Credit under the federal income tax laws. Such notice shall be provided in accordance with the requirements set forth in Internal Revenue Service Notice No. 1015.

8.33 Notice to Employees Regarding the Safely Surrendered Baby Law

The contractor shall notify and provide to its employees, and shall require each subcontractor to notify and provide to its employees, information regarding the Safely Surrendered Baby Law, its implementation in Los Angeles County, and where and how to safely surrender a baby. The information is set forth in Exhibit I, Safely Surrendered Baby Law of this Contract. Additional information is available at <u>www.babysafela.org</u>.

8.34 Notices

All notices or demands required or permitted to be given or made under this Contract shall be in writing and shall be hand delivered with signed receipt or mailed by first-class registered or certified mail, postage prepaid, addressed to the parties as identified in Exhibits E -District's Administration and F - Contractor's Administration. Addresses may be changed by either party giving ten (10) days prior written notice thereof to the other party. The Fire Chief or his designee shall have the authority to issue all notices or demands required or permitted by the District under this Contract.

8.35 Prohibition Against Inducement or Persuasion

Notwithstanding the above, the contractor and the District agree that, during the term of this Contract and for a period of one year thereafter, neither party shall in any way intentionally induce or persuade any employee of one party to become an employee or agent of the other party. No bar exists against any hiring action initiated through a public announcement.

8.36 Public Records Act

- 8.36.1 Any documents submitted by the contractor; all information obtained in connection with the District's right to audit and inspect the contractor's documents, books, and accounting records pursuant to Paragraph 8.38 (Record Retention and Inspection-Audit Settlement) of this Contract; as well as those documents which were required to be submitted in response to the Request for Proposals (RFP) used in the solicitation process for this Contract, become the exclusive property of the District. All such documents become a matter of public record and shall be regarded as public records. Exceptions will be those elements in the California Government Code Section 6250 et seq. (Public Records Act) and which are marked "trade secret," "confidential," or "proprietary." The District shall not in any way be liable or responsible for the disclosure of any such records including, without limitation, those so marked, if disclosure is required by law, or by an order issued by a court of competent jurisdiction.
- 8.36.2 In the event the District is required to defend an action on a Public Records Act request for any of the aforementioned documents, information, books, records, and/or contents of a proposal marked "trade secret," "confidential," or

"proprietary," the contractor agrees to defend and indemnify the District from all costs and expenses, including reasonable attorney's fees, in action or liability arising under the Public Records Act.

8.37 Publicity

- 8.37.1 The contractor shall not disclose any details in connection with this Contract to any person or entity except as may be otherwise provided hereunder or required by law. However, in recognizing the contractor's need to identify its services and related clients to sustain itself, the District shall not inhibit the contractor from publishing its role under this Contract within the following conditions:
 - 8.37.1.1 The contractor shall develop all publicity material in a professional manner; and
 - 8.37.1.2 During the term of this Contract, the contractor shall not, and shall not authorize another to, publish or disseminate any commercial advertisements, press releases, feature articles, or other materials using the name of the District without the prior written consent of the District's Project Director. The District shall not unreasonably withhold written consent.
- 8.37.2 The contractor may, without the prior written consent of District, indicate in its proposals and sales materials that it has been awarded this Contract with the County of Los Angeles, provided that the requirements of this Paragraph 8.37 (Publicity) shall apply.

8.38 Record Retention and Inspection-Audit Settlement

8.38.1 The Contractor shall maintain accurate and complete financial records of its activities and operations relating to this Contract in accordance with generally accepted accounting principles, and which meet the requirements for contract accounting described in Auditor-Controller Contract Accounting and Administration Handbook. The Contractor shall also maintain accurate and complete employment and other records relating to its performance of this Contract. The contractor agrees that the District, or its authorized representatives, shall have access to and the right to examine, audit, excerpt, copy, or transcribe any pertinent

transaction, activity, or record relating to this Contract. All such material, including, but not limited to, all financial records, bank statements, cancelled checks or other proof of payment, timecards, sign-in/sign-out sheets and other time and employment records, and proprietary data and information, shall be kept and maintained by the contractor and shall be made available to the District during the term of this Contract and for a period of five (5) years thereafter unless the District's written permission is given to dispose of any such material prior to such time. All such material shall be maintained by the contractor at a location in Los Angeles County, provided that if any such material is located outside Los Angeles County, then, at the District's option, the contractor shall pay the District for travel, per diem, and other costs incurred by the District to examine, audit, excerpt, copy, or transcribe such material at such other location.

- 8.38.2 In the event that an audit of the contractor is conducted specifically regarding this Contract by any Federal or State auditor, or by any auditor or accountant employed by the contractor or otherwise, then the contractor shall file a copy of such audit report with the County's Auditor-Controller within thirty (30) days of the contractor's receipt thereof, unless otherwise provided by applicable Federal or State law or under this Contract. Subject to applicable law, the District shall make a reasonable effort to maintain the confidentiality of such audit report(s). Failure on the part of the contractor to comply with any of the provisions of this subparagraph 8.38 shall constitute a material breach of this Contract.
- 8.38.3 If, at any time during the term of this Contract or within five (5) years after the expiration or termination of this Contract, representatives of the District conduct an audit of the contractor regarding the work performed under this Contract, and if such audit finds that the District's dollar liability for any such work is less than payments made by the District to the contractor, then the difference shall be either: a) repaid by the contractor to the District by cash payment upon demand or b) at the sole option of the County's Auditor-Controller, deducted from any amounts due to the contractor from the District, whether under this Contract or otherwise. If such audit finds that the District's

dollar liability for such work is more than the payments made by the District to the contractor, then the difference shall be paid to the contractor by the District by cash payment, provided that in no event shall the District's maximum obligation for this Contract exceed the funds appropriated by the District for the purpose of this Contract.

8.39 Recycled Bond Paper

Consistent with the Board of Supervisors' policy to reduce the amount of solid waste deposited at the County landfills, the contractor agrees to use recycled-content paper to the maximum extent possible on this Contract.

8.40 Subcontracting

- 8.40.1 The requirements of this Contract may not be subcontracted by the contractor **without the advance written approval of the District**. Any attempt by the contractor to subcontract without the prior consent of the District may be deemed a material breach of this Contract.
- 8.40.2 If the contractor desires to subcontract, the contractor shall provide the following information promptly at the District's request:
 - 8.40.2.1 A description of the work to be performed by the subcontractor;
 - 8.40.2.2 A draft copy of the proposed subcontract; and
 - 8.40.2.3 Other pertinent information and/or certifications requested by the District.
- 8.40.3 The contractor shall indemnify, defend, and hold the County harmless with respect to the activities of each and every subcontractor in the same manner and to the same degree as if such subcontractor(s) were the contractor employees.
- 8.40.4 The contractor shall remain fully responsible for all performances required of it under this Contract, including those that the contractor has determined to subcontract, notwithstanding the District's approval of the contractor's proposed subcontract.
- 8.40.5 The District's consent to subcontract shall not waive the District's right to prior and continuing approval of any and all

personnel, including subcontractor employees, providing services under this Contract. The contractor is responsible to notify its subcontractors of this District right.

- 8.40.6 The District's Project Director is authorized to act for and on behalf of the District with respect to approval of any subcontract and subcontractor employees. After approval of the subcontract by the District, contractor shall forward a fully executed subcontract to the District for their files.
- 8.40.7 The contractor shall be solely liable and responsible for all payments or other compensation to all subcontractors and their officers, employees, agents, and successors in interest arising through services performed hereunder, notwithstanding the District's consent to subcontract.
- 8.40.8 The contractor shall obtain certificates of insurance, which establish that the subcontractor maintains all the programs of insurance required by the District from each approved subcontractor. Before any subcontractor employee may perform any work hereunder, contractor shall ensure delivery of all such documents to:

Consolidated Fire Protection District of Los Angeles County Materials Management Division / Contracts Section 5801 S. Eastern Avenue, Suite 100 Commerce, California 90040-4001

8.41 Termination for Breach of Warranty to Maintain Compliance with County's Child Support Compliance Program

Failure of the contractor to maintain compliance with the requirements set forth in Paragraph 8.14 (Contractor's Warranty of Adherence to County's Child Support Compliance Program) shall constitute default under this Contract. Without limiting the rights and remedies available to the District under any other provision of this Contract, failure of the contractor to cure such default within ninety (90) calendar days of written notice shall be grounds upon which the District may terminate this Contract pursuant to Paragraph 8.43 (Termination for Default) and pursue debarment of the contractor, pursuant to County Code Chapter 2.202.

8.42 Termination for Convenience

8.42.1 This Contract may be terminated, in whole or in part, from time to time, when such action is deemed by the District, in

its sole discretion, to be in its best interest. Termination of work hereunder shall be effected by notice of termination to the contractor specifying the extent to which performance of work is terminated and the date upon which such termination becomes effective. The date upon which such termination becomes effective shall be no less than ten (10) days after the notice is sent.

- 8.42.2 After receipt of a notice of termination and except as otherwise directed by the District, the contractor shall:
 - 8.42.2.1 Stop work under this Contract on the date and to the extent specified in such notice, and
 - 8.42.2.2 Complete performance of such part of the work as shall not have been terminated by such notice.
- 8.42.3 All material including books, records, documents, or other evidence bearing on the costs and expenses of the contractor under this Contract shall be maintained by the contractor in accordance with Paragraph 8.38 (Record Retention and Inspection-Audit Settlement).

8.43 Termination for Default

- 8.43.1 The District may, by written notice to the contractor, terminate the whole or any part of this Contract, if, in the judgment of District's Project Director:
 - 8.43.1.1 Contractor has materially breached this Contract; or
 - 8.43.1.2 Contractor fails to timely provide and/or satisfactorily perform any task, deliverable, service, or other work required either under this Contract; or
 - 8.43.1.3 Contractor fails to demonstrate a high probability of timely fulfillment of performance requirements under this Contract, or of any obligations of this Contract and in either case, fails to demonstrate convincing progress toward a cure within five (5) working days (or such longer period as the District may authorize in writing) after receipt of written notice from the District specifying such failure.

- 8.43.2 In the event the District terminates this Contract in whole or in part as provided in Paragraph 8.43.1, the District may procure, upon such terms and in such manner as the District may deem appropriate, goods and services similar to those so terminated. The contractor shall be liable to the District for any and all excess costs incurred by the District, as determined by the District, for such similar goods and services. The contractor shall continue the performance of this Contract to the extent not terminated under the provisions of this paragraph.
- Except with respect to defaults of any subcontractor, the 8.43.3 contractor shall not be liable for any such excess costs of the type identified in Paragraph 8.43.2 if its failure to perform this Contract arises out of causes beyond the control and without the fault or negligence of the contractor. Such causes may include, but are not limited to: acts of God or of the public enemy, acts of the District in either its sovereign or contractual capacity, acts of Federal or State governments in their sovereign capacities, fires, floods, restrictions. epidemics. quarantine strikes. freight embargoes, and unusually severe weather; but in every case, the failure to perform must be beyond the control and without the fault or negligence of the contractor. If the failure to perform is caused by the default of a subcontractor, and if such default arises out of causes beyond the control of both the contractor and subcontractor, and without the fault or negligence of either of them, the contractor shall not be liable for any such excess costs for failure to perform, unless the goods or services to be furnished by the subcontractor were obtainable from other sources in sufficient time to permit the contractor to meet the required performance schedule. As used in this paragraph, the term "subcontractor(s)" means subcontractor(s) at any tier.
- 8.43.4 If, after the District has given notice of termination under the provisions of Paragraph 8.43 (Termination for Default) it is determined by the District that the contractor was not in default under the provisions of Paragraph 8.43 (Termination for Default) or that the default was excusable under the provisions of subparagraph 8.43.3, the rights and obligations of the parties shall be the same as if the notice of termination had been issued pursuant to Paragraph 8.42 (Termination for Convenience).

8.43.5 The rights and remedies of the District provided in this Paragraph 8.43 (Termination for Default) shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.44 Termination for Improper Consideration

- 8.44.1 The District may, by written notice to the contractor, immediately terminate the right of the contractor to proceed under this Contract if it is found that consideration, in any form, was offered or given by the contractor, either directly or through an intermediary, to any District officer, employee, or agent with the intent of securing this Contract or securing favorable treatment with respect to the award, amendment, or extension of this Contract or the making of any determinations with respect to the contractor's performance pursuant to this Contract. In the event of such termination, the District shall be entitled to pursue the same remedies against the contractor.
- 8.44.2 The contractor shall immediately report any attempt by a District officer or employee to solicit such improper consideration. The report shall be made either to the District manager charged with the supervision of the employee or to the County Auditor-Controller's Employee Fraud Hotline at (800) 544-6861.
- 8.44.3 Among other items, such improper consideration may take the form of cash, discounts, services, the provision of travel or entertainment, or tangible gifts.

8.45 Termination for Insolvency

- 8.45.1 The District may terminate this Contract forthwith in the event of the occurrence of any of the following:
 - 8.45.1.1 Insolvency of the contractor. The contractor shall be deemed to be insolvent if it has ceased to pay its debts for at least sixty (60) days in the ordinary course of business or cannot pay its debts as they become due, whether or not a petition has been filed under the Federal Bankruptcy Code and whether or not the contractor is insolvent within the meaning of the Federal Bankruptcy Code;

- 8.45.1.2 The filing of a voluntary or involuntary petition regarding the contractor under the Federal Bankruptcy Code;
- 8.45.1.3 The appointment of a Receiver or Trustee for the contractor; or
- 8.45.1.4 The execution by the contractor of a general assignment for the benefit of creditors.
- 8.45.2 The rights and remedies of the District provided in this Paragraph 8.45 (Termination for Insolvency) shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.46 Termination for Non-Adherence of County Lobbyist Ordinance

The contractor, and each County Lobbyist or County Lobbying firm as defined in County Code Section 2.160.010 retained by the contractor, shall fully comply with the County's Lobbyist Ordinance, County Code Chapter 2.160. Failure on the part of the contractor or any County Lobbyist or County Lobbying firm retained by the contractor to fully comply with the County's Lobbyist Ordinance shall constitute a material breach of this Contract, upon which the District may in its sole discretion, immediately terminate or suspend this Contract.

8.47 Termination for Non-Appropriation of Funds

Notwithstanding any other provision of this Contract, the District shall not be obligated for the contractor's performance hereunder or by any provision of this Contract during any of the District's future fiscal years unless and until the County's Board of Supervisors appropriates funds for this Contract in the District's Budget for each such future fiscal year. In the event that funds are not appropriated for this Contract, then this Contract shall terminate as of June 30 of the last fiscal year for which funds were appropriated. The District shall notify the contractor in writing of any such non-allocation of funds at the earliest possible date.

8.48 Validity

If any provision of this Contract or the application thereof to any person or circumstance is held invalid, the remainder of this Contract and the application of such provision to other persons or circumstances shall not be affected thereby.

8.49 Waiver

No waiver by the District of any breach of any provision of this Contract shall constitute a waiver of any other breach or of such provision. Failure of the District to enforce at any time, or from time to time, any provision of this Contract shall not be construed as a waiver thereof. The rights and remedies set forth in this paragraph 8.49 shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.50 Warranty Against Contingent Fees

- 8.50.1 The contractor warrants that no person or selling agency has been employed or retained to solicit or secure this Contract upon any Contract or understanding for a commission, percentage, brokerage, or contingent fee, excepting bona fide employees or bona fide established commercial or selling agencies maintained by the contractor for the purpose of securing business.
- 8.50.2 For breach of this warranty, the District shall have the right to terminate this Contract and, at its sole discretion, deduct from the Contract price or consideration, or otherwise recover, the full amount of such commission, percentage, brokerage, or contingent fee.

8.51 Warranty of Compliance with County's Defaulted Property Tax Reduction Program

Contractor acknowledges that County has established a goal of ensuring that all individuals and businesses that benefit financially from County through contract are current in paying their property tax obligations (secured and unsecured roll) in order to mitigate the economic burden otherwise imposed upon County and its taxpayers.

Unless contractor qualifies for an exemption or exclusion, contractor warrants and certifies that to the best of its knowledge it is now in compliance, and during the term of this contract will maintain compliance, with Los Angeles County Code Chapter 2.206.

8.52 Termination for Breach of Warranty to Maintain Compliance with County's Defaulted Property Tax Reduction Program

Failure of contractor to maintain compliance with the requirements set forth in Paragraph 8.51 "Warranty of Compliance with County's Defaulted Property Tax Reduction Program" shall constitute default under this contract. Without limiting the rights and remedies available to County under any other provision of this contract, failure of contractor to cure such default within ten (10) days of notice shall be grounds upon which District may terminate this contract and/or pursue debarment of contractor, pursuant to County Code Chapter 2.206.

8.53 Time Off for Voting

The contractor shall notify its employees, and shall require each subcontractor to notify and provide to its employees, information regarding the time off for voting law (Elections Code Section 14000). Not less than ten (10) days before every statewide election, every contractor and subcontractors shall keep posted conspicuously at the place of work, if practicable, or elsewhere where it can be seen as employees come or go to their place of work, a notice setting forth the provisions of Section 14000.

8.54 Compliance with County's Zero Tolerance Policy on Human Trafficking

Contractor acknowledges that the County has established a Zero Tolerance Policy on Human Trafficking prohibiting contractors from engaging in human trafficking.

If a Contractor or member of Contractor's staff is convicted of a human trafficking offense, the District shall require that the Contractor or member of Contractor's staff be removed immediately from performing services under the Contract. District will not be under any obligation to disclose confidential information regarding the offenses other than those required by law.

Disqualification of any member of Contractor's staff pursuant to this paragraph shall not relieve Contractor of its obligation to complete all work in accordance with the terms and conditions of this Contract.

8.55 Intentionally Omitted

8.56 Compliance with Fair Chance Employment Practices

Contractor shall comply with fair chance employment hiring practices set forth in California Government Code Section 12952, Employment Discrimination: Conviction History. Contractor's violation of this paragraph of the Contract may constitute a material breach of the Contract. In the event of such material breach, District may, in its sole discretion, terminate the Contract.

8.57 Compliance with the County Policy of Equity

The contractor acknowledges that the County takes its commitment to preserving the dignity and professionalism of the workplace very seriously, as set forth in the County Policy of Equity (CPOE) (https://ceop.lacounty.gov/). The contractor further acknowledges that the County strives to provide a workplace free from discrimination, harassment, retaliation and inappropriate conduct based on a protected characteristic, and which may violate the CPOE. The contractor, its employees and subcontractors acknowledge and certify receipt and understanding of the CPOE. Failure of the contractor, its employees or its subcontractors to uphold the County's expectations of a workplace free from harassment and discrimination, including inappropriate conduct based on a protected characteristic, may subject the contractor to termination of contractual agreements as well as civil liability.

8.58 **Prohibition from Participation in Future Solicitation(s)**

A Proposer, or a Contractor or its subsidiary or Subcontractor ("Proposer/Contractor"), is prohibited from submitting a bid or proposal in a County solicitation if the Proposer/Contractor has provided advice or consultation for the solicitation. A Proposer/Contractor is also prohibited from submitting a bid or proposal in a County solicitation if the Proposer/Contractor has developed or prepared any of the solicitation materials on behalf of the County. A violation of this provision shall result in the disqualification of the Contractor/Proposer from participation in the County solicitation or the termination or cancellation of any resultant County contract. This provision shall survive the expiration, or other termination of this Agreement.

9 UNIQUE TERMS AND CONDITIONS

9.1 Intentionally Omitted

9.2 Health Insurance Portability and Accountability Act of 1996 (HIPAA)

- 9.2.1 Contractor expressly acknowledges and agrees that the provision of services under this Agreement does not require or permit access by Contractor or any of its officers, employees, or agents, to any patient medical records/patient information. Accordingly, Contractor shall instruct its officers, employees, and agents that they are not to pursue, or gain access to, patient medical records/patient information for any reason whatsoever.
- 9.2.2 Notwithstanding the forgoing, the parties acknowledge that in the course of the provision of services hereunder, Contractor or its officers, employees, and agents, may have inadvertent access to patient medical records/patient information. Contractor understands and agrees that neither it nor its officers, employees, or agents, are to take advantage of such access for any purpose whatsoever.
- 9.2.3 Additionally, in the event of such inadvertent access, Contractor and its officers, employees, and agents, shall maintain the confidentiality of any information obtained and shall notify Director that such access has been gained immediately, or upon the first reasonable opportunity to do so. In the event of any access, whether inadvertent or intentional, Contractor shall indemnify, defend, and hold harmless County, its officers, employees, and agents, from and against any and all liability, including but not limited to, actions, claims, costs, demands, expenses, and fees (including attorney and expert witness fees) arising from or connected with Contractor's or its officers', employees', or agents', access to patient medical records/patient information. Contractor agrees to provide appropriate training to its employees regarding their obligations as described hereinabove.
- 9.3 Intentionally Omitted
- 9.4 Intentionally Omitted
- 9.5 Intentionally Omitted

9.6 Data Destruction

Contractor(s) and Vendor(s) that have maintained, processed, or stored the County of Los Angeles' ("County") data and/or information, implied or expressed, have the sole responsibility to certify that the data and information have been appropriately destroyed consistent with the National Institute of Standards and Technology (NIST) Special Publication SP 800-88 titled *Guidelines for Media Sanitization*. Available at:

http://csrc.nist.gov/publications/PubsDrafts.html#SP-800-88 Rev.%201

The data and/or information may be stored on purchased, leased, or rented electronic storage equipment (e.g., printers, hard drives) and electronic devices (e.g., servers, workstations) that are geographically located within the County, or external to the County's boundaries. The District must receive within ten (10) business days, a signed document from Contractor(s) and Vendor(s) that certifies and validates the data and information were placed in one or more of the following stored states: unusable, unreadable, and indecipherable.

Vendor shall certify that any District data stored on purchased, leased, or rented electronic storage equipment and electronic devices, including, but not limited to printers, hard drives, servers, and/or workstations are destroyed consistent with the current National Institute of Standard and Technology (NIST) Special Publication SP-800-88, *Guidelines for Media Sanitization.* Vendor shall provide District with written certification, within ten (10) business days of removal of any electronic storage equipment and devices that validates that any and all District data was destroyed and is unusable, unreadable, and/or undecipherable.

9.7 Local Small Business Enterprise (LSBE) Preference Program

- 9.7.1 This Contract is subject to the provisions of the County's ordinance entitled LSBE Preference Program, as codified in Chapter 2.204 of the Los Angeles County Code.
- 9.7.2 The Contractor shall not knowingly and with the intent to defraud, fraudulently obtain, retain, attempt to obtain or retain, or aid another in fraudulently obtaining or retaining or attempting to obtain or retain certification as a LSBE.
- 9.7.3 The Contractor shall not willfully and knowingly make a false statement with the intent to defraud, whether by

affidavit, report, or other representation, to a County official or employee for the purpose of influencing the certification or denial of certification of any entity as a LSBE.

- 9.7.4 If the Contractor has obtained certification as a LSBE by reason of having furnished incorrect supporting information or by reason of having withheld information, and which knew, or should have known, the information furnished was incorrect or the information withheld was relevant to its request for certification, and which by reason of such certification has been awarded this contract to which it would not otherwise have been entitled, shall:
 - 1. Pay to the District any difference between the contract amount and what the District's costs would have been if the contract had been properly awarded;
 - In addition to the amount described in subdivision (1), be assessed a penalty in an amount of not more than ten (10) percent of the amount of the contract; and
 - 3. Be subject to the provisions of Chapter 2.202 of the Los Angeles County Code (Determinations of Contractor Non-responsibility and Contractor Debarment).

The above penalties shall also apply to any business that has previously obtained proper certification, however, as a result of a change in their status would no longer be eligible for certification, and fails to notify the State and the Department of Consumer and Business Affairs of this information prior to responding to a solicitation or accepting a contract award.

9.8 Intentionally Omitted

9.9 Intentionally Omitted

9.10 Mandatory Requirement to Register on County's WebVen

As a pre-condition to the award of this Contract, Contractor represents and warrants that it has registered in Los Angeles County's vendor registration system (hereafter "WebVen"). The WebVen contains Contractor's business profile and identifies the goods/services being provided by Contractor. Contractor shall ensure that it updates its vendor profile whenever changes occur to Contractor's operations by accessing the WebVen site located online at: <u>http://camisvr.co.la.ca.us/webven</u> County shall use the data obtained from Contractor's WebVen profile to ensure that Contractor's information is consistent with Contract records (e.g., Contractor's legal name, as reflected in its WebVen profile, shall be used in Contract documents).

9.11 Limitation on Corporate Acts

- 9.11.1 Contractor shall not amend its articles of incorporation or bylaws, move to dissolve or transfer any assets obtained using Contract Funds, or take any other steps which may materially affect the performance of this Contract without first notifying the District in writing no less than thirty (30) days prior to said action. Contractor shall notify the District's Contract Administrator immediately in writing of any change in Contractor's corporate name.
- 9.11.2 If, in the District's sole discretion, the steps taken by Contractor are determined to materially affect Contractor's performance of this Contract, the District may, at its sole discretion, take any (or all) of the following actions:
 - 9.11.2.1 Require Contractor to remedy the areas that affect Contractor's ability to perform its obligations under this Contract.
 - 9.11.2.2 Suspend Contractor from performing (and receiving payment for) Contract tasks until a remedy has been reached.
 - 9.11.2.3 Terminate this Contract pursuant to Paragraph 8.43 (Termination for Default).

9.12 Modifications

This Contract fully expresses the agreement of the parties. Any modification to this Contract must be by means of a separate written document approved by the District. No oral conversation between any officer, employee or agent of the parties shall modify or otherwise amend this Contract in any way.

9.13 Remedies of Non-Compliance

Contractor agrees to comply with the requirements set forth in the entirety of this Contract as well as the requirements contained in any applicable directives, notices, guidelines and instructions used by the District. Contractor's failure to comply with such requirements shall subject Contractor to remedies which are available under this Contract and as provided by law. These remedies include but are not limited to the following: suspension of payment(s); suspension of Service(s); assessment and collection of liquidated damages; de-obligation of Contract Funds (for purposes of this Contract, de-obligation is the partial or full removal of Contract Funds from Contractor); debarment; and/or termination of Contract. The District shall have the sole discretion to determine which remedy(ies) will be applied as a result of Contractor's noncompliance.

9.14 Suspension

- 9.14.1 Contractor may be placed on suspension if District determines that Contractor is not in compliance with any Service, Work, task, deliverable or requirement outlined in this Contract and/or Contractor has demonstrated a consistent and significant lack of achievement of the Contract goals (including, but not limited to, meeting the requirements for work performance, the Pricing Sheet, staffing, administration, etc.). The District shall notify the Contractor in writing in the event that Contractor is placed on suspension.
- 9.14.2 Suspension as used herein shall mean a specified period of time (as determined by the District) during which the District shall withhold payment from Contractor. During the suspension, Contractor has a continuing obligation to remedy the areas of non-compliance which have been identified by the District or its duly authorized representative(s). The District shall monitor Contractor's adherence to such remedy(ies) during the suspension period. When applicable, the District may also provide the Contractor with a written determination stating whether or not the Contractor may continue to provide non-suspended Services, if any, during the suspension period.
- 9.14.3 District's written notice of suspension shall set forth the conditions of Contractor's non-compliance as well as the period in which Contractor must correct noted deficiencies.

In response to the notice of suspension, Contractor shall submit a written Corrective Action Plan to the District's Contract Manager within ten (10) days of the date indicated on the notice from the District. Contractor's Corrective Action Plan shall address all of the deficiencies noted by the District.

- 9.14.4 The District shall review Contractor's Corrective Action Plan, and will determine whether it meets the requirements for District's approval. The District reserves the right to suspend/deduct payments for or to terminate all or any part of this Contract (and/or any Contractor's other contracts with the District) when Contractor submits a Corrective Action Plan that is not acceptable to the District.
- 9.14.5 Contractor shall implement the Corrective Action Plan upon receiving District's final written approval of the Corrective Action Plan. Contractor's failure to comply with an approved Corrective Action Plan will be cause for material breach of Contract upon which the District may pursue the remedies for default of Contract.

9.15 Transition of Contract Services

9.15.1 Completion of Contract

9.15.1.1 Within sixty (60) calendar days prior to the expiration of this Contract (or shorter time period if notified in writing by District), Contractor shall allow the District or a newly selected contractor a transition period for orientation purposes and the orderly transition of Contractor's current Services without additional cost to the District. Contractor shall continue to provide Services timely and accurately so that the Services are current at the expiration of this Contract.

9.15.2 Transition Plan

9.15.2.2 If this Contract (or any part thereof) is terminated pursuant to any of the termination provisions outlined herein or if it expires pursuant to Paragraph 4.0 (Term of Contract), Contractor shall provide a Transition Plan to the District. Contractor shall submit said Transition Plan to the District's Contract Administrator within the timeframe designated by the District in the notice of termination or Contractor shall submit it at least sixty (60) days prior to the expiration of this Contract as noted in Paragraph 4.0 (Term of Contract).

9.15.2.3 The District shall review Contractor's Transition Plan and will determine whether it meets the requirements for District's approval. The District reserves the right to suspend/deduct payments under this Contract and/or under any of Contractor's other contracts with the District when Contractor submits a Transition Plan that is not acceptable to the District. Contractor shall adhere to the Transition Plan which, at a minimum, shall include all of the elements outlined below.

9.15.3 Elements of the Transition Plan

- 9.15.3.1 Contractor's method to communicate with other organizations that can assist in locating alternative Services.
- 9.15.3.2 Contractor's plan to assure for the provision of adequate staff to provide continued Services through the remaining term of this Contract.
- 9.15.3.3 Any additional information which may be necessary to ensure Contract Services are being provided.

9.15.4 Implementation of the Transition Plan

Contractor shall implement the Transition Plan that is approved by the District. Contractor's failure to provide and/or implement the Transition Plan as prescribed herein shall mean that the District will provide Contractor a Transition Plan and Contractor will implement the Transition Plan provided by the District. The District will monitor Contractor's progress in carrying out all elements of the Transition Plan. IN WITNESS WHEREOF, the Board of Supervisors of the County of Los Angeles has caused this Contract to be executed by the Fire Chief of the Consolidated Fire Protection District of Los Angeles County (or designee) and approved by County Counsel, and Contractor has caused this Contract to be executed in its behalf by its duly authorized officer, this 1st day of July, 2021.

CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY

Ву _____

Fire Chief

CathyJon Enterprises, Inc. By <u>dba HB Staffing</u> Contractor

Signed: _____

Printed: Cathy Volpe

Title: President

APPROVED AS TO FORM:

RODRIGO CASTRO-SILVA County Counsel

By_

Senior Deputy County Counsel

ATTACHMENT A

CONTRACT NO.: _____

CONTRACT



BY AND BETWEEN

CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY

AND

HOWROYD-WRIGHT EMPLOYMENT AGENCY, INC. DBA APPLEONE EMPLOYMENT SERVICES

FOR

TEMPORARY PERSONNEL SERVICES

CONTRACT PROVISIONS TABLE OF CONTENTS

PARA	-		PAGE			
		EXHIBITS				
		CABLE DOCUMENTS				
		ITIONS				
	WORK					
-		OF CONTRACT	-			
		RACT SUM				
-	5.1	Maximum Contract Sum	-			
		Written Approval for Reimbursement				
	5.3	Notification of 75% of Total Contract Sum				
	5.4	No Payment for Services Provided Following Expiration-Termination of Contract				
	5.5	Invoices and Payments	7			
	5.6	Intentionally Omitted	9			
	5.7 5.8	Default Method of Payment: Direct Deposit or Electronic Funds Transf Travel, Transportation, and Parking				
6	ADMINISTRATION OF CONTRACT- DISTRICT					
	6.1	District Administration	10			
	6.2	District's Project Director				
	6.3	District's Project Manager	10			
	6.4	District's Contract Administrator	11			
7		NISTRATION OF CONTRACT-CONTRACTOR	11			
	7.1	Contractor Administration	11			
	7.2	Contractor's Project Manager	11			
	7.3	Approval of Contractor's Staff	12			
	7.4	Contractor's Staff Identification	12			
	7.5	Background and Security Investigations	12			
	7.6	Confidentiality	13			
8	STAN	STANDARD TERMS AND CONDITIONS14				
	8.1	Amendments	14			
	8.2	Assignment and Delegation/Mergers or Acquisitions	15			
	8.3	Authorization Warranty	16			
	8.4	Budget Reductions	16			
	8.5	Complaints				
	8.6	Compliance with Applicable Law	17			

8.7	Compliance with Civil Rights Laws	. 18
8.8	Compliance with the County's Jury Service Program	. 18
8.9	Conflict of Interest	. 20
8.10	Consideration of Hiring County Employees Targeted for Layoff or Re- Employment List	. 21
8.11	Consideration of Hiring GAIN-GROW Participants	. 21
8.12	Contractor Responsibility and Debarment	. 21
8.13	Contractor's Acknowledgement of County's Commitment to Safely Surrendered Baby Law	. 24
8.14	Contractor's Warranty of Adherence to County's Child Support	
	Compliance Program	. 25
8.15	District's Quality Assurance Plan	. 25
8.16	Damage to District Facilities, Buildings or Grounds	. 25
8.17	Employment Eligibility Verification	. 26
8.18	Counterparts and Electronic Signatures and Representations	. 26
8.19	Fair Labor Standards	. 27
8.20	Force Majeure	. 27
8.21	Governing Law, Jurisdiction, and Venue	. 28
8.22	Independent Contractor Status	. 28
8.23	Indemnification	. 29
8.24	General Provisions for all Insurance Coverage	. 29
8.25	Insurance Coverage	. 34
8.26	Liquidated Damages	. 35
8.27	Most Favored Public Entity	. 37
8.28	Nondiscrimination and Affirmative Action	. 37
8.29	Non Exclusivity	. 38
8.30	Notice of Delays	. 38
8.31	Notice of Disputes	. 39
8.32	Notice to Employees Regarding the Federal Earned Income Credit	. 39
8.33	Notice to Employees Regarding the Safely Surrendered Baby Law	. 39
8.34	Notices	. 39
8.35	Prohibition Against Inducement or Persuasion	. 40
8.36	Public Records Act	. 40
8.37	Publicity	. 40
8.38	Record Retention and Inspection-Audit Settlement	. 41
8.39	Recycled Bond Paper	. 42
8.40	Subcontracting	. 43

8.41	Termination for Breach of Warranty to Maintain Compliance with County's Child Support Compliance Program	. 44
8.42	Termination for Convenience	. 44
8.43	Termination for Default	. 45
8.44	Termination for Improper Consideration	. 46
8.45	Termination for Insolvency	. 47
8.46	Termination for Non-Adherence of County Lobbyist Ordinance	. 48
8.47	Termination for Non-Appropriation of Funds	. 48
8.48	Validity	. 48
8.49	Waiver	. 48
8.50	Warranty Against Contingent Fees	. 48
8.51	Warranty of Compliance with County's Defaulted Property Tax	
	Reduction Program	. 49
8.52	Termination for Breach of Warranty to Maintain Compliance with	
	County's Defaulted Property Tax Reduction Program	. 49
8.53	Time off for Voting	. 49
8.54	Compliance with County's Zero Tolerance Policy on Human Trafficking	. 50
8.55	Intentionally Omitted	. 50
8.56	Compliance with Fair Chance Employment Practices	
8.57	Compliance with the County Policy of Equity	. 50
8.58	Prohibition from Participation in Future Solicitation(s)	. 51
UNIQ	UE TERMS AND CONDITIONS	
9.1	Intentionally Omitted	. 51
9.2	Health Insurance Portability and Accountability Act of 1996 ("HIPAA")	. 51
9.3	Intentionally Omitted	. 52
9.4	Intentionally Omitted	. 52
9.5	Intentionally Omitted	. 52
9.6	Data Destruction	. 52
9.7	Intentionally Omitted	. 53
9.8	Intentionally Omitted	
9.9	Intentionally Omitted	. 53
9.10	Mandatory Requirement to Register on County's WebVen	. 53
9.11	Limitation on Corporate Acts	. 53
9.12	Modifications	. 54
9.13	Remedies of Non-Compliance	. 54
9.14	Suspension	. 54
9.15	Transition of Contract Services	. 56

9

NATURES

STANDARD EXHIBITS

- A Statement of Work
- B Pricing Sheet
- C Independent Price Determination
- D Contractor's EEO Certification
- E District's Administration
- F Contractor's Administration
- G Contractor's Confidentiality Acknowledgement
- H Jury Service Ordinance
- I Safely Surrendered Baby Law

CONTRACT BETWEEN CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY

AND

HOWROYD-WRIGHT EMPLOYMENT AGENCY, INC. DBA APPLEONE EMPLOYMENT SERVICES

FOR

TEMPORARY PERSONNEL SERVICES

This Contract ("Contract") made and entered into this 1st day of July, 2021 by and between the Consolidated Fire Protection District of Los Angeles County, hereinafter referred to as "District" and Howroyd-Wright Employment Agency, Inc. dba AppleOne Employment Services, hereinafter referred to as "Contractor." Contractor is located at 327 W. Broadway, Glendale, CA 91204.

RECITALS

WHEREAS, the District may contract with private businesses for temporary personnel services when certain requirements are met; and

WHEREAS, the Contractor is a private firm specializing in providing temporary personnel services; and

WHEREAS, the District is authorized under California Government Code Section 31000 to contract for special services, and under 31000.4 to contract with temporary help firms to assist the District during any peak load, temporary absence, or emergency other than a labor dispute; and

WHEREAS, the District is in need of temporary personnel services on an asneeded and intermittent basis and is authorized under Health and Safety Code Section 13861 to contract for services; and

NOW THEREFORE, in consideration of the mutual covenants contained herein, and for good and valuable consideration, the parties agree to the following:

1 APPLICABLE DOCUMENTS

Exhibits A, B, C, D, E, F, G, H, and I are attached to and form a part of this Contract. In the event of any conflict or inconsistency in the definition or interpretation of any word, responsibility, schedule, or the contents or description of any task, deliverable, goods, service, or other work, or otherwise between the base Contract and the Exhibits, or between Exhibits, such conflict or inconsistency shall be resolved by giving precedence first to the terms and conditions of the Contract and then to the Exhibits according to the following priority.

Standard Exhibits:

- 1.1 Exhibit A Statement of Work
- 1.2 Exhibit B Pricing Sheet
- 1.3 Exhibit C Independent Price Determination
- 1.4 Exhibit D Contractor's EEO Certification
- 1.5 Exhibit E District's Administration
- 1.6 Exhibit F Contractor's Administration
- 1.7 Exhibit G Acknowledgement and Confidentiality Agreement
- 1.8 Exhibit H Jury Service Ordinance
- 1.9 Exhibit I Safely Surrendered Baby Law

2 **DEFINITIONS**

The headings herein contained are for convenience and reference only and are not intended to define the scope of any provision thereof. The following words as used herein shall be construed to have the following meaning, unless otherwise apparent from the context in which they are used.

- 2.1 **Board of Supervisors (Board):** The Board of Supervisors of the County of Los Angeles; the governing body of the District and the County of Los Angeles.
- 2.2 **Contract:** This agreement executed between District and Contractor. Included are all supplemental agreements amending or extending the service to be performed. The Contract sets forth the terms and conditions for the issuance and performance of all tasks, deliverables, services and other work
- 2.3 **Contractor:** The person or persons, sole proprietor, partnership, joint venture, corporation or other legal entity who has entered into an agreement with the District to perform or execute the work covered by this Contract.
- 2.4 <u>**Contractor Project Manager:**</u> The person designated by the Contractor to administer the Contract operations under this Contract.
- 2.5 **<u>County</u>**: The County of Los Angeles, a political subdivision of the State of California.
- 2.6 **<u>Dav(s)</u>**: Calendar day(s) unless otherwise specified.
- 2.7 <u>District</u>: The Consolidated Fire Protection District of Los Angeles County.
- 2.8 **District Project Director:** Person designated by District with authority for District on contractual or administrative matters relating to this Contract that cannot be resolved by the District's Project Manager.
- 2.9 <u>**District Project Manager</u>:** Person designated by District's Project Director to manage the operations under this Contract.</u>
- 2.10 **Fiscal Year**: The twelve (12) month period beginning July 1st and ending the following June 30th.
- 2.11 <u>Statement of Work</u>: The directions, provisions, and requirements provided herein and special provisions pertaining to the method, frequency, manner and place of performing the contract services.

- 2.12 **<u>Subcontract</u>**: An agreement by the contractor to employ a subcontractor to provide services to fulfill this contract.
- 2.13 **Subcontractor:** Any individual, person or persons, sole proprietor, firm, partnership, joint venture, corporation, or other legal entity furnishing supplies, services of any nature, equipment, and/or materials to contractor in furtherance of contractor's performance of this contract, at any tier, under oral or written agreement.

3 WORK

- 3.1 Pursuant to the provisions of this Contract, the Contractor shall fully perform, complete and deliver on time, all tasks, deliverables, services and other work as set forth in herein.
- 3.2 If the Contractor provides any tasks, deliverables, goods, services, or other work, other than as specified in this contract, the same shall be deemed to be a gratuitous effort on the part of the contractor, and the contractor shall have no claim whatsoever against the District.

4 TERM OF CONTRACT

4.1 The term of this Contract shall be three (3) years commencing after approval by County's Board of Supervisors (Board), and execution by the Fire Chief or his designee, unless sooner terminated or extended, in whole or in part, as provided in this Contract.

The District shall have the sole option to extend this Contract term for up to two (2) additional one-year periods and twelve (12) month to month extension options, for a maximum total Contract term of six (6) years, pursuant to the same terms and conditions. Each such extension option may be exercised at the sole discretion of the Fire Chief or his designee as authorized by the Board.

- 4.2 The District maintains databases that track/monitor contractor performance history. Information entered into such databases may be used for a variety of purposes, including determining whether the District will exercise a contract term extension option.
- 4.3 The Contractor shall notify the District when this Contract is within six (6) months of the expiration of the term as provided for hereinabove. Upon occurrence of this event, the Contractor shall send written notification to the District at the address herein provided in Exhibit E District's Administration.

5 CONTRACT SUM

5.1 Maximum Contract Sum

5.1.1 The amount the District shall expend from its own funds during the Contract's entire term for as-needed Temporary Personnel Services shall not exceed \$690,000 per Contract year. The Maximum Contract Sum shall be the maximum monetary amount available that is payable by the District to the Contractor for supplying all the Services, Deliverables, Work, etc.

The annual Maximum Contract Sum under the terms of this Contract shall be \$690,000, shared amongst all contracts for temporary personnel services.

This is not a commitment or offer on the part of the District to expend the Maximum Contract Sum. The Contractor shall perform and complete all Services required of the Contractor under this Contract as set forth in Exhibit B (Pricing Sheet), but in any event, not in excess of the Maximum Contract Sum.

5.1.2 The Contractor acknowledges and agrees the Maximum Contract Sum is an all-inclusive, not-to-exceed price that cannot be adjusted for any costs or expenses whatsoever of Contractor. This Contract includes the full amount of compensation and reimbursement the District will be asked to provide to the Contractor in order for the Contractor to fully perform all of its obligations under this Contract, with such amount of compensation and reimbursement subject to any executed Amendments if applicable. The Contractor understands the District is entering into this Contract in reliance upon the premise that the Contractor shall fully perform all of its obligations under this Contract without seeking any additional compensation or reimbursement beyond that already provided for in this Contract, subject to any Amendments, if applicable. It is the Contractor's risk and responsibility to achieve and timely deliver the Services in accordance with the requirements of the Contract.

5.2 Written Approval for Reimbursement

The Contractor shall not be entitled to payment or reimbursement for any tasks or services performed, nor for any incidental or administrative expenses whatsoever incurred in or incidental to performance hereunder, except as specified herein. Assumption or takeover of any of the Contractor's duties, responsibilities, or obligations, or performance of same by any person or entity other than the Contractor, whether through assignment, subcontract, delegation, merger, buyout, or any other mechanism, with or without consideration for any reason whatsoever, shall not occur except with the District's express prior written approval.

5.3 Notification of 75% of Total Contract Sum

The Contractor shall maintain a system of record keeping that will allow the Contractor to determine when it has incurred seventy-five percent (75%) of the total contract sum under this Contract. Upon occurrence of this event, the Contractor shall send written notification to the District at the address herein provided in Exhibit E - District's Administration.

5.4 No Payment for Services Provided Following Expiration-Termination of Contract

The Contractor shall have no claim against District for payment of any money or reimbursement, of any kind whatsoever, for any service provided by the Contractor after the expiration or other termination of this Contract. Should the Contractor receive any such payment it shall immediately notify the District and shall immediately repay all such funds to the District. Payment by the District for services rendered after expiration-termination of this Contract shall not constitute a waiver of the District's right to recover such payment from the Contractor. This provision shall survive the expiration or other termination of this Contract.

5.5 Invoices and Payments

5.5.1 The Contractor shall invoice the District only for providing the tasks, deliverables, goods, services, and other work specified in Exhibit A - Statement of Work and elsewhere hereunder. The Contractor shall prepare invoices, which shall include the charges owed to the Contractor by the District under the terms of this Contract. The Contractor's payments shall be as provided in Exhibit B - Pricing Sheet and the Contractor shall be paid only for the tasks, deliverables, goods, services, and other work approved in writing by the District. If the District does not approve work in writing no payment shall be due to the Contractor for that work.

- 5.5.2 The Contractor's invoices shall be priced in accordance with Exhibit B Pricing Sheet.
- 5.5.3 The Contractor's invoices shall contain the information set forth in Exhibit A - Statement of Work describing the tasks, deliverables, goods, services, work hours, and facility and/or other work for which payment is claimed.
- 5.5.4 The Contractor shall submit the monthly invoices to the District by the 15th calendar day of the month following the month of service.
- 5.5.5 Payment to Contractor shall be made on an arrears basis, upon acceptance of completed work by the District, provided the Contractor is not in default under any provisions of this Contract. The Contractor shall email one (1) copy of the invoice to the following:
 - 1. <u>Bianca.Moya@fire.lacounty.gov</u> <u>Vinrose.Napoles@fire.lacounty.gov</u> for review and approval of all invoices; and
 - 2. <u>Fire-InvoiceSubmission@fire.lacounty.gov</u> for payment of all invoices.

The Contractor's invoices shall include the following:

- Contract Number
- Date(s) of Service
- Brief description of services
- A breakdown of labor hours and hourly rate i.e.: 3 hours @ \$20/hour = \$60.00.
 NO OVERTIME MAY BE ACCUMULATED OR PAID.
- Employee Name and Employee Number of District Employee who ordered or authorized the service.
- Signature of authorized District employee. Contractor's failure to obtain the signature of District employee authorizing the work may result in a delay of payment.

5.5.6 **District Approval of Invoices**

All invoices submitted by the Contractor for payment must have the written approval of the District's Project Manager prior to any payment thereof. In no event shall the District be liable or responsible for any payment prior to such written approval. Approval for payment will not be unreasonably withheld.

5.5.7 Local Small Business Enterprises – Prompt Payment Program

Certified Local Small Business Enterprises (LSBEs) will receive prompt payment for services they provide to the District. Prompt payment is defined as fifteen (15) calendar days after receipt of an undisputed invoice.

5.6 Intentionally Omitted

5.7 Default Method of Payment: Direct Deposit or Electronic Funds Transfer

- 5.7.1 The County, at its sole discretion, has determined the most efficient and secure default form of payment for goods and/or services provided under an agreement/ contract with the County shall be Electronic Funds Transfer (EFT) or direct deposit, unless an alternative method of payment is deemed appropriate by the Auditor-Controller (A-C).
- 5.7.2 The Contractor shall submit a direct deposit authorization request via the website https://directdeposit.lacounty.gov with banking and vendor information, and any other information that the A-C determines is reasonably necessary to process the payment and comply with all accounting, record keeping, and tax reporting requirements.
- 5.7.3 Any provision of law, grant, or funding agreement requiring a specific form or method of payment other than EFT or direct deposit shall supersede this requirement with respect to those payments.
- 5.7.4 At any time during the duration of the agreement/contract, a Contractor may submit a written request for an exemption to this requirement. Such request must be based on specific legal, business or operational needs and explain why the

payment method designated by the A-C is not feasible and an alternative is necessary. The A-C, in consultation with the contracting department(s), shall decide whether to approve exemption requests.

5.8 Travel, Transportation, and Parking

- 5.8.1 All travel-related costs are the responsibility of the Contractor. The District will not be responsible for paying or reimbursing Contractor or its employees for any travel related costs.
- 5.8.2 The District is not responsible for paying or reimbursing parking or transportation costs for temporary personnel.

6 ADMINISTRATION OF CONTRACT – DISTRICT

6.1 District Administration

A listing of all District Administration referenced in the following subparagraphs are designated in Exhibit E - District's Administration. The District will notify the Contractor in writing of any change in the names or addresses shown.

6.2 District's Project Director

The responsibilities of the District's Project Director include:

- Coordinating with Contractor and ensuring Contractor's performance of the Contract; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby; and
- Upon request of the Contractor, providing direction to the Contractor, as appropriate in areas relating to District policy, information requirements, and procedural requirements; however, in no event, shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby.

6.3 District's Project Manager

The role of the District's Project Manager is to oversee the day-today administration of this Contract; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby. The District's Project Manager's responsibilities include:

- Meeting with the Contractor's Project Manager on a regular basis; and
- Inspecting any and all tasks, deliverables, goods, services, or other work provided by or on behalf of the Contractor; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby.

The District's Project Manager is not authorized to make any changes in any of the terms and conditions of this Contract and is not authorized to further obligate District in any respect whatsoever.

6.4 District's Contract Administrator

The responsibilities of the District's Contract Administrator include:

- Ensuring that the objectives of this Contract are met; and
- Making changes in the terms and conditions of this Contract in accordance with Sub-paragraph 8.1, Amendments; and
- Providing direction to Contractor in the areas relating to District policy, information requirements, and procedural requirements.

7 ADMINISTRATION OF CONTRACT - CONTRACTOR

7.1 Contractor Administration

A listing of all of Contractor's Administration referenced in the following paragraphs is designated in Exhibit F - Contractor's Administration. The Contractor will notify the District in writing of any change in the names or addresses shown.

7.2 Contractor's Project Manager

- 7.2.1 The Contractor's Project Manager is designated in Exhibit F - Contractor's Administration. The Contractor shall notify the District in writing of any change in the name or address of the Contractor's Project Manager.
- 7.2.2 The Contractor's Project Manager shall be responsible for the Contractor's day-to-day activities as related to this

Contract and shall meet and coordinate with the District's Project Manager on a regular basis.

7.3 Approval of Contractor's Staff

District has the absolute right to approve or disapprove all of the Contractor's staff performing work hereunder and any proposed changes in the Contractor's staff, including, but not limited to, the Contractor's Project Manager.

7.4 Contractor's Staff Identification

All of the Contractor's employees assigned to District facilities are required to have a District Identification (ID) badge on their person and visible at all times. The Contractor may bear the expense of any badge not returned to the District when the temporary personnel's assignment has ended.

- 7.4.1 Contractor is responsible to ensure employees have obtained a District ID badge before they are assigned to work in a District facility. Contractor personnel may be asked by a District representative to leave a District facility if they do not have the proper District ID badge on their person and Contractor personnel must immediately comply with such request.
- 7.4.2 Contractor shall notify the District within one business day when staff is terminated from working under this Contract. Contractor shall retrieve and return an employee's District ID badge to the District on the next business day after the employee has terminated employment with the Contractor.
- 7.4.3 If District requests the removal of Contractor's staff, Contractor shall retrieve and return an employee's District ID badge to the District on the next business day after the employee has been removed from working on the District's Contract.

7.5 Background and Security Investigations

7.5.1 Each of Contractor's staff performing services under this Contract, as determined by District in District's sole discretion, shall undergo and pass a background investigation to the satisfaction of District as a condition of beginning and continuing to perform services under this Contract. Such background investigation must be obtained through fingerprints submitted to the California Department of Justice to include State, local, and federal-level review, which may include, but shall not be limited to, criminal conviction information. The fees associated with the background investigation shall be at the expense of the Contractor, regardless of whether the member of Contractor's staff passes or fails the background investigation. The District, may at its sole discretion and expense, provide live scan services for Contractor's staff assigned to District facilities. The District reserves the right to request reimbursement for the live scan services.

If a member of Contractor's staff does not pass the background investigation, the District may request that the member of Contractor's staff be removed immediately from performing services under the Contract. Contractor shall comply with District's request at any time during the term of the Contract. District will not provide to Contractor or to Contractor's staff any information obtained through the District's background investigation.

- 7.5.2 District shall immediately deny or terminate facility access to any member of Contractor's staff that does not pass such investigation to the satisfaction of the District or whose background or conduct is incompatible with District facility access.
- 7.5.3 Disqualification of any member of Contractor's staff pursuant to this Paragraph 7.5 shall not relieve Contractor of its obligation to complete all work in accordance with the terms and conditions of this Contract.

7.6 Confidentiality

- 7.6.1 Contractor shall maintain the confidentiality of all records and information in accordance with all applicable Federal, State and local laws, rules, regulations, ordinances, directives, guidelines, policies and procedures relating to confidentiality, including, without limitation, County policies concerning information technology security and the protection of confidential records and information.
- 7.6.2 Contractor shall indemnify, defend, and hold harmless County, its officers, employees, and agents, from and against any and all claims, demands, damages, liabilities, losses, costs and expenses, including, without limitation,

defense costs and legal, accounting and other expert, consulting, or professional fees, arising from, connected with, or related to any failure by Contractor, its officers, employees, agents, or subcontractors, to comply with this Paragraph 7.6, as determined by County in its sole judgment. Any legal defense pursuant to contractor's indemnification obligations under this Paragraph 7.6 shall be conducted by contractor and performed by counsel selected by Contractor and approved by District. Notwithstanding the preceding sentence, District shall have the right to participate in any such defense at its sole cost and expense, except that in the event Contractor fails to provide District with a full and adequate defense, as determined by District in its sole judgment, District shall be entitled to retain its own counsel, including, without limitation, County Counsel, and to reimbursement from Contractor for all such costs and expenses incurred by County in doing so. Contractor shall not have the right to enter into any settlement, agree to any injunction, or make any admission, in each case, on behalf of District without District's prior written approval.

- 7.6.3 Contractor shall inform all of its officers, employees, agents and subcontractors providing services hereunder of the confidentiality provisions of this Contract.
- 7.6.4 Contractor shall sign and adhere to the provisions of Exhibit G - "Contractor Acknowledgement and Confidentiality Agreement."

8 STANDARD TERMS AND CONDITIONS

8.1 Amendments

- 8.1.1 For any change which affects the scope of work, term, contract sum, payments, or any term or condition included under this Contract, an amendment to the Contract shall be prepared and executed by the contractor and by the Fire Chief or his designee.
- 8.1.2 The County's Board of Supervisors or Chief Executive Officer or designee may require the addition and/or change of certain terms and conditions in the Contract during the term of this Contract. The District reserves the right to add and/or change such provisions as required by the County's Board of Supervisors or Chief Executive Officer. To implement such changes, an Amendment to the Contract shall be prepared

and executed by the contractor and by the Fire Chief or his designee.

8.1.3 The Fire Chief or his designee may at his sole discretion, authorize extensions of time as defined in Paragraph 4 - Term of Contract. The contractor agrees that such extensions of time shall not change any other term or condition of this Contract during the period of such extensions. To implement an extension of time, an Amendment to the Contract shall be prepared and executed by the contractor and by the Fire Chief or his designee.

8.2 Assignment and Delegation/Mergers or Acquisitions

- 8.2.1 The contractor shall notify the District of any pending acquisitions/mergers of its company unless otherwise legally prohibited from doing so. If the contractor is restricted from legally notifying the District of pending acquisitions/mergers, District then it should notifv the of the actual acquisitions/mergers as soon as the law allows and provide to the District the legal framework that restricted it from notifying the District prior to the actual acquisitions/mergers.
- 8.2. 2 The contractor shall not assign its rights or delegate its duties under this Contract, or both, whether in whole or in part, without the prior written consent of District, in its discretion, and any attempted assignment or delegation without such consent shall be null and void. For purposes of this paragraph, District consent shall require a written Amendment to the Contract, which is formally approved and executed by the parties. Any payments by the District to any approved designee or assignee on any claim under this Contract shall be deductible, at District's sole discretion, against the claims, which the contractor may have against the District.
- 8.2.3 Shareholders, partners, members, or other equity holders of contractor may transfer, sell, exchange, assign, or divest themselves of any interest they may have therein. However, in the event any such sale, transfer, exchange, assignment, or divestment is effected in such a way as to give majority control of contractor to any person(s), corporation, partnership, or legal entity other than the majority controlling interest therein at the time of execution of the Contract, such disposition is an assignment requiring the prior written consent of District in accordance with applicable provisions of this Contract.

8.2.4 Any assumption, assignment, delegation, or takeover of any of the contractor's duties, responsibilities, obligations, or performance of same by any person or entity other than the contractor, whether through assignment, subcontract, delegation, merger, buyout, or any other mechanism, with or without consideration for any reason whatsoever without District's express prior written approval, shall be a material breach of the Contract which may result in the termination of this Contract. In the event of such termination. District shall be entitled to pursue the same remedies against contractor as it could pursue in the event of default by contractor.

8.3 Authorization Warranty

The contractor represents and warrants that the person executing this Contract for the contractor is an authorized agent who has actual authority to bind the contractor to each and every term, condition, and obligation of this Contract and that all requirements of the contractor have been fulfilled to provide such actual authority.

8.4 Budget Reductions

In the event the County's Board of Supervisors adopts, in any fiscal year, a County Budget which provides for reductions in the salaries and benefits paid to the majority of County employees and imposes similar reductions with respect to County contracts, the District reserves the right to reduce its payment obligation under this Contract correspondingly for that fiscal year and any subsequent fiscal year during the term of this Contract (including any extensions), and the services to be provided by the contractor under this Contract shall also be reduced correspondingly. The District's notice to the contractor regarding said reduction in payment obligation shall be provided within thirty (30) calendar days of the Board's approval of such actions. Except as set forth in the preceding sentence, the contractor shall continue to provide all of the services set forth in this Contract.

8.5 Complaints

- 8.5.1 The contractor shall develop, maintain and operate procedures for receiving, investigating and responding to complaints.
- 8.5.2 Complaint Procedures

- 8.5.2.1 Within forty-five (45) business days after the Contract effective date, the contractor shall provide the District with the contractor's policy for receiving, investigating and responding to user complaints.
- 8.5.2.2 The District will review the contractor's policy and provide the contractor with approval of said plan or with requested changes.
- 8.5.2.3 If the District requests changes in the contractor's policy, the contractor shall make such changes and resubmit the plan within five (5) business days for District approval.
- 8.5.2.4 If, at any time, the contractor wishes to change the contractor's policy, the contractor shall submit proposed changes to the District for approval before implementation.
- 8.5.2.5 The contractor shall preliminarily investigate all complaints and notify the District's Project Manager of the status of the investigation within five (5) business days of receiving the complaint.
- 8.5.2.6 When complaints cannot be resolved informally, a system of follow-through shall be instituted which adheres to formal plans for specific actions and strict time deadlines.
- 8.5.2.7 Copies of all written responses shall be sent to the District's Project Manager within three (3) business days of mailing to the complainant.

8.6 Compliance with Applicable Law

- 8.6.1 In the performance of this Contract, contractor shall comply with all applicable Federal, State and local laws, rules, regulations, ordinances, directives, guidelines, policies and procedures, and all provisions required thereby to be included in this Contract are hereby incorporated herein by reference.
- 8.6.2 Contractor shall indemnify, defend, and hold harmless County, its officers, employees, and agents, from and against any and all claims, demands, damages, liabilities, losses, costs, and expenses, including, without limitation,

defense costs and legal, accounting and other expert, consulting or professional fees, arising from, connected with, or related to any failure by contractor, its officers, employees, agents, or subcontractors, to comply with any such laws, rules, regulations, ordinances, directives, guidelines, policies, or procedures, as determined by County in its sole judgment. Any legal defense pursuant to contractor's indemnification obligations under Paragraph 8.6 (Compliance with Applicable Law) shall be conducted by contractor and performed by counsel selected by contractor and approved by County. Notwithstanding the preceding sentence, County shall have the right to participate in any such defense at its sole cost and expense, except that in the event contractor fails to provide County with a full and adequate defense, as determined by County in its sole judgment, County shall be entitled to retain its own counsel, including, without limitation, County Counsel, and to reimbursement from contractor for all such costs and expenses incurred by County in doing so. Contractor shall not have the right to enter into any settlement, agree to any injunction or other equitable relief, or make any admission, in each case, on behalf of County without County's prior written approval.

8.7 Compliance with Civil Rights Laws

The contractor hereby assures that it will comply with Subchapter VI of the Civil Rights Act of 1964, 42 USC Sections 2000 (e) (1) through 2000 (e) (17), to the end that no person shall, on the grounds of race, creed, color, sex, religion, ancestry, age, condition of physical handicap, marital status, political affiliation, or national origin, be excluded from participation in, be denied the benefits of, or be otherwise subjected to discrimination under this Contract or under any project, program, or activity supported by this Contract. The contractor shall comply with Exhibit D - Contractor's EEO Certification.

8.8 Compliance with the County's Jury Service Program

8.8.1 Jury Service Program:

This Contract is subject to the provisions of the County's ordinance entitled Contractor Employee Jury Service ("Jury Service Program") as codified in Sections 2.203.010 through 2.203.090 of the Los Angeles County Code, a copy of which

is attached as Exhibit H and incorporated by reference into and made a part of this Contract.

8.8.2 Written Employee Jury Service Policy.

- 1. Unless the contractor has demonstrated to the District's satisfaction either that the contractor is not a "contractor" as defined under the Jury Service Program (Section 2.203.020 of the County Code) or that the contractor qualifies for an exception to the Jury Service Program (Section 2.203.070 of the County Code), the contractor shall have and adhere to a written policy that provides that its Employees shall receive from the contractor, on an annual basis, no less than five days of regular pay for actual jury service. The policy may provide that Employees deposit any fees received for such jury service with the contractor or that the contractor deduct from the Employee's regular pay the fees received for jury service.
- 2. For purposes of this paragraph, "contractor" means a person, partnership, corporation or other entity which has a contract with the County or a subcontract with a County contractor and has received or will receive an aggregate sum of fifty thousand dollars (\$50,000) or more in any twelve (12) month period under one or more County contracts or subcontracts. "Employee" means any California resident who is a full-time employee of the contractor. "Full-time" means forty (40) hours or more worked per week, or a lesser number of hours if: 1) the lesser number is a recognized industry standard as determined by the District, or 2) contractor has a long-standing practice that defines the lesser number of hours as full-time. Full-time employees providing short-term, temporary services of ninety (90) days or less within a twelve (12) month period are not considered full-time for purposes of the Jury Service Program. If the contractor uses any subcontractor to perform services for the District under the Contract, the subcontractor shall also be subject to the provisions of this paragraph. The provisions of this paragraph shall be inserted into any such subcontract agreement and a copy of the Jury Service Program shall be attached to the agreement.

- 3. If the contractor is not required to comply with the Jury Service Program when the Contract commences, the contractor shall have a continuing obligation to review the applicability of its "exception status" from the Jury Service Program, and the contractor shall immediately notify the District if the contractor at any time either comes within the Jury Service Program's definition of "contractor" or if the contractor no longer gualifies for an exception to the Jury Service Program. In either event, the contractor shall immediately implement a written policy consistent with the Jury Service Program. The District may also require, at any time during the Contract and at its sole discretion, that the contractor demonstrate, to the District's satisfaction that the contractor either continues to remain outside of the Jury Service Program's definition of "contractor" and/or that the contractor continues to qualify for an exception to the Program.
- 4. Contractor's violation of this paragraph of the Contract may constitute a material breach of the Contract. In the event of such material breach, District may, in its sole discretion, terminate the Contract and/or bar the contractor from the award of future County contracts for a period of time consistent with the seriousness of the breach.

8.9 Conflict of Interest

- 8.9.1 No County employee whose position with the County enables such employee to influence the award of this Contract or any competing Contract, and no spouse or economic dependent of such employee, shall be employed in any capacity by the contractor or have any other direct or indirect financial interest in this Contract. No officer or employee of the contractor who may financially benefit from the performance of work hereunder shall in any way participate in the District's approval, or ongoing evaluation, of such work, or in any way attempt to unlawfully influence the District's approval or ongoing evaluation of such work.
- 8.9.2 The contractor shall comply with all conflict of interest laws, ordinances, and regulations now in effect or hereafter to be enacted during the term of this Contract. The contractor warrants that it is not now aware of any facts that create a conflict of interest. If the contractor hereafter becomes

aware of any facts that might reasonably be expected to create a conflict of interest, it shall immediately make full written disclosure of such facts to the District. Full written disclosure shall include, but is not limited to, identification of all persons implicated and a complete description of all relevant circumstances. Failure to comply with the provisions of this paragraph shall be a material breach of this Contract.

8.10 Consideration of Hiring County Employees Targeted for Layoff or Re-Employment List

Should the contractor require additional or replacement personnel after the effective date of this Contract to perform the services set forth herein, the contractor shall give first consideration for such employment openings to qualified, permanent County employees who are targeted for layoff or qualified, former County employees who are on a re-employment list during the life of this Contract.

8.11 Consideration of Hiring GAIN-GROW Participants

- 8.11.1 Should the contractor require additional or replacement personnel after the effective date of this Contract, the contractor shall give consideration for any such employment openings to participants in the County's Department of Public Social Services Greater Avenues for Independence (GAIN) Program or General Relief Opportunity for Work (GROW) Program who meet the contractor's minimum qualifications for the open position. For this purpose, consideration shall mean that the contractor will interview qualified candidates. The County will refer GAIN-GROW participants by job category to the contractor. Contractors shall report all job openings with job requirements to: GAINGROW@DPSS.LACOUNTY.GOV and BSERVICES@WDACS.LACOUNTY.GOV and DPSS will refer qualified GAIN/GROW job candidates.
- 8.11.2 In the event that both laid-off County employees and GAIN/GROW participants are available for hiring, County employees shall be given first priority.

8.12 Contractor Responsibility and Debarment

8.12.1 **Responsible Contractor**

A responsible contractor is a contractor who has demonstrated the attribute of trustworthiness, as well as quality, fitness, capacity and experience to satisfactorily perform the contract. It is the County's policy to conduct business only with responsible contractors.

8.12.2 Chapter 2.202 of the County Code

The contractor is hereby notified that, in accordance with Chapter 2.202 of the County Code, if the County acquires information concerning the performance of the contractor on this or other contracts which indicates that the contractor is not responsible, the County may, in addition to other remedies provided in the Contract, debar the contractor from bidding or proposing on, or being awarded, and/or performing work on County contracts for a specified period of time, which generally will not exceed five (5) years but may exceed five (5) years or be permanent if warranted by the circumstances, and terminate any or all existing contracts the contractor may have with the County.

8.12.3 Non-responsible contractor

The County may debar a contractor if the Board of Supervisors finds, in its discretion, that the contractor has done any of the following: 1) violated a term of a contract with the County or a nonprofit corporation created by the County, 2) committed an act or omission which negatively reflects on the contractor's quality, fitness or capacity to perform a contract with the County, any other public entity, or a nonprofit corporation created by the County, or engaged in a pattern or practice which negatively reflects on same, 3) committed an act or offense which indicates a lack of business integrity or business honesty, or 4) made or submitted a false claim against the County or any other public entity.

8.12.4 Contractor Hearing Board

8.12.4.1 If there is evidence that the contractor may be subject to debarment, the County will notify the contractor in writing of the evidence which is the basis for the proposed debarment and will advise the contractor of the scheduled date for a debarment hearing before the Contractor Hearing Board.

- 8.12.4.2 The Contractor Hearing Board will conduct a hearing where evidence on the proposed debarment is presented. The contractor and/or the contractor's representative shall be given an opportunity to submit evidence at that hearing. After the hearing, the Contractor Hearing Board shall prepare a tentative proposed decision, which shall contain а recommendation regarding whether the contractor should be debarred, and, if so, the appropriate length of time of the debarment. The contractor and the District shall be provided an opportunity to object to the tentative proposed decision prior to its presentation to the Board of Supervisors.
- 8.12.4.3 After consideration of any objections, or if no objections are submitted, a record of the hearing, the proposed decision, and any other recommendation of the Contractor Hearing Board shall be presented to the Board of Supervisors. The Board of Supervisors shall have the right to modify, deny, or adopt the proposed decision and recommendation of the Contractor Hearing Board.
- 8.12.4.4 If a contractor has been debarred for a period longer than five (5) years, that contractor may after the debarment has been in effect for at least five (5) years, submit a written request for review of the debarment determination to reduce the period of debarment or terminate the debarment. The County may, in its discretion, reduce the period of debarment or terminate the debarment if it finds that the contractor has adequately demonstrated one or more of the 1) elimination of the grounds for followina: which the debarment was imposed; 2) a bona fide change in ownership or management; 3) material evidence discovered after debarment was imposed; or 4) any other reason that is in the best interests of the County.
- 8.12.4.5 The Contractor Hearing Board will consider a request for review of a debarment determination only where 1) the contractor has been debarred for a period longer than five (5) years; 2) the debarment has been in effect for at least five (5)

years; and 3) the request is in writing, states one or more of the grounds for reduction of the debarment period or termination of the debarment. includes and supporting documentation. Upon receiving an appropriate request, the Contractor Hearing Board will provide notice of the hearing on the request. At the hearing, the Contractor Hearing Board shall conduct a hearing where evidence on the proposed reduction of debarment period or termination of debarment is presented. This hearing shall be conducted and the request for review decided by the Contractor Hearing Board pursuant to the same procedures as for a debarment hearing.

8.12.4.6 The Contractor Hearing Board's proposed decision shall contain a recommendation on the request to reduce the period of debarment or terminate the debarment. The Contractor Hearing Board shall present its proposed decision and recommendation to the Board of Supervisors. The Board of Supervisors shall have the right to modify, deny, or adopt the proposed decision and recommendation of the Contractor Hearing Board.

8.12.5 Subcontractors of Contractor

These terms shall also apply to subcontractors of County contractors.

8.13 Contractor's Acknowledgement of County's Commitment to Safely Surrendered Baby Law

The contractor acknowledges that the County places a high priority on the implementation of the Safely Surrendered Baby Law. The contractor understands that it is the County's policy to encourage all County contractors to voluntarily post the County's "Safely Surrendered Baby Law" poster, in Exhibit I, in a prominent position at the contractor's place of business. The contractor will also encourage its subcontractors, if any, to post this poster in a prominent position in the subcontractor's place of business. Information and posters for printing are available at www.babysafela.org.

8.14 Contractor's Warranty of Adherence to County's Child Support Compliance Program

- 8.14.1 The contractor acknowledges that the County has established a goal of ensuring that all individuals who benefit financially from the County through contracts are in compliance with their court-ordered child, family and spousal support obligations in order to mitigate the economic burden otherwise imposed upon the County and its taxpayers.
- 8.14.2 As required by the County's Child Support Compliance Program (County Code Chapter 2.200) and without limiting the contractor's duty under this Contract to comply with all applicable provisions of law, the contractor warrants that it is now in compliance and shall during the term of this Contract maintain in compliance with employment and wage reporting requirements as required by the Federal Social Security Act (42 USC Section 653a) and California Unemployment Insurance Code Section 1088.5, and shall implement all lawfully served Wage and Earnings Withholding Orders or Child Support Services Department Notices of Wage and Earnings Assignment for Child, Family or Spousal Support, pursuant to Code of Civil Procedure Section 706.031 and Family Code Section 5246(b).

8.15 District's Quality Assurance Plan

The District or its agent(s) will monitor the contractor's performance under this Contract on not less than an annual basis. Such monitoring will include assessing the contractor's compliance with all Contract terms and conditions and performance standards. Contractor deficiencies which the District determines are significant or continuing and that may place performance of the Contract in jeopardy if not corrected will be reported to the Board of Supervisors and listed in the appropriate contractor performance database. The report to the Board will include improvement/corrective action measures taken by the District and the contractor. If improvement does not occur consistent with the corrective action measures, the District may terminate this Contract or impose other penalties as specified in this Contract.

8.16 Damage to District Property, Facilities, Buildings or Grounds

8.16.1 The contractor shall repair, or cause to be repaired, at its own cost, any damage to the District's real or personal

property used or occupied by the temporary personnel beyond wear from normal usage, as determined by District, any and all damage to District facilities, buildings, or grounds caused by the contractor or employees or agents of the contractor. Such repairs shall be made immediately after the contractor has become aware of such damage, but in no event later than thirty (30) days after the occurrence.

8.16.2 If the contractor fails to make timely repairs, District may make any necessary repairs. All costs incurred by District, as determined by District, for such repairs shall be repaid by the contractor by cash payment upon demand.

8.17 Employment Eligibility Verification

- 8.17.1 The contractor warrants that it fully complies with all Federal and State statutes and regulations regarding the employment of aliens and others and that all its employees performing work under this Contract meet the citizenship or alien status requirements set forth in Federal and State statutes and regulations. The contractor shall obtain, from all employees performing work hereunder, all verification and other documentation of employment eligibility status required by Federal and State statutes and regulations including, but not limited to, the Immigration Reform and Control Act of 1986, (P.L. 99-603), or as they currently exist and as they may be hereafter amended. The contractor shall retain all such documentation for all covered employees for the period prescribed by law.
- 8.17.2 The contractor shall indemnify, defend, and hold harmless, the County, its agents, officers, and employees from employer sanctions and any other liability which may be assessed against the contractor or the County or both in connection with any alleged violation of any Federal or State statutes or regulations pertaining to the eligibility for employment of any persons performing work under this Contract.

8.18 Counterparts and Electronic Signatures and Representations

This Contract may be executed in two or more counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same Contract. The facsimile, email or electronic signature of the Parties shall be deemed to constitute original signatures, and facsimile or electronic copies hereof shall be deemed to constitute duplicate originals.

The County and the Contractor hereby agree to regard electronic representations of original signatures of authorized officers of each party, when appearing in appropriate places on the Amendments prepared pursuant to Paragraph 8.1 (Amendments) and received via communications facilities (facsimile, email or electronic signature), as legally sufficient evidence that such legally binding signatures have been affixed to Amendments to this Contract.

8.19 Fair Labor Standards

The contractor shall comply with all applicable provisions of the Federal Fair Labor Standards Act and shall indemnify, defend, and hold harmless the County and its agents, officers, and employees from any and all liability, including, but not limited to, wages, overtime pay, liquidated damages, penalties, court costs, and attorneys' fees arising under any wage and hour law, including, but not limited to, the Federal Fair Labor Standards Act, for work performed by the contractor's employees for which the County may be found jointly or solely liable.

8.20 Force Majeure

- 8.20.1 Neither party shall be liable for such party's failure to perform its obligations under and in accordance with this Contract, if such failure arises out of fires, floods, epidemics, quarantine restrictions. other natural occurrences, strikes, lockouts (other than a lockout by such party or any of such party's subcontractors), freight embargoes, or other similar events to those described above, but in every such case the failure to perform must be totally beyond the control and without any fault or negligence of such party (such events are referred to in this paragraph as "force majeure events").
- 8.20.2 Notwithstanding the foregoing, a default by a subcontractor of contractor shall not constitute a force majeure event, unless such default arises out of causes beyond the control of both contractor and such subcontractor, and without any fault or negligence of either of them. In such case, contractor shall not be liable for failure to perform, unless the goods or services to be furnished by the subcontractor were obtainable from other sources in sufficient time to permit contractor to meet the required performance

schedule. As used in this subparagraph, the term "subcontractor" and "subcontractors" mean subcontractors at any tier.

8.20.3 In the event contractor's failure to perform arises out of a force majeure event, contractor agrees to use commercially reasonable best efforts to obtain goods or services from other sources, if applicable, and to otherwise mitigate the damages and reduce the delay caused by such force majeure event.

8.21 Governing Law, Jurisdiction, and Venue

This Contract shall be governed by, and construed in accordance with, the laws of the State of California. The contractor agrees and consents to the exclusive jurisdiction of the courts of the State of California for all purposes regarding this Contract and further agrees and consents that venue of any action brought hereunder shall be exclusively in the County of Los Angeles.

8.22 Independent Contractor Status

- 8.22.1 This Contract is by and between the District and the contractor and is not intended, and shall not be construed, to create the relationship of agent, servant, employee, partnership, joint venture, or association, as between the District and the contractor. The employees and agents of one party shall not be, or be construed to be, the employees or agents of the other party for any purpose whatsoever.
- 8.22.2 The contractor shall be solely liable and responsible for providing to, or on behalf of, all persons performing work pursuant to this Contract all compensation and benefits. The District shall have no liability or responsibility for the payment of any salaries, wages, unemployment benefits, disability benefits, Federal, State, or local taxes, or other compensation, benefits, or taxes for any personnel provided by or on behalf of the contractor.
- 8.22.3 The contractor understands and agrees that all persons performing work pursuant to this Contract are, for purposes of Workers' Compensation liability, solely employees of the contractor and not employees of the District. The contractor shall be solely liable and responsible for furnishing any and all Workers' Compensation benefits to any person as a result of any injuries arising from or connected with any

work performed by or on behalf of the contractor pursuant to this Contract.

8.22.4 The contractor shall adhere to the provisions stated in Paragraph 7.6 (Confidentiality).

8.23 Indemnification

The contractor shall indemnify, defend and hold harmless the County, its Special Districts, elected and appointed officers, employees, agents and volunteers (County Indemnitees) from and against any and all liability, including but not limited to demands, claims, actions, fees, costs and expenses (including attorney and expert witness fees), to the extent caused by Contractor's negligent or gross negligence in the performance of this Contract, except for such loss or damage arising from the sole negligence or willful misconduct of the County indemnitees.

8.24 General Provisions for all Insurance Coverage

Without limiting Contractor's indemnification of District, and in the performance of this Contract and until all of its obligations pursuant to this Contract have been met, Contractor shall provide and maintain at its own expense insurance coverage satisfying the requirements specified in Paragraphs 8.24 and 8.25 of this Contract. These minimum insurance coverage terms, types and limits (the "Required Insurance") also are in addition to and separate from any other contract. The District in no way warrants that the Required Insurance is sufficient to protect the Contractor for liabilities which may arise from or relate to this Contract.

8.24.2 Evidence of Coverage and Notice to District

8.24.2.1 Certificate(s) of insurance coverage (Certificate) satisfactory to District, and a copy of an Additional Insured endorsement confirming County and its Agents (defined below) has been given Insured status under the Contractor's General Liability policy, shall be delivered to District at the address shown below and provided prior to commencing services under this Contract.

- 8.24.2.2 Renewal Certificates shall be provided to District not less than ten (10) days prior to contractor's policy expiration dates. The District reserves the right to obtain complete, certified copies of any required contractor and/or subcontractor insurance policies at any time.
- 8.24.2.3 Certificates shall identify all Required Insurance coverage types and limits specified herein, reference this Contract by name or number, and be signed by an authorized representative of the insurer(s). The Insured party named on the Certificate shall match the name of the contractor identified as the contracting party in this Contract. Certificates shall provide the full name of each insurer providing coverage, its NAIC (National Association of Insurance Commissioners) identification number, its financial rating, the amounts of any policy deductibles or self-insured retentions exceeding fifty thousand dollars (\$50,000), and list any District required endorsement forms.
- 8.24.2.4 Neither the District's failure to obtain, nor the District's receipt of, or failure to object to a noncomplying insurance certificate or endorsement, or any other insurance documentation or information provided by the contractor, its insurance broker(s) and/or insurer(s), shall be construed as a waiver of any of the Required Insurance provisions.
- 8.24.2.5 Certificates and copies of any required endorsements shall be sent to:

<u>Consolidated Fire Protection District of</u> <u>Los Angeles County</u> <u>Materials Management Division/Contracts Section</u> <u>5801 S. Eastern Avenue, Suite 100</u> Commerce, California 90040-4001

8.24.2.6 Contractor also shall promptly report to District any injury or property damage accident or incident, including any injury to a contractor employee occurring on District property, and any loss, disappearance, destruction, misuse, or theft of District property, monies or securities entrusted to contractor. Contractor also shall promptly notify District of any third-party claim or suit filed against contractor or any of its subcontractors which arises from or relates to this Contract, and could result in the filing of a claim or lawsuit against contractor and/or County.

8.24.3 Additional Insured Status and Scope of Coverage

The County of Los Angeles, it's Special Districts, Elected Officials, Officers, Agents, employees and volunteers (collectively County and its Agents) shall be provided additional insured status under contractor's General Liability policy with respect to liability arising out of contractor's ongoing and completed operations performed on behalf of the District. County and its Agents additional insured status shall apply with respect to liability and defense of suits arising out of the contractor's acts or omissions, whether such liability is attributable to the contractor or to the District. The full policy limits and scope of protection also shall apply to the County and its Agents as an additional insured, even if they exceed the County's minimum Required Insurance specifications herein. Use of an automatic additional insured endorsement form is acceptable providing it satisfies the Required Insurance provisions herein.

8.24.4 Cancellation of or Changes in Insurance

Contractor shall provide District with, or contractor's insurance policies shall contain a provision that District shall receive, written notice of cancellation or any change in Required Insurance, including insurer, limits of coverage, term of coverage or policy period. The written notice shall be provided to District at least ten (10) days in advance of cancellation for non-payment of premium and thirty (30) days in advance for any other cancellation or policy change. Failure to provide written notice of cancellation or any change in Required Insurance may constitute a material breach of the Contract, in the sole discretion of the District, upon which the District may suspend or terminate this Contract.

8.24.5 Failure to Maintain Insurance

Contractor's failure to maintain or to provide acceptable evidence that it maintains the Required Insurance shall constitute a material breach of the Contract, upon which District immediately may withhold payments due to contractor, and/or suspend or terminate this Contract. District, at its sole discretion, may obtain damages from contractor resulting from said breach. Alternatively, the District may purchase the Required Insurance, and without further notice to contractor, deduct the premium cost from sums due to contractor or pursue contractor reimbursement.

8.24.6 Insurer Financial Ratings

Coverage shall be placed with insurers acceptable to the District with A.M. Best ratings of not less than A:VII unless otherwise approved by District.

8.24.7 Contractor's Insurance Shall Be Primary

Contractor's insurance policies, with respect to any claims related to this Contract, shall be primary with respect to all other sources of coverage available to contractor. Any District maintained insurance or self-insurance coverage shall be in excess of and not contribute to any contractor coverage.

8.24.8 Waivers of Subrogation

To the fullest extent permitted by law, the contractor hereby waives its rights and its insurer(s)' rights of recovery against County under all the Required Insurance for any loss arising from or relating to this Contract. The contractor shall require its insurers to execute any waiver of subrogation endorsements which may be necessary to effect such waiver.

8.24.9 Subcontractor Insurance Coverage Requirements

Contractor shall include all subcontractors as insureds under contractor's own policies, or shall provide District with each subcontractor's separate evidence of insurance coverage. Contractor shall be responsible for verifying each subcontractor complies with the Required Insurance provisions herein, and shall require that each subcontractor name the County and contractor as additional insureds on the subcontractor's General Liability policy. Contractor shall obtain County's prior review and approval of any subcontractor request for modification of the Required Insurance.

8.24.10 Deductibles and Self-Insured Retentions (SIRs)

Contractor's policies shall not obligate the County to pay any portion of any contractor deductible or SIR. The District retains the right to require contractor to reduce or eliminate policy deductibles and SIRs as respects the District, or to provide a bond guaranteeing contractor's payment of all deductibles and SIRs, including all related claims investigation, administration and defense expenses. Such bond shall be executed by a corporate surety licensed to transact business in the State of California.

8.24.11 Claims Made Coverage

If any part of the Required Insurance is written on a claims made basis, any policy retroactive date shall precede the effective date of this Contract. Contractor understands and agrees it shall maintain such coverage for a period of not less than three (3) years following Contract expiration, termination or cancellation.

8.24.12 Application of Excess Liability Coverage

Contractors may use a combination of primary, and excess insurance policies which provide coverage as broad as ("follow form" over) the underlying primary policies, to satisfy the Required Insurance provisions.

8.24.13 Separation of Insureds

All liability policies shall provide cross-liability coverage as would be afforded by the standard ISO (Insurance Services Office, Inc.) separation of insureds provision with no insured versus insured exclusions or limitations.

8.24.14 Alternative Risk Financing Programs

The District reserves the right to review, and then approve, Contractor use of self-insurance, risk retention groups, risk purchasing groups, pooling arrangements and captive insurance to satisfy the Required Insurance provisions. The County and its Agents shall be designated as an Additional Covered Party under any approved program.

8.24.15 District Review and Approval of Insurance Requirements

The District reserves the right to review and adjust the Required Insurance provisions, conditioned upon District's determination of changes in risk exposures.

8.25 Insurance Coverage

8.25.1 Commercial General Liability insurance (providing scope of coverage equivalent to ISO policy form CG 00 01), naming County and its Agents as an additional insured, with limits of not less than:

General Aggregate: \$2 million

Products/Completed Operations Aggregate: \$1 million

Personal and Advertising Injury: \$1 million

Each Occurrence: \$1 million

- 8.25.2 Automobile Liability insurance (providing scope of coverage equivalent to ISO policy form CA 00 01) with limits of not less than \$1 million for bodily injury and property damage, in combined or equivalent split limits, for each single accident. Insurance shall cover liability arising out of contractor's use of autos pursuant to this Contract, including owned, leased, hired, and/or non-owned autos, as each may be applicable.
- 8.25.3 Workers Compensation and Employers' Liability insurance or gualified self- insurance satisfying statutory requirements, which includes Employers' Liability coverage with limits of not less than \$1 million per accident. If Contractor will provide leased employees, or, is an employee leasing or temporary staffing firm or a professional employer organization (PEO), coverage also shall include an Alternate Employer Endorsement (providing scope of coverage equivalent to ISO policy form WC 00 03 01 A) naming the County as the Alternate Employer. The written notice shall be provided to District at least ten (10) days in advance of cancellation

for non-payment of premium and thirty (30) days in advance for any other cancellation or policy change. If applicable to Contractor's operations, coverage also shall be arranged to satisfy the requirements of any federal workers or workmen's compensation law or any federal occupational disease law.

8.25.4 Unique Insurance Coverage

8.25.4.1 **Professional Liability-Errors and Omissions**

Insurance covering Contractor's liability arising from or related to this Contract, with limits of not less than \$1 million per claim and \$2 million aggregate. Further, Contractor understands and agrees it shall maintain such coverage for a period of not less than three (3) years following this Agreement's expiration, termination or cancellation.

8.25.4.2 Privacy/Network Security (Cyber) Liability

coverage providing Insurance protection against liability for (1) privacy breaches [liability arising from the loss or disclosure of confidential information no matter how it occurs]; (2) system breach; (3) denial or loss of service; (4) introduction, implantation. spread of malicious or software code; (5) unauthorized access to or use of computer systems with limits of not less than \$2 million. No exclusion/restriction for unencrypted portable devices/media may be on the policy.

8.26 Liquidated Damages

8.26.1 If, in the judgment of the Fire Chief or his designee, the contractor is deemed to be non-compliant with the terms and obligations assumed hereby, the Fire Chief or his designee, at his/her option, in addition to, or in lieu of, other remedies provided herein, may withhold the entire monthly payment or deduct pro rata from the contractor's invoice for work not performed. A description of the work not performed and the amount to be withheld or deducted from payments to the contractor from the District, will be

forwarded to the contractor by the Fire Chief or his designee, in a written notice describing the reasons for said action.

- 8.26.2 If the Fire Chief or his designee, determines that there are deficiencies in the performance of this Contract that the Fire Chief or his designee, deems are correctable by the contractor over a certain time span, the Fire Chief or his designee, will provide a written notice to the contractor to correct the deficiency within specified time frames. Should the contractor fail to correct deficiencies within said time frame, the Fire Chief, or his designee, may: (a) Deduct from the contractor's payment, pro rata, those applicable portions of the Monthly Contract Sum; and/or (b) Deduct liquidated damages. The parties agree that it will be impracticable or extremely difficult to fix the extent of actual damages resulting from the failure of the contractor to correct a deficiency within the specified time frame. The parties hereby agree that under the current circumstances a reasonable estimate of such damages is one hundred dollars (\$100) per day per infraction, or as specified in the Exhibit 2 - Performance Requirements Summary (PRS) Chart, Statement of Work Exhibits, hereunder, and that the contractor shall be liable to the District for liquidated damages in said amount. Said amount shall be deducted from the District's payment to the contractor; and/or (c) Upon giving five (5) days notice to the contractor for failure to correct the deficiencies, the District may correct any and all deficiencies and the total costs incurred by the District for completion of the work by an alternate source, whether it be District forces or separate private contractor, will be deducted and forfeited from the payment to the contractor from the District, as determined by the District.
- 8.26.3 The action noted in Paragraph 8.26.2 shall not be construed as a penalty, but as adjustment of payment to the contractor to recover the District cost due to the failure of the contractor to complete or comply with the provisions of this Contract.
- 8.26.4 This Paragraph shall not, in any manner, restrict or limit the District's right to damages for any breach of this Contract provided by law or as specified in the PRS or Paragraph 8.26.2, and shall not, in any manner, restrict or limit the District's right to terminate this Contract as agreed to herein.

8.27 Most Favored Public Entity

If the contractor's prices decline, or should the contractor at any time during the term of this Contract provide the same goods or services under similar quantity and delivery conditions to the State of California or any county, municipality, or district of the State at prices below those set forth in this Contract, then such lower prices shall be immediately extended to the District.

8.28 Nondiscrimination and Affirmative Action

- 8.28.1 The contractor certifies and agrees that all persons employed by it, its affiliates, subsidiaries, or holding companies are and shall be treated equally without regard to or because of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, in compliance with all applicable Federal and State anti-discrimination laws and regulations.
- 8.28.2 The contractor shall certify to, and comply with, the provisions of Exhibit D Contractor's EEO Certification.
- 8.28.3 The contractor shall take affirmative action to ensure that applicants are employed, and that employees are treated during employment, without regard to race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, in compliance with all applicable Federal and State anti-discrimination laws and regulations. Such action shall include, but is not limited to: employment, upgrading, demotion, transfer, recruitment or recruitment advertising, layoff or termination, rates of pay or other forms of compensation, and selection for training, including apprenticeship.
- 8.28.4 The contractor certifies and agrees that it will deal with its subcontractors, bidders, or vendors without regard to or because of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation.
- 8.28.5 The contractor certifies and agrees that it, its affiliates, subsidiaries, or holding companies shall comply with all applicable Federal and State laws and regulations to the end that no person shall, on the grounds of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, be

excluded from participation in, be denied the benefits of, or be otherwise subjected to discrimination under this Contract or under any project, program, or activity supported by this Contract.

- 8.28.6 The contractor shall allow District representatives access to the contractor's employment records during regular business hours to verify compliance with the provisions of this Paragraph 8.28 Nondiscrimination and Affirmative Action when so requested by the District.
- 8.28.7 If the District finds that any provisions of this Paragraph 8.28 (Nondiscrimination and Affirmative Action) have been violated, such violation shall constitute a material breach of this Contract upon which the District may terminate or suspend this Contract. While the District reserves the right to determine independently that the anti-discrimination provisions of this Contract have been violated, in addition, a determination by the California Fair Employment and Housing Commission or the Federal Equal Employment Opportunity Commission that the contractor has violated Federal or State anti-discrimination laws or regulations shall constitute a finding by the District that the contractor has violated the anti-discrimination provisions of this Contract.
- 8.28.8 The parties agree that in the event the contractor violates any of the anti-discrimination provisions of this Contract, the District shall, at its sole option, be entitled to the sum of five hundred dollars (\$500) for each such violation pursuant to California Civil Code Section 1671 as liquidated damages in lieu of terminating or suspending this Contract.

8.29 Non Exclusivity

Nothing herein is intended nor shall be construed as creating any exclusive arrangement with the contractor. This Contract shall not restrict District from acquiring similar, equal or like goods and/or services from other entities or sources.

8.30 Notice of Delays

Except as otherwise provided under this Contract, when either party has knowledge that any actual or potential situation is delaying or threatens to delay the timely performance of this Contract, that party shall, within one (1) business day, give notice thereof, including all relevant information with respect thereto, to the other party.

8.31 Notice of Disputes

The contractor shall bring to the attention of the District's Project Manager and/or District's Project Director any dispute between the District and the contractor regarding the performance of services as stated in this Contract. If the District's Project Manager or District's Project Director is not able to resolve the dispute, the Fire Chief or his designee shall resolve it.

8.32 Notice to Employees Regarding the Federal Earned Income Credit

The contractor shall notify its employees, and shall require each subcontractor to notify its employees, that they may be eligible for the Federal Earned Income Credit under the federal income tax laws. Such notice shall be provided in accordance with the requirements set forth in Internal Revenue Service Notice No. 1015.

8.33 Notice to Employees Regarding the Safely Surrendered Baby Law

The contractor shall notify and provide to its employees, and shall require each subcontractor to notify and provide to its employees, information regarding the Safely Surrendered Baby Law, its implementation in Los Angeles County, and where and how to safely surrender a baby. The information is set forth in Exhibit I, Safely Surrendered Baby Law of this Contract. Additional information is available at <u>www.babysafela.org</u>.

8.34 Notices

All notices or demands required or permitted to be given or made under this Contract shall be in writing and shall be hand delivered with signed receipt or mailed by first-class registered or certified mail, postage prepaid, addressed to the parties as identified in Exhibits E -District's Administration and F - Contractor's Administration. Addresses may be changed by either party giving ten (10) days prior written notice thereof to the other party. The Fire Chief or his designee shall have the authority to issue all notices or demands required or permitted by the District under this Contract.

8.35 Prohibition Against Inducement or Persuasion

Notwithstanding the above, the contractor and the District agree that, during the term of this Contract and for a period of one year thereafter, neither party shall in any way intentionally induce or persuade any employee of one party to become an employee or agent of the other party. No bar exists against any hiring action initiated through a public announcement.

8.36 Public Records Act

- 8.36.1 Any documents submitted by the contractor; all information obtained in connection with the District's right to audit and inspect the contractor's documents, books, and accounting records pursuant to Paragraph 8.38 (Record Retention and Inspection-Audit Settlement) of this Contract; as well as those documents which were required to be submitted in response to the Request for Proposals (RFP) used in the solicitation process for this Contract, become the exclusive property of the District. All such documents become a matter of public record and shall be regarded as public records. Exceptions will be those elements in the California Government Code Section 6250 et seq. (Public Records Act) and which are marked "trade secret," "confidential," or "proprietary." The District shall not in any way be liable or responsible for the disclosure of any such records including. without limitation, those so marked, if disclosure is required by law, or by an order issued by a court of competent jurisdiction.
- 8.36.2 In the event the District is required to defend an action on a Public Records Act request for any of the aforementioned documents, information, books, records, and/or contents of a proposal marked "trade secret," "confidential," or "proprietary," the contractor agrees to defend and indemnify the District from all costs and expenses, including reasonable attorney's fees, in action or liability arising under the Public Records Act.

8.37 Publicity

8.37.1 The contractor shall not disclose any details in connection with this Contract to any person or entity except as may be otherwise provided hereunder or required by law. However, in recognizing the contractor's need to identify its services and related clients to sustain itself, the District shall not inhibit the contractor from publishing its role under this Contract within the following conditions:

- 8.37.1.1 The contractor shall develop all publicity material in a professional manner; and
- 8.37.1.2 During the term of this Contract, the contractor shall not, and shall not authorize another to, publish or disseminate any commercial advertisements, press releases, feature articles, or other materials using the name of the District without the prior written consent of the District's Project Director. The District shall not unreasonably withhold written consent.
- 8.37.2 The contractor may, without the prior written consent of District, indicate in its proposals and sales materials that it has been awarded this Contract with the County of Los Angeles, provided that the requirements of this Paragraph 8.37 (Publicity) shall apply.

8.38 Record Retention and Inspection-Audit Settlement

8.38.1 The Contractor shall maintain accurate and complete financial records of its activities and operations relating to this Contract in accordance with generally accepted accounting principles, and which meet the requirements for contract accounting described in Auditor-Controller Contract Accounting and Administration Handbook. The Contractor shall also maintain accurate and complete employment and other records relating to its performance of this Contract. The contractor agrees that the District, or its authorized representatives, shall have access to and the right to examine, audit, excerpt, copy, or transcribe any pertinent transaction, activity, or record relating to this Contract. All such material, including, but not limited to, all financial records, bank statements, cancelled checks or other proof of payment, timecards, sign-in/sign-out sheets and other time and employment records, and proprietary data and information, shall be kept and maintained by the contractor and shall be made available to the District during the term of this Contract and for a period of five (5) years thereafter unless the District's written permission is given to dispose of any such material prior to such time. All such material shall be maintained by the contractor at a location in Los Angeles County, provided that if any such material is located outside Los Angeles County, then, at the District's option, the contractor shall pay the District for travel, per diem, and other costs incurred by the District to examine, audit, excerpt, copy, or transcribe such material at such other location.

- 8.38.2 In the event that an audit of the contractor is conducted specifically regarding this Contract by any Federal or State auditor, or by any auditor or accountant employed by the contractor or otherwise, then the contractor shall file a copy of such audit report with the County's Auditor-Controller within thirty (30) days of the contractor's receipt thereof, unless otherwise provided by applicable Federal or State law or under this Contract. Subject to applicable law, the District shall make a reasonable effort to maintain the confidentiality of such audit report(s). Failure on the part of the contractor to comply with any of the provisions of this subparagraph 8.38 shall constitute a material breach of this Contract.
- 8.38.3 If, at any time during the term of this Contract or within five (5) years after the expiration or termination of this Contract, representatives of the District conduct an audit of the contractor regarding the work performed under this Contract, and if such audit finds that the District's dollar liability for any such work is less than payments made by the District to the contractor, then the difference shall be either: a) repaid by the contractor to the District by cash payment upon demand or b) at the sole option of the County's Auditor-Controller, deducted from any amounts due to the contractor from the District, whether under this Contract or otherwise. If such audit finds that the District's dollar liability for such work is more than the payments made by the District to the contractor, then the difference shall be paid to the contractor by the District by cash payment, provided that in no event shall the District's maximum obligation for this Contract exceed the funds appropriated by the District for the purpose of this Contract.

8.39 Recycled Bond Paper

Consistent with the Board of Supervisors' policy to reduce the amount of solid waste deposited at the County landfills, the contractor agrees to use recycled-content paper to the maximum extent possible on this Contract.

8.40 Subcontracting

- 8.40.1 The requirements of this Contract may not be subcontracted by the contractor **without the advance written approval of the District**. Any attempt by the contractor to subcontract without the prior consent of the District may be deemed a material breach of this Contract.
- 8.40.2 If the contractor desires to subcontract, the contractor shall provide the following information promptly at the District's request:
 - 8.40.2.1 A description of the work to be performed by the subcontractor;
 - 8.40.2.2 A draft copy of the proposed subcontract; and
 - 8.40.2.3 Other pertinent information and/or certifications requested by the District.
- 8.40.3 The contractor shall indemnify, defend, and hold the County harmless with respect to the activities of each and every subcontractor in the same manner and to the same degree as if such subcontractor(s) were the contractor employees.
- 8.40.4 The contractor shall remain fully responsible for all performances required of it under this Contract, including those that the contractor has determined to subcontract, notwithstanding the District's approval of the contractor's proposed subcontract.
- 8.40.5 The District's consent to subcontract shall not waive the District's right to prior and continuing approval of any and all personnel, including subcontractor employees, providing services under this Contract. The contractor is responsible to notify its subcontractors of this District right.
- 8.40.6 The District's Project Director is authorized to act for and on behalf of the District with respect to approval of any subcontract and subcontractor employees. After approval of the subcontract by the District, contractor shall forward a fully executed subcontract to the District for their files.
- 8.40.7 The contractor shall be solely liable and responsible for all payments or other compensation to all subcontractors and their officers, employees, agents, and successors in interest

arising through services performed hereunder, notwithstanding the District's consent to subcontract.

8.40.8 The contractor shall obtain certificates of insurance, which establish that the subcontractor maintains all the programs of insurance required by the District from each approved subcontractor. Before any subcontractor employee may perform any work hereunder, contractor shall ensure delivery of all such documents to:

Consolidated Fire Protection District of Los Angeles County Materials Management Division / Contracts Section 5801 S. Eastern Avenue, Suite 100 Commerce, California 90040-4001

8.41 Termination for Breach of Warranty to Maintain Compliance with County's Child Support Compliance Program

Failure of the contractor to maintain compliance with the requirements set forth in Paragraph 8.14 (Contractor's Warranty of Adherence to County's Child Support Compliance Program) shall constitute default under this Contract. Without limiting the rights and remedies available to the District under any other provision of this Contract, failure of the contractor to cure such default within ninety (90) calendar days of written notice shall be grounds upon which the District may terminate this Contract pursuant to Paragraph 8.43 (Termination for Default) and pursue debarment of the contractor, pursuant to County Code Chapter 2.202.

8.42 Termination for Convenience

- 8.42.1 This Contract may be terminated, in whole or in part, from time to time, when such action is deemed by the District, in its sole discretion, to be in its best interest. Termination of work hereunder shall be effected by notice of termination to the contractor specifying the extent to which performance of work is terminated and the date upon which such termination becomes effective. The date upon which such termination becomes effective shall be no less than ten (10) days after the notice is sent.
- 8.42.2 After receipt of a notice of termination and except as otherwise directed by the District, the contractor shall:
 - 8.42.2.1 Stop work under this Contract on the date and to the extent specified in such notice, and

- 8.42.2.2 Complete performance of such part of the work as shall not have been terminated by such notice.
- 8.42.3 All material including books, records, documents, or other evidence bearing on the costs and expenses of the contractor under this Contract shall be maintained by the contractor in accordance with Paragraph 8.38 (Record Retention and Inspection-Audit Settlement).

8.43 Termination for Default

- 8.43.1 The District may, by written notice to the contractor, terminate the whole or any part of this Contract, if, in the judgment of District's Project Director:
 - 8.43.1.1 Contractor has materially breached this Contract; or
 - 8.43.1.2 Contractor fails to timely provide and/or satisfactorily perform any task, deliverable, service, or other work required either under this Contract; or
 - 8.43.1.3 Contractor fails to demonstrate a high probability of timely fulfillment of performance requirements under this Contract, or of any obligations of this Contract and in either case, fails to demonstrate convincing progress toward a cure within five (5) working days (or such longer period as the District may authorize in writing) after receipt of written notice from the District specifying such failure.
- 8.43.2 In the event the District terminates this Contract in whole or in part as provided in Paragraph 8.43.1, the District may procure, upon such terms and in such manner as the District may deem appropriate, goods and services similar to those so terminated. The contractor shall be liable to the District for any and all excess costs incurred by the District, as determined by the District, for such similar goods and services. The contractor shall continue the performance of this Contract to the extent not terminated under the provisions of this paragraph.
- 8.43.3 Except with respect to defaults of any subcontractor, the contractor shall not be liable for any such excess costs of

the type identified in Paragraph 8.43.2 if its failure to perform this Contract arises out of causes beyond the control and without the fault or negligence of the contractor. Such causes may include, but are not limited to: acts of God or of the public enemy, acts of the District in either its sovereign or contractual capacity, acts of Federal or State governments in their sovereign capacities, fires, floods, restrictions, epidemics, quarantine strikes. freight embargoes, and unusually severe weather; but in every case, the failure to perform must be beyond the control and without the fault or negligence of the contractor. If the failure to perform is caused by the default of a subcontractor, and if such default arises out of causes beyond the control of both the contractor and subcontractor, and without the fault or negligence of either of them, the contractor shall not be liable for any such excess costs for failure to perform, unless the goods or services to be furnished by the subcontractor were obtainable from other sources in sufficient time to permit the contractor to meet the required performance schedule. As used in this paragraph, "subcontractor(s)" the term means subcontractor(s) at any tier.

- 8.43.4 If, after the District has given notice of termination under the provisions of Paragraph 8.43 (Termination for Default) it is determined by the District that the contractor was not in default under the provisions of Paragraph 8.43 (Termination for Default) or that the default was excusable under the provisions of subparagraph 8.43.3, the rights and obligations of the parties shall be the same as if the notice of termination had been issued pursuant to Paragraph 8.42 (Termination for Convenience).
- 8.43.5 The rights and remedies of the District provided in this Paragraph 8.43 (Termination for Default) shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.44 Termination for Improper Consideration

8.44.1 The District may, by written notice to the contractor, immediately terminate the right of the contractor to proceed under this Contract if it is found that consideration, in any form, was offered or given by the contractor, either directly or through an intermediary, to any District officer, employee, or agent with the intent of securing this Contract or securing favorable treatment with respect to the award, amendment, or extension of this Contract or the making of any determinations with respect to the contractor's performance pursuant to this Contract. In the event of such termination, the District shall be entitled to pursue the same remedies against the contractor as it could pursue in the event of default by the contractor.

- 8.44.2 The contractor shall immediately report any attempt by a District officer or employee to solicit such improper consideration. The report shall be made either to the District manager charged with the supervision of the employee or to the County Auditor-Controller's Employee Fraud Hotline at (800) 544-6861.
- 8.44.3 Among other items, such improper consideration may take the form of cash, discounts, services, the provision of travel or entertainment, or tangible gifts.

8.45 Termination for Insolvency

- 8.45.1 The District may terminate this Contract forthwith in the event of the occurrence of any of the following:
 - 8.45.1.1 Insolvency of the contractor. The contractor shall be deemed to be insolvent if it has ceased to pay its debts for at least sixty (60) days in the ordinary course of business or cannot pay its debts as they become due, whether or not a petition has been filed under the Federal Bankruptcy Code and whether or not the contractor is insolvent within the meaning of the Federal Bankruptcy Code;
 - 8.45.1.2 The filing of a voluntary or involuntary petition regarding the contractor under the Federal Bankruptcy Code;
 - 8.45.1.3 The appointment of a Receiver or Trustee for the contractor; or
 - 8.45.1.4 The execution by the contractor of a general assignment for the benefit of creditors.
- 8.45.2 The rights and remedies of the District provided in this Paragraph 8.45 (Termination for Insolvency) shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.46 Termination for Non-Adherence of County Lobbyist Ordinance

The contractor, and each County Lobbyist or County Lobbying firm as defined in County Code Section 2.160.010 retained by the contractor, shall fully comply with the County's Lobbyist Ordinance, County Code Chapter 2.160. Failure on the part of the contractor or any County Lobbyist or County Lobbying firm retained by the contractor to fully comply with the County's Lobbyist Ordinance shall constitute a material breach of this Contract, upon which the District may in its sole discretion, immediately terminate or suspend this Contract.

8.47 Termination for Non-Appropriation of Funds

Notwithstanding any other provision of this Contract, the District shall not be obligated for the contractor's performance hereunder or by any provision of this Contract during any of the District's future fiscal years unless and until the County's Board of Supervisors appropriates funds for this Contract in the District's Budget for each such future fiscal year. In the event that funds are not appropriated for this Contract, then this Contract shall terminate as of June 30 of the last fiscal year for which funds were appropriated. The District shall notify the contractor in writing of any such non-allocation of funds at the earliest possible date.

8.48 Validity

If any provision of this Contract or the application thereof to any person or circumstance is held invalid, the remainder of this Contract and the application of such provision to other persons or circumstances shall not be affected thereby.

8.49 Waiver

No waiver by the District of any breach of any provision of this Contract shall constitute a waiver of any other breach or of such provision. Failure of the District to enforce at any time, or from time to time, any provision of this Contract shall not be construed as a waiver thereof. The rights and remedies set forth in this paragraph 8.49 shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.50 Warranty Against Contingent Fees

8.50.1 The contractor warrants that no person or selling agency has been employed or retained to solicit or secure this

Contract upon any Contract or understanding for a commission, percentage, brokerage, or contingent fee, excepting bona fide employees or bona fide established commercial or selling agencies maintained by the contractor for the purpose of securing business.

8.50.2 For breach of this warranty, the District shall have the right to terminate this Contract and, at its sole discretion, deduct from the Contract price or consideration, or otherwise recover, the full amount of such commission, percentage, brokerage, or contingent fee.

8.51 Warranty of Compliance with County's Defaulted Property Tax Reduction Program

Contractor acknowledges that County has established a goal of ensuring that all individuals and businesses that benefit financially from County through contract are current in paying their property tax obligations (secured and unsecured roll) in order to mitigate the economic burden otherwise imposed upon County and its taxpayers.

Unless contractor qualifies for an exemption or exclusion, contractor warrants and certifies that to the best of its knowledge it is now in compliance, and during the term of this contract will maintain compliance, with Los Angeles County Code Chapter 2.206.

8.52 Termination for Breach of Warranty to Maintain Compliance with County's Defaulted Property Tax Reduction Program

Failure of contractor to maintain compliance with the requirements set forth in Paragraph 8.51 "Warranty of Compliance with County's Defaulted Property Tax Reduction Program" shall constitute default under this contract. Without limiting the rights and remedies available to County under any other provision of this contract, failure of contractor to cure such default within ten (10) days of notice shall be grounds upon which District may terminate this contract and/or pursue debarment of contractor, pursuant to County Code Chapter 2.206.

8.53 Time Off for Voting

The contractor shall notify its employees, and shall require each subcontractor to notify and provide to its employees, information regarding the time off for voting law (Elections Code Section 14000). Not less than ten (10) days before every statewide election, every contractor and subcontractors shall keep posted conspicuously at the place of work, if practicable, or elsewhere where it can be seen as employees come or go to their place of work, a notice setting forth the provisions of Section 14000.

8.54 Compliance with County's Zero Tolerance Policy on Human Trafficking

Contractor acknowledges that the County has established a Zero Tolerance Policy on Human Trafficking prohibiting contractors from engaging in human trafficking.

If a Contractor or member of Contractor's staff is convicted of a human trafficking offense, the District shall require that the Contractor or member of Contractor's staff be removed immediately from performing services under the Contract. District will not be under any obligation to disclose confidential information regarding the offenses other than those required by law.

Disqualification of any member of Contractor's staff pursuant to this paragraph shall not relieve Contractor of its obligation to complete all work in accordance with the terms and conditions of this Contract.

8.55 Intentionally Omitted

8.56 Compliance with Fair Chance Employment Practices

Contractor shall comply with fair chance employment hiring practices set forth in California Government Code Section 12952, Employment Discrimination: Conviction History. Contractor's violation of this paragraph of the Contract may constitute a material breach of the Contract. In the event of such material breach, District may, in its sole discretion, terminate the Contract.

8.57 Compliance with the County Policy of Equity

The contractor acknowledges that the County takes its commitment to preserving the dignity and professionalism of the workplace very seriously, as set forth in the County Policy of Equity (CPOE) (<u>https://ceop.lacounty.gov/</u>). The contractor further acknowledges that the County strives to provide a workplace free from discrimination, harassment, retaliation and inappropriate conduct based on a protected characteristic, and which may violate the CPOE. The contractor, its employees and subcontractors acknowledge and certify receipt and understanding of the CPOE. Failure of the contractor, its employees or its subcontractors to uphold the County's expectations of a workplace free from harassment and discrimination, including inappropriate conduct based on a protected characteristic, may subject the contractor to termination of contractual agreements as well as civil liability.

8.58 **Prohibition from Participation in Future Solicitation(s)**

A Proposer, or a Contractor or its subsidiary or Subcontractor ("Proposer/Contractor"), is prohibited from submitting a bid or proposal in a County solicitation if the Proposer/Contractor has provided advice or consultation for the solicitation. A Proposer/Contractor is also prohibited from submitting a bid or proposal in a County solicitation if the Proposer/Contractor has developed or prepared any of the solicitation materials on behalf of the County. A violation of this provision shall result in the disqualification of the Contractor/Proposer from participation in the County solicitation or the termination or cancellation of any resultant County contract. This provision shall survive the expiration, or other termination of this Agreement.

9 UNIQUE TERMS AND CONDITIONS

9.1 Intentionally Omitted

9.2 Health Insurance Portability and Accountability Act of 1996 (HIPAA)

- 9.2.1 Contractor expressly acknowledges and agrees that the provision of services under this Agreement does not require or permit access by Contractor or any of its officers, employees, or agents, to any patient medical records/patient information. Accordingly, Contractor shall instruct its officers, employees, and agents that they are not to pursue, or gain access to, patient medical records/patient information for any reason whatsoever.
- 9.2.2 Notwithstanding the forgoing, the parties acknowledge that in the course of the provision of services hereunder, Contractor or its officers, employees, and agents, may have inadvertent access to patient medical records/patient information. Contractor understands and agrees that neither it nor its officers, employees, or agents, are to take advantage of such access for any purpose whatsoever.
- 9.2.3 Additionally, in the event of such inadvertent access, Contractor and its officers, employees, and agents, shall

maintain the confidentiality of any information obtained and shall notify Director that such access has been gained immediately, or upon the first reasonable opportunity to do so. In the event of any access, whether inadvertent or intentional, Contractor shall indemnify, defend, and hold harmless County, its officers, employees, and agents, from and against any and all liability, including but not limited to, actions, claims, costs, demands, expenses, and fees (including attorney and expert witness fees) arising from or connected with Contractor's or its officers', employees', or agents', access to patient medical records/patient information. Contractor agrees to provide appropriate training to its employees regarding their obligations as described hereinabove.

9.3 Intentionally Omitted

9.4 Intentionally Omitted

9.5 Intentionally Omitted

9.6 Data Destruction

Contractor(s) and Vendor(s) that have maintained, processed, or stored the County of Los Angeles' ("County") data and/or information, implied or expressed, have the sole responsibility to certify that the data and information have been appropriately destroyed consistent with the National Institute of Standards and Technology (NIST) Special Publication SP 800-88 titled *Guidelines for Media Sanitization*. Available at:

http://csrc.nist.gov/publications/PubsDrafts.html#SP-800-88 Rev.%201

The data and/or information may be stored on purchased, leased, or rented electronic storage equipment (e.g., printers, hard drives) and electronic devices (e.g., servers, workstations) that are geographically located within the County, or external to the County's boundaries. The District must receive within ten (10) business days, a signed document from Contractor(s) and Vendor(s) that certifies and validates the data and information were placed in one or more of the following stored states: unusable, unreadable, and indecipherable.

Vendor shall certify that any District data stored on purchased, leased, or rented electronic storage equipment and electronic devices, including, but not limited to printers, hard drives, servers, and/or workstations are destroyed consistent with the current National Institute of Standard and Technology (NIST) Special Publication SP-800-88, *Guidelines for Media Sanitization*. Vendor shall provide District with written certification, within ten (10) business days of removal of any electronic storage equipment and devices that validates that any and all District data was destroyed and is unusable, unreadable, and/or undecipherable.

9.7 Intentionally Omitted

9.8 Intentionally Omitted

9.9 Intentionally Omitted

9.10 Mandatory Requirement to Register on County's WebVen

As a pre-condition to the award of this Contract, Contractor represents and warrants that it has registered in Los Angeles County's vendor registration system (hereafter "WebVen"). The WebVen contains Contractor's business profile and identifies the goods/services being provided by Contractor. Contractor shall ensure that it updates its vendor profile whenever changes occur to Contractor's operations by accessing the WebVen site located on-line at: <u>http://camisvr.co.la.ca.us/webven</u> County shall use the data obtained from Contractor's WebVen profile to ensure that Contractor's legal name, as reflected in its WebVen profile, shall be used in Contract documents).

9.11 Limitation on Corporate Acts

- 9.11.1 Contractor shall not amend its articles of incorporation or bylaws, move to dissolve or transfer any assets obtained using Contract Funds, or take any other steps which may materially affect the performance of this Contract without first notifying the District in writing no less than thirty (30) days prior to said action. Contractor shall notify the District's Contract Administrator immediately in writing of any change in Contractor's corporate name.
- 9.11.2 If, in the District's sole discretion, the steps taken by Contractor are determined to materially affect Contractor's performance of this Contract, the District may, at its sole discretion, take any (or all) of the following actions:

- 9.11.2.1 Require Contractor to remedy the areas that affect Contractor's ability to perform its obligations under this Contract.
- 9.11.2.2 Suspend Contractor from performing (and receiving payment for) Contract tasks until a remedy has been reached.
- 9.11.2.3 Terminate this Contract pursuant to Paragraph 8.43 (Termination for Default).

9.12 Modifications

This Contract fully expresses the agreement of the parties. Any modification to this Contract must be by means of a separate written document approved by the District. No oral conversation between any officer, employee or agent of the parties shall modify or otherwise amend this Contract in any way.

9.13 Remedies of Non-Compliance

Contractor agrees to comply with the requirements set forth in the entirety of this Contract as well as the requirements contained in any applicable directives, notices, guidelines and instructions used by the District. Contractor's failure to comply with such requirements shall subject Contractor to remedies which are available under this Contract and as provided by law. These remedies include but are not limited to the following: suspension of payment(s); suspension of Service(s); assessment and collection of liquidated damages; de-obligation of Contract Funds (for purposes of this Contract, de-obligation is the partial or full removal of Contract Funds from Contractor); debarment; and/or termination of Contract. The District shall have the sole discretion to determine which remedy(ies) will be applied as a result of Contractor's noncompliance.

9.14 Suspension

9.14.1 Contractor may be placed on suspension if District determines that Contractor is not in compliance with any Service, Work, task, deliverable or requirement outlined in this Contract and/or Contractor has demonstrated a consistent and significant lack of achievement of the Contract goals (including, but not limited to, meeting the requirements for work performance, the Pricing Sheet, staffing, administration, etc.). The District shall notify the

Contractor in writing in the event that Contractor is placed on suspension.

- 9.14.2 Suspension as used herein shall mean a specified period of time (as determined by the District) during which the District shall withhold payment from Contractor. During the suspension, Contractor has a continuing obligation to remedy the areas of non-compliance which have been identified by the District or its duly authorized representative(s). The District shall monitor Contractor's adherence to such remedy(ies) during the suspension period. When applicable, the District may also provide the Contractor with a written determination stating whether or not the Contractor may continue to provide non-suspended Services, if any, during the suspension period.
- 9.14.3 District's written notice of suspension shall set forth the conditions of Contractor's non-compliance as well as the period in which Contractor must correct noted deficiencies. In response to the notice of suspension, Contractor shall submit a written Corrective Action Plan to the District's Contract Manager within ten (10) days of the date indicated on the notice from the District. Contractor's Corrective Action Plan shall address all of the deficiencies noted by the District.
- 9.14.4 The District shall review Contractor's Corrective Action Plan, and will determine whether it meets the requirements for District's approval. The District reserves the right to suspend/deduct payments for or to terminate all or any part of this Contract (and/or any Contractor's other contracts with the District) when Contractor submits a Corrective Action Plan that is not acceptable to the District.
- 9.14.5 Contractor shall implement the Corrective Action Plan upon receiving District's final written approval of the Corrective Action Plan. Contractor's failure to comply with an approved Corrective Action Plan will be cause for material breach of Contract upon which the District may pursue the remedies for default of Contract.

9.15 Transition of Contract Services

9.15.1 Completion of Contract

9.15.1.1 Within sixty (60) calendar days prior to the expiration of this Contract (or shorter time period if notified in writing by District), Contractor shall allow the District or a newly selected contractor a transition period for orientation purposes and the orderly transition of Contractor's current Services without additional cost to the District. Contractor shall continue to provide Services timely and accurately so that the Services are current at the expiration of this Contract.

9.15.2 Transition Plan

- 9.15.2.2 If this Contract (or any part thereof) is pursuant to any terminated of the termination provisions outlined herein or if it expires pursuant to Paragraph 4.0 (Term of Contract), Contractor shall provide a Transition Plan to the District. Contractor shall submit said Transition Plan to the District's Contract Administrator within the timeframe designated by the District in the notice of termination or Contractor shall submit it at least sixty (60) days prior to the expiration of this Contract as noted in Paragraph 4.0 (Term of Contract).
- 9.15.2.3 The District shall review Contractor's Transition Plan and will determine whether it meets the requirements for District's approval. The District reserves the right to suspend/deduct payments under this Contract and/or under any of Contractor's other contracts with the District when Contractor submits a Transition Plan that is not acceptable to the District. Contractor shall adhere to the Transition Plan which, at a minimum, shall include all of the elements outlined below.

9.15.3 Elements of the Transition Plan

- 9.15.3.1 Contractor's method to communicate with other organizations that can assist in locating alternative Services.
- 9.15.3.2 Contractor's plan to assure for the provision of adequate staff to provide continued Services through the remaining term of this Contract.
- 9.15.3.3 Any additional information which may be necessary to ensure Contract Services are being provided.

9.15.4 Implementation of the Transition Plan

Contractor shall implement the Transition Plan that is approved by the District. Contractor's failure to provide and/or implement the Transition Plan as prescribed herein shall mean that the District will provide Contractor a Transition Plan and Contractor will implement the Transition Plan provided by the District. The District will monitor Contractor's progress in carrying out all elements of the Transition Plan. IN WITNESS WHEREOF, the Board of Supervisors of the County of Los Angeles has caused this Contract to be executed by the Fire Chief of the Consolidated Fire Protection District of Los Angeles County (or designee) and approved by County Counsel, and Contractor has caused this Contract to be executed in its behalf by its duly authorized officer, this 1st day of July, 2021.

CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY

By Howroyd-Wright Employment Agency, In By <u>dba AppleOne Employment Service</u> Contractor	Fire Chief
Signed:	_
Printed: Michael A. Hoyal	_
Title: <u>CFO</u>	_
APPROVED AS TO FORM:	
RODRIGO CASTRO-SILVA County Counsel	
Ву	_

Senior Deputy County Counsel

ATTACHMENT A

CONTRACT NO.: _____

CONTRACT



BY AND BETWEEN

CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY

AND

TRUST TEMPORARY SERVICES, INC. DBA HELPMATES STAFFING SERVICES

FOR

TEMPORARY PERSONNEL SERVICES

CONTRACT PROVISIONS TABLE OF CONTENTS

PARA	-		PAGE		
		EXHIBITS			
		CABLE DOCUMENTS			
	APPLICABLE DOCUMENTS				
	WORK				
-	TERM OF CONTRACT				
-	5.1	Maximum Contract Sum	-		
		Written Approval for Reimbursement			
	5.3	Notification of 75% of Total Contract Sum			
	5.4	No Payment for Services Provided Following Expiration-Termination of Contract			
	5.5	Invoices and Payments	7		
	5.6	Intentionally Omitted	9		
	5.7 5.8	Default Method of Payment: Direct Deposit or Electronic Funds Transf Travel, Transportation, and Parking			
6		NISTRATION OF CONTRACT- DISTRICT	10		
	6.1	District Administration	10		
	6.2	District's Project Director			
	6.3	District's Project Manager	10		
	6.4	District's Contract Administrator	11		
7	ADMINISTRATION OF CONTRACT-CONTRACTOR		11		
	7.1	Contractor Administration	11		
	7.2	Contractor's Project Manager	11		
	7.3	Approval of Contractor's Staff	12		
	7.4	Contractor's Staff Identification	12		
	7.5	Background and Security Investigations	12		
	7.6	Confidentiality	13		
8	STAN	DARD TERMS AND CONDITIONS	14		
	8.1	Amendments	14		
	8.2	Assignment and Delegation/Mergers or Acquisitions	15		
	8.3	Authorization Warranty	16		
	8.4	Budget Reductions	16		
	8.5	Complaints			
	8.6	Compliance with Applicable Law	17		

18 20
20
21
21
21
24
25
25
25
26
26
27
27
28
28
29
29
34
35
37
37
38
38
39
39
39
39
40
40
40
41
42
43

8.41	Termination for Breach of Warranty to Maintain Compliance with County's Child Support Compliance Program	. 44
8.42	Termination for Convenience	. 44
8.43	Termination for Default	. 45
8.44	Termination for Improper Consideration	. 46
8.45	Termination for Insolvency	. 47
8.46	Termination for Non-Adherence of County Lobbyist Ordinance	. 48
8.47	Termination for Non-Appropriation of Funds	. 48
8.48	Validity	. 48
8.49	Waiver	. 48
8.50	Warranty Against Contingent Fees	. 48
8.51	Warranty of Compliance with County's Defaulted Property Tax	
	Reduction Program	. 49
8.52	Termination for Breach of Warranty to Maintain Compliance with	
	County's Defaulted Property Tax Reduction Program	
8.53	Time off for Voting	. 49
8.54	Compliance with County's Zero Tolerance Policy on Human Trafficking	. 50
8.55	Intentionally Omitted	. 50
8.56	Compliance with Fair Chance Employment Practices	
8.57	Compliance with the County Policy of Equity	. 50
8.58	Prohibition from Participation in Future Solicitation(s)	. 51
UNIQ	UE TERMS AND CONDITIONS	
9.1	Intentionally Omitted	. 51
9.2	Health Insurance Portability and Accountability Act of 1996 ("HIPAA")	. 51
9.3	Intentionally Omitted	. 52
9.4	Intentionally Omitted	. 52
9.5	Intentionally Omitted	. 52
9.6	Data Destruction	. 52
9.7	Intentionally Omitted	. 53
9.8	Intentionally Omitted	
9.9	Intentionally Omitted	. 53
9.10	Mandatory Requirement to Register on County's WebVen	. 53
9.11	Limitation on Corporate Acts	. 53
9.12	Modifications	. 54
9.13	Remedies of Non-Compliance	. 54
9.14	Suspension	. 54
9.15	Transition of Contract Services	. 56

9

NATURES

STANDARD EXHIBITS

- A Statement of Work
- B Pricing Sheet
- C Independent Price Determination
- D Contractor's EEO Certification
- E District's Administration
- F Contractor's Administration
- G Contractor's Confidentiality Acknowledgement
- H Jury Service Ordinance
- I Safely Surrendered Baby Law

CONTRACT BETWEEN CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY

AND

TRUST TEMPORARY SERVICES, INC. DBA HELPMATES STAFFING SERVICES

FOR

TEMPORARY PERSONNEL SERVICES

This Contract ("Contract") made and entered into this 1st day of July, 2021 by and between the Consolidated Fire Protection District of Los Angeles County, hereinafter referred to as "District" and Trust Temporary Services, Inc. dba Helpmates Staffing Services, hereinafter referred to as "Contractor." Contractor is located at 879 W. 190th Street, Suite 910, Gardena, CA 90248.

RECITALS

WHEREAS, the District may contract with private businesses for temporary personnel services when certain requirements are met; and

WHEREAS, the Contractor is a private firm specializing in providing temporary personnel services; and

WHEREAS, the District is authorized under California Government Code Section 31000 to contract for special services, and under 31000.4 to contract with temporary help firms to assist the District during any peak load, temporary absence, or emergency other than a labor dispute; and

WHEREAS, the District is in need of temporary personnel services on an asneeded and intermittent basis and is authorized under Health and Safety Code Section 13861 to contract for services; and

NOW THEREFORE, in consideration of the mutual covenants contained herein, and for good and valuable consideration, the parties agree to the following:

1 APPLICABLE DOCUMENTS

Exhibits A, B, C, D, E, F, G, H, and I are attached to and form a part of this Contract. In the event of any conflict or inconsistency in the definition or interpretation of any word, responsibility, schedule, or the contents or description of any task, deliverable, goods, service, or other work, or otherwise between the base Contract and the Exhibits, or between Exhibits, such conflict or inconsistency shall be resolved by giving precedence first to the terms and conditions of the Contract and then to the Exhibits according to the following priority.

Standard Exhibits:

1.1	Exhibit A -	Statement of Work
1.2	Exhibit B -	Pricing Sheet
1.3	Exhibit C -	Independent Price Determination
1.4	Exhibit D -	Contractor's EEO Certification
1.5	Exhibit E -	District's Administration
1.6	Exhibit F -	Contractor's Administration
1.7	Exhibit G -	Acknowledgement and Confidentiality Agreement
1.8	Exhibit H -	Jury Service Ordinance
1.9	Exhibit I -	Safely Surrendered Baby Law

2 **DEFINITIONS**

The headings herein contained are for convenience and reference only and are not intended to define the scope of any provision thereof. The following words as used herein shall be construed to have the following meaning, unless otherwise apparent from the context in which they are used.

- 2.1 **Board of Supervisors (Board):** The Board of Supervisors of the County of Los Angeles; the governing body of the District and the County of Los Angeles.
- 2.2 **Contract:** This agreement executed between District and Contractor. Included are all supplemental agreements amending or extending the service to be performed. The Contract sets forth the terms and conditions for the issuance and performance of all tasks, deliverables, services and other work
- 2.3 **Contractor:** The person or persons, sole proprietor, partnership, joint venture, corporation or other legal entity who has entered into an agreement with the District to perform or execute the work covered by this Contract.
- 2.4 <u>**Contractor Project Manager:**</u> The person designated by the Contractor to administer the Contract operations under this Contract.
- 2.5 **<u>County</u>**: The County of Los Angeles, a political subdivision of the State of California.
- 2.6 **<u>Dav(s)</u>**: Calendar day(s) unless otherwise specified.
- 2.7 <u>District</u>: The Consolidated Fire Protection District of Los Angeles County.
- 2.8 <u>District Project Director</u>: Person designated by District with authority for District on contractual or administrative matters relating to this Contract that cannot be resolved by the District's Project Manager.
- 2.9 <u>**District Project Manager</u>:** Person designated by District's Project Director to manage the operations under this Contract.</u>
- 2.10 **Fiscal Year**: The twelve (12) month period beginning July 1st and ending the following June 30th.
- 2.11 <u>Statement of Work</u>: The directions, provisions, and requirements provided herein and special provisions pertaining to the method, frequency, manner and place of performing the contract services.

- 2.12 **<u>Subcontract</u>**: An agreement by the contractor to employ a subcontractor to provide services to fulfill this contract.
- 2.13 **Subcontractor:** Any individual, person or persons, sole proprietor, firm, partnership, joint venture, corporation, or other legal entity furnishing supplies, services of any nature, equipment, and/or materials to contractor in furtherance of contractor's performance of this contract, at any tier, under oral or written agreement.

3 WORK

- 3.1 Pursuant to the provisions of this Contract, the Contractor shall fully perform, complete and deliver on time, all tasks, deliverables, services and other work as set forth in herein.
- 3.2 If the Contractor provides any tasks, deliverables, goods, services, or other work, other than as specified in this contract, the same shall be deemed to be a gratuitous effort on the part of the contractor, and the contractor shall have no claim whatsoever against the District.

4 TERM OF CONTRACT

4.1 The term of this Contract shall be three (3) years commencing after approval by County's Board of Supervisors (Board), and execution by the Fire Chief or his designee, unless sooner terminated or extended, in whole or in part, as provided in this Contract.

The District shall have the sole option to extend this Contract term for up to two (2) additional one-year periods and twelve (12) month to month extension options, for a maximum total Contract term of six (6) years, pursuant to the same terms and conditions. Each such extension option may be exercised at the sole discretion of the Fire Chief or his designee as authorized by the Board.

- 4.2 The District maintains databases that track/monitor contractor performance history. Information entered into such databases may be used for a variety of purposes, including determining whether the District will exercise a contract term extension option.
- 4.3 The Contractor shall notify the District when this Contract is within six (6) months of the expiration of the term as provided for hereinabove. Upon occurrence of this event, the Contractor shall send written notification to the District at the address herein provided in Exhibit E District's Administration.

5 CONTRACT SUM

5.1 Maximum Contract Sum

5.1.1 The amount the District shall expend from its own funds during the Contract's entire term for as-needed Temporary Personnel Services shall not exceed \$690,000 per Contract year. The Maximum Contract Sum shall be the maximum monetary amount available that is payable by the District to the Contractor for supplying all the Services, Deliverables, Work, etc.

The annual Maximum Contract Sum under the terms of this Contract shall be \$690,000, shared amongst all contracts for temporary personnel services.

This is not a commitment or offer on the part of the District to expend the Maximum Contract Sum. The Contractor shall perform and complete all Services required of the Contractor under this Contract as set forth in Exhibit B (Pricing Sheet), but in any event, not in excess of the Maximum Contract Sum.

5.1.2 The Contractor acknowledges and agrees the Maximum Contract Sum is an all-inclusive, not-to-exceed price that cannot be adjusted for any costs or expenses whatsoever of Contractor. This Contract includes the full amount of compensation and reimbursement the District will be asked to provide to the Contractor in order for the Contractor to fully perform all of its obligations under this Contract, with such amount of compensation and reimbursement subject to any executed Amendments if applicable. The Contractor understands the District is entering into this Contract in reliance upon the premise that the Contractor shall fully perform all of its obligations under this Contract without seeking any additional compensation or reimbursement beyond that already provided for in this Contract, subject to any Amendments, if applicable. It is the Contractor's risk and responsibility to achieve and timely deliver the Services in accordance with the requirements of the Contract.

5.2 Written Approval for Reimbursement

The Contractor shall not be entitled to payment or reimbursement for any tasks or services performed, nor for any incidental or administrative expenses whatsoever incurred in or incidental to performance hereunder, except as specified herein. Assumption or takeover of any of the Contractor's duties, responsibilities, or obligations, or performance of same by any person or entity other than the Contractor, whether through assignment, subcontract, delegation, merger, buyout, or any other mechanism, with or without consideration for any reason whatsoever, shall not occur except with the District's express prior written approval.

5.3 Notification of 75% of Total Contract Sum

The Contractor shall maintain a system of record keeping that will allow the Contractor to determine when it has incurred seventy-five percent (75%) of the total contract sum under this Contract. Upon occurrence of this event, the Contractor shall send written notification to the District at the address herein provided in Exhibit E - District's Administration.

5.4 No Payment for Services Provided Following Expiration-Termination of Contract

The Contractor shall have no claim against District for payment of any money or reimbursement, of any kind whatsoever, for any service provided by the Contractor after the expiration or other termination of this Contract. Should the Contractor receive any such payment it shall immediately notify the District and shall immediately repay all such funds to the District. Payment by the District for services rendered after expiration-termination of this Contract shall not constitute a waiver of the District's right to recover such payment from the Contractor. This provision shall survive the expiration or other termination of this Contract.

5.5 Invoices and Payments

5.5.1 The Contractor shall invoice the District only for providing the tasks, deliverables, goods, services, and other work specified in Exhibit A - Statement of Work and elsewhere hereunder. The Contractor shall prepare invoices, which shall include the charges owed to the Contractor by the District under the terms of this Contract. The Contractor's payments shall be as provided in Exhibit B - Pricing Sheet and the Contractor shall be paid only for the tasks, deliverables, goods, services, and other work approved in writing by the District. If the District does not approve work in writing no payment shall be due to the Contractor for that work.

- 5.5.2 The Contractor's invoices shall be priced in accordance with Exhibit B Pricing Sheet.
- 5.5.3 The Contractor's invoices shall contain the information set forth in Exhibit A - Statement of Work describing the tasks, deliverables, goods, services, work hours, and facility and/or other work for which payment is claimed.
- 5.5.4 The Contractor shall submit the monthly invoices to the District by the 15th calendar day of the month following the month of service.
- 5.5.5 Payment to Contractor shall be made on an arrears basis, upon acceptance of completed work by the District, provided the Contractor is not in default under any provisions of this Contract. The Contractor shall email one (1) copy of the invoice to the following:
 - 1. <u>Bianca.Moya@fire.lacounty.gov</u> <u>Vinrose.Napoles@fire.lacounty.gov</u> for review and approval of all invoices; and
 - 2. <u>Fire-InvoiceSubmission@fire.lacounty.gov</u> for payment of all invoices.

The Contractor's invoices shall include the following:

- Contract Number
- Date(s) of Service
- Brief description of services
- A breakdown of labor hours and hourly rate i.e.: 3 hours @ \$20/hour = \$60.00.
 NO OVERTIME MAY BE ACCUMULATED OR PAID.
- Employee Name and Employee Number of District Employee who ordered or authorized the service.
- Signature of authorized District employee. Contractor's failure to obtain the signature of District employee authorizing the work may result in a delay of payment.

5.5.6 **District Approval of Invoices**

All invoices submitted by the Contractor for payment must have the written approval of the District's Project Manager prior to any payment thereof. In no event shall the District be liable or responsible for any payment prior to such written approval. Approval for payment will not be unreasonably withheld.

5.5.7 Local Small Business Enterprises – Prompt Payment Program

Certified Local Small Business Enterprises (LSBEs) will receive prompt payment for services they provide to the District. Prompt payment is defined as fifteen (15) calendar days after receipt of an undisputed invoice.

5.6 Intentionally Omitted

5.7 Default Method of Payment: Direct Deposit or Electronic Funds Transfer

- 5.7.1 The County, at its sole discretion, has determined the most efficient and secure default form of payment for goods and/or services provided under an agreement/ contract with the County shall be Electronic Funds Transfer (EFT) or direct deposit, unless an alternative method of payment is deemed appropriate by the Auditor-Controller (A-C).
- 5.7.2 The Contractor shall submit a direct deposit authorization request via the website https://directdeposit.lacounty.gov with banking and vendor information, and any other information that the A-C determines is reasonably necessary to process the payment and comply with all accounting, record keeping, and tax reporting requirements.
- 5.7.3 Any provision of law, grant, or funding agreement requiring a specific form or method of payment other than EFT or direct deposit shall supersede this requirement with respect to those payments.
- 5.7.4 At any time during the duration of the agreement/contract, a Contractor may submit a written request for an exemption to this requirement. Such request must be based on specific legal, business or operational needs and explain why the

payment method designated by the A-C is not feasible and an alternative is necessary. The A-C, in consultation with the contracting department(s), shall decide whether to approve exemption requests.

5.8 Travel, Transportation, and Parking

- 5.8.1 All travel-related costs are the responsibility of the Contractor. The District will not be responsible for paying or reimbursing Contractor or its employees for any travel related costs.
- 5.8.2 The District is not responsible for paying or reimbursing parking or transportation costs for temporary personnel.

6 ADMINISTRATION OF CONTRACT – DISTRICT

6.1 District Administration

A listing of all District Administration referenced in the following subparagraphs are designated in Exhibit E - District's Administration. The District will notify the Contractor in writing of any change in the names or addresses shown.

6.2 District's Project Director

The responsibilities of the District's Project Director include:

- Coordinating with Contractor and ensuring Contractor's performance of the Contract; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby; and
- Upon request of the Contractor, providing direction to the Contractor, as appropriate in areas relating to District policy, information requirements, and procedural requirements; however, in no event, shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby.

6.3 District's Project Manager

The role of the District's Project Manager is to oversee the day-today administration of this Contract; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby. The District's Project Manager's responsibilities include:

- Meeting with the Contractor's Project Manager on a regular basis; and
- Inspecting any and all tasks, deliverables, goods, services, or other work provided by or on behalf of the Contractor; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby.

The District's Project Manager is not authorized to make any changes in any of the terms and conditions of this Contract and is not authorized to further obligate District in any respect whatsoever.

6.4 District's Contract Administrator

The responsibilities of the District's Contract Administrator include:

- Ensuring that the objectives of this Contract are met; and
- Making changes in the terms and conditions of this Contract in accordance with Sub-paragraph 8.1, Amendments; and
- Providing direction to Contractor in the areas relating to District policy, information requirements, and procedural requirements.

7 ADMINISTRATION OF CONTRACT - CONTRACTOR

7.1 Contractor Administration

A listing of all of Contractor's Administration referenced in the following paragraphs is designated in Exhibit F - Contractor's Administration. The Contractor will notify the District in writing of any change in the names or addresses shown.

7.2 Contractor's Project Manager

- 7.2.1 The Contractor's Project Manager is designated in Exhibit F - Contractor's Administration. The Contractor shall notify the District in writing of any change in the name or address of the Contractor's Project Manager.
- 7.2.2 The Contractor's Project Manager shall be responsible for the Contractor's day-to-day activities as related to this

Contract and shall meet and coordinate with the District's Project Manager on a regular basis.

7.3 Approval of Contractor's Staff

District has the absolute right to approve or disapprove all of the Contractor's staff performing work hereunder and any proposed changes in the Contractor's staff, including, but not limited to, the Contractor's Project Manager.

7.4 Contractor's Staff Identification

All of the Contractor's employees assigned to District facilities are required to have a District Identification (ID) badge on their person and visible at all times. The Contractor may bear the expense of any badge not returned to the District when the temporary personnel's assignment has ended.

- 7.4.1 Contractor is responsible to ensure employees have obtained a District ID badge before they are assigned to work in a District facility. Contractor personnel may be asked by a District representative to leave a District facility if they do not have the proper District ID badge on their person and Contractor personnel must immediately comply with such request.
- 7.4.2 Contractor shall notify the District within one business day when staff is terminated from working under this Contract. Contractor shall retrieve and return an employee's District ID badge to the District on the next business day after the employee has terminated employment with the Contractor.
- 7.4.3 If District requests the removal of Contractor's staff, Contractor shall retrieve and return an employee's District ID badge to the District on the next business day after the employee has been removed from working on the District's Contract.

7.5 Background and Security Investigations

7.5.1 Each of Contractor's staff performing services under this Contract, as determined by District in District's sole discretion, shall undergo and pass a background investigation to the satisfaction of District as a condition of beginning and continuing to perform services under this Contract. Such background investigation must be obtained through fingerprints submitted to the California Department of Justice to include State, local, and federal-level review, which may include, but shall not be limited to, criminal conviction information. The fees associated with the background investigation shall be at the expense of the Contractor, regardless of whether the member of Contractor's staff passes or fails the background investigation. The District, may at its sole discretion and expense, provide live scan services for Contractor's staff assigned to District facilities. The District reserves the right to request reimbursement for the live scan services.

If a member of Contractor's staff does not pass the background investigation, the District may request that the member of Contractor's staff be removed immediately from performing services under the Contract. Contractor shall comply with District's request at any time during the term of the Contract. District will not provide to Contractor or to Contractor's staff any information obtained through the District's background investigation.

- 7.5.2 District shall immediately deny or terminate facility access to any member of Contractor's staff that does not pass such investigation to the satisfaction of the District or whose background or conduct is incompatible with District facility access.
- 7.5.3 Disqualification of any member of Contractor's staff pursuant to this Paragraph 7.5 shall not relieve Contractor of its obligation to complete all work in accordance with the terms and conditions of this Contract.

7.6 Confidentiality

- 7.6.1 Contractor shall maintain the confidentiality of all records and information in accordance with all applicable Federal, State and local laws, rules, regulations, ordinances, directives, guidelines, policies and procedures relating to confidentiality, including, without limitation, County policies concerning information technology security and the protection of confidential records and information.
- 7.6.2 Contractor shall indemnify, defend, and hold harmless County, its officers, employees, and agents, from and against any and all claims, demands, damages, liabilities, losses, costs and expenses, including, without limitation,

defense costs and legal, accounting and other expert, consulting, or professional fees, arising from, connected with, or related to any failure by Contractor, its officers, employees, agents, or subcontractors, to comply with this Paragraph 7.6, as determined by County in its sole judgment. Any legal defense pursuant to contractor's indemnification obligations under this Paragraph 7.6 shall be conducted by contractor and performed by counsel selected by Contractor and approved by District. Notwithstanding the preceding sentence, District shall have the right to participate in any such defense at its sole cost and expense, except that in the event Contractor fails to provide District with a full and adequate defense, as determined by District in its sole judgment, District shall be entitled to retain its own counsel, including, without limitation, County Counsel, and to reimbursement from Contractor for all such costs and expenses incurred by County in doing so. Contractor shall not have the right to enter into any settlement, agree to any injunction, or make any admission, in each case, on behalf of District without District's prior written approval.

- 7.6.3 Contractor shall inform all of its officers, employees, agents and subcontractors providing services hereunder of the confidentiality provisions of this Contract.
- 7.6.4 Contractor shall sign and adhere to the provisions of Exhibit G - "Contractor Acknowledgement and Confidentiality Agreement."

8 STANDARD TERMS AND CONDITIONS

8.1 Amendments

- 8.1.1 For any change which affects the scope of work, term, contract sum, payments, or any term or condition included under this Contract, an amendment to the Contract shall be prepared and executed by the contractor and by the Fire Chief or his designee.
- 8.1.2 The County's Board of Supervisors or Chief Executive Officer or designee may require the addition and/or change of certain terms and conditions in the Contract during the term of this Contract. The District reserves the right to add and/or change such provisions as required by the County's Board of Supervisors or Chief Executive Officer. To implement such changes, an Amendment to the Contract shall be prepared

and executed by the contractor and by the Fire Chief or his designee.

8.1.3 The Fire Chief or his designee may at his sole discretion, authorize extensions of time as defined in Paragraph 4 - Term of Contract. The contractor agrees that such extensions of time shall not change any other term or condition of this Contract during the period of such extensions. To implement an extension of time, an Amendment to the Contract shall be prepared and executed by the contractor and by the Fire Chief or his designee.

8.2 Assignment and Delegation/Mergers or Acquisitions

- 8.2.1 The contractor shall notify the District of any pending acquisitions/mergers of its company unless otherwise legally prohibited from doing so. If the contractor is restricted from legally notifying the District of pending acquisitions/mergers, District then it should notifv the of the actual acquisitions/mergers as soon as the law allows and provide to the District the legal framework that restricted it from notifying the District prior to the actual acquisitions/mergers.
- 8.2. 2 The contractor shall not assign its rights or delegate its duties under this Contract, or both, whether in whole or in part, without the prior written consent of District, in its discretion, and any attempted assignment or delegation without such consent shall be null and void. For purposes of this paragraph, District consent shall require a written Amendment to the Contract, which is formally approved and executed by the parties. Any payments by the District to any approved designee or assignee on any claim under this Contract shall be deductible, at District's sole discretion, against the claims, which the contractor may have against the District.
- 8.2.3 Shareholders, partners, members, or other equity holders of contractor may transfer, sell, exchange, assign, or divest themselves of any interest they may have therein. However, in the event any such sale, transfer, exchange, assignment, or divestment is effected in such a way as to give majority control of contractor to any person(s), corporation, partnership, or legal entity other than the majority controlling interest therein at the time of execution of the Contract, such disposition is an assignment requiring the prior written consent of District in accordance with applicable provisions of this Contract.

8.2.4 Any assumption, assignment, delegation, or takeover of any of the contractor's duties, responsibilities, obligations, or performance of same by any person or entity other than the contractor, whether through assignment, subcontract, delegation, merger, buyout, or any other mechanism, with or without consideration for any reason whatsoever without District's express prior written approval, shall be a material breach of the Contract which may result in the termination of this Contract. In the event of such termination. District shall be entitled to pursue the same remedies against contractor as it could pursue in the event of default by contractor.

8.3 Authorization Warranty

The contractor represents and warrants that the person executing this Contract for the contractor is an authorized agent who has actual authority to bind the contractor to each and every term, condition, and obligation of this Contract and that all requirements of the contractor have been fulfilled to provide such actual authority.

8.4 Budget Reductions

In the event the County's Board of Supervisors adopts, in any fiscal year, a County Budget which provides for reductions in the salaries and benefits paid to the majority of County employees and imposes similar reductions with respect to County contracts, the District reserves the right to reduce its payment obligation under this Contract correspondingly for that fiscal year and any subsequent fiscal year during the term of this Contract (including any extensions), and the services to be provided by the contractor under this Contract shall also be reduced correspondingly. The District's notice to the contractor regarding said reduction in payment obligation shall be provided within thirty (30) calendar days of the Board's approval of such actions. Except as set forth in the preceding sentence, the contractor shall continue to provide all of the services set forth in this Contract.

8.5 Complaints

- 8.5.1 The contractor shall develop, maintain and operate procedures for receiving, investigating and responding to complaints.
- 8.5.2 Complaint Procedures

- 8.5.2.1 Within forty-five (45) business days after the Contract effective date, the contractor shall provide the District with the contractor's policy for receiving, investigating and responding to user complaints.
- 8.5.2.2 The District will review the contractor's policy and provide the contractor with approval of said plan or with requested changes.
- 8.5.2.3 If the District requests changes in the contractor's policy, the contractor shall make such changes and resubmit the plan within five (5) business days for District approval.
- 8.5.2.4 If, at any time, the contractor wishes to change the contractor's policy, the contractor shall submit proposed changes to the District for approval before implementation.
- 8.5.2.5 The contractor shall preliminarily investigate all complaints and notify the District's Project Manager of the status of the investigation within five (5) business days of receiving the complaint.
- 8.5.2.6 When complaints cannot be resolved informally, a system of follow-through shall be instituted which adheres to formal plans for specific actions and strict time deadlines.
- 8.5.2.7 Copies of all written responses shall be sent to the District's Project Manager within three (3) business days of mailing to the complainant.

8.6 Compliance with Applicable Law

- 8.6.1 In the performance of this Contract, contractor shall comply with all applicable Federal, State and local laws, rules, regulations, ordinances, directives, guidelines, policies and procedures, and all provisions required thereby to be included in this Contract are hereby incorporated herein by reference.
- 8.6.2 Contractor shall indemnify, defend, and hold harmless County, its officers, employees, and agents, from and against any and all claims, demands, damages, liabilities, losses, costs, and expenses, including, without limitation,

defense costs and legal, accounting and other expert, consulting or professional fees, arising from, connected with, or related to any failure by contractor, its officers, employees, agents, or subcontractors, to comply with any such laws, rules, regulations, ordinances, directives, guidelines, policies, or procedures, as determined by County in its sole judgment. Any legal defense pursuant to contractor's indemnification obligations under Paragraph 8.6 (Compliance with Applicable Law) shall be conducted by contractor and performed by counsel selected by contractor and approved by County. Notwithstanding the preceding sentence, County shall have the right to participate in any such defense at its sole cost and expense, except that in the event contractor fails to provide County with a full and adequate defense, as determined by County in its sole judgment, County shall be entitled to retain its own counsel, including, without limitation, County Counsel, and to reimbursement from contractor for all such costs and expenses incurred by County in doing so. Contractor shall not have the right to enter into any settlement, agree to any injunction or other equitable relief, or make any admission, in each case, on behalf of County without County's prior written approval.

8.7 Compliance with Civil Rights Laws

The contractor hereby assures that it will comply with Subchapter VI of the Civil Rights Act of 1964, 42 USC Sections 2000 (e) (1) through 2000 (e) (17), to the end that no person shall, on the grounds of race, creed, color, sex, religion, ancestry, age, condition of physical handicap, marital status, political affiliation, or national origin, be excluded from participation in, be denied the benefits of, or be otherwise subjected to discrimination under this Contract or under any project, program, or activity supported by this Contract. The contractor shall comply with Exhibit D - Contractor's EEO Certification.

8.8 Compliance with the County's Jury Service Program

8.8.1 Jury Service Program:

This Contract is subject to the provisions of the County's ordinance entitled Contractor Employee Jury Service ("Jury Service Program") as codified in Sections 2.203.010 through 2.203.090 of the Los Angeles County Code, a copy of which

is attached as Exhibit H and incorporated by reference into and made a part of this Contract.

8.8.2 Written Employee Jury Service Policy.

- 1. Unless the contractor has demonstrated to the District's satisfaction either that the contractor is not a "contractor" as defined under the Jury Service Program (Section 2.203.020 of the County Code) or that the contractor qualifies for an exception to the Jury Service Program (Section 2.203.070 of the County Code), the contractor shall have and adhere to a written policy that provides that its Employees shall receive from the contractor, on an annual basis, no less than five days of regular pay for actual jury service. The policy may provide that Employees deposit any fees received for such jury service with the contractor or that the contractor deduct from the Employee's regular pay the fees received for jury service.
- 2. For purposes of this paragraph, "contractor" means a person, partnership, corporation or other entity which has a contract with the County or a subcontract with a County contractor and has received or will receive an aggregate sum of fifty thousand dollars (\$50,000) or more in any twelve (12) month period under one or more County contracts or subcontracts. "Employee" means any California resident who is a full-time employee of the contractor. "Full-time" means forty (40) hours or more worked per week, or a lesser number of hours if: 1) the lesser number is a recognized industry standard as determined by the District, or 2) contractor has a long-standing practice that defines the lesser number of hours as full-time. Full-time employees providing short-term, temporary services of ninety (90) days or less within a twelve (12) month period are not considered full-time for purposes of the Jury Service Program. If the contractor uses any subcontractor to perform services for the District under the Contract, the subcontractor shall also be subject to the provisions of this paragraph. The provisions of this paragraph shall be inserted into any such subcontract agreement and a copy of the Jury Service Program shall be attached to the agreement.

- 3. If the contractor is not required to comply with the Jury Service Program when the Contract commences, the contractor shall have a continuing obligation to review the applicability of its "exception status" from the Jury Service Program, and the contractor shall immediately notify the District if the contractor at any time either comes within the Jury Service Program's definition of "contractor" or if the contractor no longer gualifies for an exception to the Jury Service Program. In either event, the contractor shall immediately implement a written policy consistent with the Jury Service Program. The District may also require, at any time during the Contract and at its sole discretion, that the contractor demonstrate, to the District's satisfaction that the contractor either continues to remain outside of the Jury Service Program's definition of "contractor" and/or that the contractor continues to qualify for an exception to the Program.
- 4. Contractor's violation of this paragraph of the Contract may constitute a material breach of the Contract. In the event of such material breach, District may, in its sole discretion, terminate the Contract and/or bar the contractor from the award of future County contracts for a period of time consistent with the seriousness of the breach.

8.9 Conflict of Interest

- 8.9.1 No County employee whose position with the County enables such employee to influence the award of this Contract or any competing Contract, and no spouse or economic dependent of such employee, shall be employed in any capacity by the contractor or have any other direct or indirect financial interest in this Contract. No officer or employee of the contractor who may financially benefit from the performance of work hereunder shall in any way participate in the District's approval, or ongoing evaluation, of such work, or in any way attempt to unlawfully influence the District's approval or ongoing evaluation of such work.
- 8.9.2 The contractor shall comply with all conflict of interest laws, ordinances, and regulations now in effect or hereafter to be enacted during the term of this Contract. The contractor warrants that it is not now aware of any facts that create a conflict of interest. If the contractor hereafter becomes

aware of any facts that might reasonably be expected to create a conflict of interest, it shall immediately make full written disclosure of such facts to the District. Full written disclosure shall include, but is not limited to, identification of all persons implicated and a complete description of all relevant circumstances. Failure to comply with the provisions of this paragraph shall be a material breach of this Contract.

8.10 Consideration of Hiring County Employees Targeted for Layoff or Re-Employment List

Should the contractor require additional or replacement personnel after the effective date of this Contract to perform the services set forth herein, the contractor shall give first consideration for such employment openings to qualified, permanent County employees who are targeted for layoff or qualified, former County employees who are on a re-employment list during the life of this Contract.

8.11 Consideration of Hiring GAIN-GROW Participants

- 8.11.1 Should the contractor require additional or replacement personnel after the effective date of this Contract, the contractor shall give consideration for any such employment openings to participants in the County's Department of Public Social Services Greater Avenues for Independence (GAIN) Program or General Relief Opportunity for Work (GROW) Program who meet the contractor's minimum qualifications for the open position. For this purpose, consideration shall mean that the contractor will interview qualified candidates. The County will refer GAIN-GROW participants by job category to the contractor. Contractors shall report all job openings with job requirements to: GAINGROW@DPSS.LACOUNTY.GOV and BSERVICES@WDACS.LACOUNTY.GOV and DPSS will refer qualified GAIN/GROW job candidates.
- 8.11.2 In the event that both laid-off County employees and GAIN/GROW participants are available for hiring, County employees shall be given first priority.

8.12 Contractor Responsibility and Debarment

8.12.1 **Responsible Contractor**

A responsible contractor is a contractor who has demonstrated the attribute of trustworthiness, as well as quality, fitness, capacity and experience to satisfactorily perform the contract. It is the County's policy to conduct business only with responsible contractors.

8.12.2 Chapter 2.202 of the County Code

The contractor is hereby notified that, in accordance with Chapter 2.202 of the County Code, if the County acquires information concerning the performance of the contractor on this or other contracts which indicates that the contractor is not responsible, the County may, in addition to other remedies provided in the Contract, debar the contractor from bidding or proposing on, or being awarded, and/or performing work on County contracts for a specified period of time, which generally will not exceed five (5) years but may exceed five (5) years or be permanent if warranted by the circumstances, and terminate any or all existing contracts the contractor may have with the County.

8.12.3 Non-responsible contractor

The County may debar a contractor if the Board of Supervisors finds, in its discretion, that the contractor has done any of the following: 1) violated a term of a contract with the County or a nonprofit corporation created by the County, 2) committed an act or omission which negatively reflects on the contractor's quality, fitness or capacity to perform a contract with the County, any other public entity, or a nonprofit corporation created by the County, or engaged in a pattern or practice which negatively reflects on same, 3) committed an act or offense which indicates a lack of business integrity or business honesty, or 4) made or submitted a false claim against the County or any other public entity.

8.12.4 Contractor Hearing Board

8.12.4.1 If there is evidence that the contractor may be subject to debarment, the County will notify the contractor in writing of the evidence which is the basis for the proposed debarment and will advise the contractor of the scheduled date for a debarment hearing before the Contractor Hearing Board.

- 8.12.4.2 The Contractor Hearing Board will conduct a hearing where evidence on the proposed debarment is presented. The contractor and/or the contractor's representative shall be given an opportunity to submit evidence at that hearing. After the hearing, the Contractor Hearing Board shall prepare a tentative proposed decision, which shall contain а recommendation regarding whether the contractor should be debarred, and, if so, the appropriate length of time of the debarment. The contractor and the District shall be provided an opportunity to object to the tentative proposed decision prior to its presentation to the Board of Supervisors.
- 8.12.4.3 After consideration of any objections, or if no objections are submitted, a record of the hearing, the proposed decision, and any other recommendation of the Contractor Hearing Board shall be presented to the Board of Supervisors. The Board of Supervisors shall have the right to modify, deny, or adopt the proposed decision and recommendation of the Contractor Hearing Board.
- 8.12.4.4 If a contractor has been debarred for a period longer than five (5) years, that contractor may after the debarment has been in effect for at least five (5) years, submit a written request for review of the debarment determination to reduce the period of debarment or terminate the debarment. The County may, in its discretion, reduce the period of debarment or terminate the debarment if it finds that the contractor has adequately demonstrated one or more of the 1) elimination of the grounds for followina: which the debarment was imposed; 2) a bona fide change in ownership or management; 3) material evidence discovered after debarment was imposed; or 4) any other reason that is in the best interests of the County.
- 8.12.4.5 The Contractor Hearing Board will consider a request for review of a debarment determination only where 1) the contractor has been debarred for a period longer than five (5) years; 2) the debarment has been in effect for at least five (5)

years; and 3) the request is in writing, states one or more of the grounds for reduction of the debarment period or termination of the debarment. includes and supporting documentation. Upon receiving an appropriate request, the Contractor Hearing Board will provide notice of the hearing on the request. At the hearing, the Contractor Hearing Board shall conduct a hearing where evidence on the proposed reduction of debarment period or termination of debarment is presented. This hearing shall be conducted and the request for review decided by the Contractor Hearing Board pursuant to the same procedures as for a debarment hearing.

8.12.4.6 The Contractor Hearing Board's proposed decision shall contain a recommendation on the request to reduce the period of debarment or terminate the debarment. The Contractor Hearing Board shall present its proposed decision and recommendation to the Board of Supervisors. The Board of Supervisors shall have the right to modify, deny, or adopt the proposed decision and recommendation of the Contractor Hearing Board.

8.12.5 Subcontractors of Contractor

These terms shall also apply to subcontractors of County contractors.

8.13 Contractor's Acknowledgement of County's Commitment to Safely Surrendered Baby Law

The contractor acknowledges that the County places a high priority on the implementation of the Safely Surrendered Baby Law. The contractor understands that it is the County's policy to encourage all County contractors to voluntarily post the County's "Safely Surrendered Baby Law" poster, in Exhibit I, in a prominent position at the contractor's place of business. The contractor will also encourage its subcontractors, if any, to post this poster in a prominent position in the subcontractor's place of business. Information and posters for printing are available at www.babysafela.org.

8.14 Contractor's Warranty of Adherence to County's Child Support Compliance Program

- 8.14.1 The contractor acknowledges that the County has established a goal of ensuring that all individuals who benefit financially from the County through contracts are in compliance with their court-ordered child, family and spousal support obligations in order to mitigate the economic burden otherwise imposed upon the County and its taxpayers.
- 8.14.2 As required by the County's Child Support Compliance Program (County Code Chapter 2.200) and without limiting the contractor's duty under this Contract to comply with all applicable provisions of law, the contractor warrants that it is now in compliance and shall during the term of this Contract maintain in compliance with employment and wage reporting requirements as required by the Federal Social Security Act (42 USC Section 653a) and California Unemployment Insurance Code Section 1088.5, and shall implement all lawfully served Wage and Earnings Withholding Orders or Child Support Services Department Notices of Wage and Earnings Assignment for Child, Family or Spousal Support, pursuant to Code of Civil Procedure Section 706.031 and Family Code Section 5246(b).

8.15 District's Quality Assurance Plan

The District or its agent(s) will monitor the contractor's performance under this Contract on not less than an annual basis. Such monitoring will include assessing the contractor's compliance with all Contract terms and conditions and performance standards. Contractor deficiencies which the District determines are significant or continuing and that may place performance of the Contract in jeopardy if not corrected will be reported to the Board of Supervisors and listed in the appropriate contractor performance database. The report to the Board will include improvement/corrective action measures taken by the District and the contractor. If improvement does not occur consistent with the corrective action measures, the District may terminate this Contract or impose other penalties as specified in this Contract.

8.16 Damage to District Property, Facilities, Buildings or Grounds

8.16.1 The contractor shall repair, or cause to be repaired, at its own cost, any damage to the District's real or personal

property used or occupied by the temporary personnel beyond wear from normal usage, as determined by District, any and all damage to District facilities, buildings, or grounds caused by the contractor or employees or agents of the contractor. Such repairs shall be made immediately after the contractor has become aware of such damage, but in no event later than thirty (30) days after the occurrence.

8.16.2 If the contractor fails to make timely repairs, District may make any necessary repairs. All costs incurred by District, as determined by District, for such repairs shall be repaid by the contractor by cash payment upon demand.

8.17 Employment Eligibility Verification

- 8.17.1 The contractor warrants that it fully complies with all Federal and State statutes and regulations regarding the employment of aliens and others and that all its employees performing work under this Contract meet the citizenship or alien status requirements set forth in Federal and State statutes and regulations. The contractor shall obtain, from all employees performing work hereunder, all verification and other documentation of employment eligibility status required by Federal and State statutes and regulations including, but not limited to, the Immigration Reform and Control Act of 1986, (P.L. 99-603), or as they currently exist and as they may be hereafter amended. The contractor shall retain all such documentation for all covered employees for the period prescribed by law.
- 8.17.2 The contractor shall indemnify, defend, and hold harmless, the County, its agents, officers, and employees from employer sanctions and any other liability which may be assessed against the contractor or the County or both in connection with any alleged violation of any Federal or State statutes or regulations pertaining to the eligibility for employment of any persons performing work under this Contract.

8.18 Counterparts and Electronic Signatures and Representations

This Contract may be executed in two or more counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same Contract. The facsimile, email or electronic signature of the Parties shall be deemed to constitute original signatures, and facsimile or electronic copies hereof shall be deemed to constitute duplicate originals.

The County and the Contractor hereby agree to regard electronic representations of original signatures of authorized officers of each party, when appearing in appropriate places on the Amendments prepared pursuant to Paragraph 8.1 (Amendments) and received via communications facilities (facsimile, email or electronic signature), as legally sufficient evidence that such legally binding signatures have been affixed to Amendments to this Contract.

8.19 Fair Labor Standards

The contractor shall comply with all applicable provisions of the Federal Fair Labor Standards Act and shall indemnify, defend, and hold harmless the County and its agents, officers, and employees from any and all liability, including, but not limited to, wages, overtime pay, liquidated damages, penalties, court costs, and attorneys' fees arising under any wage and hour law, including, but not limited to, the Federal Fair Labor Standards Act, for work performed by the contractor's employees for which the County may be found jointly or solely liable.

8.20 Force Majeure

- 8.20.1 Neither party shall be liable for such party's failure to perform its obligations under and in accordance with this Contract, if such failure arises out of fires, floods, epidemics, quarantine restrictions. other natural occurrences, strikes, lockouts (other than a lockout by such party or any of such party's subcontractors), freight embargoes, or other similar events to those described above, but in every such case the failure to perform must be totally beyond the control and without any fault or negligence of such party (such events are referred to in this paragraph as "force majeure events").
- 8.20.2 Notwithstanding the foregoing, a default by a subcontractor of contractor shall not constitute a force majeure event, unless such default arises out of causes beyond the control of both contractor and such subcontractor, and without any fault or negligence of either of them. In such case, contractor shall not be liable for failure to perform, unless the goods or services to be furnished by the subcontractor were obtainable from other sources in sufficient time to permit contractor to meet the required performance

schedule. As used in this subparagraph, the term "subcontractor" and "subcontractors" mean subcontractors at any tier.

8.20.3 In the event contractor's failure to perform arises out of a force majeure event, contractor agrees to use commercially reasonable best efforts to obtain goods or services from other sources, if applicable, and to otherwise mitigate the damages and reduce the delay caused by such force majeure event.

8.21 Governing Law, Jurisdiction, and Venue

This Contract shall be governed by, and construed in accordance with, the laws of the State of California. The contractor agrees and consents to the exclusive jurisdiction of the courts of the State of California for all purposes regarding this Contract and further agrees and consents that venue of any action brought hereunder shall be exclusively in the County of Los Angeles.

8.22 Independent Contractor Status

- 8.22.1 This Contract is by and between the District and the contractor and is not intended, and shall not be construed, to create the relationship of agent, servant, employee, partnership, joint venture, or association, as between the District and the contractor. The employees and agents of one party shall not be, or be construed to be, the employees or agents of the other party for any purpose whatsoever.
- 8.22.2 The contractor shall be solely liable and responsible for providing to, or on behalf of, all persons performing work pursuant to this Contract all compensation and benefits. The District shall have no liability or responsibility for the payment of any salaries, wages, unemployment benefits, disability benefits, Federal, State, or local taxes, or other compensation, benefits, or taxes for any personnel provided by or on behalf of the contractor.
- 8.22.3 The contractor understands and agrees that all persons performing work pursuant to this Contract are, for purposes of Workers' Compensation liability, solely employees of the contractor and not employees of the District. The contractor shall be solely liable and responsible for furnishing any and all Workers' Compensation benefits to any person as a result of any injuries arising from or connected with any

work performed by or on behalf of the contractor pursuant to this Contract.

8.22.4 The contractor shall adhere to the provisions stated in Paragraph 7.6 (Confidentiality).

8.23 Indemnification

The contractor shall indemnify, defend and hold harmless the County, its Special Districts, elected and appointed officers, employees, agents and volunteers (County Indemnitees) from and against any and all liability, including but not limited to demands, claims, actions, fees, costs and expenses (including attorney and expert witness fees), arising from and/or relating to this Contract, except for such loss or damage arising from the sole negligence or willful misconduct of the County indemnitees.

8.24 General Provisions for all Insurance Coverage

Without limiting Contractor's indemnification of District, and in the performance of this Contract and until all of its obligations pursuant to this Contract have been met, Contractor shall provide and maintain at its own expense insurance coverage satisfying the requirements specified in Paragraphs 8.24 and 8.25 of this Contract. These minimum insurance coverage terms, types and limits (the "Required Insurance") also are in addition to and separate from any other contract. The District in no way warrants that the Required Insurance is sufficient to protect the Contractor for liabilities which may arise from or relate to this Contract.

8.24.2 Evidence of Coverage and Notice to District

- 8.24.2.1 Certificate(s) of insurance coverage (Certificate) satisfactory to District, and a copy of an Additional Insured endorsement confirming County and its Agents (defined below) has been given Insured status under the Contractor's General Liability policy, shall be delivered to District at the address shown below and provided prior to commencing services under this Contract.
- 8.24.2.2 Renewal Certificates shall be provided to District not less than ten (10) days prior to

contractor's policy expiration dates. The District reserves the right to obtain complete, certified copies of any required contractor and/or subcontractor insurance policies at any time.

- 8.24.2.3 Certificates shall identify all Required Insurance coverage types and limits specified herein. reference this Contract by name or number, and be signed by an authorized representative of the insurer(s). The Insured party named on the Certificate shall match the name of the contractor identified as the contracting party in this Contract. Certificates shall provide the full name of each insurer providing coverage, its NAIC (National Association of Insurance identification number, Commissioners) its financial rating, the amounts of any policy deductibles or self-insured retentions exceeding fifty thousand dollars (\$50,000), and list any District required endorsement forms.
- 8.24.2.4 Neither the District's failure to obtain, nor the District's receipt of, or failure to object to a noncomplying insurance certificate or endorsement, or any other insurance documentation or information provided by the contractor, its insurance broker(s) and/or insurer(s), shall be construed as a waiver of any of the Required Insurance provisions.
- 8.24.2.5 Certificates and copies of any required endorsements shall be sent to:

<u>Consolidated Fire Protection District of</u> <u>Los Angeles County</u> <u>Materials Management Division/Contracts Section</u> <u>5801 S. Eastern Avenue, Suite 100</u> <u>Commerce, California 90040-4001</u>

8.24.2.6 Contractor also shall promptly report to District any injury or property damage accident or incident, including any injury to a contractor employee occurring on District property, and any loss, disappearance, destruction, misuse, or theft of District property, monies or securities entrusted to contractor. Contractor also shall promptly notify District of any third-party claim or suit filed against contractor or any of its subcontractors which arises from or relates to this Contract, and could result in the filing of a claim or lawsuit against contractor and/or County.

8.24.3 Additional Insured Status and Scope of Coverage

The County of Los Angeles, it's Special Districts, Elected Officials, Officers, Agents, employees and volunteers (collectively County and its Agents) shall be provided additional insured status under contractor's General Liability policy with respect to liability arising out of contractor's ongoing and completed operations performed on behalf of the District. County and its Agents additional insured status shall apply with respect to liability and defense of suits arising out of the contractor's acts or omissions, whether such liability is attributable to the contractor or to the District. The full policy limits and scope of protection also shall apply to the County and its Agents as an additional insured, even if they exceed the County's minimum Required Insurance specifications herein. Use of an automatic additional insured endorsement form is acceptable providing it satisfies the Required Insurance provisions herein.

8.24.4 Cancellation of or Changes in Insurance

Contractor shall provide District with, or contractor's insurance policies shall contain a provision that District shall receive, written notice of cancellation or any change in Required Insurance, including insurer, limits of coverage, term of coverage or policy period. The written notice shall be provided to District at least ten (10) days in advance of cancellation for non-payment of premium and thirty (30) days in advance for any other cancellation or policy change. Failure to provide written notice of cancellation or any change in Required Insurance may constitute a material breach of the Contract, in the sole discretion of the District, upon which the District may suspend or terminate this Contract.

8.24.5 Failure to Maintain Insurance

Contractor's failure to maintain or to provide acceptable evidence that it maintains the Required Insurance shall constitute a material breach of the Contract, upon which District immediately may withhold payments due to contractor, and/or suspend or terminate this Contract. District, at its sole discretion, may obtain damages from contractor resulting from said breach. Alternatively, the District may purchase the Required Insurance, and without further notice to contractor, deduct the premium cost from sums due to contractor or pursue contractor reimbursement.

8.24.6 Insurer Financial Ratings

Coverage shall be placed with insurers acceptable to the District with A.M. Best ratings of not less than A:VII unless otherwise approved by District.

8.24.7 Contractor's Insurance Shall Be Primary

Contractor's insurance policies, with respect to any claims related to this Contract, shall be primary with respect to all other sources of coverage available to contractor. Any District maintained insurance or self-insurance coverage shall be in excess of and not contribute to any contractor coverage.

8.24.8 Waivers of Subrogation

To the fullest extent permitted by law, the contractor hereby waives its rights and its insurer(s)' rights of recovery against County under all the Required Insurance for any loss arising from or relating to this Contract. The contractor shall require its insurers to execute any waiver of subrogation endorsements which may be necessary to effect such waiver.

8.24.9 Subcontractor Insurance Coverage Requirements

Contractor shall include all subcontractors as insureds under contractor's own policies, or shall provide District with each subcontractor's separate evidence of insurance coverage. Contractor shall be responsible for verifying each subcontractor complies with the Required Insurance provisions herein, and shall require that each subcontractor name the County and contractor as additional insureds on the subcontractor's General Liability policy. Contractor shall obtain County's prior review and approval of any subcontractor request for modification of the Required Insurance.

8.24.10 Deductibles and Self-Insured Retentions (SIRs)

Contractor's policies shall not obligate the County to pay any portion of any contractor deductible or SIR. The District retains the right to require contractor to reduce or eliminate policy deductibles and SIRs as respects the District, or to provide a bond guaranteeing contractor's payment of all deductibles and SIRs, including all related claims investigation, administration and defense expenses. Such bond shall be executed by a corporate surety licensed to transact business in the State of California.

8.24.11 Claims Made Coverage

If any part of the Required Insurance is written on a claims made basis, any policy retroactive date shall precede the effective date of this Contract. Contractor understands and agrees it shall maintain such coverage for a period of not less than three (3) years following Contract expiration, termination or cancellation.

8.24.12 Application of Excess Liability Coverage

Contractors may use a combination of primary, and excess insurance policies which provide coverage as broad as ("follow form" over) the underlying primary policies, to satisfy the Required Insurance provisions.

8.24.13 Separation of Insureds

All liability policies shall provide cross-liability coverage as would be afforded by the standard ISO (Insurance Services Office, Inc.) separation of insureds provision with no insured versus insured exclusions or limitations.

8.24.14 Alternative Risk Financing Programs

The District reserves the right to review, and then approve, Contractor use of self-insurance, risk retention groups, risk purchasing groups, pooling arrangements and captive insurance to satisfy the Required Insurance provisions. The County and its Agents shall be designated as an Additional Covered Party under any approved program.

8.24.15 District Review and Approval of Insurance Requirements

The District reserves the right to review and adjust the Required Insurance provisions, conditioned upon District's determination of changes in risk exposures.

8.25 Insurance Coverage

8.25.1 Commercial General Liability insurance (providing scope of coverage equivalent to ISO policy form CG 00 01), naming County and its Agents as an additional insured, with limits of not less than:

General Aggregate: \$2 million

Products/Completed Operations Aggregate: \$1 million

Personal and Advertising Injury: \$1 million

Each Occurrence: \$1 million

- **8.25.2** Automobile Liability insurance (providing scope of coverage equivalent to ISO policy form CA 00 01) with limits of not less than \$1 million for bodily injury and property damage, in combined or equivalent split limits, for each single accident. Insurance shall cover liability arising out of contractor's use of autos pursuant to this Contract, including owned, leased, hired, and/or non-owned autos, as each may be applicable.
- 8.25.3 Workers Compensation and Employers' Liability insurance or gualified self- insurance satisfying statutory requirements, which includes Employers' Liability coverage with limits of not less than \$1 million per accident. If Contractor will provide leased employees, or, is an employee leasing or temporary staffing firm or a professional employer organization (PEO), coverage also shall include an Alternate Employer Endorsement (providing scope of coverage equivalent to ISO policy form WC 00 03 01 A) naming the County as the Alternate Employer. The written notice shall be provided to District at least ten (10) days in advance of cancellation

for non-payment of premium and thirty (30) days in advance for any other cancellation or policy change. If applicable to Contractor's operations, coverage also shall be arranged to satisfy the requirements of any federal workers or workmen's compensation law or any federal occupational disease law.

8.25.4 Unique Insurance Coverage

8.25.4.1 **Professional Liability-Errors and Omissions**

Insurance covering Contractor's liability arising from or related to this Contract, with limits of not less than \$1 million per claim and \$2 million aggregate. Further, Contractor understands and agrees it shall maintain such coverage for a period of not less than three (3) years following this Agreement's expiration, termination or cancellation.

8.25.4.2 Privacy/Network Security (Cyber) Liability

coverage providing Insurance protection against liability for (1) privacy breaches [liability arising from the loss or disclosure of confidential information no matter how it occurs]; (2) system breach; (3) denial or loss of service; (4) introduction, implantation. spread of malicious or software code; (5) unauthorized access to or use of computer systems with limits of not less than \$2 million. No exclusion/restriction for unencrypted portable devices/media may be on the policy.

8.26 Liquidated Damages

8.26.1 If, in the judgment of the Fire Chief or his designee, the contractor is deemed to be non-compliant with the terms and obligations assumed hereby, the Fire Chief or his designee, at his/her option, in addition to, or in lieu of, other remedies provided herein, may withhold the entire monthly payment or deduct pro rata from the contractor's invoice for work not performed. A description of the work not performed and the amount to be withheld or deducted from payments to the contractor from the District, will be

forwarded to the contractor by the Fire Chief or his designee, in a written notice describing the reasons for said action.

- 8.26.2 If the Fire Chief or his designee, determines that there are deficiencies in the performance of this Contract that the Fire Chief or his designee, deems are correctable by the contractor over a certain time span, the Fire Chief or his designee, will provide a written notice to the contractor to correct the deficiency within specified time frames. Should the contractor fail to correct deficiencies within said time frame, the Fire Chief, or his designee, may: (a) Deduct from the contractor's payment, pro rata, those applicable portions of the Monthly Contract Sum; and/or (b) Deduct liquidated damages. The parties agree that it will be impracticable or extremely difficult to fix the extent of actual damages resulting from the failure of the contractor to correct a deficiency within the specified time frame. The parties hereby agree that under the current circumstances a reasonable estimate of such damages is one hundred dollars (\$100) per day per infraction, or as specified in the Exhibit 2 - Performance Requirements Summary (PRS) Chart, Statement of Work Exhibits, hereunder, and that the contractor shall be liable to the District for liquidated damages in said amount. Said amount shall be deducted from the District's payment to the contractor; and/or (c) Upon giving five (5) days notice to the contractor for failure to correct the deficiencies, the District may correct any and all deficiencies and the total costs incurred by the District for completion of the work by an alternate source, whether it be District forces or separate private contractor, will be deducted and forfeited from the payment to the contractor from the District, as determined by the District.
- 8.26.3 The action noted in Paragraph 8.26.2 shall not be construed as a penalty, but as adjustment of payment to the contractor to recover the District cost due to the failure of the contractor to complete or comply with the provisions of this Contract.
- 8.26.4 This Paragraph shall not, in any manner, restrict or limit the District's right to damages for any breach of this Contract provided by law or as specified in the PRS or Paragraph 8.26.2, and shall not, in any manner, restrict or limit the District's right to terminate this Contract as agreed to herein.

8.27 Most Favored Public Entity

If the contractor's prices decline, or should the contractor at any time during the term of this Contract provide the same goods or services under similar quantity and delivery conditions to the State of California or any county, municipality, or district of the State at prices below those set forth in this Contract, then such lower prices shall be immediately extended to the District.

8.28 Nondiscrimination and Affirmative Action

- 8.28.1 The contractor certifies and agrees that all persons employed by it, its affiliates, subsidiaries, or holding companies are and shall be treated equally without regard to or because of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, in compliance with all applicable Federal and State anti-discrimination laws and regulations.
- 8.28.2 The contractor shall certify to, and comply with, the provisions of Exhibit D Contractor's EEO Certification.
- 8.28.3 The contractor shall take affirmative action to ensure that applicants are employed, and that employees are treated during employment, without regard to race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, in compliance with all applicable Federal and State anti-discrimination laws and regulations. Such action shall include, but is not limited to: employment, upgrading, demotion, transfer, recruitment or recruitment advertising, layoff or termination, rates of pay or other forms of compensation, and selection for training, including apprenticeship.
- 8.28.4 The contractor certifies and agrees that it will deal with its subcontractors, bidders, or vendors without regard to or because of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation.
- 8.28.5 The contractor certifies and agrees that it, its affiliates, subsidiaries, or holding companies shall comply with all applicable Federal and State laws and regulations to the end that no person shall, on the grounds of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, be

excluded from participation in, be denied the benefits of, or be otherwise subjected to discrimination under this Contract or under any project, program, or activity supported by this Contract.

- 8.28.6 The contractor shall allow District representatives access to the contractor's employment records during regular business hours to verify compliance with the provisions of this Paragraph 8.28 Nondiscrimination and Affirmative Action when so requested by the District.
- 8.28.7 If the District finds that any provisions of this Paragraph 8.28 (Nondiscrimination and Affirmative Action) have been violated, such violation shall constitute a material breach of this Contract upon which the District may terminate or suspend this Contract. While the District reserves the right to determine independently that the anti-discrimination provisions of this Contract have been violated, in addition, a determination by the California Fair Employment and Housing Commission or the Federal Equal Employment Opportunity Commission that the contractor has violated Federal or State anti-discrimination laws or regulations shall constitute a finding by the District that the contractor has violated the anti-discrimination provisions of this Contract.
- 8.28.8 The parties agree that in the event the contractor violates any of the anti-discrimination provisions of this Contract, the District shall, at its sole option, be entitled to the sum of five hundred dollars (\$500) for each such violation pursuant to California Civil Code Section 1671 as liquidated damages in lieu of terminating or suspending this Contract.

8.29 Non Exclusivity

Nothing herein is intended nor shall be construed as creating any exclusive arrangement with the contractor. This Contract shall not restrict District from acquiring similar, equal or like goods and/or services from other entities or sources.

8.30 Notice of Delays

Except as otherwise provided under this Contract, when either party has knowledge that any actual or potential situation is delaying or threatens to delay the timely performance of this Contract, that party shall, within one (1) business day, give notice thereof, including all relevant information with respect thereto, to the other party.

8.31 Notice of Disputes

The contractor shall bring to the attention of the District's Project Manager and/or District's Project Director any dispute between the District and the contractor regarding the performance of services as stated in this Contract. If the District's Project Manager or District's Project Director is not able to resolve the dispute, the Fire Chief or his designee shall resolve it.

8.32 Notice to Employees Regarding the Federal Earned Income Credit

The contractor shall notify its employees, and shall require each subcontractor to notify its employees, that they may be eligible for the Federal Earned Income Credit under the federal income tax laws. Such notice shall be provided in accordance with the requirements set forth in Internal Revenue Service Notice No. 1015.

8.33 Notice to Employees Regarding the Safely Surrendered Baby Law

The contractor shall notify and provide to its employees, and shall require each subcontractor to notify and provide to its employees, information regarding the Safely Surrendered Baby Law, its implementation in Los Angeles County, and where and how to safely surrender a baby. The information is set forth in Exhibit I, Safely Surrendered Baby Law of this Contract. Additional information is available at <u>www.babysafela.org</u>.

8.34 Notices

All notices or demands required or permitted to be given or made under this Contract shall be in writing and shall be hand delivered with signed receipt or mailed by first-class registered or certified mail, postage prepaid, addressed to the parties as identified in Exhibits E -District's Administration and F - Contractor's Administration. Addresses may be changed by either party giving ten (10) days prior written notice thereof to the other party. The Fire Chief or his designee shall have the authority to issue all notices or demands required or permitted by the District under this Contract.

8.35 Prohibition Against Inducement or Persuasion

Notwithstanding the above, the contractor and the District agree that, during the term of this Contract and for a period of one year thereafter, neither party shall in any way intentionally induce or persuade any employee of one party to become an employee or agent of the other party. No bar exists against any hiring action initiated through a public announcement.

8.36 Public Records Act

- 8.36.1 Any documents submitted by the contractor; all information obtained in connection with the District's right to audit and inspect the contractor's documents, books, and accounting records pursuant to Paragraph 8.38 (Record Retention and Inspection-Audit Settlement) of this Contract; as well as those documents which were required to be submitted in response to the Request for Proposals (RFP) used in the solicitation process for this Contract, become the exclusive property of the District. All such documents become a matter of public record and shall be regarded as public records. Exceptions will be those elements in the California Government Code Section 6250 et seq. (Public Records Act) and which are marked "trade secret," "confidential," or "proprietary." The District shall not in any way be liable or responsible for the disclosure of any such records including. without limitation, those so marked, if disclosure is required by law, or by an order issued by a court of competent jurisdiction.
- 8.36.2 In the event the District is required to defend an action on a Public Records Act request for any of the aforementioned documents, information, books, records, and/or contents of a proposal marked "trade secret," "confidential," or "proprietary," the contractor agrees to defend and indemnify the District from all costs and expenses, including reasonable attorney's fees, in action or liability arising under the Public Records Act.

8.37 Publicity

8.37.1 The contractor shall not disclose any details in connection with this Contract to any person or entity except as may be otherwise provided hereunder or required by law. However, in recognizing the contractor's need to identify its services and related clients to sustain itself, the District shall not inhibit the contractor from publishing its role under this Contract within the following conditions:

- 8.37.1.1 The contractor shall develop all publicity material in a professional manner; and
- 8.37.1.2 During the term of this Contract, the contractor shall not, and shall not authorize another to, publish or disseminate any commercial advertisements, press releases, feature articles, or other materials using the name of the District without the prior written consent of the District's Project Director. The District shall not unreasonably withhold written consent.
- 8.37.2 The contractor may, without the prior written consent of District, indicate in its proposals and sales materials that it has been awarded this Contract with the County of Los Angeles, provided that the requirements of this Paragraph 8.37 (Publicity) shall apply.

8.38 Record Retention and Inspection-Audit Settlement

8.38.1 The Contractor shall maintain accurate and complete financial records of its activities and operations relating to this Contract in accordance with generally accepted accounting principles, and which meet the requirements for contract accounting described in Auditor-Controller Contract Accounting and Administration Handbook. The Contractor shall also maintain accurate and complete employment and other records relating to its performance of this Contract. The contractor agrees that the District, or its authorized representatives, shall have access to and the right to examine, audit, excerpt, copy, or transcribe any pertinent transaction, activity, or record relating to this Contract. All such material, including, but not limited to, all financial records, bank statements, cancelled checks or other proof of payment, timecards, sign-in/sign-out sheets and other time and employment records, and proprietary data and information, shall be kept and maintained by the contractor and shall be made available to the District during the term of this Contract and for a period of five (5) years thereafter unless the District's written permission is given to dispose of any such material prior to such time. All such material shall be maintained by the contractor at a location in Los Angeles County, provided that if any such material is located outside Los Angeles County, then, at the District's option, the contractor shall pay the District for travel, per diem, and other costs incurred by the District to examine, audit, excerpt, copy, or transcribe such material at such other location.

- 8.38.2 In the event that an audit of the contractor is conducted specifically regarding this Contract by any Federal or State auditor, or by any auditor or accountant employed by the contractor or otherwise, then the contractor shall file a copy of such audit report with the County's Auditor-Controller within thirty (30) days of the contractor's receipt thereof, unless otherwise provided by applicable Federal or State law or under this Contract. Subject to applicable law, the District shall make a reasonable effort to maintain the confidentiality of such audit report(s). Failure on the part of the contractor to comply with any of the provisions of this subparagraph 8.38 shall constitute a material breach of this Contract.
- 8.38.3 If, at any time during the term of this Contract or within five (5) years after the expiration or termination of this Contract, representatives of the District conduct an audit of the contractor regarding the work performed under this Contract, and if such audit finds that the District's dollar liability for any such work is less than payments made by the District to the contractor, then the difference shall be either: a) repaid by the contractor to the District by cash payment upon demand or b) at the sole option of the County's Auditor-Controller, deducted from any amounts due to the contractor from the District, whether under this Contract or otherwise. If such audit finds that the District's dollar liability for such work is more than the payments made by the District to the contractor, then the difference shall be paid to the contractor by the District by cash payment, provided that in no event shall the District's maximum obligation for this Contract exceed the funds appropriated by the District for the purpose of this Contract.

8.39 Recycled Bond Paper

Consistent with the Board of Supervisors' policy to reduce the amount of solid waste deposited at the County landfills, the contractor agrees to use recycled-content paper to the maximum extent possible on this Contract.

8.40 Subcontracting

- 8.40.1 The requirements of this Contract may not be subcontracted by the contractor **without the advance written approval of the District**. Any attempt by the contractor to subcontract without the prior consent of the District may be deemed a material breach of this Contract.
- 8.40.2 If the contractor desires to subcontract, the contractor shall provide the following information promptly at the District's request:
 - 8.40.2.1 A description of the work to be performed by the subcontractor;
 - 8.40.2.2 A draft copy of the proposed subcontract; and
 - 8.40.2.3 Other pertinent information and/or certifications requested by the District.
- 8.40.3 The contractor shall indemnify, defend, and hold the County harmless with respect to the activities of each and every subcontractor in the same manner and to the same degree as if such subcontractor(s) were the contractor employees.
- 8.40.4 The contractor shall remain fully responsible for all performances required of it under this Contract, including those that the contractor has determined to subcontract, notwithstanding the District's approval of the contractor's proposed subcontract.
- 8.40.5 The District's consent to subcontract shall not waive the District's right to prior and continuing approval of any and all personnel, including subcontractor employees, providing services under this Contract. The contractor is responsible to notify its subcontractors of this District right.
- 8.40.6 The District's Project Director is authorized to act for and on behalf of the District with respect to approval of any subcontract and subcontractor employees. After approval of the subcontract by the District, contractor shall forward a fully executed subcontract to the District for their files.
- 8.40.7 The contractor shall be solely liable and responsible for all payments or other compensation to all subcontractors and their officers, employees, agents, and successors in interest

arising through services performed hereunder, notwithstanding the District's consent to subcontract.

8.40.8 The contractor shall obtain certificates of insurance, which establish that the subcontractor maintains all the programs of insurance required by the District from each approved subcontractor. Before any subcontractor employee may perform any work hereunder, contractor shall ensure delivery of all such documents to:

Consolidated Fire Protection District of Los Angeles County Materials Management Division / Contracts Section 5801 S. Eastern Avenue, Suite 100 Commerce, California 90040-4001

8.41 Termination for Breach of Warranty to Maintain Compliance with County's Child Support Compliance Program

Failure of the contractor to maintain compliance with the requirements set forth in Paragraph 8.14 (Contractor's Warranty of Adherence to County's Child Support Compliance Program) shall constitute default under this Contract. Without limiting the rights and remedies available to the District under any other provision of this Contract, failure of the contractor to cure such default within ninety (90) calendar days of written notice shall be grounds upon which the District may terminate this Contract pursuant to Paragraph 8.43 (Termination for Default) and pursue debarment of the contractor, pursuant to County Code Chapter 2.202.

8.42 Termination for Convenience

- 8.42.1 This Contract may be terminated, in whole or in part, from time to time, when such action is deemed by the District, in its sole discretion, to be in its best interest. Termination of work hereunder shall be effected by notice of termination to the contractor specifying the extent to which performance of work is terminated and the date upon which such termination becomes effective. The date upon which such termination becomes effective shall be no less than ten (10) days after the notice is sent.
- 8.42.2 After receipt of a notice of termination and except as otherwise directed by the District, the contractor shall:
 - 8.42.2.1 Stop work under this Contract on the date and to the extent specified in such notice, and

- 8.42.2.2 Complete performance of such part of the work as shall not have been terminated by such notice.
- 8.42.3 All material including books, records, documents, or other evidence bearing on the costs and expenses of the contractor under this Contract shall be maintained by the contractor in accordance with Paragraph 8.38 (Record Retention and Inspection-Audit Settlement).

8.43 Termination for Default

- 8.43.1 The District may, by written notice to the contractor, terminate the whole or any part of this Contract, if, in the judgment of District's Project Director:
 - 8.43.1.1 Contractor has materially breached this Contract; or
 - 8.43.1.2 Contractor fails to timely provide and/or satisfactorily perform any task, deliverable, service, or other work required either under this Contract; or
 - 8.43.1.3 Contractor fails to demonstrate a high probability of timely fulfillment of performance requirements under this Contract, or of any obligations of this Contract and in either case, fails to demonstrate convincing progress toward a cure within five (5) working days (or such longer period as the District may authorize in writing) after receipt of written notice from the District specifying such failure.
- 8.43.2 In the event the District terminates this Contract in whole or in part as provided in Paragraph 8.43.1, the District may procure, upon such terms and in such manner as the District may deem appropriate, goods and services similar to those so terminated. The contractor shall be liable to the District for any and all excess costs incurred by the District, as determined by the District, for such similar goods and services. The contractor shall continue the performance of this Contract to the extent not terminated under the provisions of this paragraph.
- 8.43.3 Except with respect to defaults of any subcontractor, the contractor shall not be liable for any such excess costs of

the type identified in Paragraph 8.43.2 if its failure to perform this Contract arises out of causes beyond the control and without the fault or negligence of the contractor. Such causes may include, but are not limited to: acts of God or of the public enemy, acts of the District in either its sovereign or contractual capacity, acts of Federal or State governments in their sovereign capacities, fires, floods, restrictions, epidemics, quarantine strikes. freight embargoes, and unusually severe weather; but in every case, the failure to perform must be beyond the control and without the fault or negligence of the contractor. If the failure to perform is caused by the default of a subcontractor, and if such default arises out of causes beyond the control of both the contractor and subcontractor, and without the fault or negligence of either of them, the contractor shall not be liable for any such excess costs for failure to perform, unless the goods or services to be furnished by the subcontractor were obtainable from other sources in sufficient time to permit the contractor to meet the required performance schedule. As used in this paragraph, "subcontractor(s)" the term means subcontractor(s) at any tier.

- 8.43.4 If, after the District has given notice of termination under the provisions of Paragraph 8.43 (Termination for Default) it is determined by the District that the contractor was not in default under the provisions of Paragraph 8.43 (Termination for Default) or that the default was excusable under the provisions of subparagraph 8.43.3, the rights and obligations of the parties shall be the same as if the notice of termination had been issued pursuant to Paragraph 8.42 (Termination for Convenience).
- 8.43.5 The rights and remedies of the District provided in this Paragraph 8.43 (Termination for Default) shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.44 Termination for Improper Consideration

8.44.1 The District may, by written notice to the contractor, immediately terminate the right of the contractor to proceed under this Contract if it is found that consideration, in any form, was offered or given by the contractor, either directly or through an intermediary, to any District officer, employee, or agent with the intent of securing this Contract or securing favorable treatment with respect to the award, amendment, or extension of this Contract or the making of any determinations with respect to the contractor's performance pursuant to this Contract. In the event of such termination, the District shall be entitled to pursue the same remedies against the contractor as it could pursue in the event of default by the contractor.

- 8.44.2 The contractor shall immediately report any attempt by a District officer or employee to solicit such improper consideration. The report shall be made either to the District manager charged with the supervision of the employee or to the County Auditor-Controller's Employee Fraud Hotline at (800) 544-6861.
- 8.44.3 Among other items, such improper consideration may take the form of cash, discounts, services, the provision of travel or entertainment, or tangible gifts.

8.45 Termination for Insolvency

- 8.45.1 The District may terminate this Contract forthwith in the event of the occurrence of any of the following:
 - 8.45.1.1 Insolvency of the contractor. The contractor shall be deemed to be insolvent if it has ceased to pay its debts for at least sixty (60) days in the ordinary course of business or cannot pay its debts as they become due, whether or not a petition has been filed under the Federal Bankruptcy Code and whether or not the contractor is insolvent within the meaning of the Federal Bankruptcy Code;
 - 8.45.1.2 The filing of a voluntary or involuntary petition regarding the contractor under the Federal Bankruptcy Code;
 - 8.45.1.3 The appointment of a Receiver or Trustee for the contractor; or
 - 8.45.1.4 The execution by the contractor of a general assignment for the benefit of creditors.
- 8.45.2 The rights and remedies of the District provided in this Paragraph 8.45 (Termination for Insolvency) shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.46 Termination for Non-Adherence of County Lobbyist Ordinance

The contractor, and each County Lobbyist or County Lobbying firm as defined in County Code Section 2.160.010 retained by the contractor, shall fully comply with the County's Lobbyist Ordinance, County Code Chapter 2.160. Failure on the part of the contractor or any County Lobbyist or County Lobbying firm retained by the contractor to fully comply with the County's Lobbyist Ordinance shall constitute a material breach of this Contract, upon which the District may in its sole discretion, immediately terminate or suspend this Contract.

8.47 Termination for Non-Appropriation of Funds

Notwithstanding any other provision of this Contract, the District shall not be obligated for the contractor's performance hereunder or by any provision of this Contract during any of the District's future fiscal years unless and until the County's Board of Supervisors appropriates funds for this Contract in the District's Budget for each such future fiscal year. In the event that funds are not appropriated for this Contract, then this Contract shall terminate as of June 30 of the last fiscal year for which funds were appropriated. The District shall notify the contractor in writing of any such non-allocation of funds at the earliest possible date.

8.48 Validity

If any provision of this Contract or the application thereof to any person or circumstance is held invalid, the remainder of this Contract and the application of such provision to other persons or circumstances shall not be affected thereby.

8.49 Waiver

No waiver by the District of any breach of any provision of this Contract shall constitute a waiver of any other breach or of such provision. Failure of the District to enforce at any time, or from time to time, any provision of this Contract shall not be construed as a waiver thereof. The rights and remedies set forth in this paragraph 8.49 shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.50 Warranty Against Contingent Fees

8.50.1 The contractor warrants that no person or selling agency has been employed or retained to solicit or secure this

Contract upon any Contract or understanding for a commission, percentage, brokerage, or contingent fee, excepting bona fide employees or bona fide established commercial or selling agencies maintained by the contractor for the purpose of securing business.

8.50.2 For breach of this warranty, the District shall have the right to terminate this Contract and, at its sole discretion, deduct from the Contract price or consideration, or otherwise recover, the full amount of such commission, percentage, brokerage, or contingent fee.

8.51 Warranty of Compliance with County's Defaulted Property Tax Reduction Program

Contractor acknowledges that County has established a goal of ensuring that all individuals and businesses that benefit financially from County through contract are current in paying their property tax obligations (secured and unsecured roll) in order to mitigate the economic burden otherwise imposed upon County and its taxpayers.

Unless contractor qualifies for an exemption or exclusion, contractor warrants and certifies that to the best of its knowledge it is now in compliance, and during the term of this contract will maintain compliance, with Los Angeles County Code Chapter 2.206.

8.52 Termination for Breach of Warranty to Maintain Compliance with County's Defaulted Property Tax Reduction Program

Failure of contractor to maintain compliance with the requirements set forth in Paragraph 8.51 "Warranty of Compliance with County's Defaulted Property Tax Reduction Program" shall constitute default under this contract. Without limiting the rights and remedies available to County under any other provision of this contract, failure of contractor to cure such default within ten (10) days of notice shall be grounds upon which District may terminate this contract and/or pursue debarment of contractor, pursuant to County Code Chapter 2.206.

8.53 Time Off for Voting

The contractor shall notify its employees, and shall require each subcontractor to notify and provide to its employees, information regarding the time off for voting law (Elections Code Section 14000). Not less than ten (10) days before every statewide election, every contractor and subcontractors shall keep posted conspicuously at the place of work, if practicable, or elsewhere where it can be seen as employees come or go to their place of work, a notice setting forth the provisions of Section 14000.

8.54 Compliance with County's Zero Tolerance Policy on Human Trafficking

Contractor acknowledges that the County has established a Zero Tolerance Policy on Human Trafficking prohibiting contractors from engaging in human trafficking.

If a Contractor or member of Contractor's staff is convicted of a human trafficking offense, the District shall require that the Contractor or member of Contractor's staff be removed immediately from performing services under the Contract. District will not be under any obligation to disclose confidential information regarding the offenses other than those required by law.

Disqualification of any member of Contractor's staff pursuant to this paragraph shall not relieve Contractor of its obligation to complete all work in accordance with the terms and conditions of this Contract.

8.55 Intentionally Omitted

8.56 Compliance with Fair Chance Employment Practices

Contractor shall comply with fair chance employment hiring practices set forth in California Government Code Section 12952, Employment Discrimination: Conviction History. Contractor's violation of this paragraph of the Contract may constitute a material breach of the Contract. In the event of such material breach, District may, in its sole discretion, terminate the Contract.

8.57 Compliance with the County Policy of Equity

The contractor acknowledges that the County takes its commitment to preserving the dignity and professionalism of the workplace very seriously, as set forth in the County Policy of Equity (CPOE) (<u>https://ceop.lacounty.gov/</u>). The contractor further acknowledges that the County strives to provide a workplace free from discrimination, harassment, retaliation and inappropriate conduct based on a protected characteristic, and which may violate the CPOE. The contractor, its employees and subcontractors acknowledge and certify receipt and understanding of the CPOE. Failure of the contractor, its employees or its subcontractors to uphold the County's expectations of a workplace free from harassment and discrimination, including inappropriate conduct based on a protected characteristic, may subject the contractor to termination of contractual agreements as well as civil liability.

8.58 **Prohibition from Participation in Future Solicitation(s)**

A Proposer, or a Contractor or its subsidiary or Subcontractor ("Proposer/Contractor"), is prohibited from submitting a bid or proposal in a County solicitation if the Proposer/Contractor has provided advice or consultation for the solicitation. A Proposer/Contractor is also prohibited from submitting a bid or proposal in a County solicitation if the Proposer/Contractor has developed or prepared any of the solicitation materials on behalf of the County. A violation of this provision shall result in the disqualification of the Contractor/Proposer from participation in the County solicitation or the termination or cancellation of any resultant County contract. This provision shall survive the expiration, or other termination of this Agreement.

9 UNIQUE TERMS AND CONDITIONS

9.1 Intentionally Omitted

9.2 Health Insurance Portability and Accountability Act of 1996 (HIPAA)

- 9.2.1 Contractor expressly acknowledges and agrees that the provision of services under this Agreement does not require or permit access by Contractor or any of its officers, employees, or agents, to any patient medical records/patient information. Accordingly, Contractor shall instruct its officers, employees, and agents that they are not to pursue, or gain access to, patient medical records/patient information for any reason whatsoever.
- 9.2.2 Notwithstanding the forgoing, the parties acknowledge that in the course of the provision of services hereunder, Contractor or its officers, employees, and agents, may have inadvertent access to patient medical records/patient information. Contractor understands and agrees that neither it nor its officers, employees, or agents, are to take advantage of such access for any purpose whatsoever.
- 9.2.3 Additionally, in the event of such inadvertent access, Contractor and its officers, employees, and agents, shall

maintain the confidentiality of any information obtained and shall notify Director that such access has been gained immediately, or upon the first reasonable opportunity to do so. In the event of any access, whether inadvertent or intentional, Contractor shall indemnify, defend, and hold harmless County, its officers, employees, and agents, from and against any and all liability, including but not limited to, actions, claims, costs, demands, expenses, and fees (including attorney and expert witness fees) arising from or connected with Contractor's or its officers', employees', or agents', access to patient medical records/patient information. Contractor agrees to provide appropriate training to its employees regarding their obligations as described hereinabove.

9.3 Intentionally Omitted

9.4 Intentionally Omitted

9.5 Intentionally Omitted

9.6 Data Destruction

Contractor(s) and Vendor(s) that have maintained, processed, or stored the County of Los Angeles' ("County") data and/or information, implied or expressed, have the sole responsibility to certify that the data and information have been appropriately destroyed consistent with the National Institute of Standards and Technology (NIST) Special Publication SP 800-88 titled *Guidelines for Media Sanitization*. Available at:

http://csrc.nist.gov/publications/PubsDrafts.html#SP-800-88 Rev.%201

The data and/or information may be stored on purchased, leased, or rented electronic storage equipment (e.g., printers, hard drives) and electronic devices (e.g., servers, workstations) that are geographically located within the County, or external to the County's boundaries. The District must receive within ten (10) business days, a signed document from Contractor(s) and Vendor(s) that certifies and validates the data and information were placed in one or more of the following stored states: unusable, unreadable, and indecipherable.

Vendor shall certify that any District data stored on purchased, leased, or rented electronic storage equipment and electronic devices, including, but not limited to printers, hard drives, servers, and/or workstations are destroyed consistent with the current National Institute of Standard and Technology (NIST) Special Publication SP-800-88, *Guidelines for Media Sanitization*. Vendor shall provide District with written certification, within ten (10) business days of removal of any electronic storage equipment and devices that validates that any and all District data was destroyed and is unusable, unreadable, and/or undecipherable.

9.7 Intentionally Omitted

9.8 Intentionally Omitted

9.9 Intentionally Omitted

9.10 Mandatory Requirement to Register on County's WebVen

As a pre-condition to the award of this Contract, Contractor represents and warrants that it has registered in Los Angeles County's vendor registration system (hereafter "WebVen"). The WebVen contains Contractor's business profile and identifies the goods/services being provided by Contractor. Contractor shall ensure that it updates its vendor profile whenever changes occur to Contractor's operations by accessing the WebVen site located on-line at: <u>http://camisvr.co.la.ca.us/webven</u> County shall use the data obtained from Contractor's WebVen profile to ensure that Contractor's legal name, as reflected in its WebVen profile, shall be used in Contract documents).

9.11 Limitation on Corporate Acts

- 9.11.1 Contractor shall not amend its articles of incorporation or bylaws, move to dissolve or transfer any assets obtained using Contract Funds, or take any other steps which may materially affect the performance of this Contract without first notifying the District in writing no less than thirty (30) days prior to said action. Contractor shall notify the District's Contract Administrator immediately in writing of any change in Contractor's corporate name.
- 9.11.2 If, in the District's sole discretion, the steps taken by Contractor are determined to materially affect Contractor's performance of this Contract, the District may, at its sole discretion, take any (or all) of the following actions:

- 9.11.2.1 Require Contractor to remedy the areas that affect Contractor's ability to perform its obligations under this Contract.
- 9.11.2.2 Suspend Contractor from performing (and receiving payment for) Contract tasks until a remedy has been reached.
- 9.11.2.3 Terminate this Contract pursuant to Paragraph 8.43 (Termination for Default).

9.12 Modifications

This Contract fully expresses the agreement of the parties. Any modification to this Contract must be by means of a separate written document approved by the District. No oral conversation between any officer, employee or agent of the parties shall modify or otherwise amend this Contract in any way.

9.13 Remedies of Non-Compliance

Contractor agrees to comply with the requirements set forth in the entirety of this Contract as well as the requirements contained in any applicable directives, notices, guidelines and instructions used by the District. Contractor's failure to comply with such requirements shall subject Contractor to remedies which are available under this Contract and as provided by law. These remedies include but are not limited to the following: suspension of payment(s); suspension of Service(s); assessment and collection of liquidated damages; de-obligation of Contract Funds (for purposes of this Contract, de-obligation is the partial or full removal of Contract Funds from Contractor); debarment; and/or termination of Contract. The District shall have the sole discretion to determine which remedy(ies) will be applied as a result of Contractor's noncompliance.

9.14 Suspension

9.14.1 Contractor may be placed on suspension if District determines that Contractor is not in compliance with any Service, Work, task, deliverable or requirement outlined in this Contract and/or Contractor has demonstrated a consistent and significant lack of achievement of the Contract goals (including, but not limited to, meeting the requirements for work performance, the Pricing Sheet, staffing, administration, etc.). The District shall notify the

Contractor in writing in the event that Contractor is placed on suspension.

- 9.14.2 Suspension as used herein shall mean a specified period of time (as determined by the District) during which the District shall withhold payment from Contractor. During the suspension, Contractor has a continuing obligation to remedy the areas of non-compliance which have been identified by the District or its duly authorized representative(s). The District shall monitor Contractor's adherence to such remedy(ies) during the suspension period. When applicable, the District may also provide the Contractor with a written determination stating whether or not the Contractor may continue to provide non-suspended Services, if any, during the suspension period.
- 9.14.3 District's written notice of suspension shall set forth the conditions of Contractor's non-compliance as well as the period in which Contractor must correct noted deficiencies. In response to the notice of suspension, Contractor shall submit a written Corrective Action Plan to the District's Contract Manager within ten (10) days of the date indicated on the notice from the District. Contractor's Corrective Action Plan shall address all of the deficiencies noted by the District.
- 9.14.4 The District shall review Contractor's Corrective Action Plan, and will determine whether it meets the requirements for District's approval. The District reserves the right to suspend/deduct payments for or to terminate all or any part of this Contract (and/or any Contractor's other contracts with the District) when Contractor submits a Corrective Action Plan that is not acceptable to the District.
- 9.14.5 Contractor shall implement the Corrective Action Plan upon receiving District's final written approval of the Corrective Action Plan. Contractor's failure to comply with an approved Corrective Action Plan will be cause for material breach of Contract upon which the District may pursue the remedies for default of Contract.

9.15 Transition of Contract Services

9.15.1 Completion of Contract

9.15.1.1 Within sixty (60) calendar days prior to the expiration of this Contract (or shorter time period if notified in writing by District), Contractor shall allow the District or a newly selected contractor a transition period for orientation purposes and the orderly transition of Contractor's current Services without additional cost to the District. Contractor shall continue to provide Services timely and accurately so that the Services are current at the expiration of this Contract.

9.15.2 Transition Plan

- 9.15.2.2 If this Contract (or any part thereof) is pursuant to any terminated of the termination provisions outlined herein or if it expires pursuant to Paragraph 4.0 (Term of Contract), Contractor shall provide a Transition Plan to the District. Contractor shall submit said Transition Plan to the District's Contract Administrator within the timeframe designated by the District in the notice of termination or Contractor shall submit it at least sixty (60) days prior to the expiration of this Contract as noted in Paragraph 4.0 (Term of Contract).
- 9.15.2.3 The District shall review Contractor's Transition Plan and will determine whether it meets the requirements for District's approval. The District reserves the right to suspend/deduct payments under this Contract and/or under any of Contractor's other contracts with the District when Contractor submits a Transition Plan that is not acceptable to the District. Contractor shall adhere to the Transition Plan which, at a minimum, shall include all of the elements outlined below.

9.15.3 Elements of the Transition Plan

- 9.15.3.1 Contractor's method to communicate with other organizations that can assist in locating alternative Services.
- 9.15.3.2 Contractor's plan to assure for the provision of adequate staff to provide continued Services through the remaining term of this Contract.
- 9.15.3.3 Any additional information which may be necessary to ensure Contract Services are being provided.

9.15.4 Implementation of the Transition Plan

Contractor shall implement the Transition Plan that is approved by the District. Contractor's failure to provide and/or implement the Transition Plan as prescribed herein shall mean that the District will provide Contractor a Transition Plan and Contractor will implement the Transition Plan provided by the District. The District will monitor Contractor's progress in carrying out all elements of the Transition Plan. IN WITNESS WHEREOF, the Board of Supervisors of the County of Los Angeles has caused this Contract to be executed by the Fire Chief of the Consolidated Fire Protection District of Los Angeles County (or designee) and approved by County Counsel, and Contractor has caused this Contract to be executed in its behalf by its duly authorized officer, this 1st day of July, 2021.

CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY

Ву _____

Fire Chief

Trust Temporary Services, Inc. By <u>dba Helpmates Staffing Services</u> Contractor

Signed: _____

Printed: Rosalie Villa

Title: Chief Revenue Officer

APPROVED AS TO FORM:

RODRIGO CASTRO-SILVA County Counsel

By_

Senior Deputy County Counsel



COUNTY OF LOS ANGELES FIRE DEPARTMENT

1320 NORTH EASTERN AVENUE LOS ANGELES, CALIFORNIA 90063-3294 (323) 881-2426 www.fire.lacounty.gov

"Proud Protectors of Life, Property, and the Environment"

DARYL L. OSBY FIRE CHIEF FORESTER & FIRE WARDEN

June 29, 2021

BOARD OF SUPERVISORS

HILDA L. SOLIS FIRST DISTRICT

HOLLY J. MITCHELL SECOND DISTRICT

> SHEILA KUEHL THIRD DISTRICT

JANICE HAHN FOURTH DISTRICT

KATHRYN BARGER FIFTH DISTRICT

The Honorable Board of Supervisors County of Los Angeles 383 Kenneth Hahn Hall of Administration 500 West Temple Street Los Angeles, California 90012

Dear Supervisors:

APPROVAL OF SOLE SOURCE CONTRACT WITH ALACO LADDER COMPANY FOR WOODEN LADDER REPAIR AND MAINTENANCE SERVICES (ALL DISTRICTS) (3 VOTES)

SUBJECT

The Consolidated Fire Protection District of Los Angeles County (District) is requesting Board of Supervisors (Board) approval to establish a sole source contract with Alaco Ladder Company (Alaco) to provide repair and maintenance services for the District's National Fire Protection Association (NFPA)-compliant wooden ground ladders (ladders).

IT IS RECOMMENDED THAT THE BOARD, ACTING AS THE GOVERNING BODY OF THE CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY:

- Approve and instruct the Fire Chief, or his designee, to sign the attached contract (Attachment A) between the District and Alaco to provide repair and maintenance services for the District's ladders.
- 2. Authorize the maximum contract sum of \$900,000, including the initial contract term of five years, and twelve month-to-month extension options. The maximum contract sum includes annual expenditures not to exceed \$150,000 per contract year.

AGOURA HILLS ARTESIA AZUSA BALDWIN PARK BELL BELL GARDENS BELLFLOWER BRADBURY CALABASAS

CARSON CERRITOS CLAREMONT COMMERCE COVINA CUDAHY DIAMOND BAR DUARTE

FI MONTE GARDENA GLENDORA HAWAIIAN GARDENS HAWTHORNE HERMOSA BEACH HIDDEN HILLS HUNTINGTON PARK INDUSTRY

SERVING THE UNINCORPORATED AREAS OF LOS ANGELES COUNTY AND THE CITIES OF:

INGLEWOOD IRWINDALE LA CANADA-FLINTRIDGE I A HABRA LA MIRADA LA PUENTE LAKEWOOD LANCASTER

LAWNDALE LOMITA LYNWOOD MALIBU MAYWOOD NORWALK PALMDALE PALOS VERDES ESTATES PARAMOUNT

PICO RIVERA POMONA RANCHO PALOS VERDES ROLLING HILLS ROLLING HILLS ESTATES ROSEMEAD SAN DIMAS SANTA CLARITA

SIGNAL HILL SOUTH EL MONTE SOUTH GATE TEMPLE CITY VERNON WALNUT WEST HOLLYWOOD WESTLAKE VILLAGE WHITTIER

The Honorable Board of Supervisors June 29, 2021 Page 2

- 3. Delegate authority to the Fire Chief, or his designee, to execute amendments, suspensions, or termination if deemed necessary including the extensions as described in recommendation number two above, and in accordance with the approved contract terms and conditions, and with prior review and approval as to form by County Counsel.
- 4. Find that this contract is exempt from the provisions of the California Environmental Quality Act (CEQA).

PURPOSE/JUSTIFICATION OF RECOMMENDED ACTION

The District seeks to continue as-needed repair and maintenance services for its ladders with Alaco, the only manufacture of NFPA-compliant ladders. Since the District's inception, it has utilized the ladders and currently equips its firefighting apparatus with over 1,000 of these ladders in varying lengths. Alaco is the only company that can warrant repairs and maintain the District's ladders.

Approval of the recommended actions will enable the District to continue uninterrupted repair and maintenance services for its ladders. These services are necessary to preserve and maintain the District's ladders in proper operating condition. The District requires safe and reliable ladders to ensure the safety of the District's firefighters.

Implementation of Strategic Plan Goals

Approval of the recommended actions is consistent with the County's Strategic Plan Goal No. III.3 – Pursue Operational Effectiveness, Fiscal Responsibility, and Accountability by maximizing the use of County assets, and ensuring that resources are expended in a responsible, efficient, and strategic manner. Alaco, the manufacturer of the ladders has the expertise to provide efficient and timely service, enabling the District to provide optimum safety for its firefighters.

FISCAL IMPACT/FINANCING

Sufficient funding is available in the District's Fiscal Year 2021-22 Adopted Budget. There is no impact to net County cost. The District will continue to allocate the necessary funds to obtain the required services. This contract includes an allowance for Cost of Living Adjustments (COLAs) after the initial three-year base contract term.

FACTS AND PROVISIONS/LEGAL REQUIREMENTS

The District's apparatus are outfitted and designed for the Alaco ladders. The ladders are electrically safe, essential to ensure firefighter safety, and crucial components to the District's mission of protecting lives and property. This contract will save the District money by preventing and/or prolonging the need to purchase new ladders. It is estimated that replacing the Districts' ladders would cost upwards of \$4,500,000.

The Honorable Board of Supervisors June 29, 2021 Page 3

The District is authorized to contract these services pursuant to California Health and Safety Code Section 13861 and Public Contract Code Section 20810-20812. The Statement of Work (SOW) and contract cover categories of services required by the District which are in compliance with the U.S. Department of Labor Occupational Safety & Health Administration (OSHA) regulations, NFPA 1932 "Fire Service Ladders" requirements and OSHA Regulations/Standards 29 CFR 1926.1053.

The attached contract specifies the District has no obligation to pay Alaco for expenditures beyond the contract payment mechanisms. Alaco will not be performing services which will exceed the contract approved amount, scope of work, and/or terms. Alaco is in compliance with all Board and Chief Executive Office (CEO) requirements, including Contractor Employee Jury Service, Safely Surrendered Baby Law, and the Defaulted Property Tax Reduction Program. Alaco agrees to maintain compliance with all requirements throughout the term of the contract.

The CEO's Risk Management Section reviewed the contract prior to entering contract negotiations and concurred with the provisions relating to insurance and indemnification. The contract has been approved as to form by County Counsel and has been signed by Alaco.

ENVIRONMENTAL DOCUMENTATION

The services provided through this proposed contract will not have a significant effect on the environment; and therefore, this contract is exempt from CEQA, pursuant to Section 15061 (b) (3) of the CEQA Guidelines.

CONTRACTING PROCESS

On February 1, 2021, the District notified your Board of the expiring sole source contract with Alaco. The District also notified your Board of its intent to proceed with negotiating a new sole source contract, pursuant to Board of Supervisors' policy 5.100, Sole Source Contracts. The District has provided the Sole Source Checklist (Attachment B) approved by the CEO detailing the justification for use of a sole source contract.

Contract negotiations were finalized and with County Counsel's review and concurrence, Alaco agreed to comply with the County's negotiated contract terms and conditions. This contract is submitted to your Board for approval with confidence the negotiated terms are commercially reasonable and represent a minimal risk exposure to the District given Alaco's history in providing these services.

IMPACT ON CURRENT SERVICES

Approval of the recommended actions will support the District's ability provide safe and essential tools to ensure firefighter safety during firefighting operations.

The Honorable Board of Supervisors June 29, 2021 Page 4

Award of this contract will not result in the displacement of any County employees as these services are currently obtained from Alaco. This contract will not result in a reduction of service and there is no change in risk exposure to the County.

CONCLUSION

Upon approval by your Honorable Board, please instruct the Executive Officer of the Board to return the adopted stamped Board Letter to the following:

Consolidated Fire Protection District of Los Angeles County Executive Office - Business Operations Attention: Zuleyda Santana, Administrative Services Manager II 1320 North Eastern Avenue Los Angeles, CA 90063 Zuleyda.Santana@fire.lacounty.gov

The District's contact can be reached at (323) 881-6173.

Respectfully submitted,

DARYL L. OSBY, FIRE CHIEF

DLO:ah

Enclosures

c: Chief Executive Officer Executive Officer, Board of Supervisors County Counsel

ATTACHMENT A

CONTRACT NO.: _____

CONTRACT



BY AND BETWEEN

CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY

AND

ALACO LADDER COMPANY

FOR

ALACO WOODEN LADDER REPAIR AND MAINTENANCE SERVICES

	GRAP		PAGE
		EXHIBITS	
		ICABLE DOCUMENTS	
2		ITIONS	
L	2.1	Standard Definitions	
3		Clandara Deminiono	
4		OF CONTRACT	
5		RACT SUM	
-	5.1	Total Contract Sum	
	5.2	Written Approval for Reimbursement	
	5.3	Notification of 75% of Total Contract Sum	
	5.4	No Payment for Services Provided Following Expiration-Termination of Contract	7
	5.5	Invoices and Payments	7
	5.6	Cost of Living Adjustment (COLA's)	9
	5.7	Default Method of Payment: Direct Deposit or Electronic Funds Transf	er 9
	ADMII 6.1	NISTRATION OF CONTRACT- DISTRICT	
	6.2	District's Project Director	10
	6.3	District's Project Manager	10
	6.4	District's Contract Administrator	11
7		NISTRATION OF CONTRACT-CONTRACTOR	11
	7.1	Contractor Administration	11
	7.2	Contractor's Project Manager	11
	7.3	Approval of Contractor's Staff	12
	7.4	Contractor's Staff Identification	12
	7.5	Background and Security Investigations	12
	7.6	Confidentiality	13
8	STAN	DARD TERMS AND CONDITIONS	13
	8.1	Amendments	13
	8.2	Assignment and Delegation/Mergers or Acquisitions	14
	8.3	Authorization Warranty	15
	8.4	Budget Reductions	16
	8.5	Complaints	16

PARAGRAPH	TITLE	PAGE
8.6	Compliance with Applicable Law	17
8.7	Compliance with Civil Rights Laws	17
8.8	Compliance with the County's Jury Service Program	19
8.9	Conflict of Interest	20
8.10	Consideration of Hiring County Employees Targeted for Layoffs or a a County Re-Employment List	
8.11	Consideration of Hiring GAIN-GROW Participants	20
8.12	Contractor Responsibility and Debarment	20
8.13	Contractor's Acknowledgement of County's Commitment to Safely Surrendered Baby Law	23
8.14	Contractor's Warranty of Adherence to County's Child Support	
	Compliance Program	24
8.15	District's Quality Assurance Plan	24
8.16	Damage to District Facilities, Buildings or Grounds	24
8.17	Employment Eligibility Verification	25
8.18	Facsimile Representations	25
8.19	Fair Labor Standards	26
8.20	Force Majeure	26
8.21	Governing Law, Jurisdiction, and Venue	27
8.22	Independent Contractor Status	27
8.23	Indemnification	28
8.24	General Provisions for all Insurance Coverage	28
8.25	Insurance Coverage	33
8.26	Liquidated Damages	35
8.27	Most Favored Public Entity	36
8.28	Nondiscrimination and Affirmative Action	36
8.29	Non Exclusivity	37
8.30	Notice of Delays	38
8.31	Notice of Disputes	38
8.32	Notice to Employees Regarding the Federal Earned Income Credit	38
8.33	Notice to Employees Regarding the Safely Surrendered Baby Law	38
8.34	Notices	38
8.35	Prohibition Against Inducement or Persuasion	39

PARAGRAPH	I TITLE P	PAGE	
8.36	Public Records Act	39	
8.37	Publicity	39	
8.38	Record Retention and Inspection-Audit Settlement	40	
8.39	Recycled Bond Paper	41	
8.40	Subcontracting	42	
8.41	Termination for Breach of Warranty to Maintain Compliance with County's Child Support Compliance Program	43	
8.42	Termination for Convenience	43	
8.43	Termination for Default	44	
8.44	Termination for Improper Consideration	45	
8.45	Termination for Insolvency	46	
8.46	Termination for Non-Adherence of County Lobbyist Ordinance	46	
8.47	Termination for Non-Appropriation of Funds	47	
8.48	Validity	47	
8.49	Waiver	47	
8.50	Warranty Against Contingent Fees	47	
8.51	Warranty of Compliance with County's Defaulted Property Tax		
	Reduction Program	48	
8.52	Termination for Breach of Warranty to Maintain Compliance with		
	County's Defaulted Property Tax Reduction Program		
8.53	Time off for Voting	48	
8.54	Compliance with County's Zero Tolerance Policy on Human Traffickin	g. 49	
8.55	Intentionally Omitted	49	
8.56	Compliance with Fair Chance Employment Practices	49	
8.57	Compliance with the County Policy of Equity	49	
8.58	Prohibition from Participation in Future Solicitation(s)	50	
9 UNIQ	UE TERMS AND CONDITIONS	50	
9.1	Intentionally Omitted	50	
9.2	Intentionally Omitted	50	
9.3	Intentionally Omitted	50	
9.4	Patent, Copyright and Trade Secret Indemnification	50	
9.5	Intentionally Omitted	51	
9.6	Data Destruction	51	
Page			

PARAGRAPH TITLE PAGE 9.7 9.8 9.9 9.10 9.11 9.12 9.13 9.14 9.15

Ę

STANDARD EXHIBITS

- A Statement of Work
- B Pricing Sheet
- C Intentionally Omitted
- D Contractor's EEO Certification
- E District's Administration
- F Contractor's Administration
- G Contractor Acknowledgement and Confidentiality Agreement
- H Jury Service Ordinance
- I Safely Surrendered Baby Law

CONTRACT BETWEEN CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY AND ALACO LADDER COMPANY FOR ALACO WOODEN LADDER REPAIR AND MAINTENANCE SERVICES

This Contract ("Contract") made and entered into this 1st day of July, 2021 by and between the Consolidated Fire Protection District of Los Angeles County, hereinafter referred to as "District" and Alaco Ladder Company, hereinafter referred to as "Contractor." Alaco Ladder Company is located at 5167 G Street, Chino, CA 91710.

RECITALS

WHEREAS, the District has the responsibility for maintaining and repairing its wooden ladders to ensure the safety of its employees and to achieve its mission to protect lives and property; and

WHEREAS, the Contractor is a private firm specializing in providing wooden ladder repair and maintenance services; and

WHEREAS, the District is authorized under Health and Safety Code Section 13861 and Public Contract Code Section 20810-20812 to contract for wooden ladder repair and maintenance services on an intermittent basis when certain requirements are met; and

NOW THEREFORE, in consideration of the mutual covenants contained herein, and for good and valuable consideration, the parties agree to the following:

1 APPLICABLE DOCUMENTS

Exhibits A, B, D, E, F, G, H, and I are attached to and form a part of this Contract. In the event of any conflict or inconsistency in the definition or interpretation of any word, responsibility, schedule, or the contents or description of any task, deliverable, goods, service, or other work, or otherwise between the base Contract and the Exhibits, or between Exhibits, such conflict or inconsistency shall be resolved by giving precedence first to the terms and conditions of the Contract and then to the Exhibits according to the following priority.

Standard Exhibits:

- 1.1 Exhibit A Statement of Work
- 1.2 Exhibit B Pricing Sheet
- 1.3 Exhibit C Intentionally Omitted
- 1.4 Exhibit D Contractor's EEO Certification
- 1.5 Exhibit E District's Administration
- 1.6 Exhibit F Contractor's Administration
- 1.7 Exhibit G Contractor Acknowledgement and Confidentiality Agreement
- 1.8 Exhibit H Jury Service Ordinance
- 1.9 Exhibit I Safely Surrendered Baby Law

This contract constitutes the complete and exclusive statement of understanding between the parties and supersedes all previous contracts, written and oral, and all communications between parties relating to the subject matter of this contract. No change to this contract shall be valid unless prepared pursuant to subparagraph 8.1 – Amendments and signed by both parties.

2 **DEFINITIONS**

The headings herein contained are for convenience and reference only and are not intended to define the scope of any provision thereof. The following words as used herein shall be construed to have the following meaning, unless otherwise apparent from the context in which they are used.

- 2.1 **Board of Supervisors (Board):** The Board of Supervisors of the County of Los Angeles; the governing body of the District and the County of Los Angeles.
- 2.2 **Contract:** This agreement executed between District and Contractor. Included are all supplemental agreements amending or extending the service to be performed. The Contract sets forth the terms and conditions for the issuance and performance of all tasks, deliverables, services and other work
- 2.3 **Contractor:** The person or persons, sole proprietor, partnership, joint venture, corporation or other legal entity who has entered into an agreement with the District to perform or execute the work covered by this contract.
- 2.4 <u>**Contractor Project Manager:**</u> The person designated by the Contractor to administer the Contract operations under this Contract.
- 2.5 **<u>County</u>**: The County of Los Angeles, a political subdivision of the State of California.
- 2.6 **<u>Days</u>**: Calendar day(s) unless otherwise specified.
- 2.7 <u>District</u>: The Consolidated Fire Protection District of Los Angeles County.
- 2.8 **District Project Director:** Person designated by District with authority for District on contractual or administrative matters relating to this contract that cannot be resolved by the District's Project Manager.
- 2.9 **District Project Manager**: Person designated by District's Project Director to manage the operations under this contract.
- 2.10 **Fiscal Year:** The twelve (12) month period beginning July 1st and ending the following June 30th.
- 2.11 **<u>Statement of Work</u>**: The directions, provisions, and requirements provided herein and special provisions pertaining to the method, frequency, manner and place of performing the contract services.

- 2.12 **<u>Subcontract</u>**: An agreement by the contractor to employ a subcontractor to provide services to fulfill this contract.
- 2.13 **Subcontractor:** Any individual, person or persons, sole proprietor, firm, partnership, joint venture, corporation, or other legal entity furnishing supplies, services of any nature, equipment, and/or materials to contractor in furtherance of contractor's performance of this contract, at any tier, under oral or written agreement.

3 WORK

- 3.1 Pursuant to the provisions of this Contract, the Contractor shall fully perform, complete and deliver on time, all tasks, deliverables, services and other work as set forth in herein.
- 3.2 If the Contractor provides any tasks, deliverables, goods, services, or other work, other than as specified in this contract, the same shall be deemed to be a gratuitous effort on the part of the contractor, and the contractor shall have no claim whatsoever against the District.

4 TERM OF CONTRACT

4.1 The term of this Contract shall be five (5) years commencing after approval by County's Board of Supervisors (Board), and execution by the Fire Chief or his designee, unless sooner terminated or extended, in whole or in part, as provided in this Contract.

The District shall have the sole option to extend this Contract term for up to twelve (12) month to month extensions, for a maximum total Contract term of six (6) years, pursuant to the same terms and conditions. Each such extension option may be exercised at the sole discretion of the Fire Chief or his designee as authorized by the Board.

- 4.2 The County maintains databases that track/monitor contractor performance history. Information entered into such databases may be used for a variety of purposes, including determining whether the District will exercise a contract term extension option.
- 4.3 The Contractor shall notify the District when this Contract is within six (6) months of the expiration of the term as provided for hereinabove. Upon occurrence of this event, the Contractor shall send written notification to the District at the address herein provided in Exhibit E District's Administration.

5 CONTRACT SUM

5.1 Maximum Contract Sum

The amount the District shall expend from its own funds during the Contract's entire term for as-needed Alaco Wooden Ladder Services shall not exceed \$150,000 per Contract year.

- 5.1.1 The Maximum Contract Sum shall be the maximum monetary amount available that is payable by the District to the Contractor for supplying all the Services, Deliverables, Work, etc.
- 5.1.2 This is not a commitment or offer on the part of the District to expend the Maximum Contract Sum. The Contractor shall perform and complete all Services required of the Contractor under this Contract as set forth in Exhibit B (Pricing Sheet), but in any event, not in excess of the Maximum Contract Sum.
- 5.1.3 The Contractor acknowledges and agrees the Maximum Contract Sum is an all-inclusive, not-to-exceed price that cannot be adjusted for any costs or expenses of the This Contract includes the full amount of Contractor. compensation and reimbursement the District will be asked to provide to the Contractor in order for the Contractor to fully perform all of its obligations under this Contract, with such amount of compensation and reimbursement subject to any executed Amendments, if applicable. The Contractor understands the District is entering into this Contract in reliance upon the premise the Contractor shall fully perform all of its obligations under this Contract without seeking any additional compensation or reimbursement beyond that already provided for in this Contract, subject to any Amendments, if applicable. It is the Contractor's risk and responsibility to achieve and timely deliver the Services in accordance with the requirements of the Contract.

5.2 Written Approval for Reimbursement

The Contractor shall not be entitled to payment or reimbursement for any tasks or services performed, nor for any incidental or administrative expenses whatsoever incurred in or incidental to performance hereunder, except as specified herein. Assumption or takeover of any of the Contractor's duties, responsibilities, or obligations, or performance of same by any person or entity other than the Contractor, whether through assignment, subcontract, delegation, merger, buyout, or any other mechanism, with or without consideration for any reason whatsoever, shall not occur except with the District's express prior written approval.

5.3 Notification of 75% of Total Contract Sum

The Contractor shall maintain a system of record keeping that will allow the Contractor to determine when it has incurred seventy-five percent (75%) of the total contract sum under this Contract. Upon occurrence of this event, the Contractor shall send written notification to the District at the address herein provided in Exhibit E - District's Administration.

5.4 No Payment for Services Provided Following Expiration-Termination of Contract

The Contractor shall have no claim against District for payment of any money or reimbursement, of any kind whatsoever, for any service provided by the Contractor after the expiration or other termination of this Contract except to the extent any such services were authorized prior to Contract expiration or termination and/or knowingly accepted by the District. Any Work and/or Work Authorization Order in progress prior to the expiration or termination of the Contract shall be completed by the Contractor for full payment of services rendered. The District may at its discretion, verbally and/or by written notice direct any authorized Work to stop and the Contractor shall stop the Work promptly. The Contractor shall be entitled to payment for Work completed prior to receipt of notice to stop, and any Work performed to preserve and protect the District's property. This provision shall survive the expiration or other termination of this Contract.

5.5 Invoices and Payments

- 5.5.1 The Contractor shall invoice the District only for providing the tasks, deliverables, goods, services, and other work specified in Exhibit A Statement of Work and elsewhere hereunder. The Contractor shall prepare invoices, which shall include the charges owed to the Contractor by the District under the terms of this Contract. The Contractor's payments shall be as provided in Exhibit B Pricing Sheet and the Contractor shall be paid only for the tasks, deliverables, goods, services, and other work approved in writing by the District. If the District does not approve work in writing no payment shall be due to the Contractor for that work.
- 5.5.2 The Contractor's invoices shall be priced in accordance with Exhibit B Pricing Sheet.

- 5.5.3 The Contractor's invoices shall contain the information set forth in Exhibit A - Statement of Work describing the tasks, deliverables, goods, services, work hours, and facility and/or other work for which payment is claimed.
- 5.5.4 The Contractor shall submit the monthly invoices to the District by the 15th calendar day of the month following the month of service.
- 5.5.5 Payment to Contractor shall be made on an arrears basis, upon acceptance of completed work by the District, provided the Contractor is not in default under any provisions of this Contract. Contractor shall email one (1) copy of the invoice to the following:
 - 1. <u>EDC.ladder@fire.lacounty.gov</u> for review and approval of all invoices; and
 - 2. <u>Fire-InvoiceSubmission@fire.lacounty.gov</u> for payment of all invoices.

The Contractor's invoices shall include the following:

- Contract Number
- Date(s) of Service
- A breakdown of labor hours, hourly rate, and material costs as separate items, (i.e., labor 3 hours @ \$20/hour = \$60.00).
- Fixed fee (i.e., any flat rate job) authorized by the District Project Manager or designee
- Employee Name and Employee Number of District Employee who ordered or authorized the service.
- Brief description of services.
- A copy of sub-contractor or sublet costs with invoice if a portion of work is contracted out.
- Signature of authorized District employee. Contractor's failure to obtain the signature of District employee authorizing the work may result in a delay of payment.

5.5.6 District Approval of Invoices

All invoices submitted by the Contractor for payment must have the written approval of the District Project Manager prior to any payment thereof. In no event shall the District be liable or responsible for any payment prior to such written approval. Approval for payment will not be unreasonably withheld.

In the event the Contractor does not receive payment from the District for services rendered as prescribed in this SOW, within forty-five (45) calendar days, the invoice will be subject to a 1% late fee or a 2% late fee for invoices not paid within sixty (60) calendar days.

5.6 Cost of Living Adjustments (COLA's)

After the initial three (3) year Contract Term, and if requested by the Contractor, the contract (hourly, daily, monthly, etc.) amount may at the sole discretion of the District, be increased annually based on the most recent published percentage change in the U.S. Department of Labor, Bureau of Labor Statistics' Consumer Price Index for Urban Consumers (CPI-U) for the Los Angeles-Long Beach-Anaheim Area for the twelve (12) month period preceding the contract anniversary date, which shall be the effective date for any Cost of Living Adjustment (COLA). However, any increase shall not exceed the general salary movement granted to County employees as determined by the Chief Executive Officer as of each July 1 for the prior twelve (12) month period. Furthermore, should fiscal circumstances ultimately prevent the Board from approving any increase in County employee salaries, no COLA will be granted. Further, before any COLA increase shall take effect and become part of this Contract, it shall require a written amendment to this Contract first, that has been formally approved and executed by the parties.

5.7 Default Method of Payment: Direct Deposit or Electronic Funds Transfer

- 5.7.1 The County, at its sole discretion, has determined the most efficient and secure default form of payment for goods and/or services provided under an agreement/ contract with the County shall be Electronic Funds Transfer (EFT) or direct deposit, unless an alternative method of payment is deemed appropriate by the Auditor-Controller (A-C).
- 5.7.2 The Contractor shall submit a direct deposit authorization request via the website <u>https://directdeposit.lacounty.gov</u> with banking and vendor information, and any other information that the A-C determines is reasonably necessary to process the payment and comply with all accounting, record keeping, and tax reporting requirements.

- 5.7.3 Any provision of law, grant, or funding agreement requiring a specific form or method of payment other than EFT or direct deposit shall supersede this requirement with respect to those payments.
- 5.7.4 At any time during the duration of the agreement/contract, a Contractor may submit a written request for an exemption to this requirement. Such request must be based on specific legal, business or operational needs and explain why the payment method designated by the A-C is not feasible and an alternative is necessary. The A-C, in consultation with the contracting department(s), shall decide whether to approve exemption requests.

6 ADMINISTRATION OF CONTRACT – DISTRICT

6.1 District Administration

A listing of all District Administration referenced in the following subparagraphs are designated in Exhibit E - District's Administration. The District will notify the Contractor in writing of any change in the names or addresses shown.

6.2 District's Project Director

The responsibilities of the District's Project Director include:

- Coordinating with Contractor and ensuring Contractor's performance of the Contract; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby; and
- Upon request of the Contractor, providing direction to the Contractor, as appropriate in areas relating to District policy, information requirements, and procedural requirements; however, in no event, shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby.

6.3 District's Project Manager

The role of the District's Project Manager is to oversee the day-today administration of this Contract; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby. The District's Project Manager's responsibilities include:

- Meeting with the Contractor's Project Manager on a regular basis; and
- Inspecting any and all tasks, deliverables, goods, services, or other work provided by or on behalf of the Contractor; however, in no event shall Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby.

The District's Project Manager is not authorized to make any changes in any of the terms and conditions of this Contract and is not authorized to further obligate District in any respect whatsoever.

6.4 District's Contract Administrator

The responsibilities of the District's Contract Administrator include:

- Ensuring that the objectives of this Contract are met; and
- Making changes in the terms and conditions of this Contract in accordance with Sub-paragraph 8.1 Amendments; and
- Providing direction to Contractor in the areas relating to District policy, information requirements, and procedural requirements.

7 ADMINISTRATION OF CONTRACT - CONTRACTOR

7.1 Contractor Administration

A listing of all of Contractor's Administration referenced in the following paragraphs is designated in Exhibit F - Contractor's Administration. The Contractor will notify the District in writing of any change in the names or addresses shown.

7.2 Contractor's Project Manager

- 7.2.1 The Contractor's Project Manager is designated in Exhibit F - Contractor's Administration. The Contractor shall notify the District in writing of any change in the name or address of the Contractor's Project Manager.
- 7.2.2 The Contractor's Project Manager shall be responsible for the Contractor's day-to-day activities as related to this Contract and shall meet and coordinate with the District's Project Manager on a regular basis.

7.3 Approval of Contractor's Staff

District has the absolute right to approve or disapprove all of the Contractor's staff performing work hereunder and any proposed changes in the Contractor's staff, including, but not limited to, the Contractor's Project Manager.

7.4 Contractor's Staff Identification

Contractor shall provide, at Contractor's expense, all staff providing services under this Contract with a photo identification badge.

7.5 Background and Security Investigations

7.5.1 Each of Contractor's staff performing services under this Contract, as determined by District in District's sole discretion, shall undergo and pass a background investigation to the satisfaction of District as a condition of beginning and continuing to perform services under this Contract. Such background investigation must be obtained through fingerprints submitted to the California Department of Justice to include State, local, and federal-level review, which may include, but shall not be limited to, criminal conviction information. The fees associated with the background investigation shall be at the expense of the Contractor, regardless of whether the member of Contractor's staff passes or fails the background investigation.

If a member of Contractor's staff does not pass the background investigation, District may request that the member of Contractor's staff be removed immediately from performing services under the Contract. Contractor shall comply with District's request at any time during the term of the Contract. District will not provide to Contractor or to Contractor's staff any information obtained through the District's background investigation.

- 7.5.2 District shall immediately deny or terminate facility access to any member of Contractor's staff that does not pass such investigation to the satisfaction of the District or whose background or conduct is incompatible with District facility access.
- 7.5.3 Disqualification of any member of Contractor's staff pursuant to this Paragraph 7.5 shall not relieve Contractor of its obligation to complete all work in accordance with the terms and conditions of this Contract.

7.6 Confidentiality

- 7.6.1 Contractor shall maintain the confidentiality of all records and information in accordance with all applicable Federal, State and local laws, rules, regulations, ordinances, directives, guidelines, policies and procedures relating to confidentiality, including, without limitation, County policies concerning information technology security and the protection of confidential records and information.
- 7.6.2 Contractor shall indemnify, defend, and hold harmless County, its officers, employees, and agents, from and against any and all claims, demands, damages, liabilities, losses, costs and expenses, including, without limitation, defense costs and legal, accounting and other expert, consulting, or professional fees, arising from, connected with, or related to any failure by Contractor, its officers, employees, agents, or subcontractors, to comply with this Paragraph 7.6, as determined by County in its sole judgment. Any legal defense pursuant to contractor's indemnification obligations under this Paragraph 7.6 shall be conducted by contractor and performed by counsel selected by Contractor and approved by District. Notwithstanding the preceding sentence, District shall have the right to participate in any such defense at its sole cost and expense, except that in the event Contractor fails to provide District with a full and adequate defense, as determined by District in its sole judgment, District shall be entitled to retain its own counsel, including, without limitation, County Counsel, and to reimbursement from Contractor for all such costs and expenses incurred by County in doing so. Contractor shall not have the right to enter into any settlement, agree to any injunction, or make any admission, in each case, on behalf of District without District's prior written approval.
- 7.6.3 Contractor shall inform all of its officers, employees, agents and subcontractors providing services hereunder of the confidentiality provisions of this Contract.
- 7.6.4 Contractor shall sign and adhere to the provisions of the "Contractor Acknowledgement and Confidentiality Agreement" - Exhibit G.

8 STANDARD TERMS AND CONDITIONS

8.1 Amendments

- 8.1.1 For any change which affects the scope of work, term, contract sum, payments, or any term or condition included under this Contract, an amendment to the Contract shall be prepared and executed by the contractor and by the Fire Chief or his designee.
- 8.1.2 The County's Board of Supervisors or Chief Executive Officer or designee may require the addition and/or change of certain terms and conditions in the Contract during the term of this Contract. The District reserves the right to add and/or change such provisions as required by the County's Board of Supervisors or Chief Executive Officer. To implement such changes, an Amendment to the Contract shall be prepared and executed by the contractor and by the Fire Chief or his designee.
- 8.1.3 The Fire Chief or his designee may at his sole discretion, authorize extensions of time as defined in Paragraph 4 Term of Contract. The contractor agrees that such extensions of time shall not change any other term or condition of this Contract during the period of such extensions. To implement an extension of time, an Amendment to the Contract shall be prepared and executed by the contractor and by the Fire Chief or his designee.

8.2 Assignment and Delegation/Mergers or Acquisitions

- 8.2.1 The contractor shall notify the District of any pending acquisitions/mergers of its company unless otherwise legally prohibited from doing so. If the contractor is restricted from legally notifying the District of pending acquisitions/mergers, then it should notify the District of the actual acquisitions/mergers as soon as the law allows and provide to the District the legal framework that restricted it from notifying the District prior to the actual acquisitions/mergers.
- 8.2.2 The contractor shall not assign its rights or delegate its duties under this Contract, or both, whether in whole or in part, without the prior written consent of District, in its discretion, and any attempted assignment or delegation without such consent shall be null and void. For purposes of this paragraph, District consent shall require a written Amendment to the Contract, which is formally approved and executed by the parties. Any payments by the District to any approved delegatee or assignee on any claim under this Contract shall be deductible, at District's sole discretion, against the claims, which the contractor may have against the District.

- 8.2.3 Shareholders, partners, members, or other equity holders of contractor may transfer, sell, exchange, assign, or divest themselves of any interest they may have therein. However, in the event any such sale, transfer, exchange, assignment, or divestment is effected in such a way as to give majority control of contractor to any person(s), corporation, partnership, or legal entity other than the majority controlling interest therein at the time of execution of the Contract, such disposition is an assignment requiring the prior written consent of District in accordance with applicable provisions of this Contract.
- 8.2.4 Any assumption, assignment, delegation, or takeover of any of the contractor's duties, responsibilities, obligations, or performance of same by any person or entity other than the contractor, whether through assignment, subcontract. delegation, merger, buyout, or any other mechanism, with or for without consideration anv reason whatsoever without District's express prior written approval, shall be a material breach of the Contract which may result in the termination of this Contract. In the event of such termination. District shall be entitled to pursue the same remedies against contractor as it could pursue in the event of default by contractor.

8.3 Authorization Warranty

The contractor represents and warrants that the person executing this Contract for the contractor is an authorized agent who has actual authority to bind the contractor to each and every term, condition, and obligation of this Contract and that all requirements of the contractor have been fulfilled to provide such actual authority.

8.4 Budget Reductions

In the event the County's Board of Supervisors adopts, in any fiscal year, a County Budget which provides for reductions in the salaries and benefits paid to the majority of County employees and imposes similar reductions with respect to County contracts, the District reserves the right to reduce its payment obligation under this Contract correspondingly for that fiscal year and any subsequent fiscal year during the term of this Contract (including any extensions), and the services to be provided by the contractor under this Contract shall also be reduced correspondingly. The District's notice to the contractor regarding said reduction in payment obligation shall be provided within thirty (30) calendar days of the Board's approval of such actions. Except as set forth in the preceding sentence, the

contractor shall continue to provide all of the services set forth in this Contract.

8.5 Complaints

- 8.5.1 The contractor shall develop, maintain and operate procedures for receiving, investigating and responding to complaints.
- 8.5.2 Complaint Procedures
 - 8.5.2.1 Within ten (10) business days after the Contract effective date, the contractor shall provide the District with the contractor's policy for receiving, investigating and responding to user complaints.
 - 8.5.2.2 The District will review the contractor's policy and provide the contractor with approval of said plan or with requested changes.
 - 8.5.2.3 If the District requests changes in the contractor's policy, the contractor shall make such changes and resubmit the plan within five (5) business days for District approval.
 - 8.5.2.4 If, at any time, the contractor wishes to change the contractor's policy, the contractor shall submit proposed changes to the District for approval before implementation.
 - 8.5.2.5 The contractor shall preliminarily investigate all complaints and notify the District's Project Manager of the status of the investigation within five (5) business days of receiving the complaint.
 - 8.5.2.6 When complaints cannot be resolved informally, a system of follow-through shall be instituted which adheres to formal plans for specific actions and strict time deadlines.
 - 8.5.2.7 Copies of all written responses shall be sent to the District Project Manager within three (3) business days of mailing to the complainant.

8.6 Compliance with Applicable Law

8.6.1 In the performance of this Contract, contractor shall comply with all applicable Federal, State and local laws, rules, regulations, ordinances, directives, guidelines, policies and procedures, and all provisions required thereby to be included in this Contract are hereby incorporated herein by reference.

8.6.2 Contractor shall indemnify, defend, and hold harmless County, its officers, employees, and agents, from and against any and all claims, demands, damages, liabilities, losses, costs, and expenses, including, without limitation, defense costs and legal, accounting and other expert, consulting or professional fees, arising from, connected with, or related to any failure by contractor, its officers, employees, agents, or subcontractors, to comply with any such laws, rules, regulations, ordinances, directives, guidelines, policies, or procedures, as determined by County in its sole judgment. Any legal defense pursuant to contractor's indemnification obligations under Paragraph 8.6 - Compliance with Applicable Law shall be conducted by contractor and performed by counsel selected by contractor and approved by County. Notwithstanding the preceding sentence, County shall have the right to participate in any such defense at its sole cost and expense, except that in the event contractor fails to provide County with a full and adequate defense, as determined by County in its sole judgment, County shall be entitled to retain its own counsel, including, without limitation, County Counsel, and to reimbursement from contractor for all such costs and expenses incurred by County in doing so. Contractor shall not have the right to enter into any settlement, agree to any injunction or other equitable relief, or make any admission, in each case, on behalf of County without County's prior written approval.

8.7 Compliance with Civil Rights Laws

The contractor hereby assures that it will comply with Subchapter VI of the Civil Rights Act of 1964, 42 USC Sections 2000 (e) (1) through 2000 (e) (17), to the end that no person shall, on the grounds of race, creed, color, sex, religion, ancestry, age, condition of physical handicap, marital status, political affiliation, or national origin, be excluded from participation in, be denied the benefits of, or be otherwise subjected to discrimination under this Contract or under any project, program, or activity supported by this Contract. The contractor shall comply with Exhibit D - Contractor's EEO Certification.

8.8 Compliance with the County's Jury Service Program

8.8.1 Jury Service Program:

This Contract is subject to the provisions of the County's ordinance entitled Contractor Employee Jury Service ("Jury Service Program") as codified in Sections 2.203.010 through 2.203.090 of the Los Angeles County Code, a copy of which is attached as Exhibit H and incorporated by reference into and made a part of this Contract.

8.8.2 Written Employee Jury Service Policy.

- Unless the contractor has demonstrated to the District's 1. satisfaction either that the contractor is not a "contractor" as defined under the Jury Service Program (Section 2.203.020 of the County Code) or that the contractor qualifies for an exception to the Jury Service Program (Section 2.203.070 of the County Code), the contractor shall have and adhere to a written policy that provides that its Employees shall receive from the contractor, on an annual basis, no less than five days of regular pay for actual jury service. The policy may provide that Employees deposit any fees received for such jury service with the contractor or that the contractor deduct from the Employee's regular pay the fees received for jury service.
- 2. For purposes of this paragraph, "contractor" means a person, partnership, corporation or other entity which has a contract with the County or a subcontract with a County contractor and has received or will receive an aggregate sum of fifty thousand dollars (\$50,000) or more in any twelve (12) month period under one or more County contracts or subcontracts. "Employee" means any California resident who is a full-time employee of the contractor. "Full-time" means forty (40) hours or more worked per week, or a lesser number of hours if: 1) the lesser number is a recognized industry standard as determined by the District, or 2) contractor has a longstanding practice that defines the lesser number of hours as full-time. Full-time employees providing short-term, temporary services of ninety (90) days or less within a twelve (12) month period are not considered full-time for purposes of the Jury Service Program. If the contractor uses any subcontractor to perform services for the District under the Contract, the subcontractor shall also be subject to the provisions of this paragraph. The provisions of this paragraph shall be inserted into any

such subcontract agreement and a copy of the Jury Service Program shall be attached to the agreement.

- 3. If the contractor is not required to comply with the Jury Service Program when the Contract commences, the contractor shall have a continuing obligation to review the applicability of its "exception status" from the Jury Service Program, and the contractor shall immediately notify the District if the contractor at any time either comes within the Jury Service Program's definition of "contractor" or if the contractor no longer qualifies for an exception to the Jury Service Program. In either event, the contractor shall immediately implement a written policy consistent with the Jury Service Program. The District may also require, at any time during the Contract and at its sole discretion, that the contractor demonstrate, to the District's satisfaction that the contractor either continues to remain outside of the Jury Service Program's definition of "contractor" and/or that the contractor continues to qualify for an exception to the Program.
- 4. Contractor's violation of this paragraph of the Contract may constitute a material breach of the Contract. In the event of such material breach, District may, in its sole discretion, terminate the Contract and/or bar the contractor from the award of future County contracts for a period of time consistent with the seriousness of the breach.

8.9 Conflict of Interest

- 8.9.1 No County employee whose position with the County enables such employee to influence the award of this Contract or any competing Contract, and no spouse or economic dependent of such employee, shall be employed in any capacity by the contractor or have any other direct or indirect financial interest in this Contract. No officer or employee of the contractor who may financially benefit from the performance of work hereunder shall in any way participate in the District's approval, or ongoing evaluation, of such work, or in any way attempt to unlawfully influence the District's approval or ongoing evaluation of such work.
- 8.9.2 The contractor shall comply with all conflict of interest laws, ordinances, and regulations now in effect or hereafter to be enacted during the term of this Contract. The contractor

warrants that it is not now aware of any facts that create a conflict of interest. If the contractor hereafter becomes aware of any facts that might reasonably be expected to create a conflict of interest, it shall immediately make full written disclosure of such facts to the District. Full written disclosure shall include, but is not limited to, identification of all persons implicated and a complete description of all relevant circumstances. Failure to comply with the provisions of this paragraph shall be a material breach of this Contract.

8.10 Consideration of Hiring County Employees Targeted for Layoffs or are on a County Re-Employment List

Should the Contractor require additional or replacement personnel after the effective date of this Contract to perform the services set forth herein, the Contractor shall give first consideration for such employment openings to qualified, permanent County employees who are targeted for layoff or qualified, former County employees who are on a re-employment list during the life of this Contract.

8.11 Consideration of Hiring GAIN-GROW Participants

- 8.11.1 Should the contractor require additional or replacement personnel after the effective date of this Contract, the contractor shall give consideration for any such employment openings to participants in the County's Department of Public Social Services Greater Avenues for Independence (GAIN) Program or General Relief Opportunity for Work (GROW) Program who meet the contractor's minimum gualifications for the open position. For this purpose, consideration shall mean that the contractor will interview gualified candidates. The County will refer GAIN-GROW participants by job category to the contractor. Contractors shall report all job requirements openings with iob to: GAINGROW@DPSS.LACOUNTY.GOV and BSERVICES@WDACS.LACOUNTY.GOV and DPSS will refer qualified GAIN/GROW job candidates.
- 8.11.2 In the event that both laid-off County employees and GAIN/GROW participants are available for hiring, County employees shall be given first priority.

8.12 Contractor Responsibility and Debarment

8.12.1 **Responsible Contractor**

A responsible contractor is a contractor who has demonstrated the attribute of trustworthiness, as well as quality, fitness, capacity and experience to satisfactorily perform the contract. It is the County's policy to conduct business only with responsible contractors.

8.12.2 Chapter 2.202 of the County Code

The contractor is hereby notified that, in accordance with Chapter 2.202 of the County Code, if the County acquires information concerning the performance of the contractor on this or other contracts which indicates that the contractor is not responsible, the County may, in addition to other remedies provided in the Contract, debar the contractor from bidding or proposing on, or being awarded, and/or performing work on County contracts for a specified period of time, which generally will not exceed five (5) years but may exceed five (5) years or be permanent if warranted by the circumstances, and terminate any or all existing contracts the contractor may have with the County.

8.12.3 Non-responsible contractor

The County may debar a contractor if the Board of Supervisors finds, in its discretion, that the contractor has done any of the following: 1) violated a term of a contract with the County or a nonprofit corporation created by the County, 2) committed an act or omission which negatively reflects on the contractor's quality, fitness or capacity to perform a contract with the County, any other public entity, or a nonprofit corporation created by the County, or engaged in a pattern or practice which negatively reflects on same, 3) committed an act or offense which indicates a lack of business integrity or business honesty, or 4) made or submitted a false claim against the County or any other public entity.

8.12.4 Contractor Hearing Board

8.12.4.1 If there is evidence that the contractor may be subject to debarment, the County will notify the contractor in writing of the evidence which is the basis for the proposed debarment and will advise the contractor of the scheduled date for a debarment hearing before the Contractor Hearing Board.

- 8.12.4.2 The Contractor Hearing Board will conduct a hearing where evidence on the proposed debarment is presented. The contractor and/or the contractor's representative shall be given an opportunity to submit evidence at that hearing. After the hearing, the Contractor Hearing Board shall prepare a tentative proposed decision, which shall contain a recommendation regarding whether the contractor should be debarred, and, if so, the appropriate length of time of the debarment. The contractor and the District shall be provided an opportunity to object to the tentative proposed decision prior to its presentation to the Board of Supervisors.
- 8.12.4.3 After consideration of any objections, or if no objections are submitted, a record of the hearing, the proposed decision, and any other recommendation of the Contractor Hearing Board shall be presented to the Board of Supervisors. The Board of Supervisors shall have the right to modify, deny, or adopt the proposed decision and recommendation of the Contractor Hearing Board.
- 8.12.4.4 If a contractor has been debarred for a period longer than five (5) years, that contractor may after the debarment has been in effect for at least five (5) years, submit a written request for review of the debarment determination to reduce the period of debarment or terminate the debarment. The County may, in its discretion, reduce the period of debarment or terminate the debarment if it finds that the contractor has adequately demonstrated one or more of the following: 1) elimination of the grounds for which the debarment was imposed; 2) a bona fide change in ownership or management; 3) material evidence discovered after debarment was imposed: or 4) any other reason that is in the best interests of the County.
- 8.12.4.5 The Contractor Hearing Board will consider a request for review of a debarment determination only where 1) the contractor has been debarred for a period longer than five (5) years; 2) the debarment has been in effect for at least five (5) years; and 3) the request is in writing, states one

or more of the grounds for reduction of the debarment period or termination of the and includes supporting debarment. documentation. Upon receiving an appropriate request, the Contractor Hearing Board will provide notice of the hearing on the request. At the hearing, the Contractor Hearing Board shall conduct a hearing where evidence on the proposed reduction of debarment period or termination of debarment is presented. This hearing shall be conducted and the request for review decided by the Contractor Hearing Board pursuant to the same procedures as for a debarment hearing.

8.12.4.6 The Contractor Hearing Board's proposed decision shall contain a recommendation on the request to reduce the period of debarment or terminate the debarment. The Contractor Hearing Board shall present its proposed decision and recommendation to the Board of Supervisors. The Board of Supervisors shall have the right to modify, deny, or adopt the proposed decision and recommendation of the Contractor Hearing Board.

8.12.5 Subcontractors of Contractor

These terms shall also apply to subcontractors of County contractors.

8.13 Contractor's Acknowledgement of County's Commitment to Safely Surrendered Baby Law

The contractor acknowledges that the County places a high priority on the implementation of the Safely Surrendered Baby Law. The contractor understands that it is the County's policy to encourage all County contractors to voluntarily post the County's "Safely Surrendered Baby Law" poster, in Exhibit I, in a prominent position at the contractor's place of business. The contractor will also encourage its subcontractors, if any, to post this poster in a prominent position in the subcontractor's place of business. Information and posters for printing are available at <u>www.babysafela.org.</u>

8.14 Contractor's Warranty of Adherence to County's Child Support Compliance Program

- 8.14.1 The contractor acknowledges that the County has established a goal of ensuring that all individuals who benefit financially from the County through contracts are in compliance with their court-ordered child, family and spousal support obligations in order to mitigate the economic burden otherwise imposed upon the County and its taxpayers.
- 8.14.2 As required by the County's Child Support Compliance Program (County Code Chapter 2.200) and without limiting the contractor's duty under this Contract to comply with all applicable provisions of law, the contractor warrants that it is now in compliance and shall during the term of this Contract maintain in compliance with employment and wage reporting requirements as required by the Federal Social Security Act (42 USC Section 653a) and California Unemployment Insurance Code Section 1088.5, and shall implement all lawfully served Wage and Earnings Withholding Orders or Child Support Services Department Notices of Wage and Earnings Assignment for Child, Family or Spousal Support, pursuant to Code of Civil Procedure Section 706.031 and Family Code Section 5246(b).

8.15 District's Quality Assurance Plan

The District or its agent(s) will monitor the contractor's performance under this Contract on not less than an annual basis. Such monitoring will include assessing the contractor's compliance with all Contract terms and conditions and performance standards. Contractor deficiencies which the District determines are significant or continuing and that may place performance of the Contract in jeopardy if not corrected will be reported to the Board of Supervisors and listed in the appropriate contractor performance database. The report to the Board will include improvement/corrective action measures taken by the District and the contractor. If improvement does not occur consistent with the corrective action measures, the District may terminate this Contract or impose other penalties as specified in this Contract.

8.16 Damage to District Facilities, Buildings or Grounds

8.16.1 The contractor shall repair, or cause to be repaired, at its own cost, any and all damage to District facilities, buildings, or grounds caused by the contractor or employees or agents of the contractor. Such repairs shall be made immediately after

the contractor has become aware of such damage, but in no event later than thirty (30) days after the occurrence.

8.16.2 If the contractor fails to make timely repairs, District may make any necessary repairs. All costs incurred by District, as determined by District, for such repairs shall be repaid by the contractor by cash payment upon demand.

8.17 Employment Eligibility Verification

- 8.17.1 The contractor warrants that it fully complies with all Federal and State statutes and regulations regarding the employment of aliens and others and that all its employees performing work under this Contract meet the citizenship or alien status requirements set forth in Federal and State statutes and regulations. The contractor shall obtain, from all employees performing work hereunder, all verification and other documentation of employment eligibility status required by Federal and State statutes and regulations including, but not limited to, the Immigration Reform and Control Act of 1986, (P.L. 99-603), or as they currently exist and as they may be hereafter amended. The contractor shall retain all such documentation for all covered employees for the period prescribed by law.
- 8.17.2 The contractor shall indemnify, defend, and hold harmless, the County, its agents, officers, and employees from employer sanctions and any other liability which may be assessed against the contractor or the County or both in connection with any alleged violation of any Federal or State statutes or regulations pertaining to the eligibility for employment of any persons performing work under this Contract.

8.18 Counterparts and Electronic Signatures and Representations

This Contract may be executed in two or more counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same Contract. The facsimile, email or electronic signature of the Parties shall be deemed to constitute original signatures, and facsimile or electronic copies hereof shall be deemed to constitute duplicate originals.

The County and the Contractor hereby agree to regard electronic representations of original signatures of authorized officers of each party, when appearing in appropriate places on the Amendments prepared pursuant to Paragraph 8.1 (Amendments) and received via

communications facilities (facsimile, email or electronic signature), as legally sufficient evidence that such legally binding signatures have been affixed to Amendments to this Contract.

8.19 Fair Labor Standards

The contractor shall comply with all applicable provisions of the Federal Fair Labor Standards Act and shall indemnify, defend, and hold harmless the County and its agents, officers, and employees from any and all liability, including, but not limited to, wages, overtime pay, liquidated damages, penalties, court costs, and attorneys' fees arising under any wage and hour law, including, but not limited to, the Federal Fair Labor Standards Act, for work performed by the contractor's employees for which the County may be found jointly or solely liable.

8.20 Force Majeure

- 8.20.1 Neither party shall be liable for such party's failure to perform its obligations under and in accordance with this Contract, if such failure arises out of fires, floods, epidemics, quarantine restrictions, other natural occurrences, strikes, lockouts (other than a lockout by such party or any of such party's subcontractors), freight embargoes, or other similar events to those described above, but in every such case the failure to perform must be totally beyond the control and without any fault or negligence of such party (such events are referred to in this paragraph as "force majeure events").
- 8.20.2 Notwithstanding the foregoing, a default by a subcontractor of contractor shall not constitute a force majeure event, unless such default arises out of causes beyond the control of both contractor and such subcontractor, and without any fault or negligence of either of them. In such case, contractor shall not be liable for failure to perform, unless the goods or services to be furnished by the subcontractor were obtainable from other sources in sufficient time to permit contractor to meet the required performance schedule. As used in this subparagraph, the term "subcontractor" and "subcontractors" mean subcontractors at any tier.
- 8.20.3 In the event contractor's failure to perform arises out of a force majeure event, contractor agrees to use commercially reasonable best efforts to obtain goods or services from other

sources, if applicable, and to otherwise mitigate the damages and reduce the delay caused by such force majeure event.

8.21 Governing Law, Jurisdiction, and Venue

This Contract shall be governed by, and construed in accordance with, the laws of the State of California. The contractor agrees and consents to the exclusive jurisdiction of the courts of the State of California for all purposes regarding this Contract and further agrees and consents that venue of any action brought hereunder shall be exclusively in the County of Los Angeles.

8.22 Independent Contractor Status

- 8.22.1 This Contract is by and between the District and the contractor and is not intended, and shall not be construed, to create the relationship of agent, servant, employee, partnership, joint venture, or association, as between the District and the contractor. The employees and agents of one party shall not be, or be construed to be, the employees or agents of the other party for any purpose whatsoever.
- 8.22.2 The contractor shall be solely liable and responsible for providing to, or on behalf of, all persons performing work pursuant to this Contract all compensation and benefits. The District shall have no liability or responsibility for the payment of any salaries, wages, unemployment benefits, disability benefits, Federal, State, or local taxes, or other compensation, benefits, or taxes for any personnel provided by or on behalf of the contractor.
- 8.22.3 The contractor understands and agrees that all persons performing work pursuant to this Contract are, for purposes of Workers' Compensation liability, solely employees of the contractor and not employees of the District. The contractor shall be solely liable and responsible for furnishing any and all Workers' Compensation benefits to any person as a result of any injuries arising from or connected with any work performed by or on behalf of the contractor pursuant to this Contract.
- 8.22.4 The contractor shall adhere to the provisions stated in Paragraph 7.6 Confidentiality.

8.23 Indemnification

The contractor shall indemnify, defend and hold harmless the County, its Special Districts, elected and appointed officers, employees, agents and volunteers (County Indemnitees) from and against any and all liability, including but not limited to demands, claims, actions, fees, costs and expenses (including attorney and expert witness fees), arising from and/or relating to this Contract, except for such loss or damage arising from the sole negligence or willful misconduct of the County indemnitees.

8.24 General Provisions for all Insurance Coverage

Without limiting Contractor's indemnification of District, and in the performance of this Contract and until all of its obligations pursuant to this Contract have been met, Contractor shall provide and maintain at its own expense insurance coverage satisfying the requirements specified in Paragraphs 8.24 and 8.25 of this Contract. These minimum insurance coverage terms, types and limits (the "Required Insurance") also are in addition to and separate from any other contractual obligation imposed upon Contractor pursuant to this Contract. The District in no way warrants that the Required Insurance is sufficient to protect the Contractor for liabilities which may arise from or relate to this Contract.

8.24.2 Evidence of Coverage and Notice to District

- 8.24.2.1 Certificate(s) of insurance coverage (Certificate) satisfactory to District, and a copy of an Additional Insured endorsement confirming County and its Agents (defined below) has been given Insured status under the Contractor's General Liability policy, shall be delivered to District at the address shown below and provided prior to commencing services under this Contract.
- 8.24.2.2 Renewal Certificates shall be provided to District not less than ten (10) days prior to contractor's policy expiration dates. The District reserves the right to obtain complete, certified copies of any required contractor and/or sub-contractor insurance policies at any time.
- 8.24.2.3 Certificates shall identify all Required Insurance coverage types and limits specified herein, reference this Contract by name or number, and

be signed by an authorized representative of the insurer(s). The Insured party named on the Certificate shall match the name of the contractor identified as the contracting party in this Contract. Certificates shall provide the full name of each insurer providing coverage, its NAIC Association of (National Insurance Commissioners) identification number. its financial rating, the amounts of any policy deductibles or self-insured retentions exceeding fifty thousand dollars (\$50,000), and list any District required endorsement forms.

- 8.24.2.4 Neither the District's failure to obtain, nor the District's receipt of, or failure to object to a noncomplying insurance certificate or endorsement, or any other insurance documentation or information provided by the contractor, its insurance broker(s) and/or insurer(s), shall be construed as a waiver of any of the Required Insurance provisions.
- 8.24.2.5 Certificates and copies of any required endorsements shall be sent to:

<u>Consolidated Fire Protection District of</u> <u>Los Angeles County</u> <u>Materials Management Division/Contracts Section</u> <u>5801 S. Eastern Avenue, Suite 100</u> <u>Commerce, California 90040-4001</u>

8.24.2.6 Contractor also shall promptly report to District any injury or property damage accident or incident, including any injury to a contractor employee occurring on District property, and any loss, disappearance, destruction, misuse, or theft of District property, monies or securities entrusted to contractor. Contractor also shall promptly notify District of any third party claim or suit filed against contractor or any of its subcontractors which arises from or relates to this Contract, and could result in the filing of a claim or lawsuit against contractor and/or County.

8.24.3 Additional Insured Status and Scope of Coverage

The County of Los Angeles, it's Special Districts, Elected Officials, Officers, Agents, employees and volunteers (collectively County and its Agents) shall be provided additional insured status under contractor's General Liability policy with respect to liability arising out of contractor's ongoing and completed operations performed on behalf of the District. County and its Agents additional insured status shall apply with respect to liability and defense of suits arising out of the contractor's acts or omissions, whether such liability is attributable to the contractor or to the District. The full policy limits and scope of protection also shall apply to the County and its Agents as an additional insured, even if they exceed the County's minimum Required Insurance specifications herein. Use of an automatic additional insured endorsement form is acceptable providing it satisfies the Required Insurance provisions herein.

8.24.4 Cancellation of or Changes in Insurance

Contractor shall provide District with, or contractor's insurance policies shall contain a provision that District shall receive, written notice of cancellation or any change in Required Insurance, including insurer, limits of coverage, term of coverage or policy period. The written notice shall be provided to District at least ten (10) days in advance of cancellation for non-payment of premium and thirty (30) days in advance for any other cancellation or policy change. Failure to provide written notice of cancellation or any change in Required Insurance may constitute a material breach of the Contract, in the sole discretion of the District, upon which the District may suspend or terminate this Contract.

8.24.5 Failure to Maintain Insurance

Contractor's failure to maintain or to provide acceptable evidence that it maintains the Required Insurance shall constitute a material breach of the Contract, upon which District immediately may withhold payments due to contractor, and/or suspend or terminate this Contract. District, at its sole discretion, may obtain damages from contractor resulting from said breach. Alternatively, the District may purchase the Required Insurance, and without further notice to contractor, deduct the premium cost from sums due to contractor or pursue contractor reimbursement.

8.24.6 Insurer Financial Ratings

Coverage shall be placed with insurers acceptable to the District with A.M. Best ratings of not less than A:VII unless otherwise approved by District.

8.24.7 Contractor's Insurance Shall Be Primary

Contractor's insurance policies, with respect to any claims related to this Contract, shall be primary with respect to all other sources of coverage available to contractor. Any District maintained insurance or self-insurance coverage shall be in excess of and not contribute to any contractor coverage.

8.24.8 Waivers of Subrogation

To the fullest extent permitted by law, the contractor hereby waives its rights and its insurer(s)' rights of recovery against County under all the Required Insurance for any loss arising from or relating to this Contract. The contractor shall require its insurers to execute any waiver of subrogation endorsements which may be necessary to effect such waiver.

8.24.9 Subcontractor Insurance Coverage Requirements

Contractor shall include all subcontractors as insureds under contractor's own policies, or shall provide District with each subcontractor's separate evidence of insurance coverage. Contractor shall be responsible for verifying each subcontractor complies with the Required Insurance provisions herein, and shall require that each subcontractor name the County and contractor as additional insureds on the subcontractor's General Liability policy. Contractor shall obtain County's prior review and approval of any subcontractor request for modification of the Required Insurance.

8.24.10 Deductibles and Self-Insured Retentions (SIRs)

Contractor's policies shall not obligate the County to pay any portion of any contractor deductible or SIR. The District retains the right to require contractor to reduce or eliminate

policy deductibles and SIRs as respects the District, or to provide a bond guaranteeing contractor's payment of all deductibles and SIRs, including all related claims investigation, administration and defense expenses. Such bond shall be executed by a corporate surety licensed to transact business in the State of California.

8.24.11 Claims Made Coverage

If any part of the Required Insurance is written on a claims made basis, any policy retroactive date shall precede the effective date of this Contract. Contractor understands and agrees it shall maintain such coverage for a period of not less than three (3) years following Contract expiration, termination or cancellation.

8.24.12 Application of Excess Liability Coverage

Contractors may use a combination of primary, and excess insurance policies which provide coverage as broad as ("follow form" over) the underlying primary policies, to satisfy the Required Insurance provisions.

8.24.13 Separation of Insureds

All liability policies shall provide cross-liability coverage as would be afforded by the standard ISO (Insurance Services Office, Inc.) separation of insureds provision with no insured versus insured exclusions or limitations.

8.24.14 Alternative Risk Financing Programs

The District reserves the right to review, and then approve, Contractor use of self-insurance, risk retention groups, risk purchasing groups, pooling arrangements and captive insurance to satisfy the Required Insurance provisions. The County and its Agents shall be designated as an Additional Covered Party under any approved program.

8.24.15 District Review and Approval of Insurance Requirements

The District reserves the right to review and adjust the Required Insurance provisions, conditioned upon District's determination of changes in risk exposures.

8.25 Insurance Coverage

8.25.1 **Commercial General Liability** insurance (providing scope of coverage equivalent to ISO policy form CG 00 01), naming County and its Agents as an additional insured, with limits of not less than:

General Aggregate:	\$2 million
Products/Completed Operations Aggregate:	\$1 million
Personal and Advertising Injury:	\$1 million
Each Occurrence:	\$1 million

- 8.25.2 **Automobile Liability** insurance (providing scope of coverage equivalent to ISO policy form CA 00 01) with limits of not less than \$1 million for bodily injury and property damage, in combined or equivalent split limits, for each single accident. Insurance shall cover liability arising out of contractor's use of autos pursuant to this Contract, including owned, leased, hired, and/or non-owned autos, as each may be applicable.
- 8.25.3 Workers Compensation and Employers' Liability insurance or qualified self- insurance satisfying statutory requirements, which includes Employers' Liability coverage with limits of not less than \$1 million per accident. If Contractor will provide leased employees, or, is an employee leasing or temporary staffing firm or a professional employer organization (PEO), coverage also shall include an Alternate Employer Endorsement (providing scope of coverage equivalent to ISO policy form WC 00 03 01 A) naming the County as the Alternate Employer. The written notice shall be provided to District at least ten (10) days in advance of cancellation for non-payment of premium and thirty (30) days in advance for any other cancellation or policy change. If applicable to Contractor's operations, coverage also shall be arranged to satisfy the requirements of any federal workers or workmen's compensation law or any federal occupational disease law.

8.25.4 Unique Insurance Coverage

8.25.4.1 Intentionally Omitted

8.25.4.2 Professional Liability-Errors and Omissions

Professional Liability/Errors and Omissions coverage is required for medical and legal Contractors, as well as Contractors in nontraditional professions including, but not limited to accountants, appraisers, architects, billers, computer programmers, engineers, interpreters, staffing/ temporary services agencies, and consultants.

Insurance covering Contractor's liability arising from or related to this Contract, with limits of not less than \$1 million per claim and \$2 million aggregate. Further, Contractor understands and agrees it shall maintain such coverage for a period of not less than three (3) years following this Agreement's expiration, termination or cancellation.

8.25.4.3 Property Coverage

Contractors given exclusive use of District owned or leased property shall carry property coverage at least as broad as that provided by the ISO special causes of loss (ISO policy form CP 10 30) form. The County and its Agents shall be named as an Additional Insured and Loss Payee on contractor's interests insurance as its may appear. Automobiles and mobile equipment shall be insured for their actual cash value. Real property and all other personal property shall be insured for their full replacement value.

- 8.25.4.4 Intentionally Omitted
- 8.25.4.5 Intentionally Omitted
- 8.25.4.6 Intentionally Omitted
- 8.25.4.7 Intentionally Omitted

8.26 Liquidated Damages

- 8.26.1 If, in the judgment of the Fire Chief or his designee, the contractor is deemed to be non-compliant with the terms and obligations assumed hereby, the Fire Chief or his designee, at his/her option, in addition to, or in lieu of, other remedies provided herein, may withhold the entire monthly payment or deduct pro rata from the contractor's invoice for work not performed. A description of the work not performed and the amount to be withheld or deducted from payments to the contractor from the District, will be forwarded to the contractor by the Fire Chief or his designee, in a written notice describing the reasons for said action.
- 8.26.2 If the Fire Chief or his designee, determines that there are deficiencies in the performance of this Contract that the Fire Chief or his designee, deems are correctable by the contractor over a certain time span, the Fire Chief or his designee, will provide a written notice to the contractor to correct the deficiency within specified time frames. Should the contractor fail to correct deficiencies within said time frame, the Fire Chief, or his designee, may: (a) Deduct from the contractor's payment, pro rata, those applicable portions of the Monthly Contract Sum; and/or (b) Deduct liquidated damages. The parties agree that it will be impracticable or extremely difficult to fix the extent of actual damages resulting from the failure of the contractor to correct a deficiency within the specified time frame. The parties hereby agree that under the current circumstances a reasonable estimate of such damages is one hundred dollars (\$100) per day per infraction, or as specified in the Exhibit 2 - Performance Requirements Summary (PRS) Chart, Statement of Work Exhibits, hereunder, and that the contractor shall be liable to the District for liquidated damages in said amount. Said amount shall be deducted from the District's payment to the contractor; and/or (c) Upon giving five (5) days notice to the contractor for failure to correct the deficiencies, the District may correct any and all deficiencies and the total costs incurred by the District for completion of the work by an alternate source, whether it be District forces or separate private contractor, will be deducted and forfeited from the payment to the contractor from the District, as determined by the District.
- 8.26.3 The action noted in Paragraph 8.26.2 shall not be construed as a penalty, but as adjustment of payment to the contractor

to recover the District cost due to the failure of the contractor to complete or comply with the provisions of this Contract.

8.26.4 This Paragraph shall not, in any manner, restrict or limit the District's right to damages for any breach of this Contract provided by law or as specified in the PRS or Paragraph 8.26.2, and shall not, in any manner, restrict or limit the District's right to terminate this Contract as agreed to herein.

8.27 Most Favored Public Entity

If the contractor's prices decline, or should the contractor at any time during the term of this Contract provide the same goods or services under similar quantity and delivery conditions to the State of California or any county, municipality, or district of the State at prices below those set forth in this Contract, then such lower prices shall be immediately extended to the District.

8.28 Nondiscrimination and Affirmative Action

- 8.28.1 The contractor certifies and agrees that all persons employed by it, its affiliates, subsidiaries, or holding companies are and shall be treated equally without regard to or because of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, in compliance with all applicable Federal and State anti-discrimination laws and regulations.
- 8.28.2 The contractor shall certify to, and comply with, the provisions of Exhibit D Contractor's EEO Certification.
- 8.28.3 The contractor shall take affirmative action to ensure that applicants are employed, and that employees are treated during employment, without regard to race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, in compliance with all applicable Federal and State anti-discrimination laws and regulations. Such action shall include, but is not limited to: employment, upgrading, demotion, transfer, recruitment or recruitment advertising, layoff or termination, rates of pay or other forms of compensation, and selection for training, including apprenticeship.
- 8.28.4 The contractor certifies and agrees that it will deal with its subcontractors, bidders, or vendors without regard to or because of race, color, religion, ancestry, national origin, sex,

age, physical or mental disability, marital status, or political affiliation.

- 8.28.5 The contractor certifies and agrees that it, its affiliates, subsidiaries, or holding companies shall comply with all applicable Federal and State laws and regulations to the end that no person shall, on the grounds of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, be excluded from participation in, be denied the benefits of, or be otherwise subjected to discrimination under this Contract or under any project, program, or activity supported by this Contract.
- 8.28.6 The contractor shall allow District representatives access to the contractor's employment records during regular business hours to verify compliance with the provisions of this Paragraph 8.28 Nondiscrimination and Affirmative Action when so requested by the District.
- 8.28.7 If the District finds that any provisions of this Paragraph 8.28 Nondiscrimination and Affirmative Action have been violated, such violation shall constitute a material breach of this Contract upon which the District may terminate or suspend this Contract. While the District reserves the right to determine independently that the anti-discrimination provisions of this Contract have been violated, in addition, a determination by the California Fair Employment and Housing Commission or the Federal Equal Employment Opportunity Commission that the contractor has violated Federal or State anti-discrimination laws or regulations shall constitute a finding by the District that the contractor has violated the anti-discrimination provisions of this Contract.
- 8.28.8 The parties agree that in the event the contractor violates any of the anti-discrimination provisions of this Contract, the District shall, at its sole option, be entitled to the sum of five hundred dollars (\$500) for each such violation pursuant to California Civil Code Section 1671 as liquidated damages in lieu of terminating or suspending this Contract.

8.29 Non Exclusivity

Nothing herein is intended nor shall be construed as creating any exclusive arrangement with the contractor. This Contract shall not restrict District from acquiring similar, equal or like goods and/or services from other entities or sources.

8.30 Notice of Delays

Except as otherwise provided under this Contract, when either party has knowledge that any actual or potential situation is delaying or threatens to delay the timely performance of this Contract, that party shall, within one (1) business day, give notice thereof, including all relevant information with respect thereto, to the other party.

8.31 Notice of Disputes

The contractor shall bring to the attention of the District's Project Manager and/or District's Project Director any dispute between the District and the contractor regarding the performance of services as stated in this Contract. If the District's Project Manager or District's Project Director is not able to resolve the dispute, the Fire Chief or his designee shall resolve it.

8.32 Notice to Employees Regarding the Federal Earned Income Credit

The contractor shall notify its employees, and shall require each subcontractor to notify its employees, that they may be eligible for the Federal Earned Income Credit under the federal income tax laws. Such notice shall be provided in accordance with the requirements set forth in Internal Revenue Service Notice No. 1015.

8.33 Notice to Employees Regarding the Safely Surrendered Baby Law

The contractor shall notify and provide to its employees, and shall require each subcontractor to notify and provide to its employees, information regarding the Safely Surrendered Baby Law, its implementation in Los Angeles County, and where and how to safely surrender a baby. The information is set forth in Exhibit I, Safely Surrendered Baby Law of this Contract. Additional information is available at <u>www.babysafela.org</u>.

8.34 Notices

All notices or demands required or permitted to be given or made under this Contract shall be in writing and shall be hand delivered with signed receipt or mailed by first-class registered or certified mail, postage prepaid, addressed to the parties as identified in Exhibits E -District's Administration and F - Contractor's Administration. Addresses may be changed by either party giving ten (10) days prior written notice thereof to the other party. The Fire Chief or his designee shall have the authority to issue all notices or demands required or permitted by the District under this Contract.

8.35 Prohibition Against Inducement or Persuasion

Notwithstanding the above, the contractor and the District agree that, during the term of this Contract and for a period of one year thereafter, neither party shall in any way intentionally induce or persuade any employee of one party to become an employee or agent of the other party. No bar exists against any hiring action initiated through a public announcement.

8.36 Public Records Act

- 8.36.1 Any documents submitted by the contractor; all information obtained in connection with the District's right to audit and inspect the contractor's documents, books, and accounting records pursuant to Paragraph 8.38 - Record Retention and Inspection-Audit Settlement of this Contract; as well as those documents which were required to be submitted in response to the Request for Proposals (RFP) used in the solicitation process for this Contract, become the exclusive property of the District. All such documents become a matter of public record and shall be regarded as public records. Exceptions will be those elements in the California Government Code Section 6250 et seq. (Public Records Act) and which are marked "trade secret," "confidential," or "proprietary." The District shall not in any way be liable or responsible for the disclosure of any such records including, without limitation, those so marked, if disclosure is required by law, or by an order issued by a court of competent jurisdiction.
- 8.36.2 In the event the District is required to defend an action on a Public Records Act request for any of the aforementioned documents, information, books, records, and/or contents of a proposal marked "trade secret," "confidential," or "proprietary," the contractor agrees to defend and indemnify the District from all costs and expenses, including reasonable attorney's fees, in action or liability arising under the Public Records Act.

8.37 Publicity

8.37.1 The contractor shall not disclose any details in connection with this Contract to any person or entity except as may be otherwise provided hereunder or required by law. However, in recognizing the contractor's need to identify its services

and related clients to sustain itself, the District shall not inhibit the contractor from publishing its role under this Contract within the following conditions:

- 8.37.1.1 The contractor shall develop all publicity material in a professional manner; and
- 8.37.1.2 During the term of this Contract, the contractor shall not, and shall not authorize another to, publish or disseminate any commercial advertisements, press releases, feature articles, or other materials using the name of the District without the prior written consent of the District's Project Director. The District shall not unreasonably withhold written consent.
- 8.37.2 The contractor may, without the prior written consent of District, indicate in its proposals and sales materials that it has been awarded this Contract with the County of Los Angeles, provided that the requirements of this Paragraph 8.37 Publicity shall apply.

8.38 Record Retention and Inspection-Audit Settlement

8.38.1 The Contractor shall maintain accurate and complete financial records of its activities and operations relating to this Contract in accordance with generally accepted accounting principles, and which meet the requirements for contract described in Auditor-Controller Contract accounting Accounting and Administration Handbook. The Contractor shall also maintain accurate and complete employment and other records relating to its performance of this Contract. The contractor agrees that the District, or its authorized representatives, shall have access to and the right to examine, audit, excerpt, copy, or transcribe any pertinent transaction, activity, or record relating to this Contract. All such material, including, but not limited to, all financial records, bank statements, cancelled checks or other proof of payment, timecards, sign-in/sign-out sheets and other time and employment records, and proprietary data and information, shall be kept and maintained by the contractor and shall be made available to the District during the term of this Contract and for a period of five (5) years thereafter unless the District's written permission is given to dispose of any such material prior to such time. All such material shall be maintained by the contractor at a location in Los Angeles County, provided that if any such material is located outside

Los Angeles County, then, at the District's option, the contractor shall pay the District for travel, per diem, and other costs incurred by the District to examine, audit, excerpt, copy, or transcribe such material at such other location.

- 8.38.2 In the event that an audit of the contractor is conducted specifically regarding this Contract by any Federal or State auditor, or by any auditor or accountant employed by the contractor or otherwise, then the contractor shall file a copy of such audit report with the County's Auditor-Controller within thirty (30) days of the contractor's receipt thereof, unless otherwise provided by applicable Federal or State law or under this Contract. Subject to applicable law, the District shall make a reasonable effort to maintain the confidentiality of such audit report(s). Failure on the part of the contractor to comply with any of the provisions of this subparagraph 8.38 shall constitute a material breach of this Contract.
- 8.38.3 If, at any time during the term of this Contract or within five (5) years after the expiration or termination of this Contract, representatives of the District conduct an audit of the contractor regarding the work performed under this Contract, and if such audit finds that the District's dollar liability for any such work is less than payments made by the District to the contractor, then the difference shall be either: a) repaid by the contractor to the District by cash payment upon demand or b) at the sole option of the County's Auditor-Controller, deducted from any amounts due to the contractor from the District, whether under this Contract or otherwise. If such audit finds that the District's dollar liability for such work is more than the payments made by the District to the contractor, then the difference shall be paid to the contractor by the District by cash payment, provided that in no event shall the District's maximum obligation for this Contract exceed the funds appropriated by the District for the purpose of this Contract.

8.39 Recycled Bond Paper

Consistent with the Board of Supervisors' policy to reduce the amount of solid waste deposited at the County landfills, the contractor agrees

to use recycled-content paper to the maximum extent possible on this Contract.

8.40 Subcontracting

- 8.40.1 The requirements of this Contract may not be subcontracted by the contractor **without the advance written approval of the District**. Any attempt by the contractor to subcontract without the prior consent of the District may be deemed a material breach of this Contract.
- 8.40.2 If the contractor desires to subcontract, the contractor shall provide the following information promptly at the District's request:
 - 8.40.2.1 A description of the work to be performed by the subcontractor;
 - 8.40.2.2 A draft copy of the proposed subcontract; and
 - 8.40.2.3 Other pertinent information and/or certifications requested by the District.
- 8.40.3 The contractor shall indemnify, defend, and hold the County harmless with respect to the activities of each and every subcontractor in the same manner and to the same degree as if such subcontractor(s) were the contractor employees.
- 8.40.4 The contractor shall remain fully responsible for all performances required of it under this Contract, including those that the contractor has determined to subcontract, notwithstanding the District's approval of the contractor's proposed subcontract.
- 8.40.5 The District's consent to subcontract shall not waive the District's right to prior and continuing approval of any and all personnel, including subcontractor employees, providing services under this Contract. The contractor is responsible to notify its subcontractors of this District right.
- 8.40.6 The District's Project Director is authorized to act for and on behalf of the District with respect to approval of any subcontract and subcontractor employees. After approval of the subcontract by the District, contractor shall forward a fully executed subcontract to the District for their files.

- 8.40.7 The contractor shall be solely liable and responsible for all payments or other compensation to all subcontractors and their officers, employees, agents, and successors in interest arising through services performed hereunder, notwithstanding the District's consent to subcontract.
- 8.40.8 The contractor shall obtain certificates of insurance, which establish that the subcontractor maintains all the programs of insurance required by the District from each approved subcontractor. Before any subcontractor employee may perform any work hereunder, contractor shall ensure delivery of all such documents to:

Consolidated Fire Protection District of Los Angeles County Materials Management Division / Contracts Section 5801 S. Eastern Avenue, Suite 100 Commerce, California 90040-4001

8.41 Termination for Breach of Warranty to Maintain Compliance with County's Child Support Compliance Program

Failure of the contractor to maintain compliance with the requirements set forth in Paragraph 8.14 - Contractor's Warranty of Adherence to County's Child Support Compliance Program shall constitute default under this Contract. Without limiting the rights and remedies available to the District under any other provision of this Contract, failure of the contractor to cure such default within ninety (90) calendar days of written notice shall be grounds upon which the District may terminate this Contract pursuant to Paragraph 8.43 - Termination for Default and pursue debarment of the contractor, pursuant to County Code Chapter 2.202.

8.42 Termination for Convenience

- 8.42.1 This Contract may be terminated, in whole or in part, from time to time, when such action is deemed by the District, in its sole discretion, to be in its best interest. Termination of work hereunder shall be effected by notice of termination to the contractor specifying the extent to which performance of work is terminated and the date upon which such termination becomes effective. The date upon which such termination becomes effective shall be no less than ten (10) days after the notice is sent.
- 8.42.2 After receipt of a notice of termination and except as otherwise directed by the District, the contractor shall:

- 8.42.2.1 Stop work under this Contract on the date and to the extent specified in such notice, and
- 8.42.2.2 Complete performance of such part of the work as shall not have been terminated by such notice.
- 8.42.3 All material including books, records, documents, or other evidence bearing on the costs and expenses of the contractor under this Contract shall be maintained by the contractor in accordance with Paragraph 8.38 Record Retention and Inspection-Audit Settlement.

8.43 Termination for Default

- 8.43.1 The District may, by written notice to the contractor, terminate the whole or any part of this Contract, if, in the judgment of District's Project Director:
 - 8.43.1.1 Contractor has materially breached this Contract; or
 - 8.43.1.2 Contractor fails to timely provide and/or satisfactorily perform any task, deliverable, service, or other work required either under this Contract; or
 - 8.43.1.3 Contractor fails to demonstrate a high probability of timely fulfillment of performance requirements under this Contract, or of any obligations of this Contract and in either case, fails to demonstrate convincing progress toward a cure within five (5) working days (or such longer period as the District may authorize in writing) after receipt of written notice from the District specifying such failure.
- 8.43.2 In the event that the District terminates this Contract in whole or in part as provided in Paragraph 8.43.1, the District may procure, upon such terms and in such manner as the District may deem appropriate, goods and services similar to those so terminated. The contractor shall be liable to the District for any and all excess costs incurred by the District, as determined by the District, for such similar goods and services. The contractor shall continue the performance of this Contract to the extent not terminated under the provisions of this paragraph.
- 8.43.3 Except with respect to defaults of any subcontractor, the contractor shall not be liable for any such excess costs of the type identified in Paragraph 8.43.2 if its failure to perform this

Contract arises out of causes beyond the control and without the fault or negligence of the contractor. Such causes may include, but are not limited to: acts of God or of the public enemy, acts of the District in either its sovereign or contractual capacity, acts of Federal or State governments in sovereign capacities, fires, floods, their epidemics, quarantine restrictions, strikes, freight embargoes, and unusually severe weather; but in every case, the failure to perform must be beyond the control and without the fault or negligence of the contractor. If the failure to perform is caused by the default of a subcontractor, and if such default arises out of causes beyond the control of both the contractor and subcontractor, and without the fault or negligence of either of them, the contractor shall not be liable for any such excess costs for failure to perform, unless the goods or services to be furnished by the subcontractor were obtainable from other sources in sufficient time to permit the contractor to meet the required performance schedule. As used in this paragraph, term "subcontractor(s)" the means subcontractor(s) at any tier.

- 8.43.4 If, after the District has given notice of termination under the provisions of Paragraph 8.43 Termination for Default it is determined by the District that the contractor was not in default under the provisions of Paragraph 8.43 Termination for Default or that the default was excusable under the provisions of subparagraph 8.43.3, the rights and obligations of the parties shall be the same as if the notice of termination had been issued pursuant to Paragraph 8.42 Termination for Convenience.
- 8.43.5 The rights and remedies of the District provided in this Paragraph 8.43 Termination for Default shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.44 Termination for Improper Consideration

8.44.1 The District may, by written notice to the contractor, immediately terminate the right of the contractor to proceed under this Contract if it is found that consideration, in any form, was offered or given by the contractor, either directly or through an intermediary, to any District officer, employee, or agent with the intent of securing this Contract or securing favorable treatment with respect to the award, amendment, or extension of this Contract or the making of any determinations with respect to the contractor's performance pursuant to this Contract. In the event of such termination, the District shall be entitled to pursue the same remedies against the contractor as it could pursue in the event of default by the contractor.

- 8.44.2 The contractor shall immediately report any attempt by a District officer or employee to solicit such improper consideration. The report shall be made either to the District manager charged with the supervision of the employee or to the County Auditor-Controller's Employee Fraud Hotline at (800) 544-6861.
- 8.44.3 Among other items, such improper consideration may take the form of cash, discounts, services, the provision of travel or entertainment, or tangible gifts.

8.45 Termination for Insolvency

- 8.45.1 The District may terminate this Contract forthwith in the event of the occurrence of any of the following:
 - 8.45.1.1 Insolvency of the contractor. The contractor shall be deemed to be insolvent if it has ceased to pay its debts for at least sixty (60) days in the ordinary course of business or cannot pay its debts as they become due, whether or not a petition has been filed under the Federal Bankruptcy Code and whether or not the contractor is insolvent within the meaning of the Federal Bankruptcy Code;
 - 8.45.1.2 The filing of a voluntary or involuntary petition regarding the contractor under the Federal Bankruptcy Code;
 - 8.45.1.3 The appointment of a Receiver or Trustee for the contractor; or
 - 8.45.1.4 The execution by the contractor of a general assignment for the benefit of creditors.
- 8.45.2 The rights and remedies of the District provided in this Paragraph 8.45 Termination for Insolvency shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.46 Termination for Non-Adherence of County Lobbyist Ordinance

The contractor, and each County Lobbyist or County Lobbying firm as

defined in County Code Section 2.160.010 retained by the contractor, shall fully comply with the County's Lobbyist Ordinance, County Code Chapter 2.160. Failure on the part of the contractor or any County Lobbyist or County Lobbying firm retained by the contractor to fully comply with the County's Lobbyist Ordinance shall constitute a material breach of this Contract, upon which the District may in its sole discretion, immediately terminate or suspend this Contract.

8.47 Termination for Non-Appropriation of Funds

Notwithstanding any other provision of this Contract, the District shall not be obligated for the contractor's performance hereunder or by any provision of this Contract during any of the District's future fiscal years unless and until the County's Board of Supervisors appropriates funds for this Contract in the District's Budget for each such future fiscal year. In the event that funds are not appropriated for this Contract, then this Contract shall terminate as of June 30 of the last fiscal year for which funds were appropriated. The District shall notify the contractor in writing of any such non-allocation of funds at the earliest possible date.

8.48 Validity

If any provision of this Contract or the application thereof to any person or circumstance is held invalid, the remainder of this Contract and the application of such provision to other persons or circumstances shall not be affected thereby.

8.49 Waiver

No waiver by the District of any breach of any provision of this Contract shall constitute a waiver of any other breach or of such provision. Failure of the District to enforce at any time, or from time to time, any provision of this Contract shall not be construed as a waiver thereof. The rights and remedies set forth in this paragraph 8.49 shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.50 Warranty Against Contingent Fees

8.50.1 The contractor warrants that no person or selling agency has been employed or retained to solicit or secure this Contract upon any Contract or understanding for a commission, percentage, brokerage, or contingent fee, excepting bona fide employees or bona fide established commercial or selling agencies maintained by the contractor for the purpose of securing business. 8.50.2 For breach of this warranty, the District shall have the right to terminate this Contract and, at its sole discretion, deduct from the Contract price or consideration, or otherwise recover, the full amount of such commission, percentage, brokerage, or contingent fee.

8.51 Warranty of Compliance with County's Defaulted Property Tax Reduction Program

Contractor acknowledges that County has established a goal of ensuring that all individuals and businesses that benefit financially from County through contract are current in paying their property tax obligations (secured and unsecured roll) in order to mitigate the economic burden otherwise imposed upon County and its taxpayers.

Unless contractor qualifies for an exemption or exclusion, contractor warrants and certifies that to the best of its knowledge it is now in compliance, and during the term of this contract will maintain compliance, with Los Angeles County Code Chapter 2.206.

8.52 Termination for Breach of Warranty to Maintain Compliance with County's Defaulted Property Tax Reduction Program

Failure of contractor to maintain compliance with the requirements set forth in Paragraph 8.51 - "Warranty of Compliance with County's Defaulted Property Tax Reduction Program" shall constitute default under this contract. Without limiting the rights and remedies available to County under any other provision of this contract, failure of contractor to cure such default within ten (10) days of notice shall be grounds upon which District may terminate this contract and/or pursue debarment of contractor, pursuant to County Code Chapter 2.206.

8.53 Time Off for Voting

The contractor shall notify its employees, and shall require each subcontractor to notify and provide to its employees, information regarding the time off for voting law (Elections Code Section 14000). Not less than ten (10) days before every statewide election, every contractor and subcontractors shall keep posted conspicuously at the place of work, if practicable, or elsewhere where it can be seen as employees come or go to their place of work, a notice setting forth the provisions of Section 14000.

8.54 Compliance with County's Zero Tolerance Policy on Human Trafficking

Contractor acknowledges that the County has established a Zero Tolerance Policy on Human Trafficking prohibiting contractors from engaging in human trafficking.

If a Contractor or member of Contractor's staff is convicted of a human trafficking offense, the District shall require that the Contractor or member of Contractor's staff be removed immediately from performing services under the Contract. District will not be under any obligation to disclose confidential information regarding the offenses other than those required by law.

Disqualification of any member of Contractor's staff pursuant to this paragraph shall not relieve Contractor of its obligation to complete all work in accordance with the terms and conditions of this Contract.

8.55 Intentionally Omitted

8.56 Compliance with Fair Chance Employment Practices

Contractor shall comply with fair chance employment hiring practices set forth in California Government Code Section 12952, Employment Discrimination: Conviction History. Contractor's violation of this paragraph of the Contract may constitute a material breach of the Contract. In the event of such material breach, District may, in its sole discretion, terminate the Contract.

8.57 Compliance with the County Policy of Equity

The contractor acknowledges that the County takes its commitment to preserving the dignity and professionalism of the workplace very seriously, as set forth in the County Policy of Equity (CPOE) (https://ceop.lacounty.gov/). The contractor further acknowledges that the County strives to provide a workplace free from discrimination, harassment, retaliation and inappropriate conduct based on a protected characteristic, and which may violate the CPOE. The contractor, its employees and subcontractors acknowledge and certify receipt and understanding of the CPOE. Failure of the contractor, its employees or its subcontractors to uphold the County's expectations of a workplace free from harassment and discrimination, including inappropriate conduct based on a protected characteristic, may subject the contractor to termination of contractual agreements as well as civil liability.

8.58 **Prohibition from Participation in Future Solicitation(s)**

A Proposer, or a Contractor or its subsidiary or Subcontractor ("Proposer/Contractor"), is prohibited from submitting a bid or proposal in a County solicitation if the Proposer/Contractor has provided advice or consultation for the solicitation. A Proposer/Contractor is also prohibited from submitting a bid or proposal in a County solicitation if the Proposer/Contractor has developed or prepared any of the solicitation materials on behalf of the County. A violation of this provision shall result in the disqualification or the termination or cancellation of any resultant County contract. This provision shall survive the expiration, or other termination of this Agreement.

9 UNIQUE TERMS AND CONDITIONS

- 9.1 Intentionally Omitted
- 9.2 Intentionally Omitted
- 9.3 Intentionally Omitted

9.4 Patent, Copyright and Trade Secret Indemnification

- 9.4.1 The Contractor shall indemnify, hold harmless and defend County from and against any and all liability, damages, costs, and expenses, including, but not limited to, defense costs and attorneys' fees, for or by reason of any actual or alleged infringement of any third party's patent or copyright, or any actual or alleged unauthorized trade secret disclosure, arising from or related to the operation and utilization of the Contractor's work under this Contract. District shall inform the Contractor as soon as practicable of any claim or action alleging such infringement or unauthorized disclosure, and shall support the Contractor's defense and settlement thereof.
- 9.4.2 In the event any equipment, part thereof, or software product becomes the subject of any complaint, claim, or proceeding alleging infringement or unauthorized disclosure, such that District's continued use of such item is formally restrained, enjoined, or subjected to a risk of damages, the Contractor, at its sole expense, and providing that District's continued use of the system is not materially impeded, shall either:

- Procure for District all rights to continued use of the questioned equipment, part, or software product; or
- Replace the questioned equipment, part, or software product with a non-questioned item; or
- Modify the questioned equipment, part, or software so that it is free of claims.
- 9.4.3 The Contractor shall have no liability if the alleged infringement or unauthorized disclosure is based upon a use of the questioned product, either alone or in combination with other items not supplied by the Contractor, in a manner for which the questioned product was not designed nor intended.

9.5 Intentionally Omitted

9.6 Data Destruction

Contractor(s) and Vendor(s) that have maintained, processed, or stored the County of Los Angeles' ("County") data and/or information, implied or expressed, have the sole responsibility to certify that the data and information have been appropriately destroyed consistent with the National Institute of Standards and Technology (NIST) Special Publication SP 800-88 titled *Guidelines for Media Sanitization*. Available at:

http://csrc.nist.gov/publications/PubsDrafts.html#SP-800-88 Rev.%201

The data and/or information may be stored on purchased, leased, or rented electronic storage equipment (e.g., printers, hard drives) and electronic devices (e.g., servers, workstations) that are geographically located within the County, or external to the County's boundaries. The District must receive within ten (10) business days, a signed document from Contractor(s) and Vendor(s) that certifies and validates the data and information were placed in one or more of the following stored states: unusable, unreadable, and indecipherable.

Vendor shall certify that any District data stored on purchased, leased, or rented electronic storage equipment and electronic devices, including, but not limited to printers, hard drives, servers, and/or workstations are destroyed consistent with the current National Institute of Standard and Technology (NIST) Special Publication SP-800-88, *Guidelines for Media Sanitization.* Vendor shall provide District with written certification, within ten (10) business

days of removal of any electronic storage equipment and devices that validates that any and all District data was destroyed and is unusable, unreadable, and/or undecipherable.

9.7 Intentionally Omitted

9.8 Intentionally Omitted

9.9 Intentionally Omitted

9.10 Mandatory Requirement to Register on County's WebVen

As a pre-condition to the award of this Contract, Contractor represents and warrants that it has registered in Los Angeles County's vendor registration system (hereafter "WebVen"). The WebVen contains Contractor's business profile and identifies the goods/services being provided by Contractor. Contractor shall ensure that it updates its vendor profile whenever changes occur to Contractor's operations by accessing the WebVen site located on-line at: <u>http://camisvr.co.la.ca.us/webven</u> County shall use the data obtained from Contractor's WebVen profile to ensure that Contractor's legal name, as reflected in its WebVen profile, shall be used in Contract documents).

9.11 Limitation on Corporate Acts

- 9.11.1 Contractor shall not amend its articles of incorporation or bylaws, move to dissolve or transfer any assets obtained using Contract Funds, or take any other steps which may materially affect the performance of this Contract without first notifying the District in writing no less than thirty (30) days prior to said a ction. Contractor shall notify the District's Contract Administrator immediately in writing of any change in Contractor's corporate name.
- 9.11.2 If, in the District's sole discretion, the steps taken by Contractor are determined to materially affect Contractor's performance of this Contract, the District may, at its sole discretion, take any (or all) of the following actions:
 - 9.11.2.1 Require Contractor to remedy the areas that affect Contractor's ability to perform its obligations under this Contract.

- 9.11.2.2 Suspend Contractor from performing (and receiving payment for) Contract tasks until a remedy has been reached.
- 9.11.2.3 Terminate this Contract pursuant to Paragraph 8.43 Termination for Default.

9.12 Modifications

This Contract fully expresses the agreement of the parties. Any modification to this Contract must be by means of a separate written document approved by the District. No oral conversation between any officer, employee or agent of the parties shall modify or otherwise amend this Contract in any way.

9.13 Remedies of Non-Compliance

Contractor agrees to comply with the requirements set forth in the entirety of this Contract as well as the requirements contained in any applicable directives, notices, guidelines and instructions used by the District. Contractor's failure to comply with such requirements shall subject Contractor to remedies which are available under this Contract and as provided by law. These remedies include but are not limited to the following: suspension of payment(s); suspension of Service(s); assessment and collection of liquidated damages; deobligation of Contract Funds (for purposes of this Contract, deobligation is the partial or full removal of Contract Funds from Contractor); debarment; and/or termination of Contract. The District shall have the sole discretion to determine which remedy(ies) will be applied as a result of Contractor's non-compliance.

9.14 Suspension

- 9.14.1 Contractor may be placed on suspension if District determines that Contractor is not in compliance with any Service, Work, task, deliverable or requirement outlined in this Contract and/or Contractor has demonstrated a consistent and significant lack of achievement of the Contract goals (including, but not limited to, meeting the requirements for work performance, the Pricing Sheet, staffing, administration, etc.). The District shall notify the Contractor in writing in the event that Contractor is placed on suspension.
- 9.14.2 Suspension as used herein shall mean a specified period of time (as determined by the District) during which the District shall withhold payment from Contractor. During the

suspension, Contractor has a continuing obligation to remedy the areas of non-compliance which have been identified by the District or its duly authorized representative(s). The District shall monitor Contractor's adherence to such remedy(ies) during the suspension period. When applicable, the District may also provide the Contractor with a written determination stating whether or not the Contractor may continue to provide non-suspended Services, if any, during the suspension period.

- 9.14.3 District's written notice of suspension shall set forth the conditions of Contractor's non-compliance as well as the period in which Contractor must correct noted deficiencies. In response to the notice of suspension, Contractor shall submit a written Corrective Action Plan to the District's Contract Manager within ten (10) days of the date indicated on the notice from the District. Contractor's Corrective Action Plan shall address all of the deficiencies noted by the District.
- 9.14.4 The District shall review Contractor's Corrective Action Plan, and will determine whether it meets the requirements for District's approval. The District reserves the right to suspend/deduct payments for or to terminate all or any part of this Contract (and/or any Contractor's other contracts with the District) when Contractor submits a Corrective Action Plan that is not acceptable to the District.
- 9.14.5 Contractor shall implement the Corrective Action Plan upon receiving District's final written approval of the Corrective Action Plan. Contractor's failure to comply with an approved Corrective Action Plan will be cause for material breach of Contract upon which the District may pursue the remedies for default of Contract.

9.15 Transition of Contract Services

9.15.1 **Completion of Contract**

9.15.1.1 Within sixty (60) calendar days prior to the expiration of this Contract (or shorter time period if notified in writing by District), Contractor shall allow the District or a newly selected contractor a transition period for orientation purposes and the orderly transition of Contractor's current Services without additional cost to the District. Contractor shall continue to provide

Services timely and accurately so that the Services are current at the expiration of this Contract.

9.15.2 Transition Plan

- 9.15.2.2 If this Contract (or any part thereof) is terminated pursuant to anv of the termination provisions outlined herein or if it expires pursuant to Paragraph 4.0 - Term of Contract, Contractor shall provide a Transition Plan to the District. Contractor shall submit said Transition Plan to the District's Contract Administrator within the timeframe designated by the District in the notice of termination or Contractor shall submit it at least sixty (60) days prior to the expiration of this Contract as noted in Paragraph 4.0 - Term of Contract.
- 9.15.2.3 The District shall review Contractor's Transition Plan and will determine whether it meets the requirements for District's approval. The District reserves the right to suspend/deduct payments under this Contract and/or under any of Contractor's other contracts with the District when Contractor submits a Transition Plan that is not acceptable to the District. Contractor shall adhere to the Transition Plan which, at a minimum, shall include all of the elements outlined below.

9.15.3 Elements of the Transition Plan

- 9.15.3.1 Contractor's method to communicate with other organizations that can assist in locating alternative Services.
- 9.15.3.2 Contractor's plan to assure for the provision of adequate staff to provide continued Services through the remaining term of this Contract.
- 9.15.3.3 Any additional information which may be necessary to ensure Contract Services are being provided.

9.15.4 Implementation of the Transition Plan

Contractor shall implement the Transition Plan that is approved by the District. Contractor's failure to provide and/or implement the Transition Plan as prescribed herein shall mean that the District will provide Contractor a Transition Plan and Contractor will implement the Transition Plan provided by the District. The District will monitor Contractor's progress in carrying out all elements of the Transition Plan. IN WITNESS WHEREOF, the Board of Supervisors of the County of Los Angeles has caused this Contract to be executed by the Fire Chief of the Consolidated Fire Protection District of Los Angeles County (or designee) and approved by County Counsel, and Contractor has caused this Contract to be executed in its behalf by its duly authorized officer, this 1st day of July, 2021.

CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY

Ву _____

Fire Chief

By <u>Alaco Ladder Company</u> Contractor

Signed:	
0	

APPROVED AS TO FORM:

RODRIGO A. CASTRO-SILVA County Counsel

By_

Senior Deputy County Counsel

SOLE SOURCE CHECKLIST

Department Name:

- □ New Sole Source Contract
- Sole Source Amendment to Existing Contract
 Date Existing Contract First Approved:

Check	JUSTIFICATION FOR SOLE SOURCE CONTRACTS
(✓)	Identify applicable justification and provide documentation for each checked item.
	Only one bona fide source (monopoly) for the service exists; performance and price competition are not available. A monopoly is an "Exclusive control of the supply of any service in a given market. If more than one source in a given market exists, a monopoly does not exist."
	Compliance with applicable statutory and/or regulatory provisions.
	Compliance with State and/or federal programmatic requirements.
	Services provided by other public or County-related entities.
	Services are needed to address an emergent or related time-sensitive need.
	The service provider(s) is required under the provisions of a grant or regulatory requirement.
	Additional services are needed to complete an ongoing task and it would be prohibitively costly in time and money to seek a new service provider.
	Services are needed during the time period required to complete a solicitation for replacement services; provided services are needed for no more than 12 months from the expiration of an existing contract which has no available option periods.
	Maintenance and support services are needed for an existing solution/system during the time to complete a solicitation for a new replacement solution/ system; provided the services are needed for no more than 24 months from the expiration of an existing maintenance and support contract which has no available option periods.
	Maintenance service agreements exist on equipment which must be serviced by the original equipment manufacturer or an authorized service representative.
	It is more cost-effective to obtain services by exercising an option under an existing contract.
	It is in the best economic interest of the County (e.g., significant costs to replace an existing system or infrastructure, administrative cost savings and excessive learning curve for a new service provider, etc.) In such cases, departments must demonstrate due diligence in qualifying the cost-savings or cost-avoidance associated with the best economic interest of the County.

Sole Source Checklist – Alaco Ladders

Justification 1 – Please refer to the attached sole source letter from the vendor.

Justification 10 - Alaco Ladder Company (Alaco) is the only company that produces NFPA compliant wooden firefighting ladders for the Fire Service. Alaco is also the only manufacturer and repair facility for wooden firefighting ladders remaining in the world. Firefighting ladders are essential tools and must be maintained and repaired to ensure firefighter safety during firefighting operations.

Justification 12 – Wooden firefighting ladders have been a tradition in Los Angeles County Fire Department (Department) since the Department's inception in 1923. The Department has and continues to equip its firefighting apparatus with Alaco wooden ground ladders. Currently, Engine Companies, Truck Companies, and Quints carry over 1,000 Alaco ladders in varying lengths. The value of these ladders is upwards of \$4,500,000.00. The use of the wooden ladders provide superior safety for the firefighters.

The Department's apparatus are outfitted and designed for the Alaco wooden ground ladders. The wooden ladders are electrically safe and essential to ensure firefighter safety and to ensure the Department's ability to maintain a state of readiness for emergency responses.

June 29, 2021

The Honorable Board of Supervisors County of Los Angeles 383 Kenneth Hahn Hall of Administration 500 West Temple Street Los Angeles, California 90012

Dear Supervisors:

APPROVAL OF TRANSIT LAW ENFORCEMENT SERVICES AGREEMENT BY AND BETWEEN COUNTY OF LOS ANGELES AND FOOTHILL TRANSIT AUTHORITY (ALL DISTRICTS) (3 VOTES)

SUBJECT

The Los Angeles County (County) Sheriff's Department seeks Board approval of a Transit Law Enforcement Services Agreement (Agreement) with the Foothill Transit Authority (Foothill), effective July 1, 2021, through June 30, 2026.

IT IS RECOMMENDED THAT THE BOARD:

- 1. Delegate authority to the Sheriff or his designee, as an agent for the County, to execute Amendments to the Agreement, substantially similar to the attached Agreement, effective July 1, 2021, or upon execution by the Sheriff, whichever is later, though June 30, 2026.
- 2. Delegate authority to the Sheriff, or his designee, to execute any and all Amendments to the Agreement, ensuring any negative fiscal impact to the County is avoided.
- 3. Delegate authority to the Sheriff, or his designee, as an agent for the County, to terminate an Agreement if it is in the best interest of the County.

The Honorable Board of Supervisors June 29, 2021 Page 2

PURPOSE / JUSTIFICATION OF RECOMMENDED ACTION

The purpose of the Agreement is to provide Foothill with as-needed Services within Foothill's service area, effective July 1, 2021 to June 30, 2026. The current agreement expires June 30, 2021. The approval of this new Agreement will permit the Department to continue to provide Services to Foothill Transit Authority.

Implementation of Strategic Plan Goals

As part of your Board's commitment to the County, approval of the recommended action would enhance Los Angeles County's Strategic Plan, Goal 1, Operational Effectiveness/Fiscal Sustainability; maximize the effectiveness of process, structure, and operations to support timely delivery of customer-oriented and efficient public service, Goal 2, Community Support and Responsiveness; enrich lives of Los Angeles County residents by providing enhanced services, and effectively planning and responding to economic, social, and environmental challenges, and Goal 3, Integrated Services Delivery; maximize opportunities to measurably improve the client and community outcomes, and leverage resources through the continuous integration of public safety services.

FISCAL IMPACT/FINANCING

None. Foothill shall pay the Sheriff's Department for Services in accordance with appropriate and prevailing billing rates as determined by the Auditor-Controller each fiscal year. Foothill is aware that the rates are revised at the beginning of every fiscal year as determined by the Auditor-Controller. The hourly overtime rates are developed by the Auditor-Controller, pursuant to the requirements of California Government Code Section 51350 and the policies and procedures adopted by the Board.

FACTS AND PROVISIONS/LEGAL REQUIREMENTS

The Agreement is authorized by Section 56 ½ and/or Section 56 ¾ of the County Charter and/or California Government Code Section 51301 and/or California Corporations Code Section 10405. The Agreement allows for the provision of Transit Law Enforcement Services by the Department from July 1, 2021 to June 30, 2026. The Agreement may be terminated by either party within 60 calendar days advance written notice. The Agreement provides for the mutual indemnification of parties.

The attached Agreement has been approved as to form by County Counsel.

The Honorable Board of Supervisors June 29, 2021 Page 3

IMPACT ON CURRENT SERVICES (OR PROJECTS)

Approving the Agreement will continue to enhance the quality of law enforcement services and public safety in the incorporated cities and the unincorporated areas of Los Angeles County served by Foothill's transit system. There are no anticipated negative impacts upon the unincorporated patrol areas of the County.

CONCLUSION

Upon Board approval, it is requested the Clerk of the Board return one original adopted Board letter to the Sheriff's Department's Contract Law Enforcement Bureau.

Sincerely,

ALEX VILLANUEVA, SHERIFF

TIMOTHY K. MURAKAMI UNDERSHERIFF

TRANSIT LAW ENFORCEMENT SERVICES AGREEMENT BY AND BETWEEN COUNTY OF LOS ANGELES AND

FOOTHILL TRANSIT AUTHORITY

TABLE OF CONTENTS

PARAGRAPH	TITLE	PAGE
RECITALS		2
1.0	SCOPE OF SERVICES	2
2.0	ADMINISTRATION OF PERSONNEL	3
3.0	INDEMNIFICATION	4
4.0	TERM OF AGREEMENT	6
5.0	RIGHT OF TERMINATION	6
6.0	BILLING RATES	7
7.0	PAYMENT PROCEDURES	7
8.0	AMENDMENTS	8
9.0	ASSIGNMENT, DELEGATION, AND SUBCONTRACTING	8
10.0	AUTHORIZATION WARRANTY	8
11.0	GOVERNING LAW, JURISDICTION, AND VENUE	8
12.0	NOTICES	9
13.0	VALIDITY	9
14.0	WAIVER	9
15.0	ENTIRE AGREEMENT	10
SIGNATURES		11
EXHIBIT A – I	PUBLIC ENTITY CONTRACT – HOURLY OVERTIME RATES	

TRANSIT LAW ENFORCEMENT SERVICES AGREEMENT BY AND BETWEEN COUNTY OF LOS ANGELES AND FOOTHILL TRANSIT AUTHORUTY

This Transit Law Enforcement Services Agreement ("Agreement") is entered into this ______ day of ______, ____ by and between the County of Los Angeles ("County") and Foothill Transit Authority ("Foothill").

RECITALS

- (a) Whereas, Foothill is desirous of contracting with the County for the performance of the transit law enforcement functions described herein by the Los Angeles County Sheriff's Department ("Sheriff's Department"); and
- (b) Whereas, the County is agreeable to rendering such services on the terms and conditions set forth in this Agreement; and
- (c) Whereas, this Agreement is authorized and provided for by Section 56-3/4 of the Charter of the County of Los Angeles; and
- (d) Whereas, the County is agreeable to rendering such transit law enforcement services, as available, on the terms and conditions set forth in this Agreement.

NOW THEREFORE, in consideration of the mutual covenants contained herein, and for good and valuable consideration, the parties mutually agree as follows:

1.0 SCOPE OF SERVICES

1.1 The County agrees, as available, through the Sheriff's Department, to provide transit law enforcement services to Foothill during the term of this Agreement to the extent and in the manner set forth herein. The classification and approximate numbers of personnel provided by the County will be determined and mutually agreed upon in writing between the Sheriff's Department and Foothill. The hours of duty performed by County employees will be established and agreed upon in writing in accordance with the requested services.

- 1.2. The request for services shall be a written request on official Foothill letterhead. For the purpose of performing said services, County shall furnish and supply, as available, all necessary labor, supervision, personnel, equipment, communications, fuel, and supplies necessary to provide the services to be rendered hereunder. Notwithstanding the foregoing, Foothill may provide additional resources for the County to utilize in performance of the services. The request shall be signed by a representative of Foothill who is duly authorized to enter into such agreements for law enforcement services. The request shall be submitted to the Los Angeles County Sheriff's Department's Contract Law Enforcement Bureau located at 211 W Temple Street, 7th Floor, Los Angeles, California 90012-3205.
- 1.3 The Sheriff intends to exercise jurisdiction and provide law enforcement services, occurring on-board buses, related to the special policing concerns of Foothill. These duties may include but are not limited to, enforcement of transit fares, conducting investigations of vandalism, routine transit patrol, and general law enforcement activities related to public transportation within the limit of the County and neighboring counties related to Foothill's bus lines.
- 1.4 Except as otherwise specifically set forth in this Agreement, transit law enforcement services shall encompass duties and functions of the type coming within the jurisdiction of and customarily rendered by the Sheriff under the Charter of the County and the statutes of the State of California.

2.0 ADMINISTRATION OF PERSONNEL

- 2.1 In the event of a dispute between the parties to this Agreement as to the extent of the duties and functions to be rendered hereunder, or the minimum level or manner of performance of such service, Foothill shall be consulted and a mutual determination thereof shall be made by both the Sheriff of the County and Foothill's Chief Executive Officer.
- 2.2 The rendition of the services performed by the Sheriff's Department, the discipline of officers, and other matters incident to the performance of such services and the control of personnel so employed shall remain with the County.

- 2.3 With regard to Paragraphs 2.1 and 2.2 above, the Sheriff's Department, in an unresolved dispute over the minimum level of performance of services, shall have final and conclusive determination as between the parties hereto.
- 2.4 All Foothill employees who work in conjunction with the Sheriff's Department pursuant to this Agreement shall remain employees of Foothill and shall not have any claim or right to employment, civil service protection, salary, or benefits or claims of any kind from the County based on this Agreement. No Foothill employees shall become employees of the County.
- 2.5 Foothill shall not be called upon to assume any liability for the direct payment of any Sheriff's Department salaries, wages, or other compensation to any County personnel performing services hereunder for said Foothill. Except as herein otherwise specified, Foothill shall not be liable for compensation or indemnity to any County employee or agent of the County for injury or sickness arising out of his/her employment as a contract employee of Foothill.
- 2.6 As part of its compliance with all applicable laws and regulations relating to employee hiring, the County agrees that the County Civil Service Rules to which it is subject and which prohibit discrimination on the basis of non-merit factors, shall for purposes of this Agreement be read and understood to prohibit discrimination on the basis of sexual orientation.

3.0 INDEMNIFICATION

3.1 Subject to the limitations stated in this Section 3.0, Indemnification, or elsewhere, the County shall indemnify, defend and hold harmless Foothill, and its officers, directors, employees and agents (collectively, "Indemnified Parties") from and against any and all liability, expense (including but not limited to defense costs and attorney's fees), claims, causes of action, and lawsuits for damages, including, but not limited to, bodily injury, death, personal injury or property damage (including property of the County) arising from or connected with any negligent, intentional or reckless act or omission of the County, its officers, directors,

employees, agents, subcontractors, or suppliers while providing services under this Agreement.

- 3.2 Notwithstanding anything contained herein or stated elsewhere, the County shall have no obligation or liability, including any obligation to indemnify or defend any Indemnified Party (a) for a failure to prevent any crime or tortious act, (b) for any injury, loss or damage caused directly or indirectly by a criminal or tortious act of anyone other than the County, its officers, directors, employees, agents, sub-contractors or suppliers while providing services under this Agreement, or (c) for any injury, loss or damage caused by any means whatsoever except as the result of a failure by the County its officers, directors, employees, agents, subcontractors, or suppliers to perform the services under this Agreement.
- 3.3 Foothill understands and agrees that the law enforcement services provided hereunder are not intended or expected to accomplish patrolling or law enforcement at any particular location, more than a few times a day or less, or to prevent crime or wrongdoing from occurring at any particular place or time. The County shall have no obligation to patrol or provide law enforcement at any specific location at any particular time(s) except under a written schedule provided in advance by Foothill and agreed to by the County.
- 3.4 Notwithstanding anything contained herein, the County's obligations hereunder to Foothill or any Indemnified Party shall be limited by any immunity of freedom from suit or liability provided by law, including but not limited to those stated in California Government Code sections 818.2 and 845, as if such immunity or legal provision were incorporated in full in this Agreement and made applicable to Foothill and all Indemnified Parties.
- 3.5 Any obligation by the County or Foothill to provide defense or indemnity hereunder shall not arise until it has been finally determined by competent judicial authority that such indemnity is owed under the provisions of this Section 3.0.
- 3.6 Foothill shall indemnify, defend and hold harmless the County, and its officers, directors, employees and agents from and against any and all liability, expense (including, but not limited to defense costs and attorneys' fees), claims, causes of

action, and lawsuits for damages of any nature whatsoever, including but not limited to bodily injury, death, personal injury or property damage (including property of Foothill) arising from or connected with any alleged act and/or omission of Foothill, its officers, directors, employees, agents, sub-contractors, or suppliers.

- 3.7 It is the intent of the parties to this Agreement that nothing herein shall impose, nor shall be interpreted to impose, on the County any liability for injuries or death to any County employee greater than the liability imposed pursuant to the provisions of the worker's compensation laws.
- 3.8 This Section 3.0, Indemnification, shall survive termination of this Agreement and/or final payment thereunder.

4.0 TERM OF AGREEMENT

The term of this Agreement shall commence July 1, 2021, or upon execution by the Sheriff, whichever is later, and shall terminate on June 30, 2026, unless sooner terminated or extended in whole or in part as provided for herein.

5.0 RIGHT OF TERMINATION

- 5.1 Either party may terminate this Agreement with or without cause by giving not less than sixty (60) calendar days advance written notice to the other party.
- 5.2 Notwithstanding the foregoing, the Sheriff's Department may cancel the provision of services with only ten (10) calendar days advance notice, or less in the event of exigent circumstances, if the Sheriff's Department concludes that there are insufficient personnel to provide the agreed upon services and still perform other Sheriff's Department duties as required by law.
- 5.3 In the event of a termination, each party shall fully discharge all obligations owed to the other party accruing prior to the date of such termination, and, except as otherwise provided herein, each party shall be released from all obligations, which would otherwise accrue subsequent to the date of termination.

6.0 BILLING RATES

- 6.1 For and in consideration of the rendition of the transit law enforcement services to be performed by the County for the Public Entity under this Agreement, the Public Entity shall pay County for said services provided by County under the terms of this Agreement at the appropriate and prevailing billing rates set forth on Exhibit A, Public Entity Contract Hourly Overtime Rates, as established by the County Auditor-Controller.
- 6.2 The billing rates set forth on Attachment A, Public Entity Contract Hourly Overtime Rates, of this Agreement shall be readjusted by the County Auditor-Controller annually effective July 1 of each year, published by the County, and attached hereto as an Amendment to this Agreement, to reflect the cost of such service in accordance with the policies and procedures for the determination of such rates as adopted by the County Board of Supervisors.
- 6.3 The billing rates published, Public Entity Contract Hourly Overtime Rates, are developed and deemed appropriate by the County Auditor-Controller to reflect the County's actual costs in the compensation of employees at premium overtime rates (except Deputy Sheriff Reserves), the administration of workers' compensation benefits, and the overhead of the County attributable to the provision of services pursuant to this Agreement. The foregoing rates are adjusted annually pursuant to the policies and practices adopted by the Los Angeles County Board of Supervisors governing the determination of such actual costs.

7.0 PAYMENT PROCEDURES

- 7.1 The County, through the Sheriff's Department, shall render to Foothill a summarized invoice which covers all services performed during said month, and Foothill shall pay County for all undisputed amounts within sixty (60) calendar days after date of said invoice.
- 7.2 If such payment is not delivered to the County office, which is described on said invoice, within sixty (60) calendar days after the date of the invoice, the County is entitled to recover interest thereon. For all disputed amounts, Foothill shall provide

the County with written notice of the dispute including the invoice date, amount, and reasons for dispute within ten (10) calendar days after receipt of the invoice. The parties shall memorialize the resolution of the dispute in writing. For any disputed amounts, interest shall accrue if payment is not received within sixty (60) calendar days after the dispute resolution is memorialized.

- 7.3 Said interest shall be at a rate of ten percent (10%) per annum or any portion thereof, calculated from the date payment was due pursuant to Sections 7.1 and 7.2 above.
- 7.4 Notwithstanding the provisions of California Government Code section 907, if such payment is not delivered to the County office which is described on said invoice within sixty (60) days of the invoice, or in the case of disputed amounts, from the date the resolution is memorialized, the County may satisfy such indebtedness, including interest thereon, from any funds of Foothill on deposit with the County without giving further notice to Foothill of County's intention to do so.

8.0 AMENDMENTS

With the exception of Amendments made pursuant to Paragraph 6.0 above which do not require the signature of either party, all changes, modifications, or amendments to this Agreement must be in the form of a written Amendment duly executed by authorized personnel of the County and Foothill.

9.0 ASSIGNMENT, DELEGATION, AND SUBCONTRACTING

A party shall not assign its rights and/or subcontract, or otherwise delegate, its duties under this Agreement, either in whole or in part, without the prior written consent of the other party, and any attempted assignment or delegation without such consent shall be null and void.

10.0 AUTHORIZATION WARRANTY

Foothill represents and warrants that the person executing this Agreement for Foothill is an authorized agent who has actual Foothill to bind Foothill to each and every term, condition, and obligation of this Agreement and that all requirements of Foothill have been fulfilled to provide such actual Foothill.

11.0 GOVERNING LAW, JURISDICTION, AND VENUE

This Agreement shall be governed by, and construed in accordance with, the laws of the State of California. The parties agree and consent to the exclusive jurisdiction of the courts of the State of California for all purposes regarding this Agreement and further agree and consent that venue of any action brought hereunder shall be exclusively in the County of Los Angeles.

12.0 NOTICES

- 12.1 Unless otherwise specified herein, all notices or demands required or permitted to be given or made under this Agreement shall be in writing and shall be hand delivered with signed receipt or mailed by first class registered or certified mail, postage prepaid, addressed to the parties at the following addresses and to the attention of the person named. Addresses and persons to be notified may be changed by either party by giving ten (10) calendar days prior written notice thereof to the other party.
- 12.2 Notices to the County shall be addressed as follows:

Los Angeles County Sheriff's Department Contract Law Enforcement Bureau Attn: Foothill Transit, Contract Manager 211 W. Temple Street, 7th Floor Los Angeles, California 90012-3205

12.3 Notices to Foothill shall be addressed as follows:

Foothill Transit Attn: Chief Executive Officer or Designee 100 South Vincent Avenue, Suite 200 West Covina, California 91790

13.0 VALIDITY

If any provision of this Agreement or the application thereof to any person or circumstance is held invalid, the remainder of this Agreement and the application of such provision to other persons or circumstances shall not be affected thereby.

14.0 WAIVER

No waiver by the parties of any breach of any provision of this Agreement shall constitute a waiver of any other breach or of such provision. Failure of the parties to enforce at any time, or from time to time, any provision of this Agreement shall not be construed as a waiver thereof.

15.0 ENTIRE AGREEMENT

This Agreement, including Exhibit A, Public Entity Contract – Hourly Overtime Rates, and any executed Amendments hereto or thereto, constitute the complete and exclusive statement of understanding of the parties which supersedes all previous agreements, written or oral, and all communications between the parties relating to the subject matter of this Agreement.

[Signatures continued on page 11.]

TRANSIT LAW ENFORCEMENT SERVICES AGREEMENT **BY AND BETWEEN COUNTY OF LOS ANGELES** AND FOOTHILL TRANSIT

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed on their behalf by their duly authorized representatives on the dates indicated below.

COUNTY OF LOS ANGELES

By ______ Alex Villanueva, SHERIFF

Date

FOOTHILL TRANSIT

By _____ Chief Executive Officer or Designee

Date

APPROVED AS TO FORM: Darold D. Pieper, Attorney at Law

By

Foothill Transit General Counsel

APPROVED AS TO FORM:	
RODRIGO A. CASTRO-SILVA	
County Counsel	
ByMicheleric	/
Principal Deputy County Counsel	

11



COUNTY OF LOS ANGELES FIRE DEPARTMENT

1320 NORTH EASTERN AVENUE LOS ANGELES, CALIFORNIA 90063-3294 (323) 881-2426 www.fire.lacounty.gov

"Proud Protectors of Life, Property, and the Environment"

BOARD OF SUPERVISORS

HILDA L. SOLIS FIRST DISTRICT

HOLLY J. MITCHELL SECOND DISTRICT

> SHEILA KUEHL THIRD DISTRICT

JANICE HAHN FOURTH DISTRICT

KATHRYN BARGER FIFTH DISTRICT

DARYL L. OSBY FIRE CHIEF FORESTER & FIRE WARDEN

June 8, 2021

The Honorable Board of Supervisors County of Los Angeles 383 Kenneth Hahn Hall of Administration 500 West Temple Street Los Angeles, CA 90012

Dear Supervisors:

ADOPT RESOLUTION TO AUTHORIZE PARTICIPATION IN THE CALIFORNIA BOATING SAFETY AND ENFORCEMENT FINANCIAL AID PROGRAM FOR FISCAL YEAR 2021-22 (ALL DISTRICTS) (3 VOTES)

SUBJECT

Adopt a resolution to secure State funding through the California Department of Parks and Recreation, Division of Boating and Waterways (DBW), in support of boating safety and enforcement on waters within Los Angeles County (County). Also, authorize participation in the California Boating Safety and Enforcement Financial Aid Program (Program) by executing the Program agreement to reimburse the County in an amount not to exceed \$2.12 million.

IT IS RECOMMENDED THAT YOUR BOARD, ACTING AS THE GOVERNING BODY OF THE CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY AND THE BOARD OF SUPERVISORS:

- 1. Adopt the Resolution (Attachment I) whereby your Board:
 - Accepts the Legislature's commitment to provide future annual baseline funding for • the County and gives priority consideration to any application submitted by the County to secure State funding in support of boating safety and enforcement on waters within the County.

AGOURA HILLS ARTESIA AZUSA BALDWIN PARK BELL BELL GARDENS BELLFLOWER BRADBURY CALABASAS

CARSON CERRITOS CLAREMONT COMMERCE COVINA CUDAHY DIAMOND BAR DUARTE

FI MONTE GARDENA GLENDORA HAWAIIAN GARDENS HAWTHORNE HERMOSA BEACH HIDDEN HILLS HUNTINGTON PARK INDUSTRY

SERVING THE UNINCORPORATED AREAS OF LOS ANGELES COUNTY AND THE CITIES OF: INGLEWOOD IRWINDALE I A HABRA LA MIRADA LA PUENTE LAKEWOOD LANCASTER

LA CANADA-FLINTRIDGE

LAWNDALE LOMITA LYNWOOD MALIBU MAYWOOD NORWALK PALMDALE PALOS VERDES ESTATES PARAMOUNT

PICO RIVERA POMONA RANCHO PALOS VERDES ROLLING HILLS ROLLING HILLS ESTATES ROSEMEAD SAN DIMAS SANTA CLARITA

SIGNAL HILL SOUTH EL MONTE SOUTH GATE TEMPLE CITY VERNON WALNUT WEST HOLLYWOOD WESTLAKE VILLAGE WHITTIER

The Honorable Board of Supervisors June 8, 2021 Page 2

- Reaffirms the continued participation of the County, through the Consolidated Fire Protection District of Los Angeles County (District) and the Los Angeles County Sheriff's Department, in the Program for Fiscal Year (FY) 2021-22.
- Certifies that the County, as a participant in the Program, shall expend the equivalent of 100 percent of its revenues collected from personal property tax on vessels of boat owners within the County for boating safety and enforcement.
- Authorizes the Fire Chief and the Sheriff, or their designees, as County agents to sign and submit an application and related expenditure reimbursement claims to the DBW for State funding.
- Authorizes the State funding received through the Program to be distributed solely to the Sheriff's Department.
- 2. Authorize the Chair of the Board of Supervisors to execute the Boating Safety and Enforcement Financial Aid Program Agreement (Attachment II), which is required as part of the application package. In executing the agreement, the County agrees to submit requests for reimbursement within sixty days of the end of the fiscal quarter; and, if such requests are submitted after the sixty days has expired, the State has the option to reduce the allocation by five percent.
- 3. Find that the resolution and the funding of the County programs are exempt from the California Environmental Quality Act (CEQA) pursuant to Section 15061 (b)(3) of the CEQA Guidelines.

PURPOSE/JUSTIFICATION OF RECOMMENDED ACTION

The DBW provides financial aid from the Harbors and Watercraft Revolving Fund to counties under the Program, as authorized by Section 663.7 of the Harbors and Navigation Code. The County has been part of this Program since 1995. Since FY 2006-07, the District has received up to \$2.12 million annually.

These monies are part of an agreement with the State to provide funding to the County as part of the transfer of ownership of eight State beaches located within the County. A key part of this agreement was the State's commitment to provide long-term funding assistance for the District (Attachment III).

The financial aid application was submitted to the DBW on December 22, 2020. Agencies already participating in the Program are required to submit the completed application six months prior to the start of the new fiscal year to provide the DBW sufficient time to prepare the annual agreement for each agency. The DBW is requesting that the attached agreement for FY 2021-22 be executed and submitted no later than June 15, 2021. In addition, the Harbors and Navigation Code and application criteria require that your Board, by resolution, authorize the County's participation in the Program, and certify that the County will expend no less than 100 percent of the amount collected from personal property taxes on vessels for

The Honorable Board of Supervisors June 8, 2021 Page 3

boating safety programs during the funding year. The certification is required because counties qualify only if they expend an amount equivalent to at least 100 percent of these personal property taxes on boating safety and enforcement.

The designation of the Fire Chief and the Sheriff, or their designees, as signatories to the multi-department application, is consistent with your Board's instructions of December 8, 1994, to pursue long-term funding from the State. This State funding will offset portions of the offshore marine rescue and enforcement activities not presently met by the personal property tax on vessels. In previous years, the funds received from the financial aid have allowed the budgeted funds for these purposes to support the operations of the eight State beaches. This State funding will be received solely by the Sheriff's Department.

Implementation of Strategic Plan Goals

Approval of the recommended actions is consistent with the County's Strategic Goal No. 3, Realize Tomorrow's Government Today, Strategy III.3 - Pursue Operational Effectiveness, Fiscal Responsibility, and Accountability to continually asses our efficiency and effectiveness, maximize and leverage resources, and hold ourselves accountable to maximize revenue.

FISCAL IMPACT/FINANCING

No matching funds are required. Counties must first expend the taxes collected from its resident boat owners before being entitled to any supplemental State funding from this Program. Only the cost of the Program, which exceeds the total fees and vessel taxes collected for the year, will be supplemented up to the maximum amount obligated by the State. The State's maximum funding for FY 2021-22 is \$2.12 million.

The County's FY 2021-22 application includes a request for \$15.900 million, net of fees and taxes, to maintain the County's boating safety and enforcement activities (Attachment IV). This request more than meets the State's funding requirement.

There is no impact on net County cost.

FACTS AND PROVISIONS/LEGAL REQUIREMENTS

The DBW provides financial aid from the Harbors and Watercraft Revolving Fund to counties under the Program, as authorized by Section 663.7 of the Harbors and Navigation Code. It allows the County to receive funding so long as the County conducts boating safety and enforcement activities. The DBW requires one adopted resolution by the Board of Supervisors per fiscal year to designate one or more agencies to participate in the Program. The County has been part of this Program since 1995. The DBW is requesting that the attached agreement for FY 2021-22 be executed and submitted. The application for FY 2021-22 was timely submitted on December 22, 2020.

The Honorable Board of Supervisors June 8, 2021 Page 4

ENVIRONMENTAL DOCUMENTATION

This Resolution and the funding of the County programs are exempt from the CEQA pursuant to Section 15061 (b)(3) of the CEQA Guidelines.

IMPACT ON CURRENT SERVICES (OR PROJECTS)

The services shall continue upon execution of this agreement.

CONCLUSION

Participation in the DBW Program will continue existing funding to the County's Boating Safety and Enforcement programs for FY 2021-22 and continue financial aid to the Sheriff's Department boating safety and law enforcement efforts in equal portions. Your Board's adoption of the Resolution and execution of the financial assistance certification authorizes participation in the Program.

Upon approval by your Board, please instruct the Executive Officer to return four copies of the adopted Board letter and Attachments I and II to the following office:

Consolidated Fire Protection District of Los Angeles County Executive Office – Business Operations Attention: Zuleyda Santana, Administrative Services Manager II 1320 N. Eastern Avenue Los Angeles, CA 90063

The District's contact may be reached at (323) 881-6173.

Respectfully submitted,

DARYL L. OSBY, FIRE CHIEF

ALEX VILLANUEVA, SHERIFF

DLO:sh

Enclosures

c: Chief Executive Officer County Counsel Auditor-Controller Sheriff's Department

Attachment I

RESOLUTION

WHEREAS, the California Department of Parks and Recreation, Division of Boating and Waterways, pursuant to Section 663.7 of the Harbors and Navigation Code, provides supplemental State funding under the State Boating Safety and Enforcement Financial Aid Program to qualifying counties for boating safety and enforcement programs on waters under their jurisdiction; and

WHEREAS, the County of Los Angeles is charged with providing vital boating safety and enforcement services to a population in excess of nine (9) million people and over 60,000 registered boaters; and

WHEREAS, the County of Los Angeles, has received for boating safety and enforcement prior allocations from the California Department of Parks and Recreation, Division of Boating and Waterways, in Fiscal Years (FYs) 1995-96 through 2020-21, which has established an annual baseline funding for future participation in the program in accordance with Subdivision (g) of Section 5002.6 of the Public Resource Code and as addressed by Assembly Bill 122 (Rainey), Chapter 971, Statutes of 1996, Section 2 under the Harbors and Navigation Code Sections 85.2 and 663.7(a) and (c)(1).

1

WHEREAS, the current levels of those boating safety and enforcement services will continue through the Consolidated Fire Protection District of Los Angeles County (District) and the Los Angeles County Sheriff's Department; and

WHEREAS, allocation of said funding to any county or a public agency therein is contingent upon the County's governing body, the Board of Supervisors, adopting a resolution authorizing participation in the California Boating Safety and Enforcement Financial Aid Program and certifying that, during the funding year, an amount at least equal to the total amount collected by the County from personal property taxes on vessels will be expended on specified boating safety programs; and

WHEREAS, the Board of Supervisors of the County of Los Angeles through the District, and the Los Angeles County Sheriff's Department, wishes to participate in the California Boating Safety and Enforcement Financial Aid Program administered by the California Department of Parks and Recreation, Division of Boating and Waterways and will distribute the funds from the Boating Safety and Enforcement Financial Aid Program solely to the Los Angeles County Sheriff's Department.

NOW, THEREFORE BE IT RESOLVED by the Board of Supervisors of the County of Los Angeles that the County of Los Angeles, through the District and the Los Angeles County Sheriff's Department, is hereby authorized to participate in and apply for the California Boating Safety and Enforcement Financial Aid Program for FY 2021-22.

2

BE IT FURTHER RESOLVED that the Board of Supervisors of the County of Los Angeles hereby certifies that during FY 2021-22, the funding year, an amount equal to 100 percent of the amount received by the County in personal property taxes levied on vessels of boat owners within the County during FY 2020-21, the most recent fiscal year for which the annual total figure is available, will be expended on specified boating safety programs.

The foregoing resolution was on the 8th day of June 2021 adopted by the Board of Supervisors of the County of Los Angeles and ex officio the governing body of all other special assessments and taxing districts, agencies, and authorities for which said Board so acts.

CELIA ZAVALA, **Executive Officer** Clerk of the Board of Supervisors of the County of Los Angeles

By_____ Deputy

APPROVED AS TO FORM: **RODRIGO A. CASTRO-SILVA** County Counsel

By<u>Jenny Tam</u> Deputy



Boating Safety and Enforcement Financial Aid Program Agreement

This agreement entered into this *1ST day of July, 2021*, by and between the CALIFORNIA DEPARTMENT OF PARKS AND RECREATION, DIVISION OF BOATING AND WATERWAYS, hereinafter called "Department," and the *COUNTY OF LOS ANGELES*, hereinafter called "Agency";

WITNESSETH

WHEREAS, Contingent on approval of Governor's Fiscal Year 2021-22 Budget Act, the Department intends to agree with Agency for the purpose of performing boating safety and enforcement activities as described in Title 14, California Code of Regulations Section 6593.3; and

WHEREAS, Agency is equipped, staffed and prepared to provide such services on the terms and conditions set forth in this agreement and in accordance with Title 14, California Code of Regulations Section 6593 et seq.; and

WHEREAS, pursuant to Title 14, California Code of Regulations Section 6593.6, Department shall enter into an annual agreement with each participating agency;

NOW, THEREFORE, it is mutually agreed as follows:

I. Applicable Law

Agency shall observe and comply with all applicable federal, state, and county statutes, ordinances, regulations, directives, and laws, including, but not limited to, Harbors and Navigation Code Section 663.7 and Section 6593 et seq. of Title 14, California Code of Regulations. Agreement shall be deemed to be executed within the State of California and construed and governed by the laws of the State of California.

II. Description of Services

Agency shall conduct boating safety and enforcement activities in the jurisdiction of the Agency in consideration of the payments hereinafter set forth.

III. Payments

- A. <u>Maximum Amount</u>. The amount the Department shall be obligated to pay for services rendered under this agreement shall not exceed <u>\$2,120,000.00</u> for the agreement term in full consideration of Agency's performance of the services described in this agreement.
- B. <u>Rate of Payment</u>. The Department shall reimburse Agency in accordance with the reimbursement procedures set forth in Title 14, California Code of Regulations Section 6593.9.

- C. <u>Submission of Claims</u>. Agency shall submit claims for reimbursement to the Division contact person identified in paragraph V of this contract on a _____monthly **OR** \checkmark quarterly basis. (**Please check one**)
- D. <u>Failure to Submit Claims</u>. Claims for reimbursement shall be submitted within 60 days following the last day of the reporting period. Pursuant to Title 14, California Code of Regulations 6593.9 (i), the Department may reduce an Agency's allocation by five percent if the Agency exceeds the sixty-day billing period and an additional five percent for every thirty-day period thereafter that the Agency is late in filing a claim.

IV. Records

Agency shall maintain records pursuant to Section 6593.10 of Title 14, California Code of Regulations.

V. Notice

Notice shall be in writing and shall be deemed to have been served when it is deposited in the United States mail, first class postage prepaid, and addressed as follows:

TO DEPARTMENT	TO AGENCY	
Ms. Joanna Andrade	Theresa Barrera, Division Chief	
Department of Parks and Recreation	Financial Management Division	
Division of Boating and Waterways	Los Angeles County Fire Department	
One Capitol Mall, Suite 500	5801 S. Eastern Avenue, Suite 130	
Sacramento, CA 95814	Commerce, CA 90040	

Either party may change the address to which subsequent notice and/or other communication can be sent by giving written notice designating a change of address to the other party.

VI. Term

This agreement shall be for the term beginning July 1, 2021, and ending June 30, 2022.

VII. Prior Agreements

All prior agreements regarding this subject matter between Department and Agency are hereby terminated effective June 30 prior to the term beginning date of this agreement.

VIII. Amendment

No amendment or variation of the terms of this agreement shall be valid unless made in writing and signed by the parties hereto.

IX. Termination

Agency may terminate this agreement without cause in writing at any time. Department may terminate this agreement without cause upon a sixty (60) days written notice served upon the Agency.

X. Special Provisions

A. Agency hereby certifies that the obligations created by this agreement do not violate the provisions of Sections 1090 to 1096 of the Government Code.

B. This agreement shall have no force or effect until signed by the Department, Agency, and approved by the Department of General Services Legal Department, if required.

C. Agency shall continue with the responsibilities of this agreement during any dispute.

D. In the event of an allocation reduction for this program, an equal allotment will be decrease from every participant.

IN WITNESS WHEREOF, the parties hereto have executed this agreement as of the day and year first above written.

CALIFORNIA DEPARTMENT OF PARKS AND RECREATION, DIVISION OF BOATING AND WATERWAYS

By:_____

California Department of Parks and Recreation, Division of Boating and Waterways

Date:

"Department"

COUNTY OF LOS ANGELES

By:_____

Title: ______

Date:

"Agency"

CELIA ZAVALA, **Executive Officer** Clerk of the Board of Supervisors of the County of Los Angeles

By_____ Deputy

APPROVED AS TO FORM: RODRIGO A. CASTRO-SILVA County Counsel

By<u>Jenny Tam</u> Deputy

STATE'S COMMITMENT TO PROVIDE LONG-TERM FUNDING ASSISTANCE FOR THE DISTRICT

Public Resources Code, SECTION 1, Section 5002.6 (g), states:

"On and after June 30, 1998, it is the intent of the Legislature that any application by the County of Los Angeles Fire Department to secure State funding support for boating safety and enforcement on waters within the County of Los Angeles shall be given priority consideration by the Legislature, unless an alternative source of funding is secured prior to that date which serves the same or similar purposes."

The 1996 State Legislature addressed the intent of Section 5002.6 (g) to assist the District through Assembly Bill 122 (Rainey), Chapter 971, Statutes of 1996, which amended Section 663.7 of the Harbors and Navigation Code to establish a permanent funding allocation for current program recipients based upon their Fiscal Year 1996-97 allocation. The District's Lifeguard Rescue Safety Program is a current recipient and qualifies for this continued \$2.12 million funding by meeting all the criteria established by the Legislature and administered by DPRDBW as stated in the amended Harbors and Navigation Code Sections 85.2 and 663.7, Subdivision (a) and (c)(1), as follows:

Section 85.2 - "All money in the Harbors and Watercraft Revolving Fund shall be available, upon appropriation by the Legislature, for expenditure by the department for boating facilities development, <u>boating safety</u>, and boating regulation programs . . ."

Section 663.7 – Subdivision (a) – "Each county of the state is entitled to receive state financial aid for boating safety and enforcement programs on waters under its jurisdiction as provided in this section. A boating safety and enforcement program, as used in this section, includes search and rescue operations, recovery of drowned bodies, enforcement of state and local measures for regulation of boating activities, inspection of vessels, and supervision of organized water events."

Section 663.7 Subdivision (c)(1) – "Of the funds appropriated for boating safety and enforcement programs pursuant to Section 85.2, the department shall adopt and utilize a formula that first allocates funds to counties so that no county receives less than the amount it was allocated in the 1996-97 fiscal year, unless the county's program is reduced, or the county does not meet the eligibility requirements of this section...."

GENERAL DESCRIPTION OF BOATING SAFETY AND ENFORCEMENT PROGRAMS

CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY (DISTRICT)

The District's Lifeguard Division performs principal functions including search, rescue, recovery and emergency medical services for the maritime vicinity and inland waterways of Los Angeles County. The areas of responsibility cover 1,686 square miles of Los Angeles County inland waterways and coastal areas, including 72 miles of public beaches and open ocean waters, extending to and including the waters of Santa Catalina, San Clemente and the Santa Barbara Islands. Additionally, the District, through a Memorandum of Understanding with the United States Coast Guard, responds to an expanded area between Ventura and Orange Counties up to 50 nautical miles from shore. The district performs these functions under the following operational units.

LACoFD Lifeguard Rescue Boat Operations: The District maintains 24-hours, 365 days / yr marine rescue boat services at Avalon and Two Harbors on Santa Catalina Island, Marina del Rey, Redondo Beach King Harbor, and Cabrillo Beach in San Pedro. In addition, there are three 24-hour operation centers (Malibu, Santa Monica, and Hermosa Beach) which monitor marine emergency frequencies and dispatch on-call search and rescue vessels from the Los Angeles Harbor, Redondo Beach King Harbor, Marina del Rey and Malibu Pier. During peak boating season, there are a minimum of seven offshore rescue boats patrolling the Los Angeles coastline with the ability to augment to a total of 10 rescue boats.

The District's primary resources for marine search and rescue include ten offshore all-weather rescue boats; three trailer-based vessels for inland response; one 40-foot dedicated fire boat; one 35-foot dedicated fire boat; six rigid-hull, inflatable rescue boats; and eight rescue watercraft patrol vessels. District rescue boats are under the command of U.S. Coast Guard licensed Captains and are supplied with the necessary fire suppression equipment, emergency pumping capacity, towing gear, emergency medical equipment, helicopter lift and hoist capability, and emergency underwater rescue gear to handle any maritime rescue incident.

District Rescue Boat Emergency Responses and Maritime Services include:

- Search and Rescue Operations
- Emergency Medical Response EMT & Paramedic
- Boat Fire Suppression, overhaul and investigation
- Boat distress, emergency salvage and towing operations
- Underwater Rescue and Recovery
- Catalina Hyperbaric Chamber Paramedic Service for dive accidents
- Homeland Security Marine Domain Awareness Patrols & Documentation

- Recreational Boating and Water Sports Safety and Enforcement
- Commercial Fishing and Recreational Dive Boat Safety and Enforcement
- Fish and Wildlife education and enforcement

Mandatory training and certification requirements for District rescue boat personnel include:

- U.S. Coast Guard Merchant Mariner Credential with towing endorsements
- EMT or Paramedic with Defibrillator certification
- Marine Firefighting certification
- Rescue Boat Operations Training certification
- Rescue SCUBA certification
- PC 832 "Powers of Arrest

Supervision of water activities and events by District rescue boats and personnel include interagency training: Los Angeles International Airport (LAX) Disaster Drills, Coast Guard Search and Rescue Coordination, Oil Spill Contingency, Major Marine Disaster Plans; and supervision of various rough water swims, paddle races, beach festivals, fishing derbies, boat shows, and boat parades.

LACoFD Lifeguard Underwater Operations: The District maintains a deployable 35 person underwater rescue and recovery unit that responds to all Los Angeles County inland water ways including Castaic, Pyramid, and Puddingstone Lakes as well as the California Aqueduct and several other reservoirs within the Los Angeles Metropolitan Water District. The unit also responds to Los Angeles County Maritime areas along the 72 miles of coastline and the Catalina, San Clemente and Santa Barbara Islands. The unit's capabilities and response types include:

- Search, Rescue and Recovery Operations of people & property
- Boat Fire Responses
- Boats sinking / taking on water
- Aircraft Incidents
- Environmental Disaster Mitigation

Mandatory training and certification requirements for District underwater operations personnel include:

- Rescue SCUBA certifications
- Underwater Instructor certifications
- Full Face Mask (FFM) Communications Training
- Current & Tethered Diving Certifications
- Lifting and Salvage Training
- Helicopter deployment & rescue training

SHERIFF'S DEPARTMENT

The Los Angeles County Sheriff's Department (LASD or Department) currently serves a number of waterways within the County and has extensive experience in boating safety and enforcement services. The Department's four organizational units are:

Avalon Sheriff Station: Avalon Sheriff Station is located on Santa Catalina Island and services one city (of 3,500 residents and one-million annual visitors), three other communities on Santa Catalina Island, seven costal children/family camps, fifteen yacht clubs, one support fishing club, sixteen designated beach campgrounds, one university marine science center, and one airport.

LASD Santa Catalina Island resident deputies conduct maritime patrols upon near coastal waters around Santa Catalina and San Clemente Islands, and about 567 square miles of the Outer Santa Barbara Passage, between the two islands.

Marine responses can also extend to Santa Barbara Island and into the San Pedro Channel, toward the Southern California mainland. Avalon Sheriff Station has the largest and most remote maritime patrol area of any LASD Sheriff station, patrolling out to 65 miles south of the Port of Los Angeles.

Avalon Sheriff Station commands law enforcement, marine search and rescue, and homeland security jurisdictional authority along 209 square miles of coastal waters around Santa Catalina Island. Responsibility includes port security for Avalon Harbor (a daily port for commercial commuter ferries) and the local federal anchorage (a bi-weekly port for international cruise ships).

Avalon Station's patrol area also cover 207 square miles of coastal waters around San Clemente Island, which is part of Los Angeles County but owned by the U.S. Navy. Avalon Station maintains a MOU with the U.S. Navy at SCI.

The ocean waters around Santa Catalina and San Clemente Islands attract multitude of water enthusiasts daily from the Los Angeles metropolis and other areas.

Deputies conduct proactive water safety education and enforcement with the many sail boaters, power boaters, recreational sport fishermen (private and charter), scuba divers (private and charter), jet skiers, kayakers, paddle boarders, and live-aboard boaters in the area waters.

The rich sea beds around Santa Catalina and San Clemente Islands also draw large numbers of commercial fishing boats from many different ports.

Due to frequent rough seas and high winds, unprotected harbors, and desolate coastline: waterborne search and rescue operations, and responses to help distressed boaters are common.

Maritime calls for service include: responding to diver, swimmer, and vessel accidents; aircraft crashes; commercial fishermen regulation; fish and game violations; larceny; domestic disputes; boating under the influence (BUI) and other boating violations; narcotics violations and smuggling; environmental issues; firearms violations; homicides; and assisting Baywatch lifeguard paramedics with medical responses and medivac.

In summary, Avalon Sheriff Station's maritime service includes:

- Homeland Security and Counter Smuggling Operations;
- Search and Rescue Operations;
- Recreational Boating and Water Sports Safety and Enforcement;
- Commercial Fishing and Dive Boat Safety and Enforcement;
- Regulating Transient Live-Aboard Boaters;
- Medical Assistance;
- Responding to Vessel Fires and Distress;
- Diver Rescue and Recovery;
- Fish and Game Enforcement;
- U.S. Navy Assistance, San Clemente Island;
- Port Security; and
- Station Details.

Currently, two LASD patrol boats (with electronic navigation, radar, FLIR, and towing assets) are operated out of Two Harbors (a small community which is located 12 miles west of the City of Avalon). The LASD boats are stationed at each side of the Two Harbors Isthmus, for deployment to the north and south sides of Santa Catalina Island.

LASD's Special Enforcement Bureau, Emergency Services Detail (ESD) and Maritime Cadre are designated to support Avalon Station's maritime operations: conducting directed patrol, search and rescue, diver recovery, and counter smuggling operations within Avalon Sheriff Station's reporting districts and area of responsibility.

Special Enforcement Bureau:

The Special Enforcement Bureau consists of five details. The Special Enforcement Detail, Emergency Services Detail, Canine Services Detail, Hazmat Detail and Arson/Explosives Detail. All of these details conduct maritime operations. The Emergency Services Detail (ESD) is the Department's tactical paramedics and rescue specialists. The unit maintains a team on duty 24-hours a day. The unit's members are special weapons team members and are trained/certified in the following areas: Special weapons and tactics/tactical operations, mountain/rural rescue, ocean/swift water rescue and Public Safety divers. Additionally, each of the members is a licensed paramedic and is able to deliver advanced life support care in any of the austere environments the unit works in. The unit's members also

maintain cadres with advanced expertise involving instructor-level certifications to train its own members and those from Special Enforcement Bureau and outside agencies. ESD also functions as a Federal Emergency Management Agency asset and has been called on numerous times to offer its expertise and services in California and across the nation (Hurricane Katrina). Members of the unit have extensive tactical experience. Training to be an ESD deputy is extensive and, once started, lasts about 18 months to become operational.

The ESD is designated to conduct search, rescue, and recovery operations in the County. This unit provides support in the form of boating safety and enforcement for Marina Del Rey, Santa Catalina Island, Castaic Lake, Pyramid Lake, Puddingstone Lake, Whittier Lake, Jackson Lake, and over 100 local Los Angeles County lakes, ponds, and waterways.

The ESD has the responsibility for conducting all underwater search and recovery operations. Each year, this unit conducts multi-day recovery operations for individuals who have drowned or otherwise perished as a direct result of boating accidents. Additionally, this unit provides boat patrol for coastal areas, the Catalina Channel, and the ocean area between Catalina and San Clement Island. They also assist with Pyramid Lake, Castaic Lake and Santa Catalina Island on holidays and busy weekends.

Marina Del Rey (Harbor Patrol): The Los Angeles County Harbor Patrol was formed in 1962 as a Division of the Department of Small Craft Harbors. The Harbor Patrol provides safety and enforcement services to the boating community in the protected waters of Marina Del Rey Small Craft Harbor and adjacent beaches. Marina Del Rey is a year-round recreational area consisting of 403 acres of water and 401 acres of land. Within its confines are sixteen restaurants, twelve yacht clubs, four hotels, two major boat yards, and numerous apartment/ condominium complexes housing 12,000 permanent residents. The Sheriff's Harbor Patrol within Marina Del Rey is responsible for conducting marine safety and enforcement patrol in the protected waters of the marina, the waters of Santa Monica Bay, and patrols of the maritime domain of Los Angeles County. This region encompasses approximately 128 miles of coastline and 70 miles of coastal waters. Marina Del Rey Station staffs and deploys an off-shore patrol vessel (*Tradition*) that covers the entire county coastline from Orange County to Ventura County.

Additionally, while maintaining emergency response capabilities beyond the bay's area, the Harbor Patrol also serves as the County Disaster Coordination Center for aircraft emergencies occurring over waters adjacent to the LAX.

Pyramid Lake Recreational Area: The Pyramid Lake Recreational Area, which includes Emigrant Landing, Los Alamos Campground and Vista del Lago Center, attracts up to 500,000 persons yearly. Pyramid Lake was opened in 1974 and has 1,297 surface acres and 21 miles of shoreline. Pyramid Lake is located in the Angeles National Forest, an unincorporated area patrolled by the LASD. Deputies perform all aspects of boating safety and law enforcement. They enforce all

boating laws using verbal warnings, issuance of citations, or arrest. They educate the public through courtesy vessel inspections and boating safety videotapes. They are the first responders to all emergencies on Pyramid Lake and its associated shoreline. Boat handler deputies are emergency medical technicians and handle all medical aid incidents on this facility. Deputies prepare required reports, investigate all boating accidents, and file criminal complaints when warranted. The deputies have rescue dive capabilities, aid disabled vessels, handle salvage operations, and perform all general law enforcement in the recreational area.

On December 15, 2009, the Board of Supervisors voted and approved the consolidation of the LASD and the Los Angeles County Police. Effective July 1, 2010, the merger commenced, at which time, LASD took over all law enforcement responsibilities on and around Castaic Lake and Bonelli Lake in San Dimas.

The Castaic Lake Recreational Facility includes the Main Lake and Lower Lagoon areas. The Main Lake covers 2,235 surface acres and has 29 miles of shoreline and two launch ramp facilities. The Lower Lagoon covers 197 surface acres and has one launch ramp facility. There is an adjacent campground on the shoreline of the Lower Lagoon, which includes 90 overnight campsites. There is also a large swim beach area along the shoreline of the Lower Lagoon. The Los Angeles County lifeguards also provide trained personnel to conduct boating enforcement and aquatic and medical rescues. The estimated annual attendance of both the upper and lower lakes at Castaic is approximately 1,750,000.



COUNTY OF LOS ANGELES FIRE DEPARTMENT

1320 NORTH EASTERN AVENUE LOS ANGELES, CALIFORNIA 90063-3294 (323) 881-2426 www.fire.lacounty.gov

"Proud Protectors of Life, Property, and the Environment"

DARYL L. OSBY FIRE CHIEF FORESTER & FIRE WARDEN

June 8, 2021

BOARD OF SUPERVISORS

HILDA L. SOLIS FIRST DISTRICT

HOLLY J. MITCHELL SECOND DISTRICT

> SHEILA KUEHL THIRD DISTRICT

JANICE HAHN FOURTH DISTRICT

KATHRYN BARGER FIFTH DISTRICT

The Honorable Board of Supervisors County of Los Angeles 383 Kenneth Hahn Hall of Administration 500 West Temple Street Los Angeles, CA 90012

Dear Supervisors:

ADOPT RESOLUTION TO AUTHORIZE PARTICIPATION IN THE CALIFORNIA BOATING SAFETY AND ENFORCEMENT FINANCIAL AID PROGRAM FOR FISCAL YEAR 2021-22 (ALL DISTRICTS) (3 VOTES)

SUBJECT

Adopt a resolution to secure State funding through the California Department of Parks and Recreation, Division of Boating and Waterways (DBW), in support of boating safety and enforcement on waters within Los Angeles County (County). Also, authorize participation in the California Boating Safety and Enforcement Financial Aid Program (Program) by executing the Program agreement to reimburse the County in an amount not to exceed \$2.12 million.

IT IS RECOMMENDED THAT YOUR BOARD, ACTING AS THE GOVERNING BODY OF THE CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY AND THE BOARD OF SUPERVISORS:

- 1. Adopt the Resolution (Attachment I) whereby your Board:
 - Accepts the Legislature's commitment to provide future annual baseline funding for • the County and gives priority consideration to any application submitted by the County to secure State funding in support of boating safety and enforcement on waters within the County.

AGOURA HILLS ARTESIA AZUSA BALDWIN PARK BELL BELL GARDENS BELLFLOWER BRADBURY CALABASAS

CARSON CERRITOS CLAREMONT COMMERCE COVINA CUDAHY DIAMOND BAR DUARTE

FI MONTE GARDENA GLENDORA HAWAIIAN GARDENS HAWTHORNE HERMOSA BEACH HIDDEN HILLS HUNTINGTON PARK INDUSTRY

SERVING THE UNINCORPORATED AREAS OF LOS ANGELES COUNTY AND THE CITIES OF: INGLEWOOD IRWINDALE LA CANADA-FLINTRIDGE I A HABRA LA MIRADA LA PUENTE LAKEWOOD LANCASTER

LAWNDALE LOMITA LYNWOOD MALIBU MAYWOOD NORWALK PALMDALE PALOS VERDES ESTATES PARAMOUNT

PICO RIVERA POMONA RANCHO PALOS VERDES ROLLING HILLS ROLLING HILLS ESTATES ROSEMEAD SAN DIMAS SANTA CLARITA

SIGNAL HILL SOUTH EL MONTE SOUTH GATE TEMPLE CITY VERNON WALNUT WEST HOLLYWOOD WESTLAKE VILLAGE WHITTIER

The Honorable Board of Supervisors June 8, 2021 Page 2

- Reaffirms the continued participation of the County, through the Consolidated Fire Protection District of Los Angeles County (District) and the Los Angeles County Sheriff's Department, in the Program for Fiscal Year (FY) 2021-22.
- Certifies that the County, as a participant in the Program, shall expend the equivalent of 100 percent of its revenues collected from personal property tax on vessels of boat owners within the County for boating safety and enforcement.
- Authorizes the Fire Chief and the Sheriff, or their designees, as County agents to sign and submit an application and related expenditure reimbursement claims to the DBW for State funding.
- Authorizes the State funding received through the Program to be distributed solely to the District.
- 2. Authorize the Chair of the Board of Supervisors to execute the Boating Safety and Enforcement Financial Aid Program Agreement (Attachment II), which is required as part of the application package. In executing the agreement, the County agrees to submit requests for reimbursement within sixty days of the end of the fiscal quarter; and, if such requests are submitted after the sixty days has expired, the State has the option to reduce the allocation by five percent.
- 3. Find that the resolution and the funding of the County programs are exempt from the California Environmental Quality Act (CEQA) pursuant to Section 15061 (b)(3) of the CEQA Guidelines.

PURPOSE/JUSTIFICATION OF RECOMMENDED ACTION

The DBW provides financial aid from the Harbors and Watercraft Revolving Fund to counties under the Program, as authorized by Section 663.7 of the Harbors and Navigation Code. The County has been part of this Program since 1995. Since FY 2006-07, the District has received up to \$2.12 million annually.

These monies are part of an agreement with the State to provide funding to the County as part of the transfer of ownership of eight State beaches located within the County. A key part of this agreement was the State's commitment to provide long-term funding assistance for the District (Attachment III).

The financial aid application was submitted to the DBW on December 22, 2020. Agencies already participating in the Program are required to submit the completed application six months prior to the start of the new fiscal year to provide the DBW sufficient time to prepare the annual agreement for each agency. The DBW is requesting that the attached agreement for FY 2021-22 be executed and submitted no later than June 15, 2021. In addition, the Harbors and Navigation Code and application criteria require that your Board, by resolution, authorize the County's participation in the Program, and certify that the County will expend no less than 100 percent of the amount collected from personal property taxes on vessels for

The Honorable Board of Supervisors June 8, 2021 Page 3

boating safety programs during the funding year. The certification is required because counties qualify only if they expend an amount equivalent to at least 100 percent of these personal property taxes on boating safety and enforcement.

The designation of the Fire Chief and the Sheriff, or their designees, as signatories to the multi-department application, is consistent with your Board's instructions of December 8, 1994, to pursue long-term funding from the State. This State funding will offset portions of the offshore marine rescue and enforcement activities not presently met by the personal property tax on vessels. In previous years, the funds received from the financial aid have allowed the budgeted funds for these purposes to support the operations of the eight State beaches. This State funding will be received solely by the District.

Implementation of Strategic Plan Goals

Approval of the recommended actions is consistent with the County's Strategic Goal No. 3, Realize Tomorrow's Government Today, Strategy III.3 - Pursue Operational Effectiveness, Fiscal Responsibility, and Accountability to continually asses our efficiency and effectiveness, maximize and leverage resources, and hold ourselves accountable to maximize revenue.

FISCAL IMPACT/FINANCING

No matching funds are required. Counties must first expend the taxes collected from its resident boat owners before being entitled to any supplemental State funding from this Program. Only the cost of the Program, which exceeds the total fees and vessel taxes collected for the year, will be supplemented up to the maximum amount obligated by the State. The State's maximum funding for FY 2021-22 is \$2.12 million.

The County's FY 2021-22 application includes a request for \$15.900 million, net of fees and taxes, to maintain the County's boating safety and enforcement activities (Attachment IV). This request more than meets the State's funding requirement.

There is no impact on net County cost.

FACTS AND PROVISIONS/LEGAL REQUIREMENTS

The DBW provides financial aid from the Harbors and Watercraft Revolving Fund to counties under the Program, as authorized by Section 663.7 of the Harbors and Navigation Code. It allows the County to receive funding so long as the County conducts boating safety and enforcement activities. The DBW requires one adopted resolution by the Board of Supervisors per fiscal year to designate one or more agencies to participate in the Program. The County has been part of this Program since 1995. The DBW is requesting that the attached agreement for FY 2021-22 be executed and submitted. The application for FY 2021-22 was timely submitted on December 22, 2020.

The Honorable Board of Supervisors June 8, 2021 Page 4

ENVIRONMENTAL DOCUMENTATION

This Resolution and the funding of the County programs are exempt from the CEQA pursuant to Section 15061 (b)(3) of the CEQA Guidelines.

IMPACT ON CURRENT SERVICES (OR PROJECTS)

The services shall continue upon execution of this agreement.

CONCLUSION

Participation in the DBW Program will continue existing funding to the District Lifeguard rescue services for FY 2021-22 and continue financial aid to the Sheriff's Department boating safety and law enforcement efforts in equal portions. Your Board's adoption of the Resolution and execution of the financial assistance certification authorizes participation in the Program.

Upon approval by your Board, please instruct the Executive Officer to return four copies of the adopted Board letter and Attachments I and II to the following office:

Consolidated Fire Protection District of Los Angeles County Executive Office – Business Operations Attention: Zuleyda Santana, Administrative Services Manager II 1320 N. Eastern Avenue Los Angeles, CA 90063

The District's contact may be reached at (323) 881-6173.

Respectfully submitted,

DARYL L. OSBY, FIRE CHIEF

ALEX VILLANUEVA, SHERIFF

DLO:sh

Enclosures

c: Chief Executive Officer County Counsel Auditor-Controller Sheriff's Department

Attachment I

RESOLUTION

WHEREAS, Assembly Bill 909 (Bowen), Chapter 472, Statutes of 1995, amended Public Resources Code Section 5002.6, Subdivision (g) to read: "On and after June 30, 1998, it is the intent of the Legislature that any application by the Consolidated Fire Protection District of Los Angeles County (District) to secure State funding support for boating safety and enforcement on waters within the County of Los Angeles shall be given priority consideration by the Legislature, unless an alternative source of funding is secured prior to that date which serves the same or similar purposes."

WHEREAS, the District, has received for boating safety and enforcement prior allocations from the California Department of Parks and Recreation, Division of Boating and Waterways, in Fiscal Years (FYs) 1995-96 through 2020-21, which has established an annual baseline funding for future participation in the program in accordance with Subdivision (g) of Section 5002.6 of the Public Resource Code and as addressed by Assembly Bill 122 (Rainey), Chapter 971, Statutes of 1996, Section 2 under the Harbors and Navigation Code Sections 85.2 and 663.7(a) and (c)(1).

WHEREAS, other funding may be available to counties through the California Boating Safety and Enforcement Financial Aid Program; and

1

WHEREAS, the California Department of Parks and Recreation, Division of Boating and Waterways, pursuant to Section 663.7 of the Harbors and Navigation Code, provides supplemental State funding under the State Boating Safety and Enforcement Financial Aid Program to qualifying counties for boating safety and enforcement programs on waters under their jurisdiction; and

WHEREAS, the County of Los Angeles is charged with providing vital boating safety and enforcement services to a population in excess of nine (9) million people and over 60,000 registered boaters; and

WHEREAS, the current levels of those boating safety and enforcement services will continue through the District and the Los Angeles County Sheriff's Department; and

WHEREAS, allocation of said funding to any county or a public agency therein is contingent upon the County's governing body, the Board of Supervisors, adopting a resolution authorizing participation in the California Boating Safety and Enforcement Financial Aid Program and certifying that, during the funding year, an amount at least equal to the total amount collected by the County from personal property taxes on vessels will be expended on specified boating safety programs; and

WHEREAS, the Board of Supervisors of the County of Los Angeles through the District, and the Los Angeles County Sheriff's Department, wishes to participate in the

2

California Boating Safety and Enforcement Financial Aid Program administered by the California Department of Parks and Recreation, Division of Boating and Waterways and will distribute the funds from the Boating Safety and Enforcement Financial Aid Program solely to the District.

NOW, THEREFORE BE IT RESOLVED by the Board of Supervisors of the County of Los Angeles that the County of Los Angeles, through the District and the Los Angeles County Sheriff's Department, is hereby authorized to participate in and apply for the California Boating Safety and Enforcement Financial Aid Program for FY 2021-22.

BE IT FURTHER RESOLVED that the Board of Supervisors of the County of Los Angeles hereby certifies that during FY 2021-22, the funding year, an amount equal to 100 percent of the amount received by the County in personal property taxes levied on vessels of boat owners within the County during FY 2020-21, the most recent fiscal year for which the annual total figure is available, will be expended on specified boating safety programs.

The foregoing resolution was on the 8th day of June 2021 adopted by the Board of Supervisors of the County of Los Angeles and ex officio the governing body of all other special assessments and taxing districts, agencies, and authorities for which said Board so acts.

3

CELIA ZAVALA, **Executive Officer** Clerk of the Board of Supervisors of the County of Los Angeles

By_____ Deputy

APPROVED AS TO FORM: **RODRIGO A. CASTRO-SILVA** County Counsel

By Jenny Tam Deputy



Boating Safety and Enforcement Financial Aid Program Agreement

This agreement entered into this *1ST day of July, 2021*, by and between the CALIFORNIA DEPARTMENT OF PARKS AND RECREATION, DIVISION OF BOATING AND WATERWAYS, hereinafter called "Department," and the *COUNTY OF LOS ANGELES*, hereinafter called "Agency";

WITNESSETH

WHEREAS, Contingent on approval of Governor's Fiscal Year 2021-22 Budget Act, the Department intends to agree with Agency for the purpose of performing boating safety and enforcement activities as described in Title 14, California Code of Regulations Section 6593.3; and

WHEREAS, Agency is equipped, staffed and prepared to provide such services on the terms and conditions set forth in this agreement and in accordance with Title 14, California Code of Regulations Section 6593 et seq.; and

WHEREAS, pursuant to Title 14, California Code of Regulations Section 6593.6, Department shall enter into an annual agreement with each participating agency;

NOW, THEREFORE, it is mutually agreed as follows:

I. Applicable Law

Agency shall observe and comply with all applicable federal, state, and county statutes, ordinances, regulations, directives, and laws, including, but not limited to, Harbors and Navigation Code Section 663.7 and Section 6593 et seq. of Title 14, California Code of Regulations. Agreement shall be deemed to be executed within the State of California and construed and governed by the laws of the State of California.

II. Description of Services

Agency shall conduct boating safety and enforcement activities in the jurisdiction of the Agency in consideration of the payments hereinafter set forth.

III. Payments

- A. <u>Maximum Amount</u>. The amount the Department shall be obligated to pay for services rendered under this agreement shall not exceed <u>\$2,120,000.00</u> for the agreement term in full consideration of Agency's performance of the services described in this agreement.
- B. <u>Rate of Payment</u>. The Department shall reimburse Agency in accordance with the reimbursement procedures set forth in Title 14, California Code of Regulations Section 6593.9.

- C. <u>Submission of Claims</u>. Agency shall submit claims for reimbursement to the Division contact person identified in paragraph V of this contract on a _____monthly **OR** \checkmark quarterly basis. (**Please check one**)
- D. <u>Failure to Submit Claims</u>. Claims for reimbursement shall be submitted within 60 days following the last day of the reporting period. Pursuant to Title 14, California Code of Regulations 6593.9 (i), the Department may reduce an Agency's allocation by five percent if the Agency exceeds the sixty-day billing period and an additional five percent for every thirty-day period thereafter that the Agency is late in filing a claim.

IV. Records

Agency shall maintain records pursuant to Section 6593.10 of Title 14, California Code of Regulations.

V. Notice

Notice shall be in writing and shall be deemed to have been served when it is deposited in the United States mail, first class postage prepaid, and addressed as follows:

TO DEPARTMENT	TO AGENCY
Ms. Joanna Andrade	Theresa Barrera, Division Chief
Department of Parks and Recreation	Financial Management Division
Division of Boating and Waterways	Los Angeles County Fire Department
One Capitol Mall, Suite 500	5801 S. Eastern Avenue, Suite 130
Sacramento, CA 95814	Commerce, CA 90040

Either party may change the address to which subsequent notice and/or other communication can be sent by giving written notice designating a change of address to the other party.

VI. Term

This agreement shall be for the term beginning July 1, 2021, and ending June 30, 2022.

VII. Prior Agreements

All prior agreements regarding this subject matter between Department and Agency are hereby terminated effective June 30 prior to the term beginning date of this agreement.

VIII. Amendment

No amendment or variation of the terms of this agreement shall be valid unless made in writing and signed by the parties hereto.

IX. Termination

Agency may terminate this agreement without cause in writing at any time. Department may terminate this agreement without cause upon a sixty (60) days written notice served upon the Agency.

X. Special Provisions

A. Agency hereby certifies that the obligations created by this agreement do not violate the provisions of Sections 1090 to 1096 of the Government Code.

B. This agreement shall have no force or effect until signed by the Department, Agency, and approved by the Department of General Services Legal Department, if required.

C. Agency shall continue with the responsibilities of this agreement during any dispute.

D. In the event of an allocation reduction for this program, an equal allotment will be decrease from every participant.

IN WITNESS WHEREOF, the parties hereto have executed this agreement as of the day and year first above written.

CALIFORNIA DEPARTMENT OF PARKS AND RECREATION, DIVISION OF BOATING AND WATERWAYS

By:_____

California Department of Parks and Recreation, Division of Boating and Waterways

Date:

"Department"

COUNTY OF LOS ANGELES

By:_____

Title: ______

Date:

"Agency"

CELIA ZAVALA, **Executive Officer** Clerk of the Board of Supervisors of the County of Los Angeles

By_____ Deputy

APPROVED AS TO FORM: **RODRIGO A. CASTRO-SILVA** County Counsel

By Jenny Tam Deputy

STATE'S COMMITMENT TO PROVIDE LONG-TERM FUNDING ASSISTANCE FOR THE DISTRICT

Public Resources Code, SECTION 1, Section 5002.6 (g), states:

"On and after June 30, 1998, it is the intent of the Legislature that any application by the County of Los Angeles Fire Department to secure State funding support for boating safety and enforcement on waters within the County of Los Angeles shall be given priority consideration by the Legislature, unless an alternative source of funding is secured prior to that date which serves the same or similar purposes."

The 1996 State Legislature addressed the intent of Section 5002.6 (g) to assist the District through Assembly Bill 122 (Rainey), Chapter 971, Statutes of 1996, which amended Section 663.7 of the Harbors and Navigation Code to establish a permanent funding allocation for current program recipients based upon their Fiscal Year 1996-97 allocation. The District's Lifeguard Rescue Safety Program is a current recipient and qualifies for this continued \$2.12 million funding by meeting all the criteria established by the Legislature and administered by DPRDBW as stated in the amended Harbors and Navigation Code Sections 85.2 and 663.7, Subdivision (a) and (c)(1), as follows:

Section 85.2 - "All money in the Harbors and Watercraft Revolving Fund shall be available, upon appropriation by the Legislature, for expenditure by the department for boating facilities development, <u>boating safety</u>, and boating regulation programs . . ."

Section 663.7 – Subdivision (a) – "Each county of the state is entitled to receive state financial aid for boating safety and enforcement programs on waters under its jurisdiction as provided in this section. A boating safety and enforcement program, as used in this section, includes search and rescue operations, recovery of drowned bodies, enforcement of state and local measures for regulation of boating activities, inspection of vessels, and supervision of organized water events."

Section 663.7 Subdivision (c)(1) – "Of the funds appropriated for boating safety and enforcement programs pursuant to Section 85.2, the department shall adopt and utilize a formula that first allocates funds to counties so that no county receives less than the amount it was allocated in the 1996-97 fiscal year, unless the county's program is reduced, or the county does not meet the eligibility requirements of this section...."

GENERAL DESCRIPTION OF BOATING SAFETY AND ENFORCEMENT PROGRAMS

CONSOLIDATED FIRE PROTECTION DISTRICT OF LOS ANGELES COUNTY (DISTRICT)

The District's Lifeguard Division performs principal functions including search, rescue, recovery and emergency medical services for the maritime vicinity and inland waterways of Los Angeles County. The areas of responsibility cover 1,686 square miles of Los Angeles County inland waterways and coastal areas, including 72 miles of public beaches and open ocean waters, extending to and including the waters of Santa Catalina, San Clemente and the Santa Barbara Islands. Additionally, the District, through a Memorandum of Understanding with the United States Coast Guard, responds to an expanded area between Ventura and Orange Counties up to 50 nautical miles from shore. The district performs these functions under the following operational units.

LACoFD Lifeguard Rescue Boat Operations: The District maintains 24-hours, 365 days / yr marine rescue boat services at Avalon and Two Harbors on Santa Catalina Island, Marina del Rey, Redondo Beach King Harbor, and Cabrillo Beach in San Pedro. In addition, there are three 24-hour operation centers (Malibu, Santa Monica, and Hermosa Beach) which monitor marine emergency frequencies and dispatch on-call search and rescue vessels from the Los Angeles Harbor, Redondo Beach King Harbor, Marina del Rey and Malibu Pier. During peak boating season, there are a minimum of seven offshore rescue boats patrolling the Los Angeles coastline with the ability to augment to a total of 10 rescue boats.

The District's primary resources for marine search and rescue include ten offshore all-weather rescue boats; three trailer-based vessels for inland response; one 40-foot dedicated fire boat; one 35-foot dedicated fire boat; six rigid-hull, inflatable rescue boats; and eight rescue watercraft patrol vessels. District rescue boats are under the command of U.S. Coast Guard licensed Captains and are supplied with the necessary fire suppression equipment, emergency pumping capacity, towing gear, emergency medical equipment, helicopter lift and hoist capability, and emergency underwater rescue gear to handle any maritime rescue incident.

District Rescue Boat Emergency Responses and Maritime Services include:

- Search and Rescue Operations
- Emergency Medical Response EMT & Paramedic
- Boat Fire Suppression, overhaul and investigation
- Boat distress, emergency salvage and towing operations
- Underwater Rescue and Recovery
- Catalina Hyperbaric Chamber Paramedic Service for dive accidents
- Homeland Security Marine Domain Awareness Patrols & Documentation

- Recreational Boating and Water Sports Safety and Enforcement
- Commercial Fishing and Recreational Dive Boat Safety and Enforcement
- Fish and Wildlife education and enforcement

Mandatory training and certification requirements for District rescue boat personnel include:

- U.S. Coast Guard Merchant Mariner Credential with towing endorsements
- EMT or Paramedic with Defibrillator certification
- Marine Firefighting certification
- Rescue Boat Operations Training certification
- Rescue SCUBA certification
- PC 832 "Powers of Arrest

Supervision of water activities and events by District rescue boats and personnel include interagency training: Los Angeles International Airport (LAX) Disaster Drills, Coast Guard Search and Rescue Coordination, Oil Spill Contingency, Major Marine Disaster Plans; and supervision of various rough water swims, paddle races, beach festivals, fishing derbies, boat shows, and boat parades.

LACoFD Lifeguard Underwater Operations: The District maintains a deployable 35 person underwater rescue and recovery unit that responds to all Los Angeles County inland water ways including Castaic, Pyramid, and Puddingstone Lakes as well as the California Aqueduct and several other reservoirs within the Los Angeles Metropolitan Water District. The unit also responds to Los Angeles County Maritime areas along the 72 miles of coastline and the Catalina, San Clemente and Santa Barbara Islands. The unit's capabilities and response types include:

- Search, Rescue and Recovery Operations of people & property
- Boat Fire Responses
- Boats sinking / taking on water
- Aircraft Incidents
- Environmental Disaster Mitigation

Mandatory training and certification requirements for District underwater operations personnel include:

- Rescue SCUBA certifications
- Underwater Instructor certifications
- Full Face Mask (FFM) Communications Training
- Current & Tethered Diving Certifications
- Lifting and Salvage Training
- Helicopter deployment & rescue training

SHERIFF'S DEPARTMENT

The Los Angeles County Sheriff's Department (LASD or Department) currently serves a number of waterways within the County and has extensive experience in boating safety and enforcement services. The Department's four organizational units are:

Avalon Sheriff Station: Avalon Sheriff Station is located on Santa Catalina Island and services one city (of 3,500 residents and one-million annual visitors), three other communities on Santa Catalina Island, seven costal children/family camps, fifteen yacht clubs, one support fishing club, sixteen designated beach campgrounds, one university marine science center, and one airport.

LASD Santa Catalina Island resident deputies conduct maritime patrols upon near coastal waters around Santa Catalina and San Clemente Islands, and about 567 square miles of the Outer Santa Barbara Passage, between the two islands.

Marine responses can also extend to Santa Barbara Island and into the San Pedro Channel, toward the Southern California mainland. Avalon Sheriff Station has the largest and most remote maritime patrol area of any LASD Sheriff station, patrolling out to 65 miles south of the Port of Los Angeles.

Avalon Sheriff Station commands law enforcement, marine search and rescue, and homeland security jurisdictional authority along 209 square miles of coastal waters around Santa Catalina Island. Responsibility includes port security for Avalon Harbor (a daily port for commercial commuter ferries) and the local federal anchorage (a bi-weekly port for international cruise ships).

Avalon Station's patrol area also cover 207 square miles of coastal waters around San Clemente Island, which is part of Los Angeles County but owned by the U.S. Navy. Avalon Station maintains a MOU with the U.S. Navy at SCI.

The ocean waters around Santa Catalina and San Clemente Islands attract multitude of water enthusiasts daily from the Los Angeles metropolis and other areas.

Deputies conduct proactive water safety education and enforcement with the many sail boaters, power boaters, recreational sport fishermen (private and charter), scuba divers (private and charter), jet skiers, kayakers, paddle boarders, and live-aboard boaters in the area waters.

The rich sea beds around Santa Catalina and San Clemente Islands also draw large numbers of commercial fishing boats from many different ports.

Due to frequent rough seas and high winds, unprotected harbors, and desolate coastline: waterborne search and rescue operations, and responses to help distressed boaters are common.

Maritime calls for service include: responding to diver, swimmer, and vessel accidents; aircraft crashes; commercial fishermen regulation; fish and game violations; larceny; domestic disputes; boating under the influence (BUI) and other boating violations; narcotics violations and smuggling; environmental issues; firearms violations; homicides; and assisting Baywatch lifeguard paramedics with medical responses and medivac.

In summary, Avalon Sheriff Station's maritime service includes:

- Homeland Security and Counter Smuggling Operations;
- Search and Rescue Operations;
- Recreational Boating and Water Sports Safety and Enforcement;
- Commercial Fishing and Dive Boat Safety and Enforcement;
- Regulating Transient Live-Aboard Boaters;
- Medical Assistance;
- Responding to Vessel Fires and Distress;
- Diver Rescue and Recovery;
- Fish and Game Enforcement;
- U.S. Navy Assistance, San Clemente Island;
- Port Security; and
- Station Details.

Currently, two LASD patrol boats (with electronic navigation, radar, FLIR, and towing assets) are operated out of Two Harbors (a small community which is located 12 miles west of the City of Avalon). The LASD boats are stationed at each side of the Two Harbors Isthmus, for deployment to the north and south sides of Santa Catalina Island.

LASD's Special Enforcement Bureau, Emergency Services Detail (ESD) and Maritime Cadre are designated to support Avalon Station's maritime operations: conducting directed patrol, search and rescue, diver recovery, and counter smuggling operations within Avalon Sheriff Station's reporting districts and area of responsibility.

Special Enforcement Bureau:

The Special Enforcement Bureau consists of five details. The Special Enforcement Detail, Emergency Services Detail, Canine Services Detail, Hazmat Detail and Arson/Explosives Detail. All of these details conduct maritime operations. The Emergency Services Detail (ESD) is the Department's tactical paramedics and rescue specialists. The unit maintains a team on duty 24-hours a day. The unit's members are special weapons team members and are trained/certified in the following areas: Special weapons and tactics/tactical operations, mountain/rural rescue, ocean/swift water rescue and Public Safety divers. Additionally, each of the members is a licensed paramedic and is able to deliver advanced life support care in any of the austere environments the unit works in. The unit's members also

maintain cadres with advanced expertise involving instructor-level certifications to train its own members and those from Special Enforcement Bureau and outside agencies. ESD also functions as a Federal Emergency Management Agency asset and has been called on numerous times to offer its expertise and services in California and across the nation (Hurricane Katrina). Members of the unit have extensive tactical experience. Training to be an ESD deputy is extensive and, once started, lasts about 18 months to become operational.

The ESD is designated to conduct search, rescue, and recovery operations in the County. This unit provides support in the form of boating safety and enforcement for Marina Del Rey, Santa Catalina Island, Castaic Lake, Pyramid Lake, Puddingstone Lake, Whittier Lake, Jackson Lake, and over 100 local Los Angeles County lakes, ponds, and waterways.

The ESD has the responsibility for conducting all underwater search and recovery operations. Each year, this unit conducts multi-day recovery operations for individuals who have drowned or otherwise perished as a direct result of boating accidents. Additionally, this unit provides boat patrol for coastal areas, the Catalina Channel, and the ocean area between Catalina and San Clement Island. They also assist with Pyramid Lake, Castaic Lake and Santa Catalina Island on holidays and busy weekends.

Marina Del Rey (Harbor Patrol): The Los Angeles County Harbor Patrol was formed in 1962 as a Division of the Department of Small Craft Harbors. The Harbor Patrol provides safety and enforcement services to the boating community in the protected waters of Marina Del Rey Small Craft Harbor and adjacent beaches. Marina Del Rey is a year-round recreational area consisting of 403 acres of water and 401 acres of land. Within its confines are sixteen restaurants, twelve yacht clubs, four hotels, two major boat yards, and numerous apartment/ condominium complexes housing 12,000 permanent residents. The Sheriff's Harbor Patrol within Marina Del Rey is responsible for conducting marine safety and enforcement patrol in the protected waters of the marina, the waters of Santa Monica Bay, and patrols of the maritime domain of Los Angeles County. This region encompasses approximately 128 miles of coastline and 70 miles of coastal waters. Marina Del Rey Station staffs and deploys an off-shore patrol vessel (*Tradition*) that covers the entire county coastline from Orange County to Ventura County.

Additionally, while maintaining emergency response capabilities beyond the bay's area, the Harbor Patrol also serves as the County Disaster Coordination Center for aircraft emergencies occurring over waters adjacent to the LAX.

Pyramid Lake Recreational Area: The Pyramid Lake Recreational Area, which includes Emigrant Landing, Los Alamos Campground and Vista del Lago Center, attracts up to 500,000 persons yearly. Pyramid Lake was opened in 1974 and has 1,297 surface acres and 21 miles of shoreline. Pyramid Lake is located in the Angeles National Forest, an unincorporated area patrolled by the LASD. Deputies perform all aspects of boating safety and law enforcement. They enforce all

boating laws using verbal warnings, issuance of citations, or arrest. They educate the public through courtesy vessel inspections and boating safety videotapes. They are the first responders to all emergencies on Pyramid Lake and its associated shoreline. Boat handler deputies are emergency medical technicians and handle all medical aid incidents on this facility. Deputies prepare required reports, investigate all boating accidents, and file criminal complaints when warranted. The deputies have rescue dive capabilities, aid disabled vessels, handle salvage operations, and perform all general law enforcement in the recreational area.

On December 15, 2009, the Board of Supervisors voted and approved the consolidation of the LASD and the Los Angeles County Police. Effective July 1, 2010, the merger commenced, at which time, LASD took over all law enforcement responsibilities on and around Castaic Lake and Bonelli Lake in San Dimas.

The Castaic Lake Recreational Facility includes the Main Lake and Lower Lagoon areas. The Main Lake covers 2,235 surface acres and has 29 miles of shoreline and two launch ramp facilities. The Lower Lagoon covers 197 surface acres and has one launch ramp facility. There is an adjacent campground on the shoreline of the Lower Lagoon, which includes 90 overnight campsites. There is also a large swim beach area along the shoreline of the Lower Lagoon. The Los Angeles County lifeguards also provide trained personnel to conduct boating enforcement and aquatic and medical rescues. The estimated annual attendance of both the upper and lower lakes at Castaic is approximately 1,750,000.

ALTERNATE PUBLIC DEFENDER

	FY 2021-22 Recommended Budget	FY 2021-22 Final Changes	Variance
Appropriation	78,476,000	81,401,000	2,925,000
Intrafund Transfer	181,000	181,000	
Revenue	1,554,000	1,592,000	38,000
Net County Cost	76,741,000	79,628,000	2,887,000
Budgeted Positions	308.0	308.0	0.0

		Gross Appropriation (\$)	Intrafund Transfers (\$)	Revenue (\$)	Net County Cost (\$)	Budg Pos
ALTERNA	TE PUBLIC DEFENDER					
2021-22 R Other Chan	ecommended Budget ges	78,476,000	181,000	1,554,000	76,741,000	308.0
	ent: Reflects revised retirement estimates oon updated payroll data.	(127,000)	-	(2,000)	(125,000)	0.0
	Plan: Reflects restoration of the County tion to the 401(k) Savings Plan.	1,834,000		24,000	1,810,000	0.0
contribu represer	s: Reflects restoration of the County tion to the Thrift Plan (Horizons) for non- nted and grand-fathered represented Flex and ex employees.	1,216,000		16,000	1,200,000	0.0
Departm	ise System Maintenance: Reflects the lent's proportional share of eCAPS ance cost increases.	2,000			2,000	0.0
align fur Federal	ial Change: Reflects an adjustment to properly ding allocated for rent charges to comply with Office of Management and Budget claiming (2CFR Part 200).	-			-	0.0
	Total Changes	2,925,000		38,000	2,887,000	0.0
2021-22 Fi	nal Changes	81,401,000	181,000	1,592,000	79,628,000	308.0

DISTRICT ATTORNEY

	FY 2021-22 Recommended Budget	FY 2021-22 Final Changes	Variance
Appropriation	462,811,000	471,842,000	9,031,000
Intrafund Transfer	4,451,000	4,451,000	0
Revenue	197,312,000	199,093,000	1,781,000
Net County Cost	261,048,000	268,298,000	7,250,000
Budgeted Positions	2,130.0	2,130.0	0.0

		Gross Appropriation	Intrafund Transfers	Revenue	Net County Cost	Buda
		(\$)	(\$)	(\$)	(\$)	Pos
20	21-22 Recommended Budget	462,811,000	4,451,000	197,312,000	261,048,000	2,130.0
1.	Salaries & Employee Benefits: Reflects Board- approved adjustments to salaries and employee benefits.	12,000		1,000	11,000	0.0
2.	Retirement: Reflects revised retirement estimates based upon updated payroll data.	(3,448,000)		(239,000)	(3,209,000)	0.0
3.	Savings Plan: Reflects restoration of the County contribution to the 401(k) Savings Plan.	7,296,000		486,000	6,810,000	0.0
4.	Horizons Plan: Reflects restoration of the County contribution to the Thrift Plan (Horizons) for non-represented and grand-fathered represented Flex and MegaFlex employees.	5,159,000		344,000	4,815,000	0.0
5.	Public Safety Sales Tax (Proposition 172): Reflects a projected increase in revenue based on historical experience and anticipated trends.			927,000	(927,000)	0.0
6.	Utility Users Tax (UUT): Reflects a decrease in UUT appropriation needed to align with anticipated funding offset by an increase in appropriation and revenue from the Community Development Block Grant.			261,000	(261,000)	0.0
7.	eCAPS Maintenance: Reflects the Department's proportional share of eCAPS maintenance cost increases.	12,000		1,000	11,000	0.0
	Total Changes	9,031,000		1,781,000	7,250,000	0.0
20	21-22 Final Changes	471,842,000	4,451,000	199,093,000	268,298,000	2,130.0

FIRE DEPARTMENT

	FY 2021-22 Recommended Budget	FY 2021-22 Final Changes	Variance
Financing Sources	1,334,761,000	1,379,347,000	44,586,000
Financing Uses	1,334,761,000	1,379,347,000	44,586,000
Budgeted Positions	4,739.0	4,740.0	1.0

		Financing Uses (\$)	Financing Sources (\$)	Budg Pos
20	21-22 Recommended Budget	1,334,761,000	1,334,761,000	4,739.0
Otl	ner Changes			
1.	Grants: Reflects the carryover of Board-approved grant funding.	22,909,000	22,932,000	-
2.	Reorganizational Changes: Reflects reorganizations of the Community Outreach, Recruitment, Diversity, and Inclusion (CORDI) Section, Return to Work (RW), and Risk Management (RM) sections, and the net addition of 1.0 position in to manage RW and RM.	332,000	-	1.0
3.	Fire Camp 13 Equipment: Reflects funding for equipment for Fire Camp 13 that was damaged by the Woolsey Fire.	60,000	-	-
4.	Workers' Compensation Management: Reflects funding for a Workers' Compensation management consultant and settlement costs.	5,673,000	-	-
5.	Operational Services: Reflects funding for vehicle fleet operations, air operations, replacement personal protective equipment for fire fighters, firefighting foam, and ministerial adjustments between bureaus.	3,630,000	-	-
6.	Administrative Services: Reflects funding for CORDI recruitment efforts, Fire Fighter Trainee exam and validation services, and temporary financial staff to assist with cost recovery claims management.	1,065,000	-	-
7.	Donations: Reflects an increase in appropriation funded by donations.	25,000	25,000	-
8.	Negotiated COLA Increases and Fringe Benefits: Reflects Board-approved adjustments to salaries and employee benefits.	1,684,000	2,000	-
9.	Retirement: Reflects revised retirement estimates based upon updated payroll data.	(750,000)	(26,000)	-
10.	Savings: Reflects restoration of the County contribution to the 401(k) Savings Plan.	1,921,000	66,000	-
11.	Horizons: Reflects restoration of the County contribution to the Thrift Plan (Horizons) for non-represented and grand-fathered represented Flex and MegaFlex employees.	1,553,000	53,000	-
12.	eCAPS Maintenance: Reflects the Department's proportional share of eCAPS maintenance cost increases.	34,000	1,000	-
13.	Property Tax Increase: Reflects an increase in property taxes revenue based on a projected increase in assessed valuation.	-	9,875,000	-
14.	Special Tax Increase: Reflects an increase in special taxes based on current collection trends.	-	87,000	-
15.	Other Revenue Increase: Reflects a net increase based on the latest projections of Fee for Services revenue, Advanced Life Support revenue, and other revenue sources.	-	11,571,000	-
16.	Appropriation for Contingencies: Reflects funding for projects that are currently in development.	6,450,000	-	-
	Total Changes	44,586,000	44,586,000	1.0
202	21-22 Final Changes	1,379,347,000	1,379,347,000	4,740.0

LIFEGUARD – FIRE DEPARTMENT

	FY 2021-22 Recommended Budget	FY 2021-22 Final Changes	Variance
Appropriation	39,453,000	39,549,000	96,000
Intrafund Transfer	0	0	0
Revenue	0	0	0
Net County Cost	39,453,000	39,549,000	96,000
Budgeted Positions	0.0	0.0	0.0

		Gross Appropriation (\$)	Intrafund Transfers (\$)	Revenue (\$)	Net County Cost (\$)	Budg Pos
L	FEGUARD – FIRE DEPARTMENT		X · 7			
20	21-22 Recommended Budget	39,453,000	0	0	39,453,000	0.0
Ot	her Changes					
1.	Salaries and Employee Benefits: Reflects Board- approved adjustments to salaries and employee benefits.	2,000	-		2,000	
2.	Retirement: Reflects revised retirement estimates based upon updated payroll data.	(26,000)			(26,000)	
3.	Savings Plan: Reflects restoration of the County contribution to the 401(k) Savings Plan.	66,000			66,000	
4.	Horizons Plan: Reflects restoration of the County contribution to the Thrift Plan (Horizons) for non-represented and grand-fathered represented Flex and MegaFlex employees.	53,000			53,000	
5.	eCAPS Maintenance: Reflects funding for the Department's share of eCAPS maintenance costs.	1,000			71,000	
	Total Changes	96,000	0	0	96,000	0.0
20	21-22 Final Changes	39,549,000	0	0	39,549,000	0.0

Public Safety Cluster + Final Changes + Fiscal Year 2021-22

GRAND JURY

	FY 2021-22 Recommended Budget	FY 2021-22 Final Changes	Variance
Appropriation	1,919,000	1,933,000	14,000
Intrafund Transfer	0	0	0
Revenue	4,000	4,000	0
Net County Cost	1,915,000	1,929,000	14,000
Budgeted Positions	5.0	5.0	0.0

		Gross Appropriation (¢)	Intrafund Transfers	Revenue	Net County Cost (\$)	Budg Pos
G	RAND JURY	(\$)	(\$)	(\$)	(\$)	F 03
20	21-22 Recommended Budget	1,919,000	0	4,000	1,915,000	5.0
Oť	her Changes					
1.	Retirement: Reflects revised retirement estimates based upon updated payroll data.	7,000			7,000	
2.	Savings Plan: Reflects restoration of the County contribution to the 401(k) Savings Plan.	7,000			7,000	
	Total Changes	14,000	0	0	14,000	0.0
20	21-22 Final Changes	1,933,000	0	4,000	1,929,000	5.0

MEDICAL EXAMINER-CORONER

	FY 2021-22 Recommended Budget	FY 2021-22 Final Changes	Variance
Appropriation	46,285,000	47,370,000	1,085,000
Intrafund Transfer	3,000	3,000	0
Revenue	2,252,000	2,263,000	11,000
Net County Cost	44,030,000	45,104,000	1,074,000
Budgeted Positions	234.0	238.0	4.0

		Gross Appropriation (\$)	Intrafund Transfers (\$)	Revenue (\$)	Net County Cost (\$)	Budg Pos
М	EDICAL EXAMINER-CORONER	(*)				
20	21-22 Recommended Budget	46,285,000	3,000	2,252,000	44,030,000	234.0
Cı	itical Issues					
1.	Death Investigations: Reflects funding for 3.0 Coroner Investigators to address increased caseload.	429,000			429,000	3.0
2.	Forensic Laboratories: Reflects funding for 1.0 Senior Criminalist to address increased caseload.	182,000			182,000	1.0
Ot	her Changes					
3.	Salaries and Employee Benefits: Reflects Board- approved adjustments to salaries and employee benefits.	(31,000)		(1,000)	(30,000)	
4.	Retirement: Reflects revised retirement estimates based upon updated payroll data.	85,000		2,000	83,000	
5.	Savings Plan: Reflects restoration of the County contribution to the 401(k) Savings Plan.	249,000		6,000	243,000	
6.	Horizons Plan: Reflects restoration of the County contribution to the Thrift Plan (Horizons) for non-represented and grand-fathered represented Flex and MegaFlex employees.	170,000		4,000	166,000	
7.	eCAPS Maintenance: Reflects the Department's proportional share of eCAPS maintenance cost increases.	1,000			1,000	
	Total Changes	1,085,000	0	11,000	1,074,000	4.0
20	21-22 Final Changes	47,370,000	3,000	2,263,000	45,104,000	238.0

PROBATION DEPARTMENT

	FY 2020-21 Recommended Budget	FY 2021-22 Final Changes	Variance
Appropriation	1,012,446,000	1,013,240,000	794,000
Intrafund Transfer	3,444,000	3,444,000	0
Revenue	370,096,000	370,232,000	136,000
Net County Cost	638,906,000	639,564,000	658,000
Budgeted Positions	5,671.0	5,540.0	(131.0)

		Gross Appropriation	Intrafund Transfers	Revenue	Net County Cost	Budg
		(\$)	(\$)	(\$)	(\$)	Pos
P	ROBATION DEPARTMENT					
20	20-21 Recommended Budget	1,012,446,000	3,444,000	370,096,000	638,906,000	5,671.0
1.	Salaries and Employee Benefits: Reflects Board- approved adjustments to salaries and employee benefits.	(44,000)		(5,000)	(39,000)	
	Support Services	(44,000)		(5,000)	(39,000)	-
2.	Retirement: Reflects revised retirement estimates based upon updated payroll data.	(8,445,000)		(1,028,000)	(7,417,000)	
	Support Services	(776,000))			(776,000)	
	Juvenile Institution Services – DSB	(3,335,000)			(3,335,000)	
	Juvenile Institution Services - RTSB	(2,506,000)			(2,506,000)	
	Field Services	(1,109,000)		(1,028,000)	(81,000)	
	Special Services	(719,000)			(719,000)	
3.	Savings Plan: Reflects restoration of the County contribution to the 401(k) Savings Plan.	1,784,000		260,000	1,524,000	
	Support Services	1,045,000			1,045,000	
	Juvenile Institution Services – DSB	128,000			128,000	
	Juvenile Institution Services - RTSB	119,000			119,000	
	Field Services	375,000		260,000	115,000	
	Special Services	117,000			117,000	
4.	Horizons Plan: Reflects restoration of the County contribution to the Thrift Plan (Horizons) for non-represented and grand-fathered represented Flex and MegaFlex employees.	1,742,000		255,000	1,487,000	
	Support Services	264,000			264,000	
	Juvenile Institution Services – DSB	343,000			343,000	
	Juvenile Institution Services - RTSB	230,000			230,000	
	Field Services	663,000		255,000	408,000	
	Special Services	242,000			242,000	

PROBATION DEPARTMENT

		Gross Appropriation (\$)	Intrafund Transfers (\$)	Revenue (\$)	Net County Cost (\$)	Budg Pos
5.	eCAPS Maintenance: Reflects the Department's proportional share of eCAPS maintenance cost increases.	21,000	(_/ 	(_) 	21,000	
	Support Services	21,000			21,000	
6.	Compensatory Education Services: Reflects \$2.7 million to provide compensatory services to youths detained in the juvenile halls whose educational experiences may have been compromised, both in quality and access; fully funded by one-time Juvenile Probation Activities growth revenues. These services include tutoring, education counseling, and career planning.	2,689,000		2,689,000	-	-
	Juvenile Institution Services – DSB	2,689,000		2,689,000		
7.	Division of Juvenile Justice (DJJ) Realignment: Reflects the first year's funding of \$8.3 million from the State for the transfer of DJJ wards from the State to the Probation department. As part of the Governor's budget, the DJJ will no longer admit new youth effective July 1, 2021. These youth will be housed by the Probation department.	8,299,000		8,299,000	-	-
	Juvenile Institution Services – DSB	8,299,000		8,299,000		
8.	Department of Mental Health (DMH) Pilot: Reflects \$5.1 million to implement a one-year pilot program at Central Juvenile Hall to expand mental health staffing in certain enhanced supervision and care units to meet youths' needs on a 16-hour-a-day, 7-day per week basis. The funding will be passed through the Probation department to the Department of Mental Health (DMH), as DMH will be providing the services to Probation youth.	5,082,000			5,082,000	-
	Juvenile Institutions Services – DSB	5,082,000			5,082,000	
9.	Position Reclass: Reflects the reclass actions approved by the Board on March 9, 2021 for various positions.				-	
	Juvenile Institution Services – DSB					
10.	Fiscal Services: Reflects 1.0 Accountant III position to manage the increased workload in processing and tracking for various reimbursement claims; fully offset by the deletion of 1.0 vacant budgeted position.	-				
	Support Services	131,000			131,000	1.0
	Juvenile Institution Services - RTSB	(131,000)			(131,000)	(1.0)
11.	Revenue Shortfall: Reflects the deletion of 131.0 vacant budgeted positions to address revenue shortfalls.	(10,334,000)		(10,334,000)		(131.0)
	Support Services			(5,709,000))	5,709,000	
	Juvenile Institution Services – DSB	(6,522,000)			(6,522,000)	(88.0)
	Juvenile Institution Services – RTSB	(3,812,000)		(2,418,000))	(1,394,000)	(43.0)

PROBATION DEPARTMENT

		Gross Appropriation	Intrafund Transfers	Revenue	Net County Cost	Budg
		(\$)	(\$)	(\$)	(\$)	Pos
Special Services				(2,207,000)	2,207,000	
	Total Changes	794,000		136,000	658,000	(131.0)
2021-22 Final Changes		1,013,240,000	3,444,000	370,232,000	639,564,000	5,540.0

PUBLIC DEFENDER

	FY 2021-22 Recommended Budget	FY 2021-22 Final Changes	Variance
Appropriation	249,709,000	254,783,000	5,074,000
Intrafund Transfer	741,000	771,000	30,000
Revenue	10,821,000	10,907,000	86,000
Net County Cost	238,147,000	243,105,000	4,958,000
Budgeted Positions	1,095.0	1,095.0	0.0

		Gross Appropriation (\$)	Intrafund Transfers (\$)	Revenue (\$)	Net County Cost (\$)	Budg Pos
PU	BLIC DEFENDER					
	21-22 Recommended Budget er Changes	249,709,000	741,000	10,821,000	238,147,000	1,095.0
	Retirement: Reflects revised retirement estimates based upon updated payroll data.	(4,475,000)	(13,000)	(90,000)	(4,372,000)	0.0
	Savings Plan: Reflects restoration of the County contribution to the 401(k) Savings Plan.	5,740,000	26,000	106,000	5,608,000	0.0
	Horizons: Reflects restoration of the County contribution to the Thrift Plan (Horizons) for non-represented and grand-fathered represented Flex and MegaFlex employees.	3,803,000	17,000	70,000	3,716,000	0.0
	eCAPS Maintenance: Reflects the Department's proportional share of eCAPS maintenance cost increases.	6,000			6,000	0.0
	Total Changes	5,074,000	30,000	86,000	4,958,000	0.0
202	1-22 Final Changes	254,783,000	771,000	10,907,000	243,105,000	1,095.0

	FY 2021-22 Recommended Changes	FY 2021-22 Final Changes	Variance
Appropriation	3,419,584,000	3,453,407,000	33,823,000
Intrafund Transfer	104,001,000	108,743,000	4,742,000
Revenue	1,708,327,000	1,716,212,000	7,885,000
Net County Cost	1,607,256,000	1,628,452,000	21,196,000
Budgeted Positions	17,018.0	17,040.0	22.0

		Gross Appropriation (\$)	Intrafund Transfers (\$)	Revenue (\$)	Net County Cost (\$)	Budg Pos
S	HERIFF					
20	21-22 Recommended Budget	3,419,584,000	104,001,000	1,708,327,000	1,607,256,000	17,018.0
1.	Antelope Valley (AV) / Department of Justice (DOJ) Settlement Agreement: Reflects an increase of 4.0 positions and funding for the Sheriff Department's Compliance Unit to ensure adherence with the terms of the settlement agreement. These positions will provide dedicated full-time staff to address the increased workload associated with the DOJ's oversight functions.	778,000	_	_	778,000	4.0
	Patrol Clearing					3.0
	Patrol – Specialized and Unallocated (S&U)	619,000			619,000	
	General Support	159,000			159,000	1.0
2.	Utility Users Tax (UUT): Reflects a decrease in appropriation to align with anticipated funding.	(215,000)			(215,000)	
	Patrol Clearing					
	Patrol – S&U	(215,000)			(215,000)	
3.	eCAPS Maintenance Adjustment: Reflects the Department's proportional share of eCAPS maintenance cost increases.	92,000		26,000	66,000	
	Administration	92,000		26,000	66,000	
4.	Public Safety Sales Tax (Proposition 172): Reflects a projected increase in revenue based on historical experience and anticipated trends.			5,135,000	(5,135,000)	
	Patrol S&U			2,580,000	(2,580,000)	
	Detective			370,000	(370,000)	
	Administration			44,000	(44,000)	
	Custody			1,717,000	(1,717,000)	

SHERIFF

	Gross Appropriation	Intrafund Transfers	Revenue	Net County Cost	Budg
General Support	(\$)	(\$)	(\$) 424,000	(\$) (424,000)	Pos
 Retirement: Reflects revised retirement estimates based upon updated payroll data. 	20,165,000	732,000	1,163,000	18,270,000	
Patrol Clearing					
Patrol – Unincorporated Areas (UA)	1,274,000		1,274,000		
Patrol – Contract Cities (CC)	1,949,000		1,949,000		
Patrol S&U	4,274,000		(3,168,000)	7,442,000	
Detective	1,072,000		30,000	1,042,000	
Administration	877,000			877,000	
Custody	4,962,000		689,000	4,273,000	
Court Services	2,440,000		6,000	2,434,000	
General Support	2,026,000		22,000	2,004,000	
County Services	1,291,000	732,000	361,000	198,000	
6. Savings Plan: Reflects restoration of the County contribution to the 401(k) Savings Plan.	5,391,000	659,000	483,000	4,249,000	
Patrol Clearing					
Patrol UA	131,000		131,000		
Patrol CC	200,000		200,000		
Patrol S&U	439,000		(320,000)	759,000	
Detective	109,000		6,000	103,000	
Administration	780,000			780,000	
Custody	1,029,000		137,000	892,000	
Court Services	166,000		1,000	165,000	
General Support	1,282,000		4,000	1,278,000	
County Services	1,255,000	659,000	324,000	272,000	
7. Horizons: Reflects restoration of the County contribution to the Thrift Plan (Horizons) for non-represented and grand-fathered represented Flex and Megaflex employees.	3,366,000	38,000	145,000	3,183,000	
Patrol Clearing					
Patrol UA	184,000		184,000		
Patrol CC	282,000		282,000		
Patrol S&U	617,000		(457,000)	1,074,000	
Detective	257,000		5,000	252,000	
Administration	161,000			161,000	
Custody	1,049,000		109,000	940,000	
Court Services	341,000		1,000	340,000	
General Support	403,000		3,000	400,000	
County Services	72,000	38,000	18,000	16,000	

SHERIFF

	Gross Appropriation (\$)	Intrafund Transfers (\$)	Revenue (\$)	Net County Cost (\$)	Budg Pos
8. Position Adjustments: Reflects Board-approved reclassifications that more appropriately reflects the assigned duties and responsibilities. Also reflects adjustments to correct positions added in the prior fiscal year.	(22,000)	(22,000)			1.0
Patrol Clearing					
County Services	(22,000)	(22,000)			
Custody					1.0
9. Contract Changes: Reflects changes in positions, services and supplies, capital assets-equipment, intrafund, and revenue in various budget units primarily due to requests by contract agencies in the prior year.	3,167,000	3,335,000	(168,000)		13.0
Patrol Clearing					1.0
Patrol CC	(443,000)		(443,000)		
Patrol S&U	275,000		275,000		
County Services	3,335,000	3,335,000			12.0
 Justice Assistance Grant (JAG) 2018: Reflects an increase in overtime in the Patrol Budget units to patrol community events, fully offset by the 2018 JAG. 	36,000		36,000		
Patrol Clearing					
Patrol S&U	36,000		36,000		
 Los Angeles County Regional Identification System (LACRIS): Reflects an increase of 1.0 Deputy Sheriff position in the General Support Budget, fully offset by revenue from the Automated Fingerprint Identification System Special Fund. 	239,000		239,000		1.0
General Support	239,000		239,000		1.0
12. Los Angeles Regional Interoperable Communications System (LARICS): Reflects an increase of 3.0 non-sworn positions in the General Support Budget, fully offset by revenue from LARICS.	826,000		826,000		3.0
General Support	826,000		826,000		3.0
Total Changes	33,823,000	4,742,000	7,885,000	21,196,000	22.0
2021-22 Final Changes	3,453,407,000	108,743.000	1,716,212,000	1,628,452.000	17,040.0

TRIAL COURT OPERATIONS

	FY 2021-22 Recommended Budget	FY 2021-22 Final Changes	Variance
Appropriation	404,412,000	404,402,000	(10,000)
Intrafund Transfer	0	0	0
Revenue	63,237,000	63,227,000	(10,000)
Net County Cost	341,175,000	341,175,000	0
Budgeted Positions	50.0	50.0	0.0

	Gross Appropriation (\$)	Intrafund Transfers (\$)		Net	
			Revenue (\$)	County Cost (\$)	Budg Pos
TRIAL COURT OPERATIONS					
2021-22 Recommended Budget	404,412,000	0	63,237,000	341,175,000	50.0
Other Changes					
1. Retirement: Reflects revised retirement estimates based upon updated payroll data.	(10,000)		(10,000)		
Total Changes	(10,000)	0	(10,000)	0	0.0
2021-22 Final Changes	404,402,000	0	62,227,000	341,175,000	50.0